# ACADEMIC YEAR 2021-22

11:37 🏴 🛤 🕅	•	(	Voi) 4 LTE1 4	⊊.ı  50°	% 🚔	
A Compl	ete GAdvance	ed.pdf	- Rea	d-only		
$\leftarrow$	$\mathbb{Z}$	Q	ズビ	Ş	:	Ŧ

# Contents

1	The Recombinant DNA Technology Era. Manisha Modak, Narendra Nyayanit, Aruna Sivaram, and Nayana Patil	1
2	Vectors: Guide to Gene Delivery Vehicles. Shalmali Bivalkar, Rajeev Mehla, Aruna Sivaram, and Nayana Patil	15
3	Gene Isolation Methods: Beginner's Guide Rajendra Patil, Aruna Sivaram, and Nayana Patil	43
4	DNA Cutters in Recombinant DNA Technology	57
5	Overview of Gene Cloning Strategies Neeraj Maheshwari, Praveen Kumar, Aruna Sivaram, and Nayana Patil	69
6	Fundamental Techniques of Recombinant DNA Transfer Shriram Rajpathak, Rupali Vyawahare, Nayana Patil, and Arage Stream	79
7	election, Screening, and Analysis of Recombinant Clones	7
8	Recent Trends and Advances Shruti Desai, Nayana Patil, and Aruna Sivaram	119
9	Applications of Recombinant DNA Technology Madhura Vipra, Nayana Patil, and Aruna Siyaram	143
10	Ethical and Safety Concerns of Recombinant DNA Technology Richa Sharma, Nayana Patil, and Aruna Sivaram	159
Ind	ex	167
		xi

Ο

<

# Contributors

[]]

11:38 🖬 🏴 🛤 🔹

 $\leftarrow$ 

(n) Vei) 4G ,III 50% ■

~

: -

K Z

A Complete G...Advanced.pdf - Read-only

2

Techniques in Life Science and Biomedicine for the Non-Expert Series Editor: Alexander E. Kalyuzhny

Nayana Patil Aruna Sivaram

A Complete Guide to Gene Cloning: From Basic to Advanced



<

Techniques in Life Science and Biomedicine for the Non-Expert

Series Editor Alexander E. Kalyuzhny, University of Minnesota

Ш

Jayachandra S. Yaradoddi • Merja Hannele Kontro • Sharanabasava V. Ganachari Editors

# Actinobacteria

Ecology, Diversity, Classification and Extensive Applications

 $\Delta$  Springer

Editors
Jayachandra S. Yamdoddi
Merja Hannele Konro
Biomaterial Science
KLE Technological University
Hubballi, Kamaaka, India
Faculty of Biological and Environmental
Sciences, Deputment – Beology, and
University of Helsinki
Lahti, Finland Sharanabasawa V. Ganachari Center for Material Science Advanced Research in Nanoscience & Nanotechnology, KLE Technological University Hubballi, Kamataka, India ISSN 2523-8442 ISSN 2523-8450 (electronic) ISBN 978-981-16-3352-2 ISBN 978-981-16-3353-9 (eBook) https://doi.org/10.1007/978-981-16-335-9 https://doi.org/10.1100/9778-981-16-3333-9 O Springer-Nature Singapore-Fie La 2.021 This work is subject to copyright. All rights are reserved by the Publisher, whether he whele or part of the material is concerned, specifically the rights of translation, reprinting, more of illustration, mitation, branksaring, reproduction on microfilms or in any other physical way, and henemission or information strongs and native algebraic translation on microfilms or in any other physical way. And henemission or information branksaring, reproduction on microfilms or in any other physical way, and henemission or information branksaring, reproduction on microfilms or in any other physical way, and henemission or information branksaring, reproduction on microfilms or in any other physical way. And henemission or information through the strong of the strong The outfilms, the antibute, and therefore free for general use. The publisher, the antibute, and the others are used to assume that the advice and information in the product of the strong of the This Springer imprint is published by the registered company Springer Natare Singapore Pte Ltd. The registered company address is: 152 Beach Read, #21-01/04 Gateway East, Singapore 189721, Singapore 1

# Contents

1	Microbial Ecology. Merja H. Kontro and Jayachandra S. Yaradoddi	1
2	Actinobacteria in Marine Environments. Jayachandra S. Yaradoddi, Menja H. Kontro, Sharanabasava V. Ganachari, Nagamj R. Banaputmath, Ajaykumar Oli, Anilkumar S. Kati, and M. B. Sulochana	21
3	Terrestrial Ecology of Actinobacteria Basavanj S. Hangund, Savitha S. Desai, Kartik C. Kamath, and Gururaj B. Tennalli	39
•	Extremophilic Actino bacteria Jayachandra S. Yaradoddi, Merja H. Kontro, Nagaraj R. Banapurmath, Sharanabasava V. Ganachari, M. B. Sulochana, Basavaraj S. Hungund, Zamen Kousar Kazi, S. K. Anilkumar, and Ajaykumar Oli	55
5	Actinobacteria: Basic Adaptation to Harsh Environments Jayachandra S. Yaradoddi and Merja H. Kontro	69
6	Diversity and Classification of Streptomyces. Basavamj S. Hangund, Samay Honnangi, Savitha S. Desai, Kaveri Badiger, and Gururaj B. Tennalli	89
7	Diversity and Classification of Rare Actinomycetes	117
8	Identification of Novel Actinomycetes . Jayachandra S. Yaradoddi, Merja H. Kontro, Nagaraj R. Banapurmath, Sharanabasava V. Ganachari, and M. K. Umesh	143

# Editors and Contributors

## About the Editors



Jay achandra S. Yaradoddi is a Research Scientist in Biomaterials Laboratory, Dept. of Center for Materials Science, KLE Technological University, Hubballi, Kanataka, India. He obtained his Masters and Doctor of Philosophy of Biotechnology from the Gulbarga University, Kalbungi. He has executed research work on "diversity and molecular characterization of halophilic bacteria from Karnataka" during his Ph.D. He was served as a project fellow (JRF and SRF) under UGC major research project entitled "Sidemphore as a biocontrol agent on phytopathogens of regional cops." Recent trends in actinobacterial diversity of environment and their potential bioactive molecules" from the University of Helsinki, Finland. He has published more than 40 publications in various national and international journals of repute and filed two Indian patents. He has completed the Govt. funded project under the IDEA2POC scheme by Bangalore Bioinnovation Center and KBITS, Govt. of Karnataka. His major thust area of research is microbial taxonomy, biomaterials, industrial biotechnology, and bio-nanotechnology.

xi

# Oral Presentation: 1 Media Optimization Studies *of Chlorella sorokiniana* PCPBEC for Biofuels

# Dr. Premjyoti C Patil Assistant Professor, Department of Biotechnology Basaveshwar Engineering College, Bagalkote. e-mail: jvosnp@gmail.com Mobile number: 9741953811 Abstract

Biofuels itself is an 'acheel's heel' and algal studies are still far away from commercial reach. Microalgae has a high potential in third generation biofuels. The major concern among microalgal research is unavailability of huge biomass required for commercial production. Green alga, species belonging to Chlorophyceae has more diversity and has a greater adaptation in wide climatic conditions. The current study aimed for biomass and lipid enhancement through media optimization studies. The experiment was divided into two phases of Response Surface Methodology in which placket - Burmann as phase-1 and Box-Behnken as phase 2. In PB experiment NaNO3, CaCl2, MgSO<sub>4</sub>, KH<sub>2</sub>PO<sub>4</sub>, K<sub>2</sub>HPO<sub>4</sub>, Urea, and 2-4, D. Biomass had a better response for all the variables except Di-potassium orthophosphate. Further in Box Behnken experiment three variables NaNO3, CaCl2 and urea were chosen based on the results obtained from PB design. The results has revealed highest wet biomass which was obtained in Experiment number 1 [Concentration of NaNO<sub>3</sub> (High; +1), CaCl<sub>2</sub> (Central; 0) and Urea (Low: -1)] with biomass yield of 3.65 g/L on 14<sup>th</sup> day of inoculation followed by experiment 2 [ Concentration of NaNO<sub>3</sub> (Low: -1), CaCl<sub>2</sub>(Central; 0) and Urea (Low: -1)] with yield of 3.45 g/L and lowest was in experiment 3 i.e., control [Concentration of NaNO<sub>3</sub> 50g/L,CaCl<sub>2</sub> 2.5 g/L and no Urea] with 2.28 g/L. Highest yield obtained in control was observed on 16-17<sup>th</sup> day of inoculation where as in formulated media it was observed on 13-14<sup>th</sup> day of inoculation. Two optimized media are being formulated for biomass and lipid enhancement. The isolate further may be used for biofuel studies.

# **Oral Presentation: 3**

# Statistical Optimization of Biodiesel Production from *Pongamia pinnata* Seed Oil Using Immobilized Lipase

Shilpa K Jigajinni& Dr. Bharati S Meti Department of Biotechnology Basaveshwar Engineering College (A) Bagalkot <u>shilpajigajinni@gmail.com</u> 8722485018

# Abstract

The environmental and economic aspect of fossil fuel use at greater extent encourages to find new alternative to fossil fuel which is ecofriendly and cost effective that could meet the growing energy demand. Biodiesel is Fatty acid alkyl esterone among the alternative renewable promising biofuel. The production of biodiesel through enzymatic transesterification is more advantageous than thermochemical process. Enzymatic process is influenced by many factors such as amount of biocatalyst, molar ratio of oil to alcohol, temperature, pH, rpm, time etc. Effective variables for transesterification may vary based on the type of feedstock and catalyst used, therefore it is essential to optimize the process suitable for each type of feedstock to achieve higher yield of biodiesel. Optimization through statistical way is more advantageous than traditional method. Reaction conditions of transesterification is optimized through Response Surface Methodology by Plackett-Burman and Box-Behnken design. The factors used in the present study for PB design are molar ratio, amount of immobilized lipase, temperature, time, pH, agitation speed and water content. The result showed that among seven variables, pH (p-value 0.003), agitation speed (p-value 0.024) and amount of immobilized lipase (*p*-value 0.041) having p < 0.05 are statistically significant positively affecting the transesterification process of *Pongamia* seed oil. Further these effective variables are studied for optimization through Box-Behnken design. *p-value* developed for the model is found to be 0.007, indicating that the model is very adequate and highly significant at 95% confidence interval (p-value 0.05) approves the model for validation. The optimum parameters obtained by statistical designs such as concentration of biocatalyst (12.5%), pH (7), agitation speed (125rpm), molar ratio (1:3), temperature (45°C) and reaction time (24hr). Under optimum conditions, the model estimated biodiesel yield of 96% and experimental trial yields around of 94.8%. Fuel properties of biodiesel is tested as per ASTM standards.



# **BOOK OF ABSTRACTS**

# INTERNATIONAL VIRTUAL CONFERENCE ON SUSTAINABLE BIOTECHNOLOGY (ICSB)

Organized by

Department of Life Sciences Kristu Jayanti College (Autonomous) Bengaluru-560077







NITTE - 574 110, Karkala Taluk, Udupi District, Karnataka, India ISO 9001:2015 certified, Accredited by NAAC with 'A' Grade



# An International Conference On

Nurturing Innovative Technological Trends in Engineering BIOscience (NITTE-BIO) -2021 Under the aegis of International Conference on Emerging Trends in Engineering (ICETE -2021)

# **CERTIFICATE OF PARTICIPATION**

This is to certify that Sharada P

of Department of Biotechnology, Basaveshwar Engineering College

has sucessfully participated

in and co-authored a paper titled "Overrepresentation analysis of genes common to biological ageing and Alzheimer's Disease reveals pathways related to cellular heat stress response and CDK5 dysregulation" the International Conference Organized by the Department of Biotechnology Engineering, N.M.A.M. Institute of Technology-Nitte, during 19th & 20th November 2021.

Missal Dr. Uiwal P. NITTE BIO 2021- Program Chair 

Noni-N.L Dr. Niranjan N. Chiplunkar Principal, ICETE-2021 Chair

Lecture Notes on Data Engineering and Communications Technologies 131

1 3

G. Rajakumar Ke-Lin Du Chandrasekar Vuppalapati Grigorios N. Beligiannis *Editors* 

Intelligent Communication Technologies and Virtual Mobile Networks

Proceedings of ICICV 2022



# Metadata of the chapter that will be visualized in SpringerLink

Book little	Intelligent Communica	ation Technologies and Virtual Mobile Networks	
Series Title			
Chapter Title	User Credibility-Based Trust Model for 5G Wireless Networks		
Copyright Year	2023		
Copyright HolderName	The Author(s), under e	exclusive license to Springer Nature Singapore Pte Ltd.	
Corresponding Author	Family Name	Manjaragi	
	Particle		
	Given Name	Shivanand V.	
	Prefix		
	Suffix		
	Role		
	Division	Department of Computer Science and Engineering	
	Organization	Hirasugar Institute of Technology	
	Address	Nidasoshi, Belgaum, India	
	Email	shiva.vm@gmail.com	
Author	Family Name	Saboji	
	Particle		
	Given Name	S. V.	
	Prefix		
	Suffix		
	Role		
	Division	Department of Computer Science and Engineering	
	Organization	Basaveshwar Engineering College	
	Address	Bagalkot, India	
	Email	saboji_skumar@yahoo.com	
Abstract	The 5G wireless networks are expected to provide fastest mobile Internet connectivity, efficient network access, capable of handling large amount of data traffic, connecting large number of mobile devices with high-throughput and very-low latency. The new technologies such as cloud computing, network function virtualization (NFV), and software-defined networking (SDN) are being used in 5G wireless networks. The number of small cells in 5G networks creating a heterogeneous network environment (HetNet), where users will join and leave the network frequently causing repeated authenticated vertical handoff across the different cells leading to delay in the network. There are new security requirements and challenges in 5G mobile wireless network due to its advanced features. Therefore, in this paper to deal with secured vertical handoff, a trusted authenticating mechanism is proposed to secularly authenticate the user based on the credibility in 5G wireless networks. It is generating a trust relationship between user, base station, and home networks based on the user credibility and performs quick and secured handoff. The user credibility is comprises the direct credibility and indirect credibility calculation. Based on the user credibility, the trustworthiness of user equipment (UE) is identified and vertical handoff performed without reauthentication across different heterogeneous small cells in 5G wireless networks.		
Keywords (separated by '-')	5G wireless networks	- SDN - Trusted model - User credibility - Secure vertical handoff	

# User Credibility-Based Trust Model for 5G Wireless Networks



1

Shivanand V. Manjaragi and S. V. Saboji

**Abstract** The 5G wireless networks are expected to provide fastest mobile Internet 1 connectivity, efficient network access, capable of handling large amount of data 2 traffic, connecting large number of mobile devices with high-throughput and very-3 low latency. The new technologies such as cloud computing, network function virtu-Δ alization (NFV), and software-defined networking (SDN) are being used in 5G 5 wireless networks. The number of small cells in 5G networks creating a hetero-6 geneous network environment (HetNet), where users will join and leave the network 7 frequently causing repeated authenticated vertical handoff across the different cells 8 leading to delay in the network. There are new security requirements and challenges 9 in 5G mobile wireless network due to its advanced features. Therefore, in this paper 10 to deal with secured vertical handoff, a trusted authenticating mechanism is proposed 11 to secularly authenticate the user based on the credibility in 5G wireless networks. 12 It is generating a trust relationship between user, base station, and home networks 13 based on the user credibility and performs quick and secured handoff. The user cred-14 ibility is comprises the direct credibility and indirect credibility calculation. Based 15 on the user credibility, the trustworthiness of user equipment (UE) is identified and 16 vertical handoff performed without re-authentication across different heterogeneous 17

18 small cells in 5G wireless networks.

<sup>19</sup> Keywords 5G wireless networks · SDN · Trusted model · User credibility ·

20 Secure vertical handoff

S. V. Manjaragi (🖂)

Department of Computer Science and Engineering, Hirasugar Institute of Technology, Nidasoshi, Belgaum, India e-mail: shiva.vm@gmail.com

S. V. Saboji

Department of Computer Science and Engineering, Basaveshwar Engineering College, Bagalkot, India

© The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2023 G. Rajakumar et al. (eds.), *Intelligent Communication Technologies and Virtual Mobile Networks*, Lecture Notes on Data Engineering and Communications Technologies 131, https://doi.org/10.1007/978-981-19-1844-5\_23

AQ1

Proceedings of the International Conference on Electronics and Renewable Systems (ICEARS 2022) DVD Part Number: CFP22AV8-DVD; ISBN: 978-1-6654-8424-4

# Fault Analysis and Identification in Ubiquitous Networks using Cognitive Agents Approach

1<sup>st</sup> Vasudha V Ayyannavar dept. Computer Science and Engineering Basaveshwar Engineering College Bagalkot, India vavasudha.11@gmai.com

Abstract—The ubiquitous network has set of heterogeneous networks such as wireless sensor networks, cellular networks, mobile adhoc networks, Bluetooth networks, etc. Each heterogeneous network communicates with other to perform ubiquitous computation, but there is a need of a fault tolerance mechanism. This kind of mechanism is used to provide continuous services to users in the presence of faults. The faults may be node, link, and memory level faults in the ubiquitous networks. Such levels faults are lead to the failure of the networks. However, there is a need for substantial time and sufficient resources for providing reliability in fault tolerance mechanisms. In this work, it has been proposed that cognitive agent based fault analysis and identification of types of faults using support vector machine algorithm. Also, the various types of cognitive agents are defined as network cognitive agents, heterogeneous node cognitive agents, gateway cognitive agents, ubiquitous cognitive agents, and fault analysis and identification cognitive agents. Each agent is communicating with other for analyzing the types of faults. Finally, the proposed work is simulated in terms of delay, probability analysis and identification, throughput, packet delivery ratio, and agent computation overhead.

Index Terms—Ubiquitous Networks, Cognitive Agents, Fault identification, Fault Analysis, Support Vector Machine

# I. INTRODUCTION

Ubiquitous Networks (UNs) are used to access the services at anytime, anywhere, and at anyplace through the computing devices [1]. The dynamic and the uncertainty nature of ubiquitous network with resource constraints of the devices lead to deal with fault toleranceA system's fault tolerance is its ability to provide the desired level of functioning even when there are defects. The basic purpose of a fault tolerant system is to increase the availability and the reliability of the system. Fault tolerance should be seriously considered in many ubiquitous network systems since ubiquitous nodes are prone to failure. Each layer of the network system takes fault tolerance into account. More specifically, fault tolerance in a ubiquitous network may exist at hardware layer, software layer, network communication layer, and application layer.

There are several reasons for a communication link or a device/node to fail. The faults in any computing network can be categorized as device faults, application faults, network faults, and service faults. Device faults are considered as physical devices such as mobile devices, sensor nodes, and actuator failure. Each device has its own set of faults which

2<sup>nd</sup>Lokesh B. Bhajantri dept. Information Science and Engineering Basaveshwar Engineering College Bagalkot, India lokeshcse79@gmail.com

contribute to failure of system. These devices will exist with limited battery power and signal strength because of which the system will not be able to execute an application. Anytime and anyplace service provider with its dynamic nature of nodes leads to networks faults like link failure and congestion problem. It gets disconnected with the ubiquitous computing environment makes the application to fail and sometimes there may be problem with service. Sometimes the system works correctly but gives incorrect result or operates incorrectly because of node failure.

Cognitive agent (CA) is a lightweight process that performs the tasks based on its knowledge regularly from the ubiquitous networks environment. CA independently executes the tasks in a flexible and intelligent manner. CAs are communicates, coordinate with other agents by its experience with changing environment and place [2]. CAs based fault tolerance in ubiquitous networks will handle the fault identification by deploying CAs at each layer of an environment to handle different types faults. Also, able to search available devices with resources either during or before data processing takes place and monitor the progress of information processing.

The CAs collects the information of users, nodes, links, available devices, and an environment for generating the beliefs. Develop the plan for the actions for achieving the goals (desires) of analysis and identification of faults using predictive analytics model that is Support Vector Machine (SVM). The desires of the beliefs are executed by the CAs in the ubiquitous networks that is the intention of the proposed CAs.SVM is a classification algorithm that classifies the types of faults in a ubiquitous network. The data is mapped from the input space vector to a higher-dimensional feature space vector using the mapping function. The high dimensional feature space and hyperplane in feature space are used to maximize the classification capability of SVM [3]. SVMs can be used to do nonlinear node classification by mapping the classified data to a high-dimensional feature space with the possibility of linear classification.

In this paper, the CA based fault analysis and identification using SVM approach is presented. Also, monitor the status of the nodes, links, and memory level of each ubiquitous node as a continuously by CAs. The classifications of the faults are done by using SVM classification algorithm. SVM based classifier has better generalization property because it has principle of structural risk minimization.

The remainder of the study is organised as follows: Section 2 discusses current strategies for detecting problems in ubiquitous computing. The proposed technique of Cognitive Agents-based defect identification and analysis is demonstrated in Section 3. The simulation and results are presented in Section 4. Finally, in Section 5, the conclusion is offered.

# II. RELATED WORK

In this work, it has been carried out the related works towards proposed work that are as follows: The work given in [4] proposes a fault-tolerant anomaly detection method to improve and provide a solution to the problem where the event detection degrades when the probability of fault reaches a crucial value. The work given in [5] presents detection of defects in the Gas Station Internet of Things (GS-IoT) system and presents a bi-directional long-short term memory (BiLSTM) network algorithm combined with a support vector machine (SVM), SVM-BiLSTM. Work given in [6] described a bacterial-inspired technique for detecting unique mobile faults in wireless sensor networks (WSNs). Fault detection accuracy is considered as a parameter for analysis. The work given in [7] offers to integrate and detect hybridization between AI approaches and big data algorithms in order to improve building system monitoring and control, boosting comfort and lowering operating costs. A method has been proposed that uses AI and big data collected by building management systems to automatically detect irregular energy consumption (BMS).

The work given in [8] proposed an algorithm in which the checkpoint approach is used for executing recovery action in distributed system. The faulty node identified will be replaced and keeping the remaining part of network as it is and it handles hardware and software faults. Work given in [9]presents a data-based fault detection system for analyzing sensor node data and determining problems, preventing the related nodes from delivering data and reducing network damage. By collecting and analyzing data, the effectiveness of the suggested fault-detection model was tested on an integrated management platform based on the Internet of Things. The findings show that if a malfunctioning sensor node is not disconnected from the network, superfluous data transmission of other sensor nodes occurs as a result of ongoing abnormal data transmission, increasing energy consumption and shortening network lifetime. The work given in [10] proposes an analysis of failure propagation is used to offer a fault localization method based on sample programmes. When enough sample programmes are provided, even if the defective programme fails all test cases, the suggested methods of programme clustering and fault localisation working together can effectively find the suspicious statements, according to the experimental results. The work given in [11] proposed task based fault handling by saving the state of system periodically and restarting the job from its previously checked point instead of starting point by which the loss of information will be prevented.

The work given in [12] presented SVM grid approach for on line fault detection. Fine tuned prediction algorithm is used to get input parameter for increasing accuracy. Using the Markov State Transition Model, the work presented in [13] offered fault tolerance and QoS-based pervasive computing. A fuzzy logic approach for fault node detection with three inputs for an IOT system in presented in [14]. The parameters considered in this are false alarm rate and detection accuracy for performance analysis. The work given in [15] proposes a mechanism in WSNs for fault detection with low energy consumption distributed fault detection. It used the information collected by node will identify the fault by itself then uses neighboring node for data collection which reduces the traffic in communication. A method for detecting the attack on network by using SVM is described in [16].

# III. PROPOSED WORK

In this section, the proposed system architecture, computation of fault assessment value, and functioning scheme of the proposed scheme are discussed. The majorities of the existing techniques consider the status of nodes and have not applied intelligence techniques. In this work, for improving the efficiency of ubiquitous networks applying CAs. The major contributions of the paper includes: 1) the design of CA based SVM algorithm to fault analysis and identification. 2) Employing Belief-Desire-Intention (BDI) model using proposed CAs towards the faults of the network in terms of node or device, link, and memory level faults. 3) The usage of proposed CAs for delivering the network information for achieving the fault tolerance. 4) Evaluate the fault assessment value at each level of faults and 5) Finally, evaluated the proposed scheme in terms of performance parameters such delay, probability of fault analysis and identification, throughput, packet delivery ratio, and agent computation overhead.

# A. System Architecture

The Figure 1 depicts the proposed system architecture. This architecture comprises various types of CAs like Network Cognitive Agents (NCAs), Heterogeneous Node Cognitive Agents (HNCAs), Base Station or Gateway Cognitive Agents (BSCA/GCAs), Ubiquitous Cognitive Agents (UCAs), and Fault Analysis and Identification Cognitive Agent (FAICA). The ubiquitous network associates with set of various heterogeneous networks like WSNs, Wi-Fi networks, Bluetooth networks, cellular networks, mobile Adhoc networks, etc., so each heterogeneous network associates with CAs. Also, each network associates comprises set of ubiquitous nodes. The descriptions of the proposed CAs are as follows:

• NCA – This agent is a type of mobile CA that monitors the network conditions in terms of node/device failure, link failure, node memory, and bandwidth of the network. This agent acts as an intelligent, autonomous, and proactive in the networks. It moves from one node to another node in the network to get the node information, bandwidth, and link information by contacting with HNCA



International Conference on Electronics and Renewable Systems (ICEARS 2022) organized by St. Mother Theresa Engineering College, Tuticorin, Tamil Nadu, India held on 16-18, March 2022.





# 16-18, March 2022

# Certificate of Presentation

# Vasudha V Ayyannavar

has successfully presented the paper entitled

Fault Analysis and Identification in Ubiquitous Networks using Cognitive Agents Approach







# **Electronics and Renewable Systems**

Tuticorin, India

This is to certify that

at the



Conference Chair Dr. A. George Klington



# Behavior of Fully and Partially Encased Stone Column in Black Cotton Soil

Mahantesh Neelawani 🗠 & Prasanna Patil

Conference paper | First Online: 27 June 2022

Part of the <u>Lecture Notes in Civil Engineering</u> book series (LNCE,volume 256)

# Behavior of Fully and Partially Encased Stone Column in Black Cotton Soil



136/1088

Mahantesh Neelawani and Prasanna Patil

Abstract In general, expansive soils are subjected to high volume changes as a consequence of variations in water content. In the dry season, these soils undergo shrinkage, which causes deep cracks in the soil. Thus, these soils are extremely troublesome for geotechnical engineers, so it is essential to implement the ground improvement techniques. The single-column behavior in expansive soil with and without encasement was discussed in this laboratory model study. The load tests were conducted in a steel tank measuring 50 cm  $\times$  50 cm  $\times$  50 cm. For the whole study, the column measures 30 mm in diameter and 300 mm in depth and is made of stone chips. The addition of a column increases ultimate strength by 225%. To enhance the strength and rigidity of the column, an encasement using non-woven geotextile was adopted. According to the load tests, a partially encased column improves ultimate strength by 275% and a fully encased column improves ultimate strength by 340%. A conventional column is associated with a lower load limit ratio, whereas a fully encased column is associated with a higher ratio. However, partially encased columns have a minor improvement. The behavior of a column in black cotton soil is influenced by climatic fluctuations, particularly at the 1D to 3D depths. Therefore, the encasement is adopted and lime was injected into the soil bed up to 150 mm deep to reduce the shrinking and swelling characteristics of the soil around the column.

Keywords Stone column · Non-woven geotextile · Black cotton soil · Fully and partial encasement

P. Patil Basaveshwar Engineering College, Bagalkot, Karnataka, India

115

M. Neelawani (🖾)

Madanapalle Institute of Technology and Science, Madanapalle, Andhra Pradesh, India e-mail: mahanteshneelawani@mits.ac.in

<sup>©</sup> The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2023 L. Nandagiri et al. (eds.), *Recent Advances in Civil Engineering*, Lecture Notes in Civil Engineering 256, https://doi.org/10.1007/978-981-19-1862-9\_8

# Recent Trends in Computational Intelligence Enabled Research

Theoretical Foundations and Applications

Edited by Siddhartha Bhattacharyya, Paramartha Dutta, Debabrata Samanta, Anirban Mukherjee, Indrajit Pan



# Contents

List of contributors Preface	xiii xv	3. re
1. Optimization in the sensor cloud: Taxonomy, challenges, and survey prashant sangulagi and ashok sutagundar	1	GU SU 3.
<ul> <li>1.1 Introduction</li> <li>1.2 Background and challenges in the sensor cloud</li> <li>1.2.1 Key definitions</li> <li>1.2.2 Challenges/issues</li> <li>1.3 Taxonomy for optimization in the sensor cloud</li> <li>1.3.1 Load balancing</li> <li>1.3.2 Information classification</li> <li>1.3.3 Information transmission</li> </ul>	1 4 5 6 7 10 11	3. 3. Re
<ul> <li>1.3.4 Information processing</li> <li>1.3.5 Limitations of existing work</li> <li>1.4 Discussion and future research</li> <li>1.4.1 Discussion</li> <li>1.4.2 Future research</li> <li>1.5 Conclusion</li> <li>References</li> </ul>	13 15 16 16 18 19 19	4. or In D4
2. Computational intelligence techniques for localization and clustering in wireless sensor networks BASAVARAJ M. ANGADI, MAHABALESHWAR S. KAKKASAGERI AND SUNILKUMAR S. MANVI	23	4 <sup>2</sup> 4 <sup>2</sup>
<ul> <li>2.1 Introduction</li> <li>2.2 Wireless sensor networks</li> <li>2.2.1 Characteristics</li> <li>2.2.2 Research issues/challenges</li> </ul>	23 24 24 25	4.4
<ul><li>2.3 Localization and clustering in wireless sensor networks</li><li>2.3.1 Localization</li><li>2.3.2 Clustering</li></ul>	26 26 26	4. A Re
<ul> <li>2.4 Computational intelligence techniques</li> <li>2.4.1 Computational intelligence techniques for localization</li> <li>2.4.2 Computational intelligence techniques for clustering</li> </ul>	27 30 34	5. in sr
2.5 Future research directions References	37 38	5. 5.

3. Computational intelligent techniques for resource management schemes in wireless se	ensor 41
GURURAJ S. KORI, MAHABALESHWAR S. KAKKASAGERI A SUNILKUMAR S. MANVI	HI ND
3.1 Introduction	41
3.2 Wireless sensor networks	42
3.2.1 Characteristics	42
3.2.2 Applications	44
3.2.3 Issues/challenges	45
3.3 Resource management in wireless sensor networks	3 46
3.3.2 Litoroture current	47
3.4 Eutroperations and conclusion	47
References	56
4. Swarm intelligence based MSMOPSO for	r
optimization of resource provisioning in	
Internet of Things	61
DANESHWARI I. HATTI AND ASHOK V. SUTAGUNDAR	
4.1 Introduction	61
4.1.1. Related work	62
4.1.2 Our contributions	64
4.2. Proposed method	64
4.2.1 Network environment	64
4.3 Agency	70
4.3.1 Device agency	70
4.3.2 Fog agency	71
4.3.3 Example scenario	72
4.4 Simulation	73
4.4.1 Simulation inputs	73
4.4.2 Simulation procedure	74
4.4.3 Performance measures	74
4.4.4 Kesults	/6
4.5 Conclusion	81
References	81
References	01
5. DNA-based authentication to access	
internet of things-based healthcare data	83
SREEJA CHERILLATH SUKUMARAN	
5.1 Introduction	83
5.2 Literature survey	83

# 2

# Computational intelligence techniques for localization and clustering in wireless sensor networks

Basavaraj M. Angadi<sup>1</sup>, Mahabaleshwar S. Kakkasageri<sup>1</sup> and Sunilkumar S. Manvi<sup>2</sup>

<sup>1</sup>Department of Electronics and Communication Engineering, Basaveshwar Engineering College (Autonomous), Bagalkot, India <sup>2</sup>School of Computing and Information Technology, REVA University, Bangalore, India

# 2.1 Introduction

Wireless sensor networks (WSNs) are used to simplify and manage complex problems in most applications. Energy conservation in WSNs is a prime concern. It is important as the lifetime of the network depends mainly on the WSN energy consumption. Hence the highest priority is given to balancing and preserving the consumption of energy. Therefore the design and development of algorithms with minimum energy expenditure is a main challenge in WSNs. To achieve this, new protocols and schemes have to be designed using emerging computational techniques (Primeau, Falcon, Abielmona, & Petriu, 2018).

In a sensor network, it is very important to know about the location of the sensor node (SN) and data collected by that node which can be obtained using localization techniques. Accurate position estimation of target nodes in WSNs is highly desirable as it has a significant impact on the overall network performance. Normally, information regarding location is useful for deployment, coverage, location service, routing, target tracking, and rescue. In some industries, SNs are used to identify minute changes such as gas leaks, pressure, and temperature, and in military applications for the detection of landmines. Location information plays a key role in both of these cases.

In general, clustering means the process of grouping SNs into clusters to overcome the scalability problem.

Clustering of SNs also helps in achieving high energy efficiency and extends the network lifetime, especially in large-scale WSN environments. It has been proved that for organizing a network into a connected hierarchy, clustering is an effective topology control approach (Angadi, Kakkasageri, & Kori, 2016). Hierarchical routing and data collection protocols include cluster-based organization of SNs for data fusion and aggregation purposes (Kakkasageri & Manvi, 2014), which leads to a substantial decrease in energy consumption.

Most computational intelligent technologies are interdisciplinary in nature, such as the integration of artificial intelligence (AI) with computer science and information systems with decision support systems. Some of the intelligent techniques include natural computation, bio-inspired computational techniques, game theory, AI, neural systems/networks, fuzzy logic techniques, genetic algorithms (GAs), intelligent multiagent systems, policy-based multiagent systems, machine learning, and so on (Alomari, Phillips, Aslam, & Comeau, 2018; Baranidharan & Santhi, 2016; Mann & Singh, 2017b; Singh, Khosla, & Kumar, 2018).

The chapter is organized as follows. Section 2.2 describes WSNs. Localization and clustering are explained in Section 2.3. Computational intelligence (CI) techniques for localization and clustering are discussed in Section 2.4, and finally, Section 2.5 presents further research challenges that need to be addressed.



Theoretical machine learning models for predicting pandemic diseases in sensor cloud

# Recent Trends in Computational Intelligence Enabled Research

Theoretical Foundations and Applications

Edited by Siddhartha Bhattacharyya, Paramartha Dutta, Debabrata Samanta, Anirban Mukherjee, Indrajit Pan



# RECENT TRENDS IN COMPUTATIONAL INTELLIGENCE ENABLED RESEARCH

1

# viii

	5.2.1 Internet of things generic architecture	84
	5.2.2 Challenges in the internet of things	84
	5.2.3 Security challenges in internet of things layers	85
	5.2.4 Authentication schemes in the internet of things	86
	5.2.5 DNA cryptography	88
5.3	Methodology	88
5.4	Security analysis	88
	5.4.1 Password file compromise attack	88
	5.4.2 Dictionary attack	91
	5.4.3 Replay attack	91
5.5	Scyther analysis	92
5.6	Conclusion	92
Ref	erences	92
6.	Computational intelligence techniques for	
car	ncer diagnosis	95
NIM	IRITA KOUL AND SUNILKUMAR S. MANVI	
6.1	Introduction	95
6.2	Background	95
	6.2.1 Cancer research data	95
	6.2.2 Genomic data for cancers	96
	6.2.3 Imaging data for cancers	97
6.3	Approaches to computational intelligence	97
	6.3.1 Evolutionary computation	98
	6.3.2 Learning theory	98
	6.3.3 Artificial neural networks	98
	6.3.4 Probabilistic methods	99
6.4	Computational intelligence techniques for feature	
	selection in cancer diagnosis	99
	6.4.1 Advantages of feature selection	100
	6.4.2 Rough sets for feature selection	100
	6.4.3 Genetic algorithms for feature selection	101
	6.4.4 Adaptive network fuzzy inference system	102
	6.4.5 Deep learning for cancer diagnosis	103
	6.4.6 Autoencoders for feature extraction	105
	6.4.7 Particle swarm optimization for feature selection	105

		0.5 Г
6.4.3 Genetic algorithms for feature selection	101	\$
6.4.4 Adaptive network fuzzy inference system	102	
6.4.5 Deep learning for cancer diagnosis	103	8
6.4.6 Autoencoders for feature extraction	105	
6.4.7 Particle swarm optimization for feature selection	105	8
6.5 Computational intelligence methods for cancer		8.4 C
classification	107	8
6.5.1 Classification methods	107	8
6.6 Conclusion	108	8.5 C
Acknowledgment	108	Appe
References	108	sal

7. Security and privacy in the internet of things: computational intelligent techniques-based approaches	111
POORNIMA M. CHANAL, MAHABALESHWAR S. KAKKASAGERI AND SUNIL KUMAR S. MANVI	
7.1 Introduction	111
7.2 Internet of things	112
7.2.1 Architecture	112
7.3 Characteristics	114
7.4 Research issues/challenges	115
7.5 Applications	116
7.6 Security and privacy in the internet of things	118

## Contents

7.6.1 Security	118
7.6.2 Privacy	119
7.7 Computational intelligent techniques	119
7.7.1 Artificial intelligence	120
7.7.2 Neural networks	120
7.7.3 Evolutionary computation	120
7.7.4 Artificial immune systems	121
7.7.5 Fuzzy system	121
7.7.6 Machine learning	121
7.7.7 Bio-inspired algorithm	121
7.8 Computational intelligent techniques to provide	
security and privacy for the internet of things	122
7.8.1 Confidentiality	122
7.8.2 Integrity	122
7.8.3 Authentication	123
7.8.4 Availability	123
7.9 Future research direction	124
References	124

8. Automatic enhancement of coronary arteries using convolutional gray-level templates and	
path-based metaheuristics	129
MIGUEL-ANGEL GIL-RIOS, IVAN CRUZ-ACEVES, FERNANDO CERVANTES-SANCHEZ, IGOR GURYEV AND JUAN-MANUEL LÔPEZ-HERNÁNDEZ	
8.1 Introduction	129
8.2 Background	131
8.2.1 Iterated local search	131
8.2.2 Tabú search	135
8.2.3 Simulated annealing	137
8.2.4 Univariate marginal distribution algorithm	138
8.3 Proposed method	139
8.3.1 Automatic generation of convolutional gray-level template	139
8.3.2 Binary classification of the gray-level filter	
response	141
8.3.3 Image postprocessing	141
8.4 Computational experiments	142
8.4.1 Results of vessel imaging enhancement	142
8.4.2 Postprocessing procedure	143
8.5 Concluding remarks	146
Appendix 1 Matlab code of the tabú search for the traveler	
salesman problem	147
References	152

<ol><li>Smart city development: Theft handling of public vehicles using image analysis and cloud network</li></ol>	155
HIMADRI BISWAS, VASKAR SARKAR, PRIYAJIT SEN AND DEBABRATA SARDDAR	
9.1 Introduction	155
9.2 Motivation scenario	155
9.3 Issues and challenges of image authentication through	
Internet of Things-based cloud framework	156
9.3.1 Biometric system	156
9.3.2 Internet of Things	160

# Block Chain Interventions for Pharmaceutical Supply Chain Management: A Literature Survey

Dr. J.D. Mallapur<sup>1</sup>, Anuradha Hiremath<sup>2</sup>

<sup>1</sup> Professor in the Department of Electronica and Communication, Basaveshwar Engineering College, Bagalkot (Autonomous)

<sup>2</sup> M. Tech Student in the Department of Electronica and Communication, Basaveshwar Engineering College,

Bagalkot (Autonomous)

<sup>1</sup>Email: <u>bdmallapur12@gmail.com</u>

<sup>2</sup>Email: <u>anuradha.math007@gmail.com</u>

Abstract : The patient care processes in hospitals are supported by a range of operational activities including inventory management and distribution of supplies to point of care locations. Hospitals carry large amounts and a great variety of items, and the issues of storing and distributing these items throughout the hospital supply chain are of great importance to providing high-quality patient service. Hence to provide more security to the supply chain system we proposed a model by integrating the blockchain with timestamp and location. To understand and improve the patient safety by combining

Key words: Block chain; Pharmaceutical supply chain: Barcode; Drug Transaction; Packet registry

# **Introduction:**

The pharmaceutical supply chain is a system through which the prescribed medicines are manufactured and delivered to the patients, a wide range of stakeholders are involved in the chain including manufacturers, wholesale distributors ,. Supply chain management in the pharma sector can refresh the organization to make better use of resources to gives rise profits. to boost shareholders value, and positively respond to the customers demand, lack of effective supply chain management can cause problem for organization. Pharmaceutical supply chain should provide medicines in the right quantity with desired quality to the right place and customers, at the right time with optimum cost . and also it should make benefits for its stockholders .when supply chain partners are connected, it allows them to work together which provides end to end visibility and an opportunity to interact with other business. And the complexity of this system requires of large scope approach and collaborations. To increase the efficiency of the pharmaceutical supply chain we embed the block chain. it can enable more transparent and accurate end to end tracking in the supply chain and has the potential to solve the accurate issues of data storage and security transaction processing. As a decentralised shared ledger that requires cryptographic signatures for access and modification, block chain can create transparent, traceable system of recording transactions across the supply chain .

# Security Issues :

# 1] Block chain technology in pharmaceutical industry to prevent counterfeit drugs

The proposed system is used to track the drugs from its manufacturing unit to patients, the effect of the drug on the patient has recorded to a database where only permissioned block chain is used. For implementation cryptographic alpha numeric key is used, but it has not given solution for security as counterfeit manufacturers uses reasonable brand logo.

# 2] A secure logistics model based on blockchain

The proposed secure logistic model analyses the security and efficiency of the model. a logistic block chain is build on decentralised and distributed platform using peer to peer network. python based Lbuntu Linux tool is used but it has not given overall security.

# A Cloud Interfaced Social Ski Driver Optimized Routing for Future Electric Motor Vehicles



Shivanand C. Hiremath and Jayashree D. Mallapur

Abstract India is revamping towards building smart cities along with collaboration of information and communication technologies as a future mode of transportation and with a vision of adapting electric motor vehicles (EMVs) more than 90% to make electric vehicle nation by 2030. With scarcity and increased floating prices of combustion fuels, it has procured enormous intrigue among various researchers as the EMVs are termed as an upcoming incredible mode of transportation. In addition, EMVs protect environment by satisfying go-green initiative and it is vital to protect conventional fuels and use renewable sources. However, the EMVs have shorter driving ranges, which are limited by inadequate charge storage in batteries, EMVs are economical compared to using gasoline but due to irregularly distributed charging stations causing lack of interest and hindrance among people in adoption of EMVs. This paper presents unique strategy to assist EMVs users through optimum routing directions to charging stations with ease of access by using cloud hosted on VANET. The goal is to locate the shortest and with least traffic routes for charging EMVs through the cloud-based vehicular ad hoc network (VANET) model. Here, optimum routes for acquiring the nearest charging station find out using proposed social ski driver (SSD) algorithm and a comparative analysis is done with particle swarm optimization (PSO) algorithm by considering traffic density, battery power and distance parameters. The experimental result of proposed SSD algorithm outperforms the PSO.

Keywords Electric motor vehicle (EMVs) · SSD · PSO · VANET · CS

S. C. Hiremath (🖂)

R. N. Shetty Polytechnic College, Belagavi, Karnataka, India

J. D. Mallapur Basaveshwar Engineering College, Bagalkot, Karnataka, India

© The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022 M. Tuba et al. (eds.), *ICT Systems and Sustainability*, Lecture Notes in Networks and Systems 321, https://doi.org/10.1007/978-981-16-5987-4\_39

385

D Springer Link

Search 📿 🍃

ICT Systems and Sustainability pp 385-395 | Cite as

# A Cloud Interfaced Social Ski Driver Optimized Routing for Future Electric Motor Vehicles

Authors Authors and affiliations

Shivanand C. Hiremath, Jayashree D. Mallapur

Conference paper First Online: 01 January 2022



Downloads

Part of the <u>Lecture Notes in Networks and Systems</u> book series (LNNS, volume 321)

# Abstract

India is revamping towards building smart cities along with collaboration of information and communication technologies as a future mode of transportation and with a vision of





Design, Simulation, and Development of DC-DC Converter for SPV-Powered DC Distribution System in Domestic Loads

# Basanagouda F. Ronad

Conference paper | <u>First Online: 20 March 2022</u> 341 Accesses

Part of the <u>Lecture Notes in Electrical Engineering</u> book series (LNEE,volume 844)

# Abstract

Solar photovoltaic (SPV) systems together with DC microgrids are promising option for remotely located domestic loads, and this combination has many advantages over conventional AC distribution system. This paper presents design, simulation, and development of single input multiple output (SIMO) DC-DC converter for SPV-powered DC distribution system employed for domestic loads. Simulation model is built for DC-DC converter with 24 V input and resulting 12, 24, 32, and 48 V DC output voltages. Regularly used domestic loads which basically operate on DC voltages are considered for

# Development of Smart Solar Tracker-An Approach towards Self Governed Tracking Mechanism

🖪 PDF

Publisher: IEEE

<< Results

Adithya Bhuti ; Jyoti Kadakolmath ; Basanagouda F. Ronad All Authors

Cite This

8<C=+ More Like This 22 Alerts Full Text Views Overview of maximum power point tracking technologies for photovoltaic power systems Manage Content Alerts IECON 2011 - 37th Annual Conference of the Add to Citation Alerts IEEE Industrial Electronics Society Published: 2011 Statistic and Parallel Testing Procedure for Abstract 노 Evaluating Maximum Power Point Tracking Downl Algorithms of Photovoltaic Power Systems Document PDF IEEE Journal of Photovoltaics Sections Published: 2013 Abstract: This paper presents a novel smart dual axis solar tracker to optimize I. Introduction the energy consumption in tracking mechanism. In conventional approach high Show More II. Review of torque motors are employ ... View more Tracking Mechanisms Metadata Abstract: III. Methodology This paper presents a novel smart dual axis solar tracker to optimize the energy IV. Experimental consumption in tracking mechanism. In conventional approach high torque Setup motors are employed to continuously hold and mobilize the solar PV panels, which lead to excess energy consumption by motors, resulting in less efficient V. Results and tracking systems. This issue is addressed by optimizing the panel rotation and Discussions detachment of motor shaft after directing the panel to sun's direction. The Show Full Outline tracker includes two parts; main SPV plant of 60 W and miniaturized tracker with 10 W panels. LDR Sensors are located on miniaturized panel to track the sun's direction and this tracking mechanism of miniature model is utterly Authors dependent on direction of irradiance only. Further, DC motor coupled to main SPV panels will rotate as per the signals of the miniaturized model and thus Figures miniaturized model acts as decision making unit. Comparative analysis of results revealed that, performance of the system has increased by 15-35% at References varying irradiance values. Keywords 2024 5+ Metrics Communication and Aerospace Technology (ICECA)

# **Optimization of Hybrid Renewable Electric Energy** System Components for Covid Hospitals

Publisher: IEEE

Cite This 🖪 PDF

<< Results

Jyothi Kumbar; Neelaganga Goudar; Sahana Bhavi; Shruti Walikar; Basanagouda F.... All Authors

**22** Full Text Views

Alerts Manage Content Alerts

Add to Citation Alerts

B <C ≥ ↓

More Like This

Multiobjective Optimization of System Configuration and Component Capacity in an AC Minigrid Hybrid Power System IEEE Transactions on Industry Applications Published: 2022

Analysis and Power Management Autonomous Hybrid Power AHPS) Using Biogeography Based on (BBO) Algorithm actions on Smart Grid 2013

Show More

Abstract		Economic A
Document Sections	Down! PDF	System (SA Optimizatio
I. Introduction	Abstract: This paper presents HOMER simulation models for optimizing renewable energy system components in Off grid and Grid connected systems	Published:
II. Resources Required for	for Covid-19 hospitals. Load profiles View more	
Modeling	▶ Metadata	
III. Homer	Abstract:	
Simulation	This paper presents HOMER simulation models for optimizing renewable	
Models and	energy system components in Off grid and Grid connected systems for Covid-	
Results	19 hospitals. Load profiles of 100 bed and 50 bed Covid hospitals are	
IV. Comparative	established critically. Two simulation models are built for each load profile.	
Analysis of the	Optimization of energy system components is conducted based on net present	
Results	cost and cost of energy. In process of optimization, HOMER simulates all	
V. Conduciono	possible configurations and ranks feasible combinations according to TNPC,	
v. Conclusions	and lists teasible options. Ten combinations are identified for analysis in each	
A	moder. The chucal analysis of results revealed that, grid connected system	
Authors	proves to be most economic and reliable option for hospital loads. The cost of	

energy obtained in grid connected system is Rs.9/kWh, whereas for diesel generated connected system costing Rs.22/kWh. This difference is appeared Figures mainly due to higher cost of diesel and variation in the solar and wind energy References potential with time of the day. Thus it is concluded that, renewable energy sources can be effectively employed for energizing the Covid-19 hospital in association with grid. Keywords Metrics

Published in: 2021 IEEE Mysore Sub Section International Conference Loading [MathJax]/extensions/MathMenuris

More Like This

# An Investigation on Torque Control Strategies for Switched Reluctance Motor

PDF

Publisher: IEEE

E Cite This

<< Results | Next >



Publisher: IEEE	Cite This 🔀 PDF		
Kirankumar Y. Bendig	eri;Jayashree D. Mallapur;S. B. Kumba	lavati All Authors	
		ø≮©⊨♠	
52 Full		Alerts	More Like This
Text Views		Manage Content Alerts	Energy Conservation in Progressive
		Add to Citation Alarte	Networks for Pervasive Computing
		And to oldation Altors	IEEE Systems Journal
Abstract			Published: 2017
	Downl		Energy Conservation Strategies in Wireless
Document	PDF		Sensor Networks: A Review
I. Introduction	Abstract: This paper discusses a nove	el node placement strategy to be applied in Wireless Sensor Network (WSN)	Trends in Informational Conference of Modern Trends in Information and Communication Technology Industry (MTICTI)
II. Literature Review	emphasizing over the energy eff Vie	w more	Published: 2021
III. Proposed Work	▶ Metadata		Show More
IV. Algorithm Implementation	Abstract: This paper discusses a novel node pla agriculture-based application in Wirele	acement strategy to be applied over the acs Sensor Network (WSN) emphasizing	
V. Result Analysis	over the energy efficiencies and increa	ased network lifetime. Most of the	
Show Full Outline	researches do not concentrate on nod parameter to estimate the performanc	le placement, which is first and critical e of network. The paper discusses the	
Authors	problems associated with conventiona research methodology, algorithm desc comparative performance analysis of	I random/grid deployment, motivation, cription. This paper finally discusses the proposed system of relay node placement	
Figures	simulation demonstrates that propose	de placement strategies in WSN. The d system maintains a good balance	
References	between communication and energy e	fficiency.	
Keywords	Published in: 2021 International Con Smart Systems (ICAIS)	ference on Artificial Intelligence and	
Metrics			
Mora Lika Thic	Date of Conference: 25-27 March	INSPEC Accession Number: 20633472	

# Performance Assessment of SPV Powered DC Irrigation Pumps with Solar Tracking Mechanism

🖹 PDF

Publisher: IEEE

Cite This

Basanagouda F. Ronad ; Santosh B. Kumbalavati All Authors

<b>37</b> Full Text Views	Alerts	
Text Views	Alerts	More Like This
	Manage Content Alerts Add to Citation Alerts	Energy, Economic, and Environmental (3E) Analysis of PV Water Pump: Case Study of the Tapioca Drip Irrigation System 2021 International Conference on Power, Energy and Innovations (ICPEI)
Abstract		Published: 2021
Document Sections I. Introduction	Downl PDF Abstract: This paper presents the performance of SPV DC irrigation pumps with and without color tracking mechanism. HOMER software is used for	Solar Powered Irrigation Pumps with Optimum Power Transfer 2019 IEEE PES GTD Grand International Conference and Exposition Asia (GTD Asia) Published: 2019
II. Resources for g	simulation. Simulation model is built View more	Show More
III. Different Methods for Solar Tracking	Metadata Abstract: This paper presents the performance of SPV DC irrigation pumps with and	
IV. Homer Simulation Model for 5HP DC Pump	without solar tracking mechanism. HOMER software is used for simulation. Simulation model is built for a 5 HP DC irrigation pump load. Local solar radiation data of Bagalkot, India are employed for the study. Developed HOMER model is standalone system with support of batteries. Constant DC load profile for source hour is employed for the applying. Further	
V. Comparision of Simulation Results Show Full Outline	total profile for seven nours/day is employed for the anarysis. Further, simulations are carried out with different types of solar tracking mechanisms. SPV panel tracking with Monthly, Weekly, Daily & Continuous Adjustment of Horizontal Axis, Continuous Adjustment of Vertical Axis and Two Axis Adjustment are employed to attain the simulation results. For each tracking	
Authors	method, optimal configurations of SPV panels & batteries are listed and various costs corresponding to initial capital, operating, net present and cost of energy, capacity shortage & electricity generation details are tabulated. Comparative analysis amongst different types tracking is taken up. Passills revealed that	
Figures	anarysis antongst unterent types tracking is taken up. Results revealed that tracking of panels will enhance power production and in particular Two Axis	
References	Adjustment method will yield highest amongst all the tracking mechanisms.	

# Chapter 37

A Well-Organized Safeguarded Access on Key Propagation by Malleable Optimization in Blend	
With Double Permutation	06
A. Sathish, M.A.M. College of engineering, Trichy-Chennai Trunk Road, Siruganur, India	
S. Ravimaran, M.A.M. College of engineering, Trichy-Chennai Trunk Road, Siruganur, India	
S. Jerald Nirmal Kumar, St. Annes College of Engineering and Technology, Anguchettypalyam, India	
Chapter 38	
Business Transaction Privacy and Security Issues in Near Field Communication	29
Chapter 39	
Fast Chaotic Encryption Using Circuits for Mobile and Cloud Computing: Investigations Under	
the Umbrella of Cryptography	48
Shalini Stalin, Aisect University, India	
Priti Maheshwary, Aisect University, India	
Piyush Kumar Shukla, University Instituter of Technology RGPV, India	
Akhilesh Tiwari, Madhav Institute of Technology, India	
Ankur Khare, M. K. Ponda College of Business and Management, India	
Chapter 40	
Secure Data Deduplication of Encrypted Data in Cloud	73
Sumit Kumar Mahana, National Institute of Technology Kurukshetra, India	
Rajesh Kumar Aggarwal, National Institute of Technology Kurukshetra, India	
Chapter 41	
A Quantum Key Distribution Technique Using Quantum Cryptography 8	90
Meenakshi Sharma Galgotias University India	70
Sonia Thind. Chandigarh University. India	
Chapter 42	
A Comprehensive Review of Access Control Mechanism Based on Attribute Based Encryption	
Scheme for Cloud Computing	99
Lokesh B. Bhajantri, Department of Information Science and Engineering, Basaveshwar	
Engineering College, Bagalkot, India Tabagayan N. Mujawan Baranga Adik Institute of Technology, Navi Mumbai, India	
Tabassum N. Mujawar, Ramrao Aaik Institute of Technology, Navi Mumbai, Inata	
Chapter 43	
Aggregate Searchable Encryption With Result Privacy	22
Dhruti P. Sharma, S.V. National Institute of Technology, Surat, India	
Devesh C. Jinwala, S.V. National Institute of Technology, Surat, India	
Charten 44	
Unapter 44 On Securing Cloud Storage Using a Homomorphic Framework	16
Daya Sagar Gupta Indian Institute of Technology Dhanhad India	40
G. P. Biswas. Indian Institute of Technology Dhanbad. India	
G. I. Diswas, maian institute of Technology Dhandaa, mata	

# Chapter 42 A Comprehensive Review of Access Control Mechanism Based on Attribute Based Encryption Scheme for Cloud Computing

# Lokesh B. Bhajantri

b https://orcid.org/0000-0002-3947-4292

Department of Information Science and Engineering, Basaveshwar Engineering College, Bagalkot, India

# Tabassum N. Mujawar

Ramrao Adik Institute of Technology, Navi Mumbai, India

# ABSTRACT

Cloud computing is the most prevailing paradigm, which provides computing resources and services over the Internet. Due to immense development in services provided by cloud computing, the trend to share large-scale and confidential data on cloud has been increased. Though cloud computing provides many benefits, ensuring security of the data stored in cloud is the biggest challenge. The security concern about the data becomes main barrier for adoption of cloud. One of the important security aspects is fine grained access control mechanism. The most widely used and efficient access control scheme for cloud computing is Attribute Based Encryption (ABE). The Attribute Based Encryption (ABE) scheme provides a new technique for embedding access policies cryptographically into encryption process. The article presents an overview of various existing attribute-based encryption schemes and traditional access control models. Also, the comparison of existing ABE schemes for cloud computing, on basis of various criteria is presented in the article.

DOI: 10.4018/978-1-7998-7705-9.ch042

# INTRODUCTION

In today's era, cloud computing has become an attracting technology, which has brought extreme changes to IT industry. Cloud Computing enables network access to various computing resources such as servers, storage, networks, applications and services. It is basically a paradigm that provides access to shared pool of resources online on demand (Armbrust et al., 2010). It is computing over internet. The users can store any amount of data on cloud and then can access it at any time, from anywhere. The most fascinating benefit of cloud computing is that it provides cost saving as the users will pay only for their usage. Cloud computing is also termed as distributed computing over a network. The term cloud can be defined as collection of servers delivering computing resources as a service on demand. Generally, the cloud comprises various interfaces, networks, hardware, storage devices etc. (Majumder et al., 2014).

Security is the biggest barrier for adoption of cloud computing. The data is present in shared environment, where other users can also access it. The users do not have complete control over the transit of data stored in cloud as both users and data are present in different domain. Hence the privacy concern arises for user's data and many users cannot completely trust the cloud environment. (KPMG, 2010; Hashizume et al., 2013, Tabassum et al., 2017). The security challenges can be faced at different levels such as architectural level, communication level and contractual and legal level (Ali et al., 2015). After adoption of cloud computing the organization cannot apply traditional security mechanism such as authentication, authorization in similar way as they exist. The reason is that the security requirements of organization with cloud environment are very much different than the traditional organization (Li and Ping, 2009).

Generally, in cloud environment users share their sensitive data with other users. In order to access the data, user must possess necessary permissions or credentials. Access control is a mechanism, which decides who can use a specific system, resource or application. It defines a way to allow, deny or restrict user access to system or its resources (Khan, 2012). Access control mechanism ensures that data must be accessed by authorized users only. In cloud environment, the owner of data and the data are present in different administrative domains. Thus, it is extremely important to ensure authorized access to data and manage user's identity. Due to the distributive and dynamic nature of cloud computing access control becomes very complex task. In traditional method the data is stored on some third-party server and access control mechanism is employed statically. However, this method does not guarantee the confidentiality of data because the server storing the data can be un-trusted entity. Therefore, providing better access control mechanism is a very important component of cloud security (Majumder et al., 2014). The main goal of access control is to restrict user to access what he/she should be able to do and prevent unauthorized access. The access control is defined as a mechanism to determine correct access to data by legitimate user depending on access privileges and permissions that are already defined in security policies (Younis et al., 2014). The major endeavor of this paper is to present brief overview of access control mechanisms used for cloud computing. Here, the classification of access control models applied for cloud computing that includes some traditional models and various Attribute Based Encryption schemes is elaborated.

The classification of access control models for cloud computing includes two categories as traditional models and the models based on cryptographic approaches. The taxonomy of access control models applied for cloud computing is depicted in Figure 1. The Discretionary Access Control (DAC), Mandatory Access Control (MAC) and Role based Access Control (RBAC) model come under traditional access control models.
### ACADEMIC YEAR 2020-21

### Shilpa K Jigajinni<sup>1</sup> and Dr.Bharati S Meti<sup>1</sup>

<sup>1</sup>Department of Biotechnology, Basaveshwar Engineering College (Autonomous), S Nijalingappa Vidyanagar, Bagalkot 587103, Karnataka.\* Corresponding author: skjbt@becbgk.edu 8722485018

### Abstract

Lipase is one of the promising biocatalyst in field of Enzymatic Biodiesel production. Screening of lipase producing bacteria and its media optimization facilitates enhanced lipase production. Considering the advantages of enzymatic biodiesel production using lipases the objective of present work was to isolate lipase producing bacteria and its media optimization. Bacterial strain were isolated from oil rich sample showing maximum zone of hydrolysis on TBA Media was selected and identified as Lysinibacillus macroides FS1 by 16SrDNA gene sequence analysis. Lipase production was carried out in production media and lipase activity of 3.1U/ ml was assayed by titrimetric method. Various media parameters were optimized and maximal lipase activity of 16.75U/ml were observed at 48 hr of incubation in optimized media with Honge oil as inducer, Galactose as carbon source, Ammonium Chloride and Beef extract as source of nitrogen, pH 7 and 37°C temperature with agitation speed of 120 rpm. Under optimized condition 5.4 fold increases in the lipase activity than innate activity was observed. 50.74% yield was observed with specific activity of 21.25U/mg after purification with 30% ammonium sulphate. Purified lipase was stable in presence of methanol at 37°C. Results obtained shows that Lipase from Lysinibacillus macroides FS1 was promising biocatalyst for Biodiesel production.

**Keywords** Isolation, *Lysinibacillus macroides* FS1, Lipase, Optimization, Biodiesel.

### Introduction

Lipases (Triacylglycerol acyl hydrolases, E.C. 3.1.1.3) are group of hydrolases which catalyses the hydrolysis of triacylglycerol to release fatty acids and glycerol. Besides the hydrolytic reactions, lipases are able to catalyze many reactions like transesterification, interesterification, esterification, acidolysis, aminolysis, alcoholysis by using wide-range of substrates with high specificity (1,2). Lipase is believed to be very versatile biocatalyst and potential enzyme for the biotechnology industries becoming a target of several researches. Lipases are used in many different industrial applications, such as household detergent additive, textile industry, cosmetic industry, paper industry, pharma industry, synthesis of biopolymers and biodiesel, agrochemicals, flavor compounds and pharmaceutical compound

synthesis etc (3,4). The study of the enzymatic synthesis of biodiesel was encouraged due to increasing pollution and by product separation problems of the chemical catalyzed process. (5).

More research focus is required to isolate industrial important lipase producing microorganisms from different habitats. Lipases are widely employed group of biocatalyst in different biological processes because of their broad substrate specificity and stereo selectivity in reaction (6).Lipases are produced from different species of plants, animals and microorganisms and they exist in different environments like compost heaps, decaying food, dairies, industrial wastes, oil-processing factories, oil seeds, soil contaminated with oil, waste water and petrol spilled soil (7, 8). The oil rich environment with organic substrates may furnish a suitable habitat for isolation of lipase producing bacteria (9). Lipase producers can be distinguished by showing a clear halo zone, turbid zone or white crystals around colonies of microbes on specific media such as Tributyrin agar(TBA), Phenol red agar, Tween20/80, Rhodamine B agar plates containing lipidic compounds like triolein, tributyrin, olive oil, tween 20, tween 80 etc. Microbial lipases occupied more industrial applications than plants and animal sources because of its hydrolytic and synthetic activities, high yield, ease in genetic manipulation, regular supply and easy cultivation of microbes on inexpensive media (10). Among microbial lipases bacterial lipases are more economical and stable (3). Most of the Bacterial species secretes lipase extracellular which are majorly affected by nutritional and physicochemical factors such as temperature, pH, nitrogen source and carbon source, inorganic salts, agitation and DO concentration (11,12,13,14).Media containing lipidic carbon, like edible and non-edible oils, fatty acids, glycerol or tweens in the presence of an organic nitrogen source facilitates the lipase production(15). Among many species some of them are efficient extracellular lipase producers mainly Achromobacter, Alcaligenes, Arthrobacter, Pseudomonas, Staphylococcus, Chromobacterium, Bacillus and Pseudomonas spp. (7). Few analyses have been shown the abilities of Lysinibacillus genera with several enzymatic activities including lipolytic, proteolytic, cellulolytic, pectinolytic and amylolytic activities (16). Solvent stable lipases are one of the promising biocatalysts in non-aqueous biological transformations hence focus need to isolate



### Handbook of Research on Emerging Trends and Applications of Machine Learning

Arun Solanki (Gautam Buddha University, India), Sandeep Kumar (Amity University, Jaipur, India) and Anand Nayyar (Duy Tan University, Da Nang, Vietnam)

Release Date: December, 2019 Copyright: © 2020 Pages: 674 DOI: 10.4018/978-1-5225-9643-1 ISBN13: 9781522596431 ISBN10: 1522596437 EISBN13: 9781522596455

Hardcover:	\$415.00
(/ISI @kkitkabidbook-research-emerging-trends-applications/222603?f=hardcover) Benefits & Incentives	~
E-Book: (Multi-User License)	\$373.50 List Price: <del>\$415.00</del>
(/Is division book-research-emerging-trends-applications/222603?f=e-book) Benefits & Incentives	~
Hardcover + E-Book: (Multi-User License)	\$495.00
(/I@div#iabidbook-research-emerging-trends-applications/222603?f=hardcover-e-book) Benefits & Incentives	~
OnDemand: (Individual Chapters)	\$37.50
(/Ige/w/#iablebook-research-emerging-trends-applications/222603#table-of-contents) Benefits & Incentives	~

### **Description & Coverage**

#### **Description:**

As today's world continues to advance, Artificial Intelligence (AI) is a field that has become a staple of technological development and led to the advancement of numerous professional industries. An application within AI that has gained attention is machine learning. Machine learning uses statistical techniques and algorithms to give computer systems the ability to understand and its popularity has circulated through many trades. Understanding this technology and its countless implementations is pivotal for scientists and researchers across the world.

The Handbook of Research on Emerging Trends and Applications of Machine Learning provides a high-level understanding of various machine learning algorithms along with modern tools and techniques using Artificial Intelligence. In addition, this book explores the critical role that machine learning plays in a variety of professional fields including healthcare, business, and computer science. While highlighting topics including image processing, predictive analytics, and smart grid management, this book is ideally designed for developers, data scientists, business analysts, information architects, finance agents, healthcare professionals, researchers, retail traders, professors, and araduate students seeking current research on the benefits, implementations, and trends

https://www.igi-global.com/book/handbook-research-emerging-trends-applications/222603#table-of-contents

Jayakumar Kaliappan, Karpagam S	Sundararajan	
Machine learning is a part of artific	al intelligence in which the learning was done using the data available in the environment.	
Machine learning Sample PDF (/viewtitlesample.aspx?id	i=247575&ptid=222603&t=Machine Learning in Video Games&isxn=9781522596431)	
Chapter 21		\$37.5
Multi-Layer Hybrid Credit Scoring	Jodel Based on Feature Selection, Ensemble Learning, and Ensemble Classifier	
(/chapter/multi-layer-hybrid-credit-s	coring-model-based-on-feature-selection-ensemble-learning-and-ensemble-classifier/247576)	
(pages 444-460)		
Diwakar Tripathi, Alok Kumar Shuk	la, Ramchandra Reddy B., Ghanshyam S. Bopche	
Credit scoring is a process to calcu industry	late the risk associated with a credit product, and it directly affects the profitability of that	
Sample PDF (/viewtitlesample.aspx?ic Ensemble Learning, and Ensemble C	i=247576&ptid=222603&t=Multi-Layer Hybrid Credit Scoring Model Based on Feature Selection, assifier&isxn=9781522596431)	
Chapter 22		\$37.50
Overview of Machine Learners in C	Sassifying of Speech Signals (/chapter/overview-of-machine-learners-in-classifying-of-speech-	
signals/247577) (pages 461-489)		
Hemanta Kumar Palo, Lokanath S	arangi	
Machine learning (ML) remains a b	uzzword during the last few decades due to the requirement of a huge amount of data for	
adequate processing, the		
Sample PDF (/viewtitlesample.aspx?id Signals&isxn=9781522596431)	I=247577&ptid=222603&t=Overview of Machine Learners in Classifying of Speech	
Chapter 23		\$37.50
Self-Driving Cars: Role of Machine	Learning (/chapter/self-driving-cars/247578) (pages 490-507)	
Pooja Jha, K. Sridhar Patnaik		
Human errors are the main cause	of vehicle crashes. Self-driving cars bear the promise to significantly reduce accidents by taking	
the human factor Sample PDF (/viewtitlesample.aspx?id	d=247578&ptid=222603&t=Self-Driving Cars: Role of Machine Learning&isxn=9781522596431)	
Chapter 24		\$37.50
Sentiment Analysis on Social Medi (pages 508-527)	a: Recent Trends in Machine Learning (/chapter/sentiment-analysis-on-social-media/247579)	
Ramesh S. Wadawadagi, Veerapp	a B. Pagi	
Due to the advent of Web 2.0, the	size of social media content (SMC) is growing rapidly and likely to increase faster in the near	
future. Social		
Sample PDF (/viewtitlesample.aspx?id Learning&isxn=9781522596431)	=247579&ptid=222603&t=Sentiment Analysis on Social Media: Recent Trends in Machine	
Chapter 25		\$37.50
Social Big Data Mining: A Survey F	ocused on Sentiment Analysis (/chapter/social-big-data-mining/247580) (pages 528-549)	
Anisha P. Rodrigues, Niranjan N. C	hiplunkar, Roshan Fernandes	
Social media is used to share the o	lata or information among the large group of people. Numerous forums, blogs, social networks,	
news reports		
Sample PDF (/viewtitlesample.aspx?id Analysis&isxn=9781522596431)	=247580&ptid=222603&t=Social Big Data Mining: A Survey Focused on Sentiment	
Chapter 26		\$37.50
Tool Condition Monitoring Using Ar models/247581) (pages 550-576)	tificial Neural Network Models (/chapter/tool-condition-monitoring-using-artificial-neural-network-	
Srinivasa P. Pai, Nagabhushana T.	Ν.	
Tool wear is a major factor that affe	cts the productivity of any machining operation and needs to be controlled for achieving	
automation. It affects		
Sample PDF (/viewtitlesample.aspx?id Models&isxn=9781522596431)	I=247581&ptid=222603&t=Tool Condition Monitoring Using Artificial Neural Network	
Chapter 27		\$37.50
Chapter 27 Computer Vision-Based Assistive	echnology for Helping Visually Impaired and Blind People Using Deep Learning Framework	\$37.50

Mohamamd Farukh Hashmi, Vishal Gupta, Dheeravath Vijay, Vinaybhai Rathwa

Millions of people in this world can't understand environment because they are blind or visually impaired. They also have navigation...

Sample PDF (/viewtitlesample.aspx?id=247582&ptid=222603&t=Computer Vision-Based Assistive Technology for Helping Visually Impaired and Blind People Using Deep Learning Framework&isxn=9781522596431)

### Chapter 24 Sentiment Analysis on Social Media: Recent Trends in Machine Learning

### Ramesh S. Wadawadagi

b https://orcid.org/0000-0002-6669-7344 Basaveshwar Engineering College, Bagalkot, India

Veerappa B. Pagi Basaveshwar Engineering College, Bagalkot, India

### ABSTRACT

Due to the advent of Web 2.0, the size of social media content (SMC) is growing rapidly and likely to increase faster in the near future. Social media applications such as Instagram, Twitter, Facebook, etc. have become an integral part of our lives, as they prompt the people to give their opinions and share information around the world. Identifying emotions in SMC is important for many aspects of sentiment analysis (SA) and is a top-level agenda of many firms today. SA on social media (SASM) extends an organization's ability to capture and study public sentiments toward social events and activities in real time. This chapter studies recent advances in machine learning (ML) used for SMC analysis and its applications. The framework of SASM consists of several phases, such as data collection, pre-processing, feature representation, model building, and evaluation. This survey presents the basic elements of SASM and its utility. Furthermore, the study reports that ML has a significant contribution to SMC mining. Finally, the research highlights certain issues related to ML used for SMC.

### **OVERVIEW**

In recent days, social media applications have emerged as leading mass media, as they allow users to work collaboratively and publish their content (Wadawadagi & Pagi, in press; Anami et al. 2014). Accordingly, large volumetric semantically rich information is being generated and accumulated every day in the form of tweets, posts, blogs, news, comments, reviews, etc. Investigating hidden but potentially useful patterns

DOI: 10.4018/978-1-5225-9643-1.ch024

Advances in Intelligent Systems and Computing 1176

Vikrant Bhateja Sheng-Lung Peng Suresh Chandra Satapathy Yu-Dong Zhang *Editors* 

### Evolution in Computational Intelligence

Frontiers in Intelligent Computing: Theory and Applications (FICTA 2020), Volume 1



### Optimal Image Feature Ranking and Fusion for Visual Question Answering

Sruthy Manmadhan, Binsu C. Kovoor Pages 103-113

### An Investigation on Indoor Navigation Systems

J. Akilandeswari, A. Naveenkumar, R. S. Sabeenian, P. Iyyanar, M. E. Paramasivam, G. Jothi Pages 115-124

### Conceptualization and Design of Remotely-Accessible Hardware Interface (RAHI) Laboratory

Shivam Mahesh Potdar, Vanshika Gupta, Pruthviraj Umesh, K. V. Gangadharan Pages 125-134

### A Non-invasive approach for Driver Drowsiness Detection using Convolutional Neural Networks

K. K. Sreelakshmi, J. Jennifer Ranjani Pages 135-143

### Disaster Severity Analysis from Micro-Blog Texts Using Deep-NN

Ramesh Wadawadagi, Veerappa Pagi Pages 145-157

### WEKA Result Reader—A Smart Tool for Reading and Summarizing WEKA Simulator Files

Ranjit Panigrahi, Samarjeet Borah, Udit Kumar Chakraborty Pages 159-167

### Disaster Severity Analysis from Micro-Blog Texts Using Deep-NN



Ramesh Wadawadagi and Veerappa Pagi

Abstract The current decade has witnessed a significant amount of research in the field of sentiment analysis (SA). Several applications have emerged to evidence the necessity of research in this area. On the contrary, the size of micro-blogs content is overgrowing and likely to increase even faster shortly. Social media applications have become part and parcel of our daily lives, as they urge the public to express their opinions and share information around the world. Especially during disasters, people are likely to utilize social media to communicate their hindrances. In this article, we investigate the severity of disaster events from micro-blog messages posted by people during natural calamities and emergencies using deep learning techniques. In particular, the work employs a joint model that combines the features of convolutional neural networks (CNN) with recurrent neural networks (RNN), taking account of the coarse-grained local features generated via CNN and long-range dependencies learned through RNN for analysis of small text messages. Furthermore, the proposed model is evaluated for both binary and fine-grained analyses tested over two different datasets. The accuracy of 87% is observed for binary classification and up to 65% for a three-class problem. The intended work finds usefulness in many instants of disaster relief and crisis management.

**Keywords** Deep learning · Sentiment analysis · Convolutional neural networks · Recurrent neural networks

### **1** Introduction

Recently, micro-blog sites have emerged as leading mass media platform, as users are authorized to work collectively and publish their content [1]. Consequently, a large volumetric and semantically rich information is being generated and accumu-

V. Pagi e-mail: veereshpagi@gmail.com

R. Wadawadagi (🖂) · V. Pagi

Basaveshwar Engineering College, Baglkot 587102, India e-mail: rswlib@yahoo.co.in

<sup>©</sup> The Editor(s) (if applicable) and The Author(s), under exclusive license to Springer 145 Nature Singapore Pte Ltd. 2021 V. Bhateja et al. (eds.), *Evolution in Computational Intelligence*, Advances in Intelligent Systems and Computing 1176,

https://doi.org/10.1007/978-981-15-5788-0\_14



Scie	enceDirect
Den Mener Manager Mener Mener Manager Mener Mener M	nystifying Big Data, Machine Learning, and Deep Learning for Healthcare lytics
Book • 2021	
Edited by: Pradeep N, Sand	leep Kautish and Sheng-Lung Peng
Browse	book content
About t	he book
Search	in this book
	Search in this book
Table of c	ontents
• Full text access Front Matter,	Copyright, Dedication, Contributors, Editors biography, Foreword, Preface, Overview
> Section 1:	Big data in healthcare analytics
✓ Section 2:	Machine learning and deep learning for healthcare
Book chapter O A	bstract only
Chapter 7 - A	comprehensive review on deep learning techniques for a BCI-based communication system
M. Bhuvaneshwa Pages 131-157	ari, E. Grace Mary Kanaga, S. Thomas George
Purchase	View abstract 🗸
Book chapter O A	bstract only
Chapter 8 - Cl	inical diagnostic systems based on machine learning and deep learning
Sanjeevakumar I Pages 159-183	M. Hatture and Nagaveni Kadakol
Purchase	View abstract 🗸
Book chapter O A Chapter 9 - An	<sup>bstract only</sup> n improved time-frequency method for efficient diagnosis of cardiac arrhythmias
Sandeep Raj Pages 185-213	
Purchase	View abstract 🗸
Book chapter 0 A Chapter 10 - I	bstract only Local plastic surgery-based face recognition using convolutional neural networks
Roshni Khedgao Pages 215-246	nkar, Kavita Singh and Mukesh Raghuwanshi
Purchase	View abstract 🗸
Book chapter O A Chapter 11 - I	bstract only Machine learning algorithms for prediction of heart disease
Rashmi Rachh, S Pages 247-275	Shridhar Allagi and B.K. Shravan

FEEDBACK 🖓

### Identification of Intra-abdominal Organs Using Deep Learning Techniques

Sanjeevakumar M. Hatture <sup>1</sup>⊠

Email smhatture@gmail.com

Nagaveni Kadakol <sup>1</sup>

Email n\_kadakol@rediffmail.com

<sup>1</sup> Basaveshwar Engineering College (Autonomous), Bagalkot, Karnataka, 587103, India

### Abstract

Presently deep learning techniques are playing important role in making healthcare systems more intelligent, efficient, and effective. Proposed methodology is an advisory system in medical imaging which helps in clinical diagnosis. In medical imaging, ultrasound imaging is most frequently used as it is safe, painless, not exposed to ionizing radiation, and it allows real-time imaging. Ultrasound imaging takes more time in diagnosis and well-trained radiologist for interpreting and understanding. Hence, proposed system acts as an advisory system in identifying intra-abdominal organs and abnormalities if any. In this proposed system, the data was collected from intra-abdominal ultrasound images that do not contain any exploring information about the patient. Using filters, noise in ultrasound images is removed. Organ is segmented from ultrasound image and is identified by using deep neural network, and using shape and texture features, abnormalities are identified if any. At the end, various challenges that exist with deep neural network and ultrasound images are discussed.

AQ1

### Keywords

Intra-abdominal organs Deep learning Ultrasound image Feature extraction Segmentation

1/11

Advances in Intelligent Systems and Computing 1270

Milan Tuba Shyam Akashe Amit Joshi *Editors* 

## ICT Systems and Sustainability

### Proceedings of ICT4SD 2020, Volume 1



Х

### Contents

Automated Dietary Monitoring System: A Novel Framework Samiul Mamud, Punyasha Chatterjee, Saubhik Bandyopadhyay, and Suchandra Bhandari	535
Intruder Insinuation and Smart Surveillance Using Face Detectionand Mask DetectionM. Rajesh Khanna, G. Prakash Raj, S. Prem Kumar,and N. S. Vignesh Raaj	543
Misinformation Analysis During Covid-19 Pandemic	553
An Agent-Based Cloud Service Negotiation in Hybrid Cloud Computing	563
IoT-Based Smart Farming Application for Sustainable Agriculture Sanieevakumar M. Hatture and Pallavi V. Yankati	573
Follow Me: A Human Following Robot Using Wi-Fi Received         Signal Strength Indicator         V. Geetha, Sanket Salvi, Gurdeep Saini, Naveen Yadav,         and Rudra Pratap Singh Tomar	585
<b>Implementation of Arithmetic Unit for RNS Using <math>2^n</math>+ 3 as Base</b> Nagaraj R. Aiholli, Uday V. Wali, and Rashmi Rachh	595
Hybridization of Artificial Bee Colony Algorithm with Estimationof Distribution Algorithm for Minimum Weight DominatingSet ProblemSima Shetgaonkar and Alok Singh	607
Privacy Preservation for Data Sharing with Multi-party Access Control in Cloud Computing: A Review Paper Shubhangi Tanaji Khot, Amrita A. Manjrekar, and Rashmi J. Deshmukh	621
<b>Pixel-Based Attack on ODENet Classifiers</b>	629
Survey on Advanced Spectrum Sharing Using Cognitive Radio Technique	639
An AI-Based Pedagogical Tool for Creating Sketched Representation of Emotive Product Forms in the Conceptual Design Stages Sunny Prakash Prajapati, Rahul Bhaumik, Tarun Kumar, and Unais Sait	649

### IoT Based Smart Farming Application For Sustainable Agriculture

Sanjeevakumar M. Hatture<sup>1</sup> and Pallavi V. Yankati<sup>2</sup>

<sup>1</sup>Basaveshwar Engineering College (Autonomous), Bagalkot – 587103, Karnataka, India smhatture@gmail.com

<sup>2</sup>Basaveshwar Engineering College (Autonomous), Bagalkot – 587103, Karnataka, India pallavivijay3575@gmail.com

Abstract. The farmers are struggling to obtain higher rate of yield due to lack of poor knowledge about the soil and water nutrients and suitability of the crop for the soil. To overcome these issues and to promote Digital Agriculture concept we propose an IoT enabled sensor system for monitoring soil nutrient [NPK] and pH of irrigation water to reduce the manual laboratory method of testing and get the results via mobile application. Smart contract farming based mobile application will further process these nutrients value to predict and suggests the suitable crop to grow and the usage of appropriate amount of fertilizer to increase the soil fertility. The mobile application also helps in establishing the connection between farmers and Agricultural Produce Market Committee (APMC) in order to avoid fragmentation of profit shares and attain Pricing uncertainty and marketing of the yields by avoiding the middle man. APMC is a state government body which ensures safeguard to the farmers from exploitation by large retailers. Thus, the proposed work helps the farmers in adopting stress-free farming practice by signing the contract offered from APMC to select the suitable crop for their agriculture land, dispensing the required amount of fertilizer to the soil based on soil and water nutrients analysis using IoT enabled sensor, funding/insurance to the crops in case of occurrence of unpredictable natural disaster in future and direct marketing facility without middle man.

**Keywords:** Sustainable agriculture, Smart contract farming, Soil and Water Nutrients, IoT, APMC, Crop insurance.

### **1** INTRODUCTION

India is a country where agriculture is given a lot of importance as it is the backbone of the Indian economy. It provides employment opportunity to more than half of the work-force of the country and also contributes 16.5% to total gross domestic product (GDP), and also. Earlier days there was no scarcity for food and water as it was used in efficient manner. But now a day there is scarcity of food and water due to increase in population growth of the country. To satisfy their needs fertilizers are used to increase crop production. Fertilizer is any substance which is added to soil for supplying nutrients externally which are crucial to plant growth. Too much use of fertilizers shatters the growth of crops and little usage will result in less yield therefore it is necessary for farmers to understand the correct usage of fertilizer to increase the soil fertility.

Nutrients are important factors for plant growth and are divided into micro and macro nutrients. Totally soil consists sixteen nutrients among them the three primary nutrients Lecture Notes on Data Engineering and Communications Technologies 61

Harish Sharma Mukesh Saraswat Sandeep Kumar Jagdish Chand Bansal *Editors* 

# Intelligent Learning for Computer Vision

Proceedings of Congress on Intelligent Systems 2020



### Contents

Development of Inter-ethnic Harmony Search Algorithm Based on Inter-ethnic Reconciliation	1
Hyun Woo Jung, Young Hwan Choi, Donghwi Jung, and Joong Hoon Kim	
A Low-Cost Embedded Computer Vision System for the Classification of Recyclable Objects	11
An Optimal Feature Selection Approach Using IBBO for Histopathological Image Classification Mukesh Saraswat, Raju Pal, Roop Singh, Himanshu Mittal, Avinash Pandey, and Jagdish Chand Bansal	31
Accuracy Evaluation of Plant Leaf Disease Detection and Classification Using GLCM and Multiclass SVM Classifier K. Rajiv, N. Rajasekhar, K. Prasanna Lakshmi, D. Srinivasa Rao, and P. Sabitha Reddy	41
A Deep Learning Technique for Automatic Teeth Recognition in Dental Panoramic X-Ray Images Using Modified Palmer Notation System Fahad Parvez Mahdi and Syoji Kobashi	55
Detection of Parkinson's Disease from Hand-Drawn Images Using Deep Transfer Learning Akalpita Das, Himanish Shekhar Das, Arijeet Choudhury, Anupal Neog, and Sourav Mazumdar	67
An Empirical Analysis of Hierarchical and Partition-Based Clustering Techniques in Optic Disc Segmentation	85

Contents
----------

Multi-class Support Vector Machine-Based Household Object Recognition System Using Features Supported by Point Cloud Library	97
Smita Gour, Pushpa B. Patil, and Basavaraj S. Malapur	
Initialization of MLP Parameters Using Deep Belief Networks for Cancer Classification	109
Barış Dinç, Yasin Kaya, and Serdar Yıldırım	
An Improved Inception Layer-Based Convolutional Neural Networkfor Identifying Rice Leaf DiseasesB. Baranidharan, C. N. S. Vinoth Kumar, and M. Vasim Babu	119
Design and Implementation of Traffic Sign Classifier Using Machine	
Learning Model	131
<b>Designing Controller Parameter of Wind Turbine Emulator</b> <b>Using Artificial Bee Colony Algorithm</b> Ajay Sharma, Harish Sharma, Ashish Khandelwal, and Nirmala Sharma	143
Text Document Orientation Detection Using ConvolutionalNeural Networks	153
A Deep Learning-Based Segregation of Housing Image Data for Real Estate Application Annu Kumari, Vinod Maan, and Dhiraj	165
Improved Image Super-resolution Using Enhanced GenerativeAdversarial Network a Comparative StudyB. V. Balaji Prabhu and Omkar Subburao Jois Narasipura	181
Comparative Study of Supervised Machine Learning Algorithms for Healthcare Dataset Using Orange Vaibhav Bhatnagar and Ramesh C. Poonia	195
Maximum Power Point Tracking of Photovoltaic SystemUsing Artificial Neural NetworkKusum Lata Agarwal and Shubham Sharma	209
<b>IoT Security: A Survey of Issues, Attacks and Defences</b>	219
Detecting the Nuclei in Different Pictures Using Region Convolutional Neural Networks Naiswita Parmar	237
Chaotic Henry Gas Solubility Optimization Algorithm Nand Kishor Yadav and Mukesh Saraswat	247

### Multi-class Support Vector Machine-Based Household Object Recognition System Using Features Supported by Point Cloud Library



Smita Gour, Pushpa B. Patil, and Basavaraj S. Malapur

**Abstract** The proposed system aims to design and develop an object recognition system with the help of the Point Cloud Library (PCL). The object recognition problem is addressed with a three-stage mechanism. In the initial stage using PCL, the object image undergoes segmentation methods which make the image suitable for extracting the features. In the second stage, a segmented image is used to extract suitable shape-based features that can separate each object type. The last stage involves the classification/recognition of the object of the particular type using support vector machine (SVM). The system reached the expected results using Point Cloud Library (PCL) and support vector machine (SVM) as a classifier. The system has given 94% accuracy for 10 different household object types. Ten samples for each object type are used for training the SVM, and to perform testing, complete different five samples from training samples are considered.

Keywords Object recognition · Point cloud library · Support vector machine

### 1 Introduction

Object recognition is a major research area in the field of computer vision since the appearance of objects is different in its size, shape, etc. due to the varying distance between the camera and the object. Also, there is a drastic change in its visual parameters due to the use of different sources of illumination and capturing conditions during image acquisition. The main goal of the object recognition system is to make

Basaveshwar Engineering College, Bagalkot 587102, Karnataka, India e-mail: smita.gour@gmail.com

B. S. Malapur e-mail: basu.malapur@gmail.com

P. B. Patil BLDEA's V P Dr PG Halakatti College of Engineering & Technology, Vijayapura 586101, Karnataka, India e-mail: pushpa\_ms@rediffmail.com

© The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2021 H. Sharma et al. (eds.), *Intelligent Learning for Computer Vision*, Lecture Notes on Data Engineering and Communications Technologies 61, https://doi.org/10.1007/978-981-33-4582-9\_8 97

S. Gour (🖂) · B. S. Malapur

Lecture Notes on Data Engineering and Communications Technologies 61

Harish Sharma Mukesh Saraswat Sandeep Kumar Jagdish Chand Bansal *Editors* 

# Intelligent Learning for Computer Vision

Proceedings of Congress on Intelligent Systems 2020



### Contents

Development of Inter-ethnic Harmony Search Algorithm Based on Inter-ethnic Reconciliation	1
Hyun Woo Jung, Young Hwan Choi, Donghwi Jung, and Joong Hoon Kim	
A Low-Cost Embedded Computer Vision System for the Classification of Recyclable Objects	11
An Optimal Feature Selection Approach Using IBBO for Histopathological Image Classification Mukesh Saraswat, Raju Pal, Roop Singh, Himanshu Mittal, Avinash Pandey, and Jagdish Chand Bansal	31
Accuracy Evaluation of Plant Leaf Disease Detection and Classification Using GLCM and Multiclass SVM Classifier K. Rajiv, N. Rajasekhar, K. Prasanna Lakshmi, D. Srinivasa Rao, and P. Sabitha Reddy	41
A Deep Learning Technique for Automatic Teeth Recognition in Dental Panoramic X-Ray Images Using Modified Palmer Notation System Fahad Parvez Mahdi and Syoji Kobashi	55
Detection of Parkinson's Disease from Hand-Drawn Images Using Deep Transfer Learning Akalpita Das, Himanish Shekhar Das, Arijeet Choudhury, Anupal Neog, and Sourav Mazumdar	67
An Empirical Analysis of Hierarchical and Partition-Based Clustering Techniques in Optic Disc Segmentation	85

Contents
----------

Multi-class Support Vector Machine-Based Household Object Recognition System Using Features Supported by Point Cloud Library	97
Smita Gour, Pushpa B. Patil, and Basavaraj S. Malapur	
Initialization of MLP Parameters Using Deep Belief Networks for Cancer Classification	109
Barış Dinç, Yasin Kaya, and Serdar Yıldırım	
An Improved Inception Layer-Based Convolutional Neural Networkfor Identifying Rice Leaf DiseasesB. Baranidharan, C. N. S. Vinoth Kumar, and M. Vasim Babu	119
Design and Implementation of Traffic Sign Classifier Using Machine	
Learning Model	131
<b>Designing Controller Parameter of Wind Turbine Emulator</b> <b>Using Artificial Bee Colony Algorithm</b> Ajay Sharma, Harish Sharma, Ashish Khandelwal, and Nirmala Sharma	143
Text Document Orientation Detection Using ConvolutionalNeural NetworksShivam Aggarwal, Safal Singh Gaur, and Manju	153
A Deep Learning-Based Segregation of Housing Image Data for Real Estate Application Annu Kumari, Vinod Maan, and Dhiraj	165
Improved Image Super-resolution Using Enhanced GenerativeAdversarial Network a Comparative StudyB. V. Balaji Prabhu and Omkar Subburao Jois Narasipura	181
Comparative Study of Supervised Machine Learning Algorithms for Healthcare Dataset Using Orange Vaibhav Bhatnagar and Ramesh C. Poonia	195
Maximum Power Point Tracking of Photovoltaic SystemUsing Artificial Neural NetworkKusum Lata Agarwal and Shubham Sharma	209
<b>IoT Security: A Survey of Issues, Attacks and Defences</b>	219
Detecting the Nuclei in Different Pictures Using Region Convolutional Neural Networks Naiswita Parmar	237
Chaotic Henry Gas Solubility Optimization Algorithm Nand Kishor Yadav and Mukesh Saraswat	247

### Multi-class Support Vector Machine-Based Household Object Recognition System Using Features Supported by Point Cloud Library



Smita Gour, Pushpa B. Patil, and Basavaraj S. Malapur

**Abstract** The proposed system aims to design and develop an object recognition system with the help of the Point Cloud Library (PCL). The object recognition problem is addressed with a three-stage mechanism. In the initial stage using PCL, the object image undergoes segmentation methods which make the image suitable for extracting the features. In the second stage, a segmented image is used to extract suitable shape-based features that can separate each object type. The last stage involves the classification/recognition of the object of the particular type using support vector machine (SVM). The system reached the expected results using Point Cloud Library (PCL) and support vector machine (SVM) as a classifier. The system has given 94% accuracy for 10 different household object types. Ten samples for each object type are used for training the SVM, and to perform testing, complete different five samples from training samples are considered.

Keywords Object recognition · Point cloud library · Support vector machine

### 1 Introduction

Object recognition is a major research area in the field of computer vision since the appearance of objects is different in its size, shape, etc. due to the varying distance between the camera and the object. Also, there is a drastic change in its visual parameters due to the use of different sources of illumination and capturing conditions during image acquisition. The main goal of the object recognition system is to make

Basaveshwar Engineering College, Bagalkot 587102, Karnataka, India e-mail: smita.gour@gmail.com

B. S. Malapur e-mail: basu.malapur@gmail.com

P. B. Patil BLDEA's V P Dr PG Halakatti College of Engineering & Technology, Vijayapura 586101, Karnataka, India e-mail: pushpa\_ms@rediffmail.com

© The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2021 H. Sharma et al. (eds.), *Intelligent Learning for Computer Vision*, Lecture Notes on Data Engineering and Communications Technologies 61, https://doi.org/10.1007/978-981-33-4582-9\_8 97

S. Gour (🖂) · B. S. Malapur

Lecture Notes in Networks and Systems 154

Simon Fong Nilanjan Dey Amit Joshi *Editors* 

# ICT Analysis and Applications

Proceedings of ICT4SD 2020, Volume 2



### Contents

Optimisation of Articulated Vehicular RADAR Antenna ArrayParameters Using Machine Learning Algorithms5Asif K. Karim and G. A. Shanmugha Sundaram	517
Traffic Congestion Detection from Twitter Using word2vec5Mohammed Ahsan Raza Noori and Ritika Mehra	527
Advanced Signal Processing for Ground-Penetrating Radar5Meghana Nagesha, H. Chetan, and Prasad Borannavar	535
Identification of Intra-abdominal Organs Using Deep Learning	
Techniques 5	547
Sanjeevakumar M. Hatture and Nagaveni Kadakol	
E-Commerce in Indian Retail Industry: Its Proliferation	
and Performance	555
Design of a Dabat for Healtheans Assistance in the Times	
of COVID-19	563
A Brief Survey of Challenge–Response Authentication	
Mechanisms       5         Prashant Kushwaha, Harshita Sonkar, Fahiem Altaf, and Soumyadev Maity	573
A Model for Automated Food Logging Through Food Recognition and Attribute Estimation Using Deep Learning	583
A Deep Learning Cenerative Annroach for Speech-to-Scene	
Generation       5         Abhijith Venugopal, Adapa Shivani, M. Neha, and H. R. Mamatha	593
A Comparative Analysis of Genes Responsible for Cataract	505
A Framework for Banana Plantation Growth Using Blockchain	
Technology       6         B. Geethanjali and B. L. Muralidhara       6	515
<b>Relative Study Between Technology to Perceive Hand Gestures</b> 6 Chetan H., S. Praveen, S. Shreyas, Samridhi Singh, and R. Urvi	521
Implementation of Electronic Records Management Systems:         Potential and Challenges a Case Study of the Water and Power         Development Authority (WAPDA) in Pakistan	529

### Clinical diagnostic systems based on machine learning and deep learning

Sanjeevakumar M. Hatture and Nagaveni Kadakol

Basaveshwar Engineering College (Autonomous), Bagalkot, Karnataka, India

### 8.1 Introduction

Revolutions in technology, including Internet applications, embedded systems, the Internet of Things (IoT), and others have made medical services easier with handheld devices. Many mobile healthcare applications are becoming the essential services of mankind. A healthcare system offers wellness programs in order to provide a healthy life to all people. A healthcare system is the organization of people, institutions, and resources for improving or maintaining health by preventing, diagnosing, and treating as well as helping in recovery from disease, illness, and physical and mental impairments. Advancement of health wards that is well-being for the people, openness to the hopes of the residents, reasonable means of finance procedures are depends on four important functions such as giving wellness programs, generation of resources, invest in, and stewardship. Healthcare is envisaged for institutions and individuals. An individual means health specialists and associated health works, working as an employee in hospital, clinic, or other healthcare institutes, or they can be freelancing. Nowadays, there have been huge technical improvements in computational power, fast data storage, and parallelization. These improvements allow machine learning (ML) and deep learning (DL) technologies to be used effectively and efficiently in healthcare systems. Since 2016, a significant amount of investment in artificial intelligence (AI) research has been done in healthcare applications as compared with other sectors. ML and DL help clinicians in early detection, identification, and treatment of diseases as well as the prediction and prognosis of a particular disease. In healthcare systems, ML and DL techniques are used significantly in medical imaging, electronic health records (EHR), genomics, treatment design, consultation using digital media, simulated nurses, management of medication, drug creation, monitoring health, analyses of healthcare systems, sensing, and the online communication of health. ML and DL techniques offer efficient and effective systems for medical diagnosis. Clinical diagnosis is the method of determining the disease or the state that defines symptoms and signs. Various diagnostic procedures are employed in diagnosis, including analysis at the cellular and chemical levels,

Demystifying Big Data, Machine Learning, and Deep Learning for Healthcare Analytics. https://doi.org/10.1016/B978-0-12-821633-0.00011-8 Copyright © 2021 Elsevier Inc. All rights reserved.





	International e-Conference	School Association
	Recent Advancement on Civil Engineering.	6
Contraction of the second s	Construction Techniques and Materials	
	RACECTAM - 2020	(S) store
	CERTIFICATE	
	OF PARTICIPATION	
	This is to certify that	
	and a so carry use.	
	SHANKAR SANNI	
Pret/Dr./Mr./Ms.	hasp	orticipated/presented
paper titled	PERFORMANCE OF GEOPOLYMER CONCRETE IN SULPHATE EI	NVIRONMENT
in International e-Co	inference on Tecoret Advancement on Civil Engineering, Construction Techn	rimes and Materials"
organized by Depart	ment of Ciril Engineering, JET, Jodhpur on 21-32 July, 2020.	
1	- 1 0	
KKL	2 To Miner Children	Age
Dest (De ) K K Balland	Dr. Arvind Dewargan Mis Kings Blood Prol. Karrlesh Pathar	Trui, Abhishek Area





National Conference on Cost Effective Civil Engineering Practices for Rural and Urban Infrastructural Development In association with International Journal of Scientific Research in Civil Engineering © 2021 IJSRCE | Volume 7 | Issue 1 | ISSN : 2456-6667

### Experimental Investigation of M-Sand in Concrete

Dr. Shankar H. Sanni<sup>1</sup>, Prabhu Gurunathappa Sheelavantar<sup>8</sup>

<sup>1</sup>Associate Professor, Dept. of Civil Engg., Basaveshwar Engg., College, Bagalkot, Karnataka, India <sup>2</sup>Dept. of Civil Engg., Basaveshwar Engg., College, Bagalkot, Karnataka, India

### ABSTRACT

The natural river and was the cheapest resource of and. However the excessive mining of river bed to meet the increasing demand for sand in construction industry has led to the ecological imbalance in the country. Now the sand available in the river bed is very coarse and contains very high percentage of silt and clay. The silt and clay present in the sand reduce the strength of the concrete and holds dampness. A few alternatives have come up for the industry to bank on of which manufactured sand or M-sand (manufactured sand), as it is called, is found to be the most suitable one to replace river sand. M-sand has caught the attention of the construction industry and environmentalists alike for its quality and the minimum damages it causes to nature. Usage of M-Sand can drastically reduce the cost since like river sand, it does not contain impurities and wastages is nil since it is made with modern technology and machinery. Once them sand becomes more popular in the construction industry, the demand for river sand and illegal sandmining would come down. M-sand that is available is graded, sieved and washed. The particles are more rounded and granular and do not have sharp edge. Usage of M-Sand can overcome the defects occurring in concrete such as sugregation, voids, capillary, etc. The main purpose of this investigation is to replace and in concrete with M-Sand for both M-30 and M-40 grades. The test results were compared to that of conventional concrete for 7 days and 28 days. Thus from result it is concluded that m-sand can be effectively used instead of river sand in concrete. **Keywords** 1 M-Sand, Compressive strength, Split tensile strength and Flexoural Strength



### Biodegradation of Organic Fraction of Municipal Solid Waste (Leaf litter) under Aerobic Conditions

B.R.Hiremath\*, Sudha Goel\*

\*Civil Engineering Department, IIT Kharagpur, West Bengal,India.

E-mail:brhiremath1@gmail.com

### Abstract:

The objective of this study was to determine the extent of biodegradation of the most common organic fraction of MSW, i.e., leaf litter. A batch biodegradation study was conducted with leaf litter for 180 days under aerobic conditions. Three sets of samples were monitored and included biologically active samples (S), autoclaved controls (C) and sodium azide (a biocide) controls (SAC). Removals of TSS for S, SAC, and C samples were 70.65%, 22.4%, and 17.65 %, respectively and removals ofVSS were 63.1%, 24.29%, and 21.52%, respectively. Dissolved Oxygen (DO) concentrations recorded were between 7.12 mg/L to 4.38 mg/L indicating aerobic conditions. Greatest reductions in TSS and VSS were observed in the biotic samples while some reductions were observed in both control sets. These results indicate that biodegradation was responsible for most of the reduction in TSS and VSS concentrations in biotic samples while reductions in controls sets were due mainly to abiotic processes.

### Speech Intelligibility Improvement based on Noise Reduction and Frequency Compression Technique

Rajani S. Pujar, P. N. Kulkarni

Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587103, Karnataka, India rajani\_pujar@rediffmail.com, pnk\_bewoor@yahoo.com

Abstract A combined structure of noise reduction and multi band frequency compression (MFC) technique has been developed to compensate for the spectral masking effect in people with sensorineural hearing impairment. Increased spectral masking reduces the speech intelligibility that affects the speech perception especially under adverse listening conditions. Noise reduction techniques namely wiener filter and spectral subtraction methods have been used in the combined structure to reduce background noise. To quantify the intelligibility of perceived speech, listening test using Modified Rhyme Test (MRT) was conducted for normal hearing and hearing impaired persons in the presence of additive noise. Test material consists of 300 words. Each word starts with phrase "would you write------". Results of MRT, when processed by cascading wiener filter with a MFC scheme, for the compression factor of 0.6 hearing-impaired subjects conducted on suggest improvement in speech intelligibility of 25.92%, to 30.134 % and decrease in response times of 0.815 to 1.626 seconds for SNR values of + 6 dB to - 6 dB in periods of + 3 dBrespectively. For spectral subtraction technique when cascaded with MFC scheme, an improvement in speech intelligibility of 20.22 % to 25.804 % and decrease in response times of 0.452 to 0.995 seconds for SNR values of + 6 dB to - 6 dB in periods of + 3 dB respectively. The results of this study showed an improvement in speech intelligibility and a reduction in the load of perception.

**Keywords** Sensorineural Hearing Impairment; Spectral Masking; MFC; Wiener Filter; Spectral Subtraction Technique; MRT

### I. INTRODUCTION

Sensorineural Hearing impaired persons have broader auditory filters [1] [2]. Hearing disabled people have trouble understanding speech because the wider auditory filters are very flat. The intelligibility of speech is decreased because of background noise. Hence, under complex acoustic conditions, Sensorineural Hearing impaired persons have problem in understanding speech. Therefore Noise reduction techniques along with signal processing methods are essential in hearing aids for improving the speech perception.

The noise minimizing methods boost required speech signal in noisy environments so as to get the listening comfort by increasing SNR.

It is crucial for hearing devices to include both noise reduction and frequency compression techniques to improve speech perception, but usually these techniques are developed and tested independently of each other.

Hence the current study is to evaluate the usefulness of combined configuration of de-noising and MFC in improving auditory processing in monaural hearing aids. We have tested different de-noising techniques (spectral subtraction and wiener filter) cascaded with MFC technique. Comparisons of the methods w.r.t improvement in intelligibility of the output speech were carried out by means of a modified rhyme test (MRT). Section II describes the related research, the proposed work is discussed in Section III, Listening tests using MRT are given in Section IV and the results are analyzed in Section V, the spectrographic analysis is done in Section VI and finally, Section VII ends with potential future work.

### **II. RELATED WORKS**

Deafness is most common at higher frequencies. Existing hearing devices do not offer any advantage for highfrequency sounds and consequently limit perceivability of sounds, mostly for profoundly hard of hearing people [3]. Transposition methods are used to transfer speech signal samples identification of speech in silent and in noise for children with from higher to lower frequency range using a fixed compression factor [4] [5]. Effect of nonlinear frequency compression (NLFC) on hearing loss was evaluated [6]. The filter bank summation approach (FBS) is researched for dichotic listening with frequency compression to reduce spectrum masking [7]. A cascaded noise reduction and time-splitting method using half cosine fading function for speech signals is explained [8]. For modest sensorineural hearing loss, spectral splitting with a wiener filter for correcting the effects of spectral masking is described. [9] [10]. Frequency compression along with noise minimizing method to enhance speech perception is addressed [11].

Single channel noise reduction techniques make use of variations in spectral and temporal features of speech and noise sources. In past, many single-channel noise reducing methods have been investigated. Spectral Subtraction (SS) [12] is the most often used technique, which involves subtraction of distorted power spectrum from auditory power spectrum to get enhanced speech. The



### Industry 4.0, AI, and Data Science Research Trends and Challenges

Edited by Vikram Bali, Kakoli Banerjee, Narendra Kumar, Sanjay Gour, and Sunil Kumar Chawla



CRC Press is an imprint of the Taylor & Francis Group, an informa business

vi	Contents
Chapter 8	Intelligent Surveillance System Using Machine Learning
	Anshuman Jaiswal, Lavkush Sharma, Akshay Kumar, and Suryansh Chauhan
Chapter 9	Cyber Security: An Approach to Secure IoT from Cyber Attacks Using Deep Learning
	Sapna Juneja, Abhinav Juneja, Vikram Bali, and Hemant Upadhyay
Chapter 10	Learning the Dynamic Change of User Interests from Noise Web Data
	Julius Onyancha and Valentina Plekhanova
Chapter 11	Artificial Intelligence Techniques Based Routing Protocols in VANETs: A Review
	Mamata J. Sataraddi, Mahabaleshwar S. Kakkasageri, and Sunilkumar S. Manvi
Chapter 12	A Comparison of Different Consensus Protocols: The Backbone of the Blockchain Technology
	Arvind Panwar and Vishal Bhatnagar
Chapter 13	Blockchain in AI: Review of Decentralized Smart System
	Vikas Goel, Narendra Kumar, Amit Kumar Gupta, and Sachin Kumar
Chapter 14	Financial Portfolio Optimization: An AI Based Decision-Making Approach
	Mansi Gupta, Sonali Semwal, and Shivani Bali
Chapter 15	Intelligent Framework and Metrics for Assessment of Smart Cities



#### Chapter

### Artificial Intelligence Techniques Based Routing Protocols in VANETs

### A Review

By Mamata J. Sataraddi, Mahabaleshwar S. Kakkasageri, Sunil kumar S. Manvi

### Book Industry 4.0, AI, and Data Science

Edition	1st Edition
First Published	2021
Imprint	CRC Press
Pages	26
eBook ISBN	9781003097181

~ Share

### ABSTRACT

Vehicle Ad Hoc Network (VANET) is a type of wireless network consisting of vehicles and roadside communication devices. As VANET is associated with life-critical applications, it is important to consider each component of VANET design, functionality, software, and problems before introducing them. The important goal of VANET is to offer seamless communication for people traveling on the road to collect and relay message from every nearby vehicles in the event of urgent situations such as severe traffic jams, collisions, lane shift, speed limit, hazard or road condition alerts, position alert services, and in the event of climatic disasters, etc. Due to the high mobility of the vehicles, frequent network disconnection happens and many problems emerge in VANETs such as routing, synchronization, network congestion, network control, information management, privacy, security, etc. This chapter reviews the current state of the research on recent artificial intelligence techniques based VANET routing. The chapter also addresses on-going research works on the usage of artificial intelligence techniques for routing in VANETs.



### 2<sup>nd</sup> International Conference on Innovative Data Communication Technologies and Application (ICIDCA 2020)

03-04, September 2020 | icidca.com | icidca.conf@gmail.com

### Acceptance Letter

Mr/Mrs Kirankumar Y. Bendigeri, Jayashree D. Mallapur, Santosh B. Kumbalavati

Herewith, the conference committee of the "2<sup>ND</sup> International Conference on Innovative Data Communication Technologies and Application (ICIDCA 2020)" is pleased to inform you that the peer-reviewed research paper "**PaperID**: **ICIDCA023 Real Time Monitoring of Crop in Agriculture using Wireless Sensor Networks**" has been accepted for oral presentation and publication at ICIDCA 2020 Organized by RVS College of Engineering and Technology, to be held during September 03-04, 2020 at Hotel Arcadia, Coimbatore, India.

ICIDCA 2020 is an interdisciplinary forum for researchers from all over the globe to share and exchange their valuable research insights on emerging intelligent data communication practices concerning the future of communication technologies domain. ICIDCA 2020 encourages only the active participation of highly qualified delegates to share and exchange various innovative research ideas, where all the accepted and registered research manuscripts will be solely dedicated for publication in Springer Lecture Notes in Data Engineering and Communication Technologies.

We congratulate you on being successfully selected for the presentation of your research work at our esteemed conference

Regards

Dr. S. Smys Organizing Secretary ICIDCA 2020



**Our Previous Publication** 2019 Publication Link

International Conference on

Intelligent Data Communication Technologies and Internet of Things

(ICICI 2020)

27-28, August 2020 | lendel.org/2020/ | friel2k17@genall.com

### **Technical Comments**

International Conference on

Intelligent Data Communication Technologies and Internet of Things (ICICI 2020) 27-28, August 2020 | JCT College of Engineering & Technology, Coimbatore, India

Paper ID: ICICI010Paper Title: Wireless Sensor Networks and its application for Agriculture

### **Review Comments 1:**

ICICI

2020

### **TECHNICAL COMMENTS**

1. The algorithms to be explained with pseudo-code/flowchart.

2. Only one sensor/node is shown in photograph. The node is also seen connected to laptop via a cable. With this in place, how do the authors claim that routing algorithms have been used as routing algorithms require a minimum of 5-7 nodes?

- 3. The title and results are confusing.
- 4. Were any fault detection or any other parameters analysed?
- 5. Screenshots of measurements in laptop to be included.
- 6. Few qualitative results to be tabulated with explanations.
- 7. Cite the following papers in reference:

i). Kumar, A. Dinesh, and S. Smys. " An energy efficient and secure data forwarding scheme for wireless body sensor network." International Journal of Networking and Virtual Organisations 21, no. 2 (2019): 163-186.

(ii). Mugunthan, S. R. (2019),"Security And Privacy Preserving Of Sensor Data Localization Based On Internet Of Things", Journal of ISMAC, 1(02) 81-92.

### GENERAL COMMENTS

- 8. Check English grammar and Springer Journal Paper formatting.
- 9. Use equation editor. All equations must be continuously numbered.
- 10. All figures must be titled and correctly numbered.
- 11. Use higher resolution images. Give citation for all images.
- 12. Avoid first and third person pronouns.

**Proceedings** by

Springer

### **Our Previous Publications**

ICICI 2018 Publication link

ICICI 2019 Publication link


# **Location Estimation of Nodes** in Underwater Acoustic Sensor Networks

B. S. Halakarnimath and A. V. Sutagundar

<sup>1</sup> Research Scholar of VTU, Department of Computer Science and Engineering, S.G. Balekundri Institute of Technology, Belagavi, Karnataka, India

<sup>2</sup> Department of Electronics and Communication, Basaveshwar Engineering College, Bagalkot, Karnataka, India

https://doi.org/10.26636/jtit.2021.145720

Abstract-The paper presents a location estimation scheme for underwater acoustic sensor networks. During the first phase, the sink node begins the trapezoid formation process by activating the trapezoid formation agent. It stores relevant information in the sink's knowledge base and in the node's knowledge base, and also develops the search data structure required for locating the node. During the second phase, the position of the node is determined by utilizing the search data structure. Identification of the location of all nodes by traveling across the trajectory may be performed as well, as an alternative approach. When identifying the location of one node, the estimation is performed based on the search data structure. When determining the position of all nodes, the sink node agent travels along the defined trajectory and transmits beacon messages which contain the real-time location at specific points. The anchor node agent measures the signal strength and localizes itself and begins estimating the locations of other nodes within the trapezoids, using location estimation techniques. Various performance parameters are used to validate the proposed scheme.

Keywords-location estimation, trapezoid, UASN.

#### 1. Introduction

Location estimation for large scale mobile underwater acoustic sensor networks (UASNs) is intriguing because of harsh aqueous environments. Even though acoustic methods are suitable for underwater communication. such features as moderate bandwidth and considerable failure rate impose specific constraints on location estimation schemes [1]. Propagation delays, movement-caused Doppler shifts, amplitude and phase fluctuations, and multipath obstruction are all factors that need to be taken into consideration in location determination procedures. Some of the localization-related issues are presented below [2]:

- need for a proper sound-speed variation model used for location estimation,
- · immersed sensor nodes need precise time synchronization,
- · efficient node mobility pattern for dynamic underwater conditions,

JOURNAL OF TELECOMMUNICATIONS AND INFORMATION TECHNOLOGY 1/2021

- impacts related to medium access control (MAC), including contention fixing, transmission overhead, localization accuracy and latency,
- · implications of location estimation protocols for location-based routing and clustering techniques.

In this paper, a computational geometrical-based localization technique is presented. The proposed location estimation scheme works in the following manner.

During the first stage, the sink node begins the trapezoid formation process on the sea surface by activating the trapezoid formation agent (TFA) and by deploying an autonomous underwater vehicle (AUV) to reach a particular depth below the surface of the sea. The AUV travels across the linear trajectory of a fixed length, at a specific depth, and the TFA creates trapezoids in the upper and lower portions of the path. The TFA stores the relevant information in the sink knowledge base (SKB) and in the node knowledge base (NKB). In the meantime, the TFA develops also the SDS in order to locate the node in the easiest manner possible.

During the second phase, position of the node is determined by relying on two methods. The first method determines the node's location by utilizing the SDS, and the other consists in finding the location of all the nodes by traveling across the trajectory. In any case, the sink triggers the localization agent (LA) and deploys the AUV to a specific depth under the surface of the ocean.

In the case of finding the location of one node, the LA moves directly to the trapezoid, which is given per the SDS, and performs the localization process.

In the case of determining the position of all nodes, the AUV traverses along the continuing trajectory and transmits beacon messages which contain real-time locations at specific points. The anchor agent (AA) at the anchor node receives these beacon messages, measures the signal strength and localizes itself based on the position of the broadcast point and the received signal strength. The LA begins the location estimating activity with the associated trapezoids of the anchor nodes, relying on location estimation techniques.

All agents keep updating the information to the respective knowledge bases whenever the data is modified.

8/2020	Stability Oriented	Cluster Dynamism in	VANET (SOCDV)	- IEEE Confe	rence Publication
EEE.org   IEEE Xpi	lore Digital Library   IEEE-SA   IEEE Spe	ctrum   More Sites			Cart   Create Account   Personal Sig
		Institutional Sig	ın In		
Browse	My Settings	Get Help	Subscribe		
					< Previous   Back to Results   Next :
Conferences > 2016	nternational Conference				
Stability Or	iented Cluster Dynamis	m in VANET	(SOCDV)		
Publisher: IEEE	Cite This		(		
<< Results   < Pr	evious   Next >				
3 Author(s) A.V	Sutagundar ; Pampapati Hubballi : Ravika	nt Belagali - View All .	Authors		Mora Lika Thia
		J= 1011 / 111			Clustering-based algorithm for connectivi
			₹x		maintenance in Vehicular Ad-Hoc Networ 2014 14th International Conference on
4 172 Paper Full				Alerts	Innovations for Community Services (I4C Published: 2014
Citations Text V	iews				A new mobility-based clustering algorithm
				Mariage Content Alerts	tor vehicular ad hoc networks (VANETs) NOMS 2016 - 2016 IEEE/IFIP Network Operations and Management Symposium
				Add to Citation	Published: 2016
				Alerts	View Mc
					Top Organizations with Patents
Abstract					on Technologies Mentioned in This Article
Document Sections	Lac m B artist				OPSANZATIONA
I. Introduction					ORGANIZATION 3
II. Stability Oriented	Abstract: Network stability and con have tremendous impact on the net	munication stability a work performance in t	are the major conce the recent advance	erns, which d networks.	OFIGANIZATION 2
in Vanet (SOCDV)	Vehicular A View more	·			ORGANIZATION 1
III. Algorithm for	Metadata				•
Clustering	Abstract: Network stability and communication	n stability are the maj	or concerns, which	have	
IV. Simulation and Results	tremendous impact on the network p Vehicular Ad-boc Network (VANET)	erformance in the re	cent advanced net	vorks.	
V. Conclusion	communication services to the users	. In this paper, we pr	oposed the Stability	/ Oriented	
	Cluster Dynamism in VANET (SOCE objective of SOCDV is to present an	IV) to enhance the clip approach for cluster	uster stability. The head selection bas	main ed on	
Authors	different parameters. Due to vehicles	s mobility, VANET fac	ing stability issue in	clustering,	
Figures	mobility, direction, density, distance	and connectivity. Since	e proposed SOCD	icea by V scheme	
References	forms the cluster and elects the Clus factors to eliminate the clustering ins	ter Head (CH) by cor tability. The SOCDV	nsidering these influ scheme is simulate	enced d in C++ to	
	test the performance in terms of clus	ter head selection tim	ne and cluster lifetir	ne.	
Citations	Published in: 2016 International Co	nference on Compute	ation System and Ir	formation	
Keywords	Technology for Sustainable Solutions	(CSITSS)	allon Gyolem and I	normatio()	
Metrics	Date of Conference: 6-8 Oct. 2016	INSPEC Acc	ession Number: 1	6525814	

IEEE websites place cookies on your device to give you the besites of the place cookies on your websites, ISBN Information: you agree to the placement of these cookies. To learn more read our Privacy Policy alore, India

3/8/2020	Wavelet Based In	nage Indexing and Retrieval	- IEEE Conference	ce Publication
IEEE.org   IEEE Xpl	lore Digital Library   IEEE-SA   IEEE Spectrur	n   More Sites		Cart   Create Account   Personal Sign In
		nstitutional Sign In		
Browse	My Settings G	et Help Subs	cribe	
				< Previous   Back to Results   Next >
Conferences > 2008 F	First International Conf			
Wavelet Ba	sed Image Indexing and R	etrieval		
Publisher: IEEE	Cite This			
<< Results   < Pr	evious   Next >			
3 Author(s) A.V	. Sutagundar ; S.S. Manvi ; S.R. Bharamgoudar	View All Authors		More Like This
<b>107</b> Full Text Views		Ex,	Alerts	A review on feature extraction techniques in Content Based Image Retrieval 2016 International Conference on Wireless Communications, Signal Processing and Networking (WiSPNET) Published: 2016
			Manage Content Alerts Add to Citation	Content based image retrieval system by using HSV color histogram, discrete wavelet transform and edge histogram descriptor 2018 International Conference on Computing, Mathematics and Engineering Technologies (COMET)
Abstract				Published: 2018 View More
Document Sections 1. Introduction	PDF			Top Organizations with Patents on Technologies Mentioned in This Article
2. Image Indexing and Retrieval system	Abstract: In this paper we present wavel for color images. Highly de-correlated wa search View more	et based image indexing and i velet coefficient planes are us	retrieval system ed to acquire a	ORGANIZATION 4
3. Simulation and Results	Metadata Abstract:			ORGANIZATION 2
4. Conclusions and Future work	In this paper we present wavelet based in images. Highly de-correlated wavelet coer efficient feature space. In this work low level the state of t	age indexing and retrieval system ficient planes are used to acq rel feature, dominant color of t	stem for color uire a search	
Authors	images is extracted and keys are generate content with respect to other images in the vector of the gupy image is permuted as	ed. The image key represents database. At the time of the	the image query, feature	
Figures	feature vectors. All the images correspond to be similar images.	ing to the matched feature ve	r the matching ctors are found	
References	Published in 2000 First state			
Keywords	and Technology	terence on Emerging Trends	in Engineering	
Metrics	Date of Conference: 16-18 July 2008	INSPEC Accession Numbe	er: 10141129	
More Like This	Date Added to IEEE Xplore: 29 July 2008	DOI: 10.1109/ICETET.2008.	129	
	CD: 978-0-7695-3267-7	Publisher: IEEE		
	ISSN Information:	Conference Location: Nag	pur,	

IEEE websites place cookies on your device to give you the best user experience. By using our websites, you agree to the placement of these cookies. To learn more, read our Privacy Policy.

3/8/2020	Remote information g	athering in wireless satelli	te based sensor netw	orks - IEEE Conference Publication
IEEE.org   IEEE Xplor	e Digital Library   IEEE-SA   IEE	E Spectrum   More Sites		Cart   Create Account   Personal Sign In
		Institutional Sig	n In	
Browse	My Settings	Get Help	Subscribe	
				< Previous   Back to Results   Next >
Conferences > 2017 Inte	ernational Conference			
Remote info	rmation gathering i	n wireless satel	lite based sei	nsor
networks				
Publisher: IEEE	Cite This			
<< Results   < Prev	rious   Next >			
3 Authoris) Mani	unath H. Pujari - Ashok V. Sutagun	lar · Padmaia Kuruba - View	All Authors	More Like This
	and the gall, notice to daugar			Conservation Techniques for Wireless Sensor Networks
			$\mu_{0} = \frac{1}{2\pi^{2}} \left[ \frac{1}{2} \right]^{2} \mu_{0}$	Communications and Information Systems (ICMCIS)
1 95 Paper Full Citation Text View	vs		C'o"-bra' - A	lerts
			Mar	nage energy conservation in wireless sensor networks
			Ade	2014 International Conference on Control, Instrumentation, Communication and Computational Technologies (ICCICCT)
			Ale	ts
				AIRM WOLG
Abstract	Disa			Top Organizations with Patents on Technologies Mentioned in
Document Sections	PDP			This Article
I. Introduction	Abstract: This paper discuss	es an energy efficient remo	te information gatherin	g in Organization 3
Communication	wireless satellite based sense Satellite Senso View more	r networks to enhance the	lifetime of networks. Th	ORGANIZATION 2
III. Cluster Based Information	Metadata			
Gathering in Terristrial WSN	Abstract: This paper discusses an ener	gy efficient remote informat	ion gathering in wireles	3S
IV. Proposed Method for Information	satellite based sensor networl Sensor Networks (SSNs) is co	ts to enhance the lifetime o mposed of group of senso	f networks. The Satelli nodes, sink nodes, ea	le arth
Gathering in Satellite Sensor	station, satellite as gateway a information and encapsulated	nd destination. Each of sen to cluster head in monitorir	sor nodes collects the ng area. The encapsula	physical ated
Networks	information has been transmit to satellite with uplink frequen	ted to sink node. The sink r cy. Once connection establi	ode is transmitting info	ormation
V. Performance Evaluation	ES, information exchange will paper an effort has made to in	take place with up/down lin prove network life time wit	k frequency of satellite h efficient energy cons	e. In this umption.
	Proposed method has two par Inter satellite link establishme	ts; establishing connection ht. The proposed method is	between satellite and to improve the networ	ES and k lífe
Authors	with balanced energy consum	ption. The reference metric	s are packet delivery ra	atio,
Figures	performance in terms of above	mentioned metrics.	as nar proposed metr	
References	Published in: 2017 Internatio	nal Conference on Comput	ing Methodologies and	
Citations	Communication (ICCMC)			

IEEReyebajtes place Determine and the second state and the second s

you agree to the placement of these cookies. To learn more creation of the placement of these cookies.

A (n

# Machine Learning based Vehicle-to-Infrastructure Communication in VANETs

Mamata J. Sataraddi, Mahabaleshwar S. Kakkasageri Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, INDIA mjsec@becbgk.edu, mskec@becbgk.edu

Abstract-Vehicular network plays a major role in understanding the detail study of vehicle communications. Multiple vehicles in local communication range need to exchange the safety and infotainment information via common roadside infrastructure in Vehicular Ad hoc Networks (VANETs). Vehicle-to-Infrastructure (V2I) communication model help to improve the efficiency of intelligent transport system by providing safety warnings and reducing vehicle collisions. Machine learning is an artificial intelligence component that gives the machine an ability to automatically learn without being expressly trained to improve from experience. Since VANET is imprecise and uncertain in nature, Machine Learning (ML) and Software Agents (SAs) combining approaches resolve the issues of V2I communication challenges in VANETs. This paper proposes ML based V2I Communication in VANETs using software agent approach. The proposed agent-based model is made up of both static and mobile agents. Proposed model executes decision tree algorithm and Q-Learning algorithm to identify the event as non critical or critical and the destination vehicle respectively to improve bandwidth utilization, packet delivery ratio and end-to-end delay.

Keywords: Vehicular ad hoc networks, V2I routing, Software agents, Decision tree, Q-Learning

#### I. INTRODUCTION

An Intelligent Transportation System (ITS) is a network that uses applications to track, control and develop transportation systems to improve the quality of transportation. In recent years traffic incidents, road congestion, excess fuel use and environmental pollution have become serious global problems due to the large number of vehicles. VANETs concentrate on road safety and effective traffic control for public roads, while offering drivers and passengers comfort and entertainment during their journeys. VANETs are highly dynamic networks and have restricted access to the network infrastructure. Vehicle to Vehicle (V2V), Vehicle to Infrastructure (V2I), and Hybrid [1] [2] are the three communication modes in VANET. The communication media used in V2V has a short latency and a fast transmission rate. In a V2I network, applications that use infrastructure points (RSUs) to multiply resources via popular internet portals are taken into account. Hybrid mode is a fusion of both V2V and V2I techniques.

V2I networking refers to wireless sharing of data among vehicles and road infrastructure, which is facilitated through a system of software, hardware and firmware. It is bi-directional, with infrastructure components including lane markers, traffic lights and road signs. It is primarily intended for safety applications such as avoiding motor crashes, ambulance assistance and accomplishing a wider range of mobility.

Computational intelligence and machine learning allow computers to learn from data iteratively rather than being directly programmed, allowing them to uncover hidden insights [3]. There are three types of machine learning approaches currently available: supervised, unsupervised and reinforcement learning.

A software agent is a goal-oriented computer simulation that responds to its situation and runs independently to perform a task for an end user [4] [5]. Cognitive agents are objects that use a Belief-Desire-Intention (BDI) architecture and have recently gained popularity due to their human-like reasoning [6]. Cognitive agents, on the other hand, are usually static agents that need more calculation and databases.

The operational sequence of proposed ML based V2I communication in VANETs is as follows. (1) EDA in vehicle agency detects the event as critical or non-critical based on decision tree algorithm using vehicle sensor outputs. (2) EIA communicates the generated event to the corresponding RSU. (3) RSUMA employs the Q-Learning algorithm to select the best neighbor RSU until the destination RSU is identified. (4) RSUIA builds a route between the source and destination vehicle by tracing the path back to the source RSU.

The remainder of the paper is laid out as follows. Section II reviews the ongoing research works. Section III gives insight of the construction of proposed scheme in detail. The proposed scheme simulation model is outlined in Section IV. Results are analyzed in Section V. Finally, Section VI concludes the work.

#### **II. RELATED WORKS**

This section provide a brief survey of related works on routing in VANETs, Machine Learning (ML) techniques, and cognitive agent based solutions. Routing protocols for V2V and V2I communication with their advantages and disadvantages are presented in [7]. A routing algorithm for V2I scenario using distributed clustering algorithm is discussed in [8]. V2I communications was established to Conferences > 2020 2nd PhD Colloquium on Et... ?

## Stochastic Power Flow Analysis of Unbalanced Distribution System based on cluster of SWTG-SPV



Cite This 🛛 🔀 PDF

<< Results

Sangamesh Y. Goudappanavar ; Suresh H. Jangamshetti All Authors

	<b>0 &lt;\$</b> © <b>≥ ♦</b>	
1 56 Paper Fu	Alerts	More Like This
Citation Te:	d Views Manage Content Alerts Add to Citation Alerts	An analytical planning approach for determining the maximum penetration of renewable distributed generation in the distribution network 2017 IEEE Conference on Energy Internet and Energy System Integration (EI2)
Abstract	Ri .	Published: 2017
Document Sections I. Introduction II. Single Value Decomposition (SVM) Based	Downl PDF Abstract: This paper presents stochastic 3-Ф power flow analysis of real-time unbalanced distribution system based on the aggregated and clustered model of Small Wind Turbine Gener View more	Optimized ESS Operation for Peak Shaving based on Probabilistic Load Prediction 2018 IEEE Innovative Smart Grid Technologies - Asia (ISGT Asia) Published: 2018 Show More
Clustering and Aggregated Model of Swtg and Spv in the Distribution System III. Results and Discussions	Abstract: This paper presents stochastic 3-Φ power flow analysis of real-time unbalanced distribution system based on the aggregated and clustered model of Small Wind Turbine Generator (SWTG) and Solar Photo Voltaic (SPV) using OpenDSS. Single value decomposition based allocation and probabilistic approach of power-generation balance in every bus is modelled. 15 mins interval of wind velocity, solar irradiance and load data of Basaveshwar Engineering College (A) Bagalkot (BEC) profiles are recorded from Distribution	
IV. Conclusion & Future Enhancement	Automation SCADA research laboratory for preparation of loadshapes. The proposed method is tested and validated on IEEE 15 and Europe LV Networks. Results for the studied High & Low Voltage (HV/LV) system indicate that buses with clustered (Distributed Energy Resources (DERs) have greater impact on voltage profile and thermal ampacity.	
Figures	Published in: 2020 2nd PhD Colloquium on Ethically Driven Innovation and Technology for Society (PhD EDITS)	

#### Identification of Fault Area in the Distribution System based on Rotor Angle behaviour of Diesel Generator

PDF

Publisher: IE	EE
---------------	----

<< Results

Sangamesh Y. Goudappanavar; Suresh H. Jangamshetti; Jai Jagadeesh O All Authors

25
Full
Text Views

Abstract

Document

I. Introduction

II. Methodology

Sections

노

Downi

PDF

**Cite This** 

**8**<۩⇒**≜** Alerts

Manage Content Alerts

Add to Citation Alerts

#### More Like This

Comparative Analysis of Integral-Based Indices for On-line Assessment of Power System Transient Stability 2021 IEEE PES Innovative Smart Grid Technologies Europe (ISGT Europe) Published: 2021

A Continuation-Based Method to Compute the Relevant Unstable Equilibrium Points for Power System Transient Stability Analysis IEEE Transactions on Power Systems Published: 2009

Show More

III. Single Line Diagram of BEC 11 Bus Distribution System	Metadata Abstract: This paper presents identification of fault location based on coherency of Diesel Generators (DG) rotor angle in distribution system. Coherent behavior of DGs is
IV. Results	identified by similar characteristics and modelled using network reduction technique and performs stability analysis. Modelling of coherency based
V. Conclusions	network reduction method for transient stability study of distribution system is developed using MiPower simulation package. Performance of proposed
Authors	method has been tested on a Basaveshwar Engineering College (A), Bagalkot (BEC) 11-Bus distribution system and carried out a fault point/fault area identification. Proposed method is verified for connected load of BEC 11 bus
Figures	distribution system. Further, analysis has been carried out for hourly load data measured on 6th May 2019, SCADA for Distribution Automation Research
References	Laboratory, BEC. Three phase to ground fault is considered as a disturbance at different buses in BEC 11 Bus distribution system and correlation factors of
Keywords	DGs rotor angle is identified. It is observed from results that, time frame of small signal stability is within 20 seconds and fault location is identified. The
Metrics	outcome/characteristics of proposed method aids for operation and secure

Abstract: This paper presents identification of fault location based on

Coherent behavior of DGs is identifi... View more

coherency of Diesel Generators (DG) rotor angle in distribution system.

maintenance of existing distribution system

# Performance Assessment of SPV Powered DC Irrigation Pumps with Solar Tracking Mechanism

🖹 PDF

Publisher: IEEE

E Cite This

<< Results

Basanagouda F. Ronad ; Santosh B. Kumbalavati All Authors

<b>37</b> Full Text Views	Alerts Manage Content Alerts Add to Citation Alerts	More Like This Energy, Economic, and Environmental (3E) Analysis of PV Water Pump: Case Study of the Tapioca Drip Irrigation System 2021 International Conference on Power, Energy and Innovations (ICPEI)
Abstract		Published: 2021
Document Sections I. Introduction II. Resources for	Downl PDF Abstract: This paper presents the performance of SPV DC irrigation pumps with and without solar tracking mechanism. HOMER software is used for simulation. Simulation model is built View more	Solar Powered Irrigation Pumps with Optimum Power Transfer 2019 IEEE PES GTD Grand International Conference and Exposition Asia (GTD Asia) Published: 2019
Modeling III. Different Methods for	Metadata     Abstract:     This paper presents the performance of SPV DC irrigation numps with and	Show More
IV. Homer Simulation Model for 5HP DC Pump	without solar tracking mechanism. HOMER software is used for simulation. Simulation model is built for a 5 HP DC irrigation pump load. Local solar radiation data of Bagalkot, India are employed for the study. Developed HOMER model is standalone system with support of batteries. Constant DC load profile for seven hours/day is employed for the analysis. Further,	
V. Comparision of Simulation Results	simulations are carried out with different types of solar tracking mechanisms. SPV panel tracking with Monthly, Weekly, Daily & Continuous Adjustment of Horizontal Axis. Continuous Adjustment of Vertical Axis and Two Axis	
Show Full Outline	Adjustment are employed to attain the simulation results. For each tracking method, optimal configurations of SPV panels & batteries are listed and various	
Authors	costs corresponding to initial capital, operating, net present and cost of energy, capacity shortage & electricity generation details are tabulated. Comparative	
Figures	analysis amongst different types tracking is taken up. Results revealed that tracking of panels will enhance power production and in particular Two Axis	



Published in: 2020 5th International Conference on Computing, Communication and Security (ICCCS)

Intelligence Algorithms With UAV-



# <section-header><section-header><section-header><section-header><section-header><text><text><text>





#### D Springer Link

Springer Link		Search Q 🍃 Log ir
Book cover Innovative Data Communication Technologies and Application pp 773–785 Cite as		
Real-Time Monitoring of Crop in Agriculture Using Wireless Sensor Networks         Kirankumar V. Bendigeri <sup>CD</sup> , Jayashree D. Mallapur & Santosh B. Kumbalavati         Conference paper   First Online: 03 February 2021         396 Accesses   3 <u>Citations</u> Part of the Lecture Notes on Data Engineering and Communications Technologies book series (UNDECT, volume 59)	<ul> <li>Chapter</li> <li>DOI: 10.1007/978-981-15</li> <li>Chapter length: 13 pages</li> <li>Instant PDF download</li> <li>Readable on all devices</li> <li>Own it forever</li> <li>Exclusive effer for individu.</li> <li>Tax calculation will be final</li> </ul>	EUR 24.95 Price excludes VAT (India) -9651-3_02 uals only lised during checkout
Abstract	> eBook	EUR 213.99
The paper works on the issue of wireless sensor network (WSN) employed in the agriculture	> Softcover Book	EUR 249.99
application. Most of research works in WSN are dedicated for various applications like military,	> Hardcover Book	EUR 249.99
medical, home automation, and so on, where the primary intention was to introduce a technology for agriculture with the evolving sensor technologies. Such technology can be considered as E-agriculture. Proposed work consideres sugarcane as a crop under monitoring, where the work mainly aims at monitoring humidity (H), temperature (T), pressure (P), and light	Learn about institutional subs	riptions is References
intensity (LI) required for the intended crop and also on the land using LBR and MBR routing	Abstract	

e Editorial Board Journal Topics Archives About the Journal	Submissions Privacy Statement Contact	Q. Search
ome / Archives / Vol. 29 No. 10s (2020) / Articles	a using Dry Sand and	International Journal of Advanced Science and
Viper to Save Water	te using Dry sand and	Not yet assigned quartile
rathi Nashi , Husnara Pinjar, Deepti Challagidad, Savita Girijavvagol, rishnamurthy Bhat	PDF	SJR 2020 O powered by scimagojr.com
bstract Dah and plate wash is an art and it is an integral part of everyday life. Dishes in a plates were cleaned manually for a long time by humans. As the technology diances, man-made sami and fully automated machines are invested to ease and anhance the life style. Be it monunal or automated machine wash, the eases of cleaning requires large amount of waters: In detto to supervision.	Hore to Cite Arahi Nashi Husnara Pinjar, Deepsi Challegida, Santa Gripovigol, Kishamaruthy Bhar, (2020). Fater Wijer: A Novel Flate Water, Brenzione Juscraf and Wiere to Save Water. Brenzionen Juscraf of Advonced Science and Richology, 2010b. 4134-4138. Bertrievel From http://wsr.com/goundsi/index.php/lk57/artii devivew/21103	Make a Submission
ater, scientific community needs to invert machines which require less or no ater for the said purpose. In this paper, a novel method of dish plate cleaning iachine which uses sand instead of water is proposed. It is a microcontroller used July automatic machine and the cleaning extent is expected to be highly	More Citation Formats	E.S.L
ccaptable. Plate wather is an electrically operated machine which wathers the these schemply in less time with minimum ar no water and can be reused mediately. The proposed machine can some lots of waters. In the water of wing precious water and highly affordable cost, it has potential market value to . Two variants to be though of 0 fore for the general kitchen use and other or sustaines racine cleanon. The imitermentation of this monitories is derech-	<u>Yol. 29 No. 10s (2020)</u> Section Articles	ELSEVIER Scopus







#### GPS AND IOT BASED VEHICLE ON-ROAD BEHAVIOR MONITORING AND WARNING SYSTEM SUITABLE FOR TRANSPORT AUTHORITY

Shweta Malpani<sup>\*</sup>, Krishnamurthy Bhat<sup>\*</sup> <sup>\*</sup>Student, <sup>\*</sup>Faculty Department of Electronics and Instrumentation Engineering Basaveshwar Engineering College, Bagalkot – S87302, Karnataka, India shwetamalpani2000@gmail.com, knbei@becbgk.edu

#### ABSTRACT

Objective/Aim: India and other populated countries are facing a threat of increasing road accidents and casualties. Most of the road accidents are caused due to the offences and violation of rules either by the accused or by the victim. Main cause for the road accidents are due to traffic rule violations. This paper describes an innovative and advanced method of monitoring the behavior of vehicles on the road and subsequently data base generation at the authority end to control the offenses.

Methods and materials: A robust and effective mechanism of monitor and control for the purpose of vehicle Methods and materials: A robust and effective mechanism of monitor and control for the purpose of vehicle behavior monitoring is not available for Regional Transport Office (RTO) in India. A tare-proof electronic system consisting of many sensors and signal conditioning circuits is to be a part of the vehicles implemented by the manufacturers. This electronic system with an apt electronic controller will be installed in individual vehicle. The surveillance and monitoring system in places like one way, no parking and silent zones are to be installed by RTOs. Any violation of traffic rules such as over speed, one-way traffic violation, blowing horn in silent zones (hospitals, schools, parks), drunken driving, and parking in prohibited areas will be automatically informed to the authority of RTO over the internet. Monitoring the important traffic rules and regulations for each vehicle on the road is done through set of protocols. The monitored data will be available on cloud for RTOs. **Results:** This is proposed work and implementation is already initiated by the authors. This will result into a robust and proper mechanism to support the authority to control the road offences and accidents.

sions: An electronic system for the effective surveillance and monitoring of vehicles on the road is the need of hour, especially in India. We are proposing one such system here.

ords: Road accidents; vehicles; surveillance; traffic; electronic monitoring. Keyw

25 | Page



Springer Link		Search Q 📮 Log
Book cover Intelligent Data Communication Technologies and Internet of Things pp 673–687 <u>Cite as</u>		
Wireless Sensor Networks and Its Application for	✓ Chapter	EUR 24.95
Agriculture		Price excludes VAT (India)
ngriculture	• DOI: 10.1007/978-981-15-	9509-7_55
Kirankumar <u>Y. Bendigeri</u> 🖂, J <u>ayashree D. Mallapur</u> & <u>Santosh B. Kumbalavati</u>	Chapter length: 15 pages	
	Instant PDF download	
Conference paper   First Online: 13 February 2021	Readable on all devices     Own it forever	
220 Accesses   3 <u>Citations</u>	Exclusive offer for individu	als only
Part of the Lecture Notes on Data Engineering and Communications Technologies book series	<ul> <li>Tax calculation will be final</li> </ul>	ised during checkout
(LNDECT,volume 57)		
	Buy C	hapter
Abstract	> eBook	EUR 181.89
The wireless sensor network (WSN) is the most proven technology in today's world. WSN has	> Softcover Book	EUR 219.99
gained its applications that are different from other networks and even the usage of WSN has		

IEEE.org | IEEE Xplore | IEEE-SA | IEEE Spectrum | More Sites Cart Create Acc IEEE Xplore<sup>®</sup> Browse ∽ My Settings ∽ Help ∽ **∲IEEE** All ferences > 2020 3rd International Confer... 🚱 Performance Assessment of SPV Powered DC Irrigation Pumps with Solar Tracking Mechanism IEEE Publisher: IEEE Cite This DPDF **ENGLISH** for eLEARNING COURSE PROGRAM Basanagouda F. Ronad ; Santosh B. Kumbalavati All Authors 27 Full Text Views ß < © Ļ More Like This Solar Powered Irrigation Pumps with Optimum Power Transfer 2019 IEEE PES GTD Grand Internationa Conference and Exposition Asia (GTD Asia) Published: 2019 Abstract Abstract: Abstract: This paper presents the performance of SPV DC irrigation pumps with and without solar tracking mechanism. HOMER software is used for simulation, Simulation model is built for a 5 HP DC irrigation pump load. Local solar radiation data of Bagalikot, India are employed for the study. Developed HOMER model is standarione system with support of batteries. Constant DC load profile for seven hoursking is employed for the analysis. Further, simulations are carried out with different types of solar tracking mechanisms. SPV panel tracking with Monthly, Weekly, Daily & Continuous Adjustment of Horizontal Axis, Document Sections I. Introduction II. Resources for Modeling

IEEE.org   IEEE Xplore   IEEE-SA	IEEE Spectrum   More Sites			Cart Create Account Personal Sign In
IEEE Xplore® Browse	✓ My Settings ✓ Help ✓	Access provided by: Basaveshwar Engineering College	a.	♦IEEE
	All	•	Q Advanced search	
Conferences > 2021 International (	Conference 🔞			
Direction Based Publisher: IEEE Cite Thie Kirankumar Y. Bendigeri ; Jaya 43 Full Text Views	Node Placement in W	ireless Sensor Netwo	rk 0 < © 늘	LEEE Global Communitation Receivering the Upper Receivering the Upper A - Brownerbar 2022 Receivering the Upper RepErson
Abstract	Abstract:			More Like This
Document Sections	This paper discusses a novel node pla Network (WSN) emphasizing over the	cement strategy to be applied over the a	griculture-based application in Wireless	Sensor Energy Conservation in
I. Introduction	concentrate on node placement, which	n is first and critical parameter to estimate	the performance of network. The pape	Hop Wireless Sensor Networks
II. Literature Review	discusses the problems associated wi description. This paper finally discusse	th conventional random/grid deployment as the comparative performance analysis	motivation, research methodology, algo of proposed system of relay node place	orithm for Pervasive Computing ement Environment
III. Proposed Work	and energy modeling with existing nor maintains a good balance between co	le placement strategies in WSN. The sim mmunication and energy efficiency.	ulation demonstrates that proposed sys	item IEEE Systems Journal Published: 2017
IV. Algorithm Implementation				Performance Analysis of Energy
V Result Applysis	Published in: 2021 International Cont	ference on Artificial Intelligence and Sma	t Systems (ICAIS)	Contention Transie for

International Journal of Advanced Pervasive and Ubiquitous Computing Volume 11 • Issue 3 • July-September 2019

# A Comprehensive Review of Access Control Mechanism Based on Attribute Based Encryption Scheme for Cloud Computing

Lokesh B. Bhajantri, Basaveshwar Engineering College, Bagalkot, India Tabassum N. Mujawar, Ramrao Adik Institute of Technology, Navi Mumbai, India

#### ABSTRACT

Cloud computing is the most prevailing paradigm, which provides computing resources and services over the Internet. Due to immense development in services provided by cloud computing, the trend to share large-scale and confidential data on cloud has been increased. Though cloud computing provides many benefits, ensuring security of the data stored in cloud is the biggest challenge. The security concern about the data becomes main barrier for adoption of cloud. One of the important security aspects is fine grained access control mechanism. The most widely used and efficient access control scheme for cloud computing is Attribute Based Encryption (ABE). The Attribute Based Encryption (ABE) scheme provides a new technique for embedding access policies cryptographically into encryption process. The article presents an overview of various existing attribute-based encryption schemes and traditional access control models. Also, the comparison of existing ABE schemes for cloud computing, on basis of various criteria is presented in the article.

#### **KEYWORDS**

Access Control, Attribute-Based Encryption, Cloud Computing, Outsourcing

#### INTRODUCTION

In today's era, cloud computing has become an attracting technology, which has brought extreme changes to IT industry. Cloud Computing enables network access to various computing resources such as servers, storage, networks, applications and services. It is basically a paradigm that provides access to shared pool of resources online on demand (Armbrust et al., 2010). It is computing over internet. The users can store any amount of data on cloud and then can access it at any time, from anywhere. The most fascinating benefit of cloud computing is that it provides cost saving as the users will pay only for their usage. Cloud computing is also termed as distributed computing over a network. The term cloud can be defined as collection of servers delivering computing resources as a service on demand. Generally, the cloud comprises various interfaces, networks, hardware, storage devices etc. (Majumder et al., 2014).

DOI: 10.4018/IJAPUC.2019070103

Copyright © 2019, IGI Global. Copying or distributing in print or electronic forms without written permission of IGI Global is prohibited.

**Composites Science and Technology** 

# Mohammad Jawaid Anish Khan *Editors*

# Vegetable Fiber Composites and their Technological Applications



Editors Mohammad Jawaid Laboratory of Biocomposite Technology INTROP Universiti Putra Malaysia Serdang, Selangor, Malaysia Anish Khan Center of Excellence for Advanced Materials Research King Abdulaziz University Jeddah, Saudi Arabia

ISSN 2662-1819 ISSN 2662-1827 (electronic) Composites Science and Technology ISBN 978-981-16-1853-6 ISBN 978-981-16-1854-3 (eBook) https://doi.org/10.1007/978-981-16-1854-3

© The Editor(s) (if applicable) and The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2021

This work is subject to copyright. All rights are solely and exclusively licensed by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, expressed or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

This Springer imprint is published by the registered company Springer Nature Singapore Pte Ltd. The registered company address is: 152 Beach Road, #21-01/04 Gateway East, Singapore 189721, Singapore

# **Tribological Behaviour of Glass Fiber Reinforced Polyamide Gears**



Sandeep C. Dhaduti, S. G. Sarganachari, Arun Y. Patil, and Anish Khan

#### **1** Introduction to Gears

Gears are wheels with tooth used for motion, power transmission and in most industrial rotating machine systems Gears are likewise utilized for changing speed of rotation of a machine shaft, alter the direction of the axis of revolution and change rotating to linear movement and the other way around. These are preferred to transmit power over a short center distance positively with constant velocity ratio. Because of a portion of the constraint of metallic gears like lubrication, noise, weight, cost, productivity and so forth, center has now been quickly changing upon polymer gears since most recent past years. Consequently, the assessment of gear failure is significant for upkeep wanting to fundamentally decrease downtime and cost [1]. Polymer gears pose many advantages upon metallic gears and are preferred because of no or less lubrication requirement, low noiseless operation, light weight, economic in production, good vibration absorbing property, impervious to erosion. But on the other hand these polymer gear drivers have low load carrying capacity, limited operating temperatures and less dimensionally stable.

The suitable element for plastic gears ought to have posed enhanced mechanical properties like modulus of rupture, resistance for deformation and compressive loads, low coefficient of friction, wear and fracture resistance of gear tooth. Thermoplastics

S. C. Dhaduti

S. G. Sarganachari

Mechanical Engineering Department, Basaveshwar Engineering College, Bagalkot, India

A. Y. Patil (🖂)

School of Mechanical Engineering, KLE Technological University, Hubballi, India

A. Khan

© The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2021 M. Jawaid and A. Khan (eds.), Vegetable Fiber Composites and their Technological Applications, Composites Science and Technology, https://doi.org/10.1007/978-981-16-1854-3\_15

339

Mechanical Engineering Department, KLE Institute of Technology, Hubballi, India

Center of Excellence for Advanced Materials Research, King Abdulaziz University, Jeddah 21589, Saudi Arabia

Waste Management and Application of Coconut Biomass and Fibre Ana Cristina Curia, Aline Ramalho Dias Souza, Janiel Rodrigo Zaro, Magali Rejane Rigon Barbieri, Genyr Kappler, and Carlos Alberto Mendes Moraes	215
Effect of Hybridization and Chemical Modification on the Water-Absorption Behaviour of Banana Fibre–Reinforced Polyester Composites Deeksha Jaiswal, Ishan Srivastava, Himani Agarwal, Anish Khan, and G. L. Devnani	239
Properties of Hemp Fibre Reinforced Polymer Composites	255
Sustainable Product Packaging Using Vegetables Fibres and Its Composite Guravtar Singh, Sumit Sharma, Kamalpreet Sandhu, Abdullah M. Asiri, and Anish Khan	275
Natural Composites: Vegetable Fiber Modification Nidhi Shukla, Himani Agrawal, Ishan Srivastava, Anish Khan, and G. L. Devnani	303
Bionanocomposite of Ag Nanoparticles/Jute Fibers as an Efficient Fungi-Free Material for the Automobile Industry Raúl A. Morales-Luckie, Sergio L. Palacios-Lozano, Víctor Sánchez-Mendieta, Oscar F. Olea-Mejia, and María G. González-Pedroza	327
Tribological Behaviour of Glass Fiber Reinforced Polyamide Gears Sandeep C. Dhaduti, S. G. Sarganachari, Arun Y. Patil, and Anish Khan	339
Identification of Vegetable Fiber Origin Sumit Sharma, Guravtar Singh, Abdullah M. Asiri, and Anish Khan	351
Hierarchical Vegetal Fiber Reinforced Composites	379
Development of Vegetable Fibre-Mortar Composites of Improved Durability	413
The Roles of Vegetable Fibres in Green Chemistry A. T. Ajiboye, K. O. Otun, H. F. Babamale, and L. B. Abdulra'uf	425

# ACADEMIC YEAR 2019-20

Advances in Intelligent Systems and Computing 1133

Niranjan N. Chiplunkar Takanori Fukao *Editors* 

# Advances in Artificial Intelligence and Data Engineering

**Select Proceedings of AIDE 2019** 



Pages 601-613

# A Survey on State-of-the-Art Applications of Variable Length Chromosome (VLC) Based GA

Ravi Domala, Upasna Singh Pages 615-630

A Multi-level Access Technique for Privacy-Preserving Perturbation in Association Rule Mining

N. Komal Kumar, D. Vigneswari Pages 631-645

# LSB and RLE Based Approach for Increasing Payload and Security of Stego Images

Rupali Sanjay Pawar Pages 647-658

Adaptive MoD Chatbot: Toward Providing Contextual Corporate Summarized Document as Suggestions and Reported Issue Ticket Routing

Shiva Prasad Nayak, Archana Rai, Kiran Vankataramanappa, Jalak Arvindkumar Pansuriya, Joerg Singler Pages 659-674

# **Classification of Text Documents**

Pushpa B. Patil, Dakshayani M. Ijeri Pages 675-685

# Fine-Grained Sentiment Rating of Online Reviews with Deep-RNN

Ramesh Wadawadagi, Veerappa Pagi Pages 687-700

# **Fine-Grained Sentiment Rating of Online Reviews with Deep-RNN**



Ramesh Wadawadagi and Veerappa Pagi

**Abstract** Increasing volume of customer reviews over the commercial Web sites have created a demand for the construction of automated content analysis systems. However, present techniques mainly focus on traditional bag-of-words (BOW) and statistical language models, ignoring semantic compositions. In contrast, deep neural networks (DNN) have exhibited greater stability in equipping on-scale sentiment prediction. Particularly, deep recursive neural networks (Deep-RNN) have been consistently used for capturing semantic compositions in natural language content when represented with structured formats (e.g., parse trees). Improved word spaces (word-embeddings) on the other hand proved to be efficient in comprehending fine-grained semantic regularities. In this paper, a fine-grained sentiment rating of online reviews based on Deep-RNN is proposed. The performance of the proposed model is evaluated through the conduction of experiments over Stanford sentiment treebank (SST) dataset. Furthermore, the effect of tuning hyper-parameters on the performance of the network is studied. The experimental results reveal that Deep-RNN exhibits better prediction accuracy compared to the traditional shallow counterparts.

**Keywords** Recursive neural networks · Sentiment analysis · Word-embeddings · Fine-grained sentiment rating · Deep learning

#### **1** Overview

Online consumer reviews and ratings may influence positively or negatively any enterprise in this business world. This being the case, many online review systems offer star rating in addition to free text reviews. In spite of the fact that, different users

Basaveshwar Engineering College, Baglkot 587102, India e-mail: rswlib@yahoo.co.in

V. Pagi e-mail: veereshpagi@gmail.com

© Springer Nature Singapore Pte Ltd. 2021

687

R. Wadawadagi (🖂) · V. Pagi

Department of Computer Science and Engineering,

N. N. Chiplunkar and T. Fukao (eds.), *Advances in Artificial Intelligence and Data Engineering*, Advances in Intelligent Systems and Computing 1133, https://doi.org/10.1007/978-981-15-3514-7\_52

# ICETETM-2019 26<sup>th</sup> - 27<sup>th</sup> April 2019 PDIT, Hoscipete International Conference on Emerging trends in Engineering, Technology and Management Organized by Proudhadeveraya Institute of Technology (PDIT) and Institute For Engineering Research and Publication (IFERP) 🔮 +91 73054 77495 🌐 www.icetetm.com info@icetetm.com

# CONTENTS

## SL.NO

## TITLES AND AUTHORS

PAGE NO

28.	Agriculture Price Prediction Using Data Mining > Veeresh Kadlimatti > S V Saboji	28
29.	Certain identities on class one Infinite series <i>Vidya H. C</i>	29
30.	Water quality evaluation in term of WQI River Tungabhadra, Karnataka, India. > S.Ranjith > Dr. Anand.V. Shivapur > Dr. P. Shiva Keshava Kumar > Chandrashekarayya.G. Hiremath > Santhosh Dhungana	30
31.	Pollution Monitoring and Control Using IoT > Firdosh Parveen.S > Farook Khan > M. Puneeth > Malatesh. C > Somashekhara	31
32.	Seismic Response of Building using Fluid Viscous Dampers: A Review	32
33.	Processing and property evaluation of Nano Al2O3 reinforced copper- 5% tin composites for bearing applications. > Dr B Adaveesh > Mohan Kumar T S > Deeepa	33
34.	Experimental investigation on dry sliding wear behaviour of Hyper- eutectoid steel using Taguchi Technique > Veerabhadrappa Algur > Mohan > Dr. V R Kabadi > Dr. C. Thotappa	34
35.	Performance analysis of variable data rate reconfigurable architecture for SDR Receiver	35
36.	Monitoring and Effective utilization of Renewable Energy by using Arduino and Lora > UShantha Kumar > Spandana.T > Noor Mahammad.S	36

# International Conference on Emerging Trends in Engineering, Technology and Management

Hosapete, Karnataka, 26<sup>th</sup> – 27<sup>th</sup>April 2019

# **Agriculture Price Prediction Using Data Mining**

Veeresh Kadlimatti, Basaveshwar Engineering College, Bagalkot S V Saboji, Basaveshwar Engineering College, Bagalkot

#### Abstract:--

Agriculture is the main source and backbone of Indian Economy and plays a vital role in individual life. In the total Gross Domestic Product (GDP) agriculture nearly contributes sixteen percent and for increasing foreign exchange it contributes nearly ten percent to the total country exports. As the population continuously increasing and to manage the livelihood of the country there's requires a proper utilization and management of agriculture products. Data mining is a better technique and best choice in predicting the accurate prices of the agriculture based on previous data. In this work various data mining algorithms are applied on the dataset to predict the future prices of the agriculture products. Keywords: Agriculture, Data-Mining, Prediction.

#### Keywords

Agriculture, Data Mining, Prediction, Regression.

26<sup>th</sup> - 27<sup>th</sup> April 2019

ICETETM – 19

ISBN: 978-81-940803-6-7

Organized by: Proudhadevaraya Institute of Technology (PDIT) In Association with Institute For Engineering Research and Publication (IFERP)

Page | 28

MALNAD COLLEGE OF ENGINEERING (An Autonomous Institution under VTU, Belagaum) (Approved by AICTE, New Delhi, Accredited by National Board of Accreditation, Accredited by NAAC) CERTIFICATE
This is to certify that Mr. /Ms. Dr. S.V. Saboji
of Basaveshusar Engineering College, Bagalkote
Dr. Ramesh B. Conference Chair McE, HASSAN McE, HASSAN Dr. K.S. Jayantha Professor & Head McE, HASSAN McE, HASSAN



Malnad College of Engineering (An Autonomous Institution under VTU, Belagavi, Approved by AICTE, New Delhi Accredited by National Board of Accreditation) Salagame Road, Hassan-573202, Karnataka



Department of Computer Science and Engineering "Fifth National Conference on Multimedia **Technology and Communication System** (NCMTCS-2019)" Sponsered by TEQIP-III TEOIP NPIU

> on 10<sup>th</sup> & 11<sup>th</sup> May, 2019 INVITATION



To,

# Establishing Trust Worthy Reliable Path In Mobile Adhoc Network Using Mobile Agent

Rakshitha N Hiremath

Department of CSE Basaveshwar Engineering College, Bagalkot, Karnataka rakshithahiremath@gmail.com

Mobile Ad-hoc Network (MANET) is a collection of independent mobile nodes that communicate with each other by forming multi-hop radio network. Routing is the primary problem in MANET. There are some limitations due to mobility of nodes in MANET. Because of the active topology and decentralized structure of a network. sometimes nodes may break routes. Frequent link failure results in packet loss, increase in delay. So the proper selection of an intermediate node for packet transmission is a bit difficult job. E-TWRP (Establishing Trust Worthy Reliable Path) in MANET. This protocol empower nodes not only relay the packets but also retain the route

Dr. S. V. Saboji Department of CSE Basaveshwar engineering College, Bagalkot, Karnataka sabojishivakumar@gmail.com

reliability. Trusted party maintains trust updation and credit account. Those nodes who are actively participating in the network they get rewarded. Their establishes reliable and optimal path between the nodes.

Keywords: Trusted party, Adhoc Network, Reliable route.

#### I. INTRODUCTION

In Mobile Ad-hoc Network nodes area unit unengaged to move anyplace within the network with none mounted topology and centralized structure. Figure shows that nodes A and B changes their position with totally different path. In Edouard Manet any node will enter and leave the network at any

See discussions, stats, and author profiles for this publication at: https://www.researchgate.net/publication/331453273

# Multimodal Biometrics Database for Person Authentication :VTU-BEC-DB

Conference Paper · September 2018

CITATION:	5	READS 223			
2 autho	rs:				
0	Shanmukhappa. A. Angadi Visvesvaraya Technological University 50 PUBLICATIONS 308 CITATIONS SEE PROFILE		Sanjeevakumar Hatture Basaveshwar Engineering College, Bagalkot 20 PUBLICATIONS 86 CITATIONS SEE PROFILE		
Some of the authors of this publication are also working on these related projects:					

Project research\_vilas View project

research\_hatture View project

All content following this page was uploaded by Sanjeevakumar Hatture on 12 March 2019.

Proceedings of the International Conference on Intelligent Computing and Sustainable System (ICICSS 2018) DVD Part Number: CFP18003-DVD; ISBN;978-1-5386-4344-0

# Multimodal Biometrics Database for Person Authentication :VTU-BEC-DB

Shanmukhappa A. Angadi Department of Computer Science and Engineering Visvesvaraya Technological University Belagavi - 590018, Karnataka State, India vinay\_angadi@yahoo.com

Abstract—This Though many standard databases are available for Biometric research, a robust multimodal database is not found. In this paper, a new multimodal biometric database, constructed with three traits namely face, hand and voice is described. These biometric traits are acquired from 100 persons(i.e. 36 males and 64 females) with age range between 18 and 50 years. The database (VTU-BEC-DB) is created at the Department of Computer Science and Engineering at Basaveshwar Engineering College (BEC), Bagalkot under (VTŬ), Belagavi, Technological University Visvesvarava Karnataka, India. Face images are collected indoor in controlled scenarios with white background and variation in illumination and pose and also by varying the pixel resolution of the camera. The peg free hand images from both right and left hands are collected from the book-scanner in the indoor environment with natural light and also by varying the illumination through fluorescent lights. The voice samples are recorded in two languages namely Kannada i.e. Indian language and English numerals by uttering zero to thirty. These dataset can be used separately for experimentation by the research community.

Keywords — Multimodal Biometric Database, Biometric traits, Face, Hand, Voice

#### I. INTRODUCTION

In the modern society use of the information technology based systems has pervaded all walks of life. These systems are more vulnerable to hacking and usurpation of personal identification. Security and identification are very essential for proper usage of IT based systems. Biometric technologies are becoming the basis of a wide array of highly secure identification and personal verification solutions. The biometric system allows identification/verification of individuals based on the physical(i.e. anatomical) or behavioral characteristics which cannot be stolen easily[1]. Various biometric technologies are developed such as fingerprint, face, iris, retina, voice, signature, gait, keystrokes and hand geometry. The biometric systems are categorized as either unimodal or multimodal biometric systems. Unimodal biometrics systems rely on the evidence of a single source of information i.e. physiological or behavioral information for authentication [2]. The unimodal systems have to contend with a variety of problems such as, noise in sensed data, intraclass variations, inter-class similarities, non-universality, and possibilities of spoof attacks. The inherent properties of biometric traits and the constraints of sensing technologies will limit the performance of the unimodal biometric system. Some of the limitations imposed by unimodal biometric systems can be overcome by including multiple sources of Sanjeevakumar M. Hatture Department of Computer Science and Engineering Basaveshwar Engineering College, Bagalkot - 587103, Karnataka State, India smhatture@yahoo.com

physiological or behavioral information for establishing identity [3]. Such systems, known as multimodal biometric systems, are expected to be more reliable due to the presence of multiple, (fairly) independent pieces of evidence [4]. These systems are able to meet the stringent performance requirements imposed by various applications. They address the problem of non-universality, since multiple traits ensure sufficient population coverage. They also deter spoofing since it would be difficult for an impostor to spoof multiple biometric traits of a genuine user simultaneously. The researchers are also making efforts in developing the databases along with deploying the biometric systems. The researchers constructed the "chimeric" multimodal databases by combining the biometric data from different unimodal databases to perform multimodal biometric research [5]. Such multimodal databases fail to represent real multimodal samples of the users. Hence, the construction of the "chimeric" multimodal databases are not preferred as the biometric data are uncorrelated. The multibiometric data may be necessarily correlated [6], hence there is a scope to build a new multibiometric database by acquiring the multiple trait data from the users. In the proposed work, the multimodal database (VTU-BEC-DB) is constructed by collecting the face, hand and voice biometric trait data from the 100 persons with age range between 18 and 50 years. Face images are collected indoor in a controlled scenarios with white background and variation in illumination and pose. The peg free right hand and left hand images are collected from the Book scanner with black-colour background. From every person 12 samples of the right hand and 12 samples of the left hand images are collected. Similarly, Voice database is a collected in two sessions over a period of one year. In each session, five set of recordings of English and five set of recordings of Kannada languages numerals utterance of zero(0) to Thirty(30) are recorded.

The rest of the paper is organized into four sections: section 2 reviews the developments in multimodal biometric databases. The construction of the proposed VTU-BEC-DB multimodal database is described in section 3. The scope and limitations of Database created are enlisted in section 4. Section 5 provides the conclusion of the work.

#### II. REVIEW OF RELATED WORK

Several multimodal databases are available for the researchers for experimenting and developing multimodal

Agro Guardian: A Smart Agriculture Framework for Precision Farming: Environment & Agriculture Book Chapter | IGI Global

## 10% Discount on All E-Books through IGI Global's Online Bookstore Extended

(10% discount on all e-books cannot be combined with most offers. Discount is valid on purchases made directly through IGI Global Online Bookstore (<u>www.igi-global.com (https://www.igi-global.com/)</u>)

and may not be utilized by booksellers and distributors. Offer does not apply to e-Collections and exclusions of select titles may apply. Offer expires December 31, 2022.)

### Browse Titles (https://www.igi-global.com/search/?p=&ctid=1)



# Agro Guardian: A Smart Agriculture Framework for Precision Farming

Sanjeevakumar M. Hatture (Basaveshwar Engineering College (Autonomous), India) and Susen P. Naik (Basaveshwar Engineering College (Autonomous), India)

Source Title: Modern Techniques for Agricultural Disease Management and Crop Yield Prediction (/book/modern-techniquesagricultural-disease-management/222503) Copyright: © 2020

Pages: 24

DOI: 10.4018/978-1-5225-9632-5.ch008

OnDemand PDF Download:		\$37.50
() ♥ Available	Current Special Offers	

# Abstract

The mechanization of the process creates agriculture-based jobs for farmers, providing financial support and facilitating affordable agriculture equipment and machineries. Fruits markets are subject of opportunity and it is important to the suppliers to identify the quality of fruits based on the ripeness level of fruits before selling out in order to get higher level of profit. The proposed framework is an Android application in native language of the farmer to help the jobless farmers to find agriculture-based jobs suitable to their skill set and receive investments from various investors across the country. Further, it finds investment for the needy farmers and create suitable agricultural employment for jobless farmers so that there is an increase in the progress in the field of agriculture. It also facilitates the farmers with advanced equipment for performing various agricultural tasks, obtains the land on lease, and determines various stages of ripeness of fruit and provides the information about the government project and funding facilities.

**Chapter Preview** 

# Introduction

Agriculture is acceptably in which people have begun to be lethargic, forgetting so far what is keeping them energetic and alive. Although there are many numbers of hardworking, obsessive farmers and their life runs or rests only on farming or agriculture. However, there's is lot of corruption keep's on increasing these days due the intervention of the third party in marketing the agriculture products. The main motivation behind Agricultural Marketing and Department of Agricultural Business is to ensure

Тор

#### 7/13/22, 10:21 PM

Agro Guardian: A Smart Agriculture Framework for Precision Farming: Environment & Agriculture Book Chapter | IGI Global

feasible price to the agriculture product and helps the farmer in competitive marketing scheme and also in implementing modern technologies that reduces the losses to farmer and encourages in more cultivation. India is most populated country, as compared to different economic category. Agriculture is important sector for economic in many countries like India. Presently the agriculture area contributes 18% of Gross Domestic Production (GDP). Various private and government agencies have straightly involved in agriculture area for develop the economics of the country. There is no success is obtained in agriculture sectors due to poor agriculture price, difficulties in contacting the dealers, lack of technology information and network connectivity etc. No proper mechanism or system to alleviate these complications.

The fundamental desire of farming is to bring improved marketing facilities, reduce trade performance, and eliminate market expenditure which facilitates the prominent budget to the efforts made by the farmers. Farmer is one of the most important people in every human being's life since the beginning of civilization. India gains food only if farmer cultivates the crops and manages many agriculture activities. There is much responsibility to farmer which makes their life tough and difficult. Even though farmer is rich enough to serve the food for every individual of the society but poor himself due to improper pricing for their cultivation, illiteracy, lack of the global knowledge, dependency on single source. Farmer remains happy if the cultivated crops are in good quality by the favorable environmental condition and better pricing. Since, the farmer's family is depending on cultivation if crops are not as per the demands then Farmer life becomes depressed. Farming work is a low admired job, especially regular agricultural workers. In this modern era industry and construction sectors have occupied majority of agriculture's labors due to the high wages paid. Most of the times small scales farmer depend on the external sources like relatives, companions, friends and others and they end up in giving high interest rates (36 to 120% per year) for requirements that might be in the form of money. Despite of the large machineries and modern technology involved in agriculture some parts of the country agricultural operations and mechanisms are carried out by human hands or labor using simple and traditional tools for implementing wooden, plough, seeds sowing etc. It results in huge investments of money on human labor. Land is the most valuable property in Framer life. For many new farmers, especially in areas where lands are quite expensive, it becomes very difficult to afford for their own the land to farmers and searching land for cultivation becomes complicated. The entire factor provides bad impact on the economic situation.

Growths of local fruit or vegetables industries in India are very immense. But, exporting fruits to other is country is less compared to other products. Fruits market is always option for consumers. It is very essential for suppliers and farmers to identify and deal the quality of fruits before sending out to market. Currently, human beings are experts' in grade the agriculture products that is by examining on the visibility features. However, physical analysis provides the inaccurate, incompatible and ineffectual on denigrating the quality of farming products. The physical analysis after a certain period of times the farmer gets board of this job. In order to overcome this problem, there should be a automatic analysis for determining the quality of fruits which provide the results in more accurate and agreeable output. By these results, it is beneficial in saving time and manual labor which in turn helps in the development of economic.

# Complete Chapter List

Search this Book:	Full text search terms				
Reset					
Editorial Advisor	ry Board			View Full PDF (/pdf.aspx? tid=234088&ptid=222503&ctid=15&t=Editorial Advisory Board&isxn=9781522596325)	
Table of Conten	ts			View Full PDF (/pdf.aspx? tid=234089&ptid=222503&ctid=15&t=Table of Contents&isxn=9781522596325)	
Detailed Table o	f Contents			View Full PDF (/pdf.aspx? tid=234090&ptid=222503&ctid=15&t=Detailed Table of Contents&isxn=9781522596325)	
Foreword Rajanish Kamalakar Kamat			View Full PDF (/pdf.aspx? tid=234091&ptid=222503&ctid=15&t=Foreword&isxn=9781522596325)		
Preface		View Full PDF (/pdf.aspx? tid=234092&ptid=222503&ctid=15&t=Preface&isxn=9781522596325)			
Acknowledgmer N. Pradeep, Sande	nt eep Koutish, C. R. Nirmala, Vishal Goyal	View Full PDF (/pdf.aspx? tid=234093&ptid=222503&ctid=15&t=Acknowledgment&isxn=9781522596325) Iirmala, Vishal Goyal, Sonia Abdeliatif			
7/13/22, 10:21 PM

Chapter 1	\$37.50
Issues and Challenges in Smart Farming for Sustainable Agriculture (/chapter/issues-and-challenges-in-smart-farming-for-sustainable- agriculture/234094) (pages 1-22)	
Immanuel Zion Ramdinthara. Shanthi Bala P.	
Sample PDF (/viewtitlesample.aspx?id=234094&ptid=222503&t=Issues and Challenges in Smart Farming for Sustainable Agriculture&isxn=9781522596325)	
Chapter 2	\$37.50
Image Processing Techniques Aiding Smart Agriculture (/chapter/image-processing- techniques-aiding-smart-agriculture/234095) (pages 23-48)	
Aspira S. Tripathy, Deepak Kumar Sharma	
Sample PDF (/viewtitlesample.aspx?id=234095&ptid=222503&t=Image Processing Techniques Aiding Smart Agriculture&isxn=9781522596325)	
Chapter 3	\$37.50
Expert System Design for Diagnosis of Diseases for Paddy Crop (/chapter/expert- system-design-for-diagnosis-of-diseases-for-paddy-crop/234096) (pages 49-65)	
Sreekantha Desai Karanam, Deepthi M. B.	
Sample PDF (/viewtitlesample.aspx?id=234096&ptid=222503&t=Expert System Design for Diagnosis of Diseases for Paddy Crop&isxn=9781522596325)	
Chapter 4	\$37.50
Deep Learning and Computer Vision in Smart Agriculture (/chapter/deep-learning- and-computer-vision-in-smart-agriculture/234097) (pages 66-88)	
Shiv Kumar, Agrima Yadav, Deepak Kumar Sharma	
Sample PDF (/viewtitlesample.aspx?id=234097&ptid=222503&t=Deep Learning and Computer Vision in Smart Agriculture&isxn=9781522596325)	
Chapter 5	\$37.50
Computer Vision for Green Plant Segmentation and Leaf Count (/chapter/computer- vision-for-green-plant-segmentation-and-leaf-count/234098) (pages 89-110)	
Praveen Kumar J., Domnic S.	
Sample PDF (/viewtitlesample.aspx?id=234098&ptid=222503&t=Computer Vision for Green Plant Segmentation and Leaf Count&isxn=9781522596325)	
Chapter 6	\$37.50
Automatic Data Acquisition and Spot Disease Identification System in Plants Pathology Domain: Agricultural Intelligence System in Plant Pathology Domain (/chapter/automatic-data-acquisition-and-spot-disease-identification-system-in- plants-pathology-domain/234099) (pages 111-141)	
Rajesh T. M., Kavyashree Dalawai, Pradeep N.	
Sample PDF (/viewtitlesample.aspx?id=234099&ptid=222503&t=Automatic Data Acquisition and Spot Disease Identification System in Plants Pathology Domain: Agricultural Intelligence System in Plant Pathology Domain&isxn=9781522596325)	

Chapter 7	\$37.50
Applications of Data Mining Techniques in Smart Farming for Sustair (/chapter/applications-of-data-mining-techniques-in-smart-farming-fo agriculture/234100) (pages 142-178)	able Agriculture r-sustainable-
Anandhavalli Muniasamy	
Sample PDF (/viewtitlesample.aspx?id=234100&ptid=222503&t=Applicatio Techniques in Smart Farming for Sustainable Agriculture&isxn=978152259	ns of Data Mining 6325)
Chapter 8	\$37.50
Agro Guardian: A Smart Agriculture Framework for Precision Farmin (/chapter/agro-guardian/234101) (pages 179-202)	9
<mark>Sanjeevakumar M. Hatture, Susen P. Nai</mark> k	
Sample PDF (/viewtitlesample.aspx?id=234101&ptid=222503&t=Agro Gua Agriculture Framework for Precision Farming&isxn=9781522596325)	rdian: A Smart
Chapter 9	\$37.50
A Study on Technology-LED Solutions for Fruit Grading to Address F Handling Issues of Horticultural Crops (/chapter/a-study-on-technolog for-fruit-grading-to-address-post-harvest-handling-issues-of-horticulture crops/234102) (pages 203-221)	'ost-Harvest gy-led-solutions- ıral-
Arun Kumar R., Vijay S. Rajpurohit, Sandeep Kautish	
Sample PDF (/viewtitlesample.aspx?id=234102&ptid=222503&t=A Study o LED Solutions for Fruit Grading to Address Post-Harvest Handling Issues o Crops&isxn=9781522596325)	n Technology- f Horticultural
Chapter 10	\$37.50
Optimized Data Mining Techniques for Outlier Detection, Removal, a Zone Delineation for Yield Prediction (/chapter/optimized-data-mining outlier-detection-removal-and-management-zone-delineation-for-yiel prediction/234103) (pages 222-258)	nd Management g-techniques-for- d-
Roopa G. M., Arun Kumar G. H., Naveen Kumar K. R., Nirmala C. R	
Sample PDF (/viewtitlesample.aspx?id=234103&ptid=222503&t=Optimized Techniques for Outlier Detection, Removal, and Management Zone Delinea Prediction&isxn=9781522596325)	l Data Mining ation for Yield
About the Contributors	View Full PDF (/pdf.aspx? tid=234105&ptid=222503&ctid=17&t=About the Contributors&isxn=9781522596325)
Index	View Full PDF (/pdf.aspx? tid=234106&ptid=222503&ctid=17&t=Index&isxn=9781522596325)

179

## Chapter 8 Agro Guardian: A Smart Agriculture Framework for Precision Farming

### Sanjeevakumar M. Hatture

Basaveshwar Engineering College (Autonomous), India

### Susen P. Naik

Basaveshwar Engineering College (Autonomous), India

### ABSTRACT

The mechanization of the process creates agriculture-based jobs for farmers, providing financial support and facilitating affordable agriculture equipment and machineries. Fruits markets are subject of opportunity and it is important to the suppliers to identify the quality of fruits based on the ripeness level of fruits before selling out in order to get higher level of profit. The proposed framework is an Android application in native language of the farmer to help the jobless farmers to find agriculture-based jobs suitable to their skill set and receive investments from various investors across the country. Further, it finds investment for the needy farmers and create suitable agricultural employment for jobless farmers so that there is an increase in the progress in the field of agriculture. It also facilitates the farmers with advanced equipment for performing various agricultural tasks, obtains the land on lease, and determines various stages of ripeness of fruit and provides the information about the government project and funding facilities.

### INTRODUCTION

Agriculture is acceptably in which people have begun to be lethargic, forgetting so far what is keeping them energetic and alive. Although there are many numbers of hardworking, obsessive farmers and their life runs or rests only on farming or agriculture. However, there's is lot of corruption keep's on increasing these days due the intervention of the third party in marketing the agriculture products. The main motivation behind Agricultural Marketing and Department of Agricultural Business is to ensure feasible price

DOI: 10.4018/978-1-5225-9632-5.ch008

Ashish Kumar Luhach · Dharm Singh Jat · Kamarul Bin Ghazali Hawari · Xiao-Zhi Gao · Pawan Lingras (Eds.)

Communications in Computer and Information Science 1076

# Advanced Informatics for Computing Research

Third International Conference, ICAICR 2019 Shimla, India, June 15–16, 2019 Revised Selected Papers, Part II









Х

### **Contents – Part II**

### Hardware

Energy Harvesting System from Household Waste Heat Employing Thermoelectric Generator	3
Design of Microstrip Patch Antenna Using E-Shaped Slots for Multiband Applications Deepa Pundir and Narinder Sharma	13
An Efficient Design of Staircase Circular Slot Shaped Antenna for Wireless Application	23
Reduction of Fault Tolerant Memory Testing Problem Using Dynamic Fault Injection Technique	35
Low-Voltage Low-Power Bulk-Driven and Bulk-Driven-Quasi-floating- Gate Neural Amplifier Design for EEG Lipika Gupta, Amod Kumar, and Sartajvir Singh	45
Transfer Time Optimization Between CPU and GPU for Virus Signature Scanning Apurva Anil Dhake and Sandip M. Walunj	55
Impact of Dummy Logic Insertion on Xilinx Family for Hardware         Trojan Prevention         Navneet Kaur Brar, Anaahat Dhindsa, and Sunil Agrawal	64
Development of EMG Controlled Electric Wheelchair Using SVM and kNN Classifier for SCI Patients Bhupender Kumar, Yogesh Paul, and Ram Avtar Jaswal	75
Design and Optimization of Synchronous Counter Using Majority Gate-Based JK Flip-Flop	84
Smart Mirror Using Raspberry Pi for Human Monitoring and Home Security	96

### Smart Mirror using Raspberry Pi for Human Monitoring and Home Security

Raju A Nadaf<sup>1</sup>, Sanjeev Kumar Hatture<sup>2</sup>, Praveen S Challigidad<sup>3</sup>, and Vasudha M Bonal<sup>4</sup>

Basaveshwar Engineering College, Department of Computer Science and Engineering, Bagalkot 587 102, India

 $[raj.enggs, smhatture, praveensc] @gmail.com, vasudha\_125 @rediffmail.com] \\$ 

Abstract. The Smart mirror is a system in which the regular mirror is converted into a smart device. The Smart mirror is designed using Raspberry Pi and a touch enabled screen. The designed system is capable of acting like a regular mirror in case of normal mode of operation and it acts like a smart mirror in a triggered mode of operation. The Smart mirror thus designed is an interactive system which is capable of accepting two modes of input namely touch and mobile based controls. The system is designed to display weather information, temperature and latest news on the mirror. The system is primarily designed as a home Security and Human Monitoring system. The proposed design is thought of as a package bundled with better features, which not only just displays information over screen, but also can be used for providing security. The system is built using hardware units like Raspberry Pi 3 model, touch screen, mobile device, camera and Python coding is used for software part. The system provides security against Intrusion in home. It is done using Background Subtraction along with Simple Frame Difference Approach. Once Intrusion is detected the administrator is sent alert message along with the photo of intruder. The Human Monitoring is implemented using Machine learning technique Yolo with OpenCV. During human monitoring mode of operation, if the Human under monitoring moves out of the vision range of camera, the administrator of Smart Mirror will be sent alert message. Using the mobile commands, the administrator can see the video streaming using camera fixed on the Smart Mirror.

**Keywords:** Smart Mirror, Raspberry Pi, Security, Intrusion Detection, Human Monitoring Mirror

### 1 Introduction

In the present world, the advancement in technology has converted almost every device as smart device. Ranging from household things to most advanced electronic gadgets, almost all are becoming smart. So having this thought in mind, here a smart is proposed, which is not only capable of displaying customized information on the display screen but also act smartly and provide security at home when needed. It is also capable of Monitoring Humans. People use mirrors



|| Jai Sri Gurudev || Sri Adichunchanagiri Shikshana Trust ®







ADICHUNCHANAGIRI INSTITUTE OF TECHNOLOGY CHIKKAMAGALURU

## DEFARTMENT OF COMPUTER SCIENCE & ENGINEERING

## FIRST IEEE INTERNATIONAL CONFERENCE ON ADVANCES IN INFORMATION TECHNOLOGY

## 24th - 27th JULY 2019













Sl. No.	Paper ID [PID]	Paper Title	Author's	Page No.
13	ICAIT-031	Sarcasm Detection in Text Data to Bring Out Genuine Sentiments for Sentimental Analysis	Adarsh M J Dr. Pushpa Ravikumar	94-98
14	ICAIT-034	A Survey on Chat-Bot System for Agriculture Domain	Prashant Y. Niranjan, Vijay S. Rajpurohit Rasika Malgi	99-103
15	ICAIT-041	Cricket Squad Analysis Using Multiple Random Forest Regression	Nigel Rodrigues Nelson Sequeira Stephen Rodrigues Varsha Shrivastava	104-108
16	ICAIT-043	Agro-Guardian: A Framework for Smart Agriculture	Sanjeevakumar M. Hatture Susen P. Naik	109-115
17	ICAIT-044	Smart Mirror Using Raspberry Pi for Human Monitoring and Intrusion Detection	Raju A. Nadaf Rubina M Sujatha P Vasudha M. Bonal	116-121
18	ICAIT-046	Driver Distraction Monitoring and Alerting System	Abhishek Walavakar Satyam Singh Roshan Salian Varsha Shrivastava	122-127
19	ICAIT-048	Segmentation of Arecanut Bunches using YCgCr Color Model	Dhanesha R, Dr. Shrinivasa Naika C. L. Dr. Kantharaj Y	50-53
20	ICAIT-054	Hybrid Cross Layer Mechanism for High Reliability in Under Water Wireless Sensor Network	Vidyalakshmi K Dr. M. Siddappa Dr. B. Shanmukha	128-134
21	ICAIT-056	Object Detection and Human Identification using Raspberry Pi	Hemaya S Fathima Khan Gowtham Ghatge R Dr. Sunitha M R	135-139
22	ICAIT-057	Survey of Hybrid VANET Design for Provisioning Infotainment Application	Shalini S Annapurna P Patil	140-145
23	ICAIT-058	Secure Transfer of Image-Acquired Text Using a Combination of Cryptography and Steganography	S.J.Fiona G.Sathiaraj Grishma P. Pingale Souvik Majumdar Saniya J.Shaikh Bhushan S. Thakare	146-152
24	ICAIT-063	Restoration of Blur Video using Blind Deconvolution Method	Manasa T. K Kavya K. P Lakshmi B Harshini B. Sangareddy B K	153-158
25	ICAIT-065	Noise Cancellation and Noise Reduction Techniques: A Review	Shraddha C, Chayadevi M L Anusuya M A	159-166
26	ICAIT-068	A Framework for Image Selection for Image Fusion Using Crowd Sourced Data	Pavan Kunchur Rohan Raju Dhanakshirur Sameer Ambekar Sadhana Bangarashetti	167-173

## KRISHI-SUVIDHA: A Framework For Smart Agriculture

Sanjeevakumar M. Hatture Department of Computer Science and Engineering Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India smhatture@gmail.com Susen P Naik Department of Computer Science and Engineering Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India susennaik55@gmail.com

Abstract. Agriculture is the backbone of the country. About 70% percentage of India population belongs to agriculture family. Due to the digital India programme, agricultural applications have direct impact on the agricultural sectors. The way people go for buying the agricultural products is tedious. Often buyers or customers have to travel distant places to buy agricultural products and sometimes even the right quality is not ensured. Besides, farming is the main or primary occupation in India. Farmers are usually deceiveed by the middleman in today's market which leads to scarcity of grains. Fruits markets are subject of opportunity and it is important to the suppliers to identify the quality of fruits based on the ripeness level of fruits before selling out in market, in-order to get higher level of profit. The main objective of this work is to help farmers. An android application in native or regional language of the farmer to help the jobless farmers to find agriculture based jobs suitable to their skill set and receive investments from various investors across the country. Further to find investment for the needy farmers and create suitable agricultural employment for jobless farmers so that there is an increase in the progress in the field of agriculture. The prposed work of an android application is also facilitate the farmers with advanced equipments for performing various agricultural tasks, obtain the land on lease and it helps buying and selling agricultural products using a computerized approach, provide the scheme available to farmer and to determine various stages of ripeness of fruit. The application is simple and easy to use by the farmers and accumulate several agri based information at single place in multilingual applications.

*Keywords*— Agriculture jobs, Android application, Cultivation, Finance and investments, Land lease, Marketing, Ripe track

### 1. INTRODUCTION

Agriculture is the primary job in India. Although there are huge number of hardworking, obsessive farmers and their life runs only on farming but agriculture is not given primary importance. However there's is lot of corruption increasing these days due to the intervention of the third party in marketing the agriculture products. Agriculture is important sector for the increase in the economic for many developing countries like India. Presently the agriculture area contributes 18% of Gross Domestic Production (GDP) [1]. Various private and government agencies have directly involved in agriculture fields for developing the economics of the country. There is no direct success is obtained in agriculture sectors due to poor agriculture price, difficulties in contacting the dealers, lack of technology information and network connectivity etc. Farming work is a low admired job, especially regular agricultural workers. In this modern era industry and construction sectors have occupied majority of agriculture's labors due to the high

wages being paid. Despite of the large machineries and modern technology involved in agriculture, some parts of the country agricultural operations and mechanisms are carried out by human hands or labor using simple and traditional tools for implementing wooden, plough, seeds sowing etc. It results in huge investments of money on human labor. Land is the most valuable property in Framers' life. For many new farmers, especially in areas where lands are quite expensive, it becomes very difficult to afford for their own the land to farmers and searching land for cultivation becomes complicated. Fruits market is always option for consumers. It is very essential for suppliers and farmers to identify and deal with the quality of fruits before sending out to market. Currently, human beings are experts in grading the agriculture products that is by examining on the visibility features. However, physical analysis provides the inaccurate, incompatible and ineffectual on denigrating the quality of farming products. In order to overcome this problem there should be an automatic analysis for determining the quality of fruits which provide the results in more accurate and agreeable output. By these results, it is beneficial in saving time and manual labor which in turn helps in the development of economy.

In the 21<sup>th</sup> century all the human beings live in modern, digital and technical adapted for their daily requirements. The smart phone is most commonly used all over, and farmers have started to use mobile phone for communication [2]. But they need to recognize the significance of smart phone and new technology, so it can help them in direct communication involving the farmers and buyer or customers. By investigating and considering the entire problem faced by farmers, the author has introduced an android application which involves various features.

Farmers need to download this application on their mobile phones so that they can easily search for jobs online and also post ads regarding the jobs. Most farmers now have to understand the operating and maintaining sophisticated machinery. So it provides them with an opportunity to post their machineries for rent to other farmers and also help the other farmers to find machineries for their work [3]. This application also helps the farmers in finding the investors who are ready to invest on lands also there is another feature which helps them in finding lands for lease. It helps them in selling and buying agriculture commodities and services locally without middlemen through an add/listing which they can post right from their mobile. It also helps the farmer to get most updated schemes available to farmer. Finally it helps thefarmer to check fruit ripeness level easily at any place and anytime.



|| Jai Sri Gurudev || Sri Adichunchanagiri Shikshana Trust ®







ADICHUNCHANAGIRI INSTITUTE OF TECHNOLOGY CHIKKAMAGALURU

## DEFARTMENT OF COMPUTER SCIENCE & ENGINEERING

## FIRST IEEE INTERNATIONAL CONFERENCE ON ADVANCES IN INFORMATION TECHNOLOGY

## 24th - 27th JULY 2019













## **Table of Contents**

Sl. No.	Paper ID [PID]	Paper Title	Author's	Page No.
1	ICAIT-007	Impact of Bandwidth on LANDSAT-7 ETM+ Image Quality using Gaussian Filter: Bangalore, Karnataka State, INDIA	Dr. Chethan K S Roopa K Swamy Sinchana G S Sowkya H K Sujith J Dr. Choodarathnakara A L	1-7
2	ICAIT-008	Recommender System Using Machine Learning	Priyanka M B Sharathkumar S Pruthvi M Manushree M.R Shubham Ankit Karan	8-12
3	ICAIT-010	Age Estimation of Person Based on Face Features	Sanjeevakumar M Hatture Ashwini Gouripur	13-18
4	ICAIT-012	Secure Video Setganography Technique Using DWT and H.264	Renuka B Dr. N Manja Naik	19-23
5	ICAIT-016	Automated Detection of White Blood Cells Cancer Disease	Lakshmi K.G Dr. N Manja Naik	24-28
6	ICAIT-017	Design and Realization of Two Stage LNA for X-Band TTC Receiver	Vinay Doki Alamelu Sunder Devadas Shenoy Y Dr. Venkataramana D	29-32
7	ICAIT-020	Profiling Social Media Users, a Content-Based Data Mining Technique for Twitter Users	Dr. Vasanthakumar G U Dr. Shashikumar D R Dr. Suresh L	33-38
8	ICAIT-023	Voice Assistant and Facial Analysis Based Approach to Screen Test Clinical Depression	Pragna Mallikarjuna Swamy Prithviraj Janardhan Kurapothula Rachana Ravikumar Sachin Venkatesha Murthy Harini S Karthik Kashyap	39-44
9	ICAIT-024	Readability Analysis of Kannada Language	Vishwaas Narasinh	45-49
10	ICAIT-026	Bone Cancer Detection Using K-Means Segmentation and KNN Classification	Ranjitha M M Taranath N L Arpitha C N Dr. C K Subbaraya	76-80
11	ICAIT-027	Survey on Deep Reinforcement Learning Protocol in VANET	Shweta S Doddalinganavar P V Tergundi Rudragouda S. Patil	81-86
12	ICAIT-030	Analysis of Human Intelligence in Identifying Persons Native through the Features of Facial Image	Vani A. Hiremani Kishore Kumar Senapati	87-93

## Age Estimation of Person Based on Face Features

Sanjeevakumar M. Hatture Department of Computer Science and Engineering Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India smhatture@gmail.com

Abstract— In the era of information technology the social media and other sources of evidence are usually used to spot suspects and victims of crimes by police. Investigators generally use individual characteristics like gender, age, weight, height, etc. from unknown speculate to testimony in their perception. For example in child abuse investigations victim essential in the detection age is of the offenses' classification. Machine learning methods are used in the analysis of digital photographs to identify soft biometric. Investigator benefits from this to improve their cases. This paper presents an efficient intelligent system for age estimation of a person's by employing the facial features. The facial information employing the deep learning technique that is transfer learning based on Convolutional Neural Networks will estimate the age of the person. The method contains pretrained model, convolutional base, and classifier. The proposed method is efficient and outer performs the state-of-the-art techniques.

### Keywords— Age, Convolutional Neural Network, Deep Learning, Face, Intelligent system, Transfer Learning,

### I. INTRODUCTION

Biometric is branch of computer science that studies the traits of humans for identification, access control or surveillance purposes. Digital image accomplish various aspects like gender, age and facial expression by quick glance. From the last few years, facial identification is the main stream media for various products particularly in deliverance of Apple iPhone X. In this product, primary authentication technology that is facial recognition that overcomes the traditional finger print authentication. Among the various applications sophisticated facial recognition is in effective improvement all over worldwide. Facial recognition technology has used by china in several applications such as pay with smile and jaywalker identification. Also, used in the churches to track worshipers and to stop shoplifters. Age estimation is used for broad area of offenses particularly in child abuse inspection, adult entertainment; purchase age limited things such as liquor and tobacco, and also restricted age videos. Among many fields researchers are age estimation of subjects is often difficult, since it is sophisticated and annoying task to urge done and by applying the various methods of machine learning algorithms constantly rising the precise obtained outcome. Moreover, differentiate the marginally under-age from the marginally super annulled is required but present techniques are not trustworthy to accomplish job. For rhetorical investigation scanning the surface of disk for important price has long been time overwhelming task. Nevertheless, techniques of machine learning and computer vision may be based on digital rhetorical specialist achieves automatic file techniques, flag entirely completely different quite information relevant to case in court and in digital rhetorical laboratories scale back the exposure of child abuse material.

Ashwini Gouripur Department of Computer Science and Engineering Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India ashwinigouripur@gmail.com

Throughout the globe abuse inquiry is usual incident on social control bureau, during this image classification based on age, gender is difficult and usually for digital detector crucial to decisive each image location. Equally, classifying the identical person whereas particularly not aware of their identification is may be very important potentiality. Shortage of data for training and testing is foremost needed issues in machine learning. Hence, there is a scope to develop algorithm/method to deal with this issue. This paper presents an efficient intelligent system for age estimation of person's employing the facial features. Facial information bv employing the deep learning technique will estimate the age of the person. The method collects the real time facial information of a person for estimating the gender and age the proposed method for age estimation in this paper is further evaluated against the state-of-the-art techniques to validate its performance. The rest of the paper is arranged in such way that section 2 describes literature review, section 3 explains issues and challenges, section 4 describes proposed system, section 5 explains experimentation and results and conclusions are drawn in section 6.

### II. RELATED WORK

Many researchers enumerable the age of person's by were arising with the formula for state-of-the-art techniques. Varieties of the recent techniques conferred within the papers are summarized as follows:

In a various case, exponentially developing storage capacities and at the same time there is need for digital rhetorical analysis, as result of this digital rhetorical expertise requires enormously accumulated amount of data than present capacities in social control bureau around the globe. Due to inadequate evidence there is delay in court cases and this results cases being dismissed [1]. There is less tendency of pursuance because of the anomaly in determinatives in the digital image the age of a victim depict [2]. Throughput of digital forensic laboratories influenced by the particular factors so for this reason blocking digital rhetorical laboratories in the future [3]. Among the right estimation of faces on own-age bias there are 114 participants were extra corrected to estimate the particular ages within their particular age among in an agegroup (10-15, 21-31 and 61-75 years old) found [4],[5]. A number of factors such as gender and facial expressions are negatively impacted the accuracy of human age estimation. The highest accuracy results in neutral expression but less accuracy results in other expression [6], [7].

The victim age is difficult to figure out in extreme cases like Child Abuse Material (CAM), in these cases in era where plenty of digital investigators wasted time in processing like this cases. Soft biometric of age attribute is hard to predict because of variation between face and body choices, i



### International Conference on Computational Intelligence and Data Science (ICCIDS 2019)

6<sup>th</sup>-07<sup>th</sup> September, 2019

## The NorthCap University, Gurugram

### All the accepted papers will be published in Procedia Computer Science Journal, Elsevier (https://www.journals.elsevier.com/procedia-computer-science)

(https://www.journals.elsevier.com/procedia-computer-science) (https://www.journals.elsevier.com/procedia-computerscience)

(https://www.journals.elsevier.com/procediacomputer-science)



(http://www.ncuindia.edu)



UNIVERSITY of



	Uttam Singh Singh, Shweta Kedas, Sikakollu Prasanth, Arun Kumar, Vijay Bhaskar Semwal and Vinay Anand	
590	Tikkiwal	Design of A Recurrent Neural Network Model for Machine Reading Comprehension
593	Vandana Singh, U. Srinivasulu Reddy and G. Manju Bhargavi	A Generic and Robust System for Automated Detection of Different Classes of Arrhythmia
594	Siva Raja P M and Ramanan K	Immune Cloud Storage In Iot By Feature Selection Rely On Learning With Crypto Based Steganography
		Session # 2.7
	07th September, 20	)19; Saturday, Room No: RN 309
	Time:	9:00 AM to 11:00 AM
	Raju A Nadaf, Sanjeevakumar M Hatture, Vasudha M	
595	Bonal and Susen P Naik	Home Security against Human Intrusion using Raspberry Pi
598	Pardeep Garg and Sunil Datt Sharma	Sensitivity Enhancement of DWT Based Algorithm for Detection of CpG Islands in DNA Sequences
603	M Sridevi and Mayuri Kharde	Video Summarization Using Highlight Detection and Pairwise Deep Ranking Model
	Divyani Mittal, Pooja Suthar, Pranaya Padamata, Madhuri	
608	Patil, Dipti Rana and Bharat Tidke	Social Network Influencer Rank Recommender Using Diverse Features from Topical Graph
611	Rashmi Bhardwaj and Dimple Pruthi	Evolutionary Techniques for Optimizing Air Quality Model
569	Ayushi Verma, Neetu Sardana and Sangeeta Lal	Developer Recommendation for Stack Exchange Software Engineering Q&A Website based on K- Means clustering and Developer Social Network Metric
612	Pooja Sarin, Arpan Kar, Kiran Kewat and P.Vigneswara Ilavarasan	Factors Affecting Future of Work: Insights from Social Media Analytics
		Sector # 2.9
	07th Sontombor 20	Session # 2.8
	Time	9.00 AM to $11.00$ AM
613	Tauseef Khan and Ayatullah Faruk Mollah	Text non-text classification based on area occupancy of equidistant pixels
616	Sangeeta Lal, Lipika Tiwari, Ravi Ranjan, Ayushi Verma, Neetu Sardana and Rahul Mourya	Analysis and Classification of Crime Tweets
623	Neeti Kashyap, A. Charan Kumari and Rita Chhikara	Multiobjective Optimization for service composition in IoT using NSGA II
279	Sudarshan Sonawane and Satish Kolhe	TCSD: Term Co-occurrence Based Sarcasm Detection from Twitter Trends



Available online at www.sciencedirect.com



Procedia Computer Science 167 (2020) 1811-1820



www.elsevier.com/locate/procedia

### International Conference on Computational Intelligence and Data Science (ICCIDS 2019)

### Home Security against Human Intrusion using Raspberry Pi

Raju A Nadaf<sup>a</sup>, S.M. Hatture<sup>a</sup>, Vasudha M Bonal<sup>a</sup>, Susen P Naik<sup>b</sup>

<sup>a</sup>Basaveshwar Engineering College, Vidyagiri, Bagalkot and 587 102, India <sup>b</sup>KLEIT, Gokul, Airport Road, Hubli and 580 030, India

### Abstract

Technology has changed the world. There is a huge growth in technology connected to almost every field. The pace and diversity of advancement is very rapid. Hence, new technology and techniques are coming up and are put to use as per the need. In such advanced scenario, providing security to home has also become a major point of concern. Presently security cameras can be used for the same. But such cameras may be visible to intruders and there is possibility that cameras may be damaged. Hence, in order to serve the purpose, there is a need to find an alternative secure, accurate and quick method. Hence, an effort is made in this paper to design a home security system using Raspberry Pi. System requires Raspberry Pi, Camera, Touch screen and android mobile as hardware components. Python, Node.js and OpenCV libraries are used for software coding. Usually most of the intrusion detection systems click the photo of intruders as soon as the human is detected. But, such systems fail to click photos, so that the intrusion detection is accurate and photos thus captured are clear, so that intruder is clearly visible. The proposed system is designed as a smart mirror which will provide both information and home security. The system is developed to accept touch and mobile commands. As soon as the intrusion is detected, an alert message along with the identifiable and clear photo (Face view) of intruder will be sent to the owner's/administrator's mobile. The mirror owner can also see the video of deployed environment through the camera fitted on smart mirror.

© 2020 The Authors. Published by Elsevier B.V.

This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/) Peer-review under responsibility of the scientific committee of the International Conference on Computational Intelligence and Data Science (ICCIDS 2019).

Keywords: Smart Mirror; Home Security; Intrusion Detection; Raspberry Pi; Frontal Face Detection;

### 1. Introduction

The comfort zone of human beings is increasing, as the facilities are increasing. With the incorporation of many technologies into day today activities, life has become easier. On the other side, as far as the technologies are concerned, it also creates security related issues. For example, we use ATMs for doing banking transactions, the security of device and the working accuracy of device matters a lot. Similarly, with the advancement in technology, it is evident that skills of thieves, robbers also have increased. Hence, it is a great challenge to design a foolproof system. Usually people make use of security cameras in order to secure home. The cameras record the activities that

\* Corresponding author. Ph.: 9741656378 ; *E-mail address:* raj.enggs@gmail.com

1877-0509 © 2020 The Authors. Published by Elsevier B.V.

This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/) Peer-review under responsibility of the scientific committee of the International Conference on Computational Intelligence and Data Science (ICCIDS 2019).

10.1016/j.procs.2020.03.200

E NORTHCAP UNIVERSITY (Formerly ITM University, Gurugram) NAAC ACCREDITED International Conference on Computational Intelligence and Data Science (ICCIDS 2019) 6<sup>th</sup> -7<sup>th</sup> September 2019 Certificate This is to certify that Prof./Dr./Mr./Ms. <u>Braveen Challagidad</u> of <u>Basaveshwan Engineers</u> ng (<u>a llege (Antonom aus)</u>, <u>Bagalkot</u> has presented/participated a paper titled <u>Efficient Nulli- auth writy</u> Arrens (and Deligation in Youd Febreage. (<u>authal using Attribute - based Encryphion</u>in the "International Conference on Computational Intelligence and Data Science (ICCIDS 2019)" organized by the Dept. of Computer Science & Engineering , The NorthCap University, Gurugram, in association with University of Dayton, Ohio, USA on 6<sup>th</sup>-7<sup>th</sup> September 2019. Dr Meghna Sharma Dr Kavita Khanna Dr Vijendra Singh (HOD, CSE&IT) (Organizing Chair) (Organizing Chair) The NorthCap University, Sec-23A, Gurugram-122017 • www.ncuindia.edu

### Experimental Investigation of High-Strength Self-Compacting Fibre-Reinforced Concrete



Aijaz Zendeo and R. B. Khadiranaikaro

Abstract The improvement and development of new techniques in material science are progressively rapid. An experimental programme was conducted to assess the influence of addition of steel fibres to self-compacting concrete (SCC). The aim of present experimental work is to study the effect of superplasticizer and steel fibres on the properties of fresh and hardened SCC. Based on the properties of constituent materials, various trials mixes were produced by varying material proportions, w/ e ratio and superplasticizer dosage. An optimum mix design satisfying all requirements of SCC was obtained. Steel fibres of varying percentages from 0 to 5% were added to optimum mix to carry out study on the properties of fresh and hardened SCC. Slump flow, V-fannel and L-Box tests were conducted to study the properties of fresh SCC. Cubes and cylinders were cast cured for 28 days and tested for compressive strength and split tensile strength to study the properties of hardened SCC. The results obtained indicate a direct impact of dosage of superplasticizer and percentage content of fibres on the properties of fresh and hardened SCC.

Keywords Self-compacting - High strength - Steel fibres - Silica fume

© Springer Nature Singapore Pie Ltd. 2019 B. B. Das and N. Neithalath (eds.), *Sustainable Construction and Building Materialis*, Lecture Notes in Civil Engineering 25, https://doi.org/10.1007/978-981-13-3317-0\_32

345

A. Zende (0-3)

A. Zende (202) Department of Civil Engineering, BLDEA's V.P. Dr. P.G. Halakatti College of Engineering and Technology. (Affiliated to Visvesvaraya Technological University, Belagavi), Vijayapar 586103, India e-mail: aijar.52964@gmail.com

R. B. Khadiranaikar

Department of Civil Engineering, Basaveshwar Engineering College, The Government Aided Institution Affiliated to Visvesvuraya Technological University Belgaum, Bagalkot 587102, India





2019 IEEE Bombay Section Signature Conference (IBSSC)

## Congestion Control in Cloud Computing Network for Load balancing using Portability

Tej C. Hiremath Department of Computer Science & Engineering B G M Institute of Technology Mudhol, India tej.h21@gmail.com

Abstract--- The major obstacles in cloud usage is its non-flexible attribute with respect to portability of applications and congestion due to enormous usage of cloud applications by users. Customers of cloud computing are unable to access the services offered by one cloud over the other. Applications running on one cloud environment is bound to the ordinance and principles of provisioning of the same. The customers we are referring here are mobile in nature. The non-flexible attribute we are referring here is cloud vendor lock-in. We have proposed a new scheme called Cloud Application Migration Management Model (CAM3) that encompasses service Cloning for flexible and versatile usage. In this scheme, we plan to clone the application proffered by cloud vendor and install in another homogeneous hybrid cloud environment. The service application migrated can be made accessible by Application Re-engineering and Code Re-factoring techniques, thus making the cloud environment elastic and versatile.

Keywords- Cloud Computing, Portability, Cloning

#### I. INTRODUCTION

Over the past decade, cloud computing technology has flared-up as an auspicious model with the principles of coherent and multi-occupant serving nature. With its widespread service spectrum stack, provisioning propensity and pay-per-use themes, it has positioned itself among the top five valuable as well as cost-effective service delivering technologies of the present IT market. Cloud computing can be illustrated as a multi-dimensional offering cum computing services delivery model, where in necessary resources are provisioned to the customers over the internet with guaranteed Quality of Service (QoS). The diverse computing services offered by the cloud stack encompass on-demand self-utility, offhand maintenance, large-scale network accessibility, swift pliability to dynamic needs, adequacy with respect to systematic measuring of customer usage and much more. The reckon services and resources will be conveyed to the users based on the concept of Virtualized service/resources include virtualization. hypervisors, virtual machines/containers, storage, network, service platform, application specific processing and computing software's.

Dr. Jayashree D. Mallapur Department of Electronics & Communication Engineering Basaveshwar Engineering College Bagalkot, India bdmallapur@yahoo.co.in

Although cloud technology provides sophisticated, costeffective services to its customers and has as a compelling impact on the present IT industry. However, lags with scarcity of open artefacts between homogeneous cloud providers. Standardization with respect to complete cloud service spectrum is a challenging task, as different providers have dissimilar compositions, costs, principles and domains. Vendor lock-in is considered as one of the gigantic threat and defiance for both the customers who are inside the cloud and who want to enter into another cloud. The lock-in problem further leads to the rise of portability and migration overheads. Cloud portability and migration is a barrier caused by cloud providers holding proprietorship with respect to APIs, platforms, versions, database schemas.

In this paper we analyze the vendor lock-in obstacle with respect to migration and portability of homogeneous cloud artifacts between different cloud environments and vendors. Further we plan to resolve the same by cloud application cloning, software re-engineering and re-factoring techniques. First, we present a cloud computing portability framework which exhibits an all angle possible migration scenarios. Second, we introduce CAM3 (Cloud Application Mobility Management Model) which can be extensively employed in order to achieve application mobility across multiple cloud ecosystems with minimum effort. CAM3 is unification of modules namely application cloning, transmission, application reengineering and installing techniques. Finally, to establish credibility of the proposed scheme, we carry out application refactoring and analysis of the same with respect to parameters and metrics of performance.

Proceeding segments of this paper is organized as follows: Section II describes the previous work carried out to solve cloud portability, migration and vendor lock-in problems. Further, various open standard models, scenarios, porting semantics are systematically analyzed. Section III presents a framework for cloud portability semantics. In section IV, we introduce CAM3 to overcome application migration with minimal effort. Section V presents the results and analysis on the same. Lastly, Section VI explains contribution and future direction to the work.

### 2019 IEEE Bombay Section Signature Conference (IBSSC)

## Optimization of Parking Dynamics in Smart City using Cloud Networks

Shivanand C. Hiremath Department of Electronics and Communication R.N.Shetty Polytechnic College Belagavi,India shivch612@gmail.com Dr. Jayashree D. Mallapur, Professor Department of Electronics and Communication Basaveshwar Engineering College Bagalkot,India bdmallapur@yahoo.co.in

Abstract—The modern India is proposing and developing smart cities, so that Indian standard of living to grow and reach on par with other countries. The smart cities should be embedded with all new technologies such as sensors, environmental, cloud access, IoT interface, data acquisition, etc. In our proposed work, we are planning to interface cloud with a smart city traffic management. In metro cities people go to cluster area such as Temples, Shopping malls, Cine theatres, etc and waste time in searching parking lots and withdrawing themselves with no solutions. We have come up with the concept that uses smart dynamic parking provision mechanism that can be accessed before the hand at target place or nearby sub parking possibilities. This solution reduces traffic turbulence and pleasure of utilizing possible parking lots as alternate solution rather than withdrawal.

Index Terms---VCCN, Public Cloud, Traffic overcrowding, Dynamic Provision

### I. INTRODUCTION

The rapid growth of infrastructure and modernization has lead to the development of sophisticated modern information and communication technologies, living standards of people can be enhanced according to the vision of smart city development schemes such as automation of smart vehicle parking plays a requisite role in daily routine life which can be done through collaborating cloud with vehicle networks to give rise to a concept Vehicle Cloud Computing Networks (VCCN). The vision of smart city is an ingenious concept that uses recently emerged information and communication technologies to enlighten all spheres of metro life. To make the effectiveness of such an ingenious concept is through implementation, several concepts and technologies are proposed and engulfed to enhance the perceptive of overcoming the complex challenges such as traffic overcrowding, autonomous traffic management, smart parking system, one can migrate to cloud computing to avail the intellectual cloud assisted services and can establish the dynamicity of provisioning resources ranging from traditional servers to hosted cloud solutions to the dormant computational potential of edge devices. Number of motor vehicles and private cars are drastically increasing

and parking spaces are gradually increasing and everybody wants to enjoy with family by moving out in their vehicles at unknown places and searching for parking becomes a natural issue mainly in Indian metropolitan cities like Mumbai, Kolkata, Bengaluru etc. conventional parking lots are accessed by searching and it availability depends on luck and sometimes end up parking on road side. Overcoming these issues has become a challenging job to cope up with demands and needs to acquire parking lot especially in a densely populated areas , one paperback solution to such a problem is by acquiring the parking availability through cloud assisted multiple cluster areas interrogation by vehicles in a desired location, where within a span of 2Km range one can find the shortest distance with minimum time reach path to the targeted cluster space nearby destination.

#### II. LITERATURE REVIEW

P.Harris et al. [1] deliberates about reservation based Smart parking availability by checking the availability in a single parking lot at a desired location by evaluating the number of free slots efficiently through sensors and aurdino kits as local units to communicate to the cloud environment. But does not take it to an account to contact multiple parking lots in a given desired area for availability. The real problem of traffic overcrowding can be resolved dynamically and probability of availing the parking increases by considering multiple parking lots in a desired area.

M. Hemalatha et al. [2] deliberates about intelligent parking system using vehicular cloud services and proposes a scenario of all vehicles communicating to RSUs and these are further synchronized with cloud and through an android app it confirms about vacancy of slots in a parking lot but this system works for single parking lots and lacks the dynamicity of providing the availability of the parking resource before the hand in a location with more probability. M. Siam et al. [3] demonstrates the FPGA based parking management system with three constituents mainly camera, database management and user interface it implements with the help of camera for capturing vacant parking spaces, with and uses deep neural

978-1-5386-7401-7/19/\$31.00 ©2019 IEEE

### Design and Simulation of Carbon Nanotube based Piezoresistive Pressure Sensor for Patient Monitoring Application

Ajayakumar C Katageri, B G Sheeparamatti Department of Electronics and Communication Engineering Basaveshwar Engineering College, Bagalkot, Karnataka, INDIA Email: ajaykatageri@yahoo.co.in, sheepar@yahoo.com

Abstract- The aim of this paper is to design a high sensitive piezoresistive pressure sensor for sensing a pressure in the range of 1 Pa to 1 MPa (required for patient monitoring application). The design of proposed sensor using piezoresistive mechanism includes piezoresistors that are made of carbon nanotube. When the input pressure is applied on the diaphragm, a subsequent deformation occurs which in turn changes the dimension of the piezoresistors that are configured in Wheatstone bridge format resulting in change of resistance of the carbon nanotubes. The basic principle of carbon nanotube based piezoresistive pressure sensor is the measurement of change in the resistance across the carbon nanotubes due to pressure applied. A contact is established between the nanotubes and the platinum electrodes, thus measuring the resistance of the nanostructures. The proposed model is analysed by considering various parameters such as deflection of the diaphragm, change in the output voltage across the bridge, sensitivity, etc.

Keywords- Pressure Sensor, Piezoresistor, Carbon Nanotube (CNT), Wheatstone bridge, Gauge factor.

### **1. INTRODUCTION**

Pressure sensor of the dimensions of microns referred to as MEMS pressure sensor is a device that measures pressure, ordinarily of gasses or fluids. Any pressure sensor consists of a sensing element such as diaphragm which converts applied pressure into a suitable physical movement. A sensing element is followed by a suitable transducer that converts mechanical movement of sensing element into a readable electrical signal. Based on the transduction mechanism employed, pressure sensors can be classified as capacitive, resonant, piezoelectric, thermal, piezoresistive, etc. In this paper, piezoresistive transduction mechanism is employed due to the fact that it has mature fabrication process, linear operation over wide range of pressure, simple read out circuitry, and it can be made highly sensitive.

In a piezoresistive pressure sensor, the piezoresistive effect is a change in the electrical conductivity (resistivity) of a piezoresistive material when mechanical strain is applied. Almost all semiconductors and metals can be used as piezoresistive materials; however semiconductors are more preferred over the metals, since semiconductors exhibit higher gauge factor. Widely used materials that are used as piezoresistive materials are metal foil (1 to 5), thin film metals (2), diffused semiconductors (80 to 200), polysilicon (30), polycrystalline SiC (3 to 5), single crystal silicon carbide (10 to 30), carbon nanotubes (200 to 1000), etc. It can be seen that CNT show a very high gauge factor of up to 1000 which recently led to research that uses CNT as piezoresistive element instead of silicon. Hence in this paper, semiconducting multiwalled carbon nanotube (MWNT) bundle is used as piezoresistive material. The advantage of MWNT bundles is the ease with which they can be formed across the electrodes on top of diaphragm.

Carbon nanotube is a tube-shaped material, composed of carbon atoms linked in hexagonal shapes, with each carbon atom covalently bonded to three other neighbouring carbon atoms. It exhibits excellent mechanical and electrical properties such as 200 times the strength and 5 times the elasticity of steel, 5 times the electrical conductivity, 15 times the thermal conductivity and 1,000 times the current capacity of copper, half the density of aluminium and it has a tensile strength of 63 GPa. The resistance of a CNT [1] is given by

$$R = R_c + \frac{1}{|t|^2} \frac{h}{8e^2} \left[ 1 + exp\left(\frac{E_{Gap}^0 + \frac{dE_{Gap}}{d\varepsilon}}{kT}\right) \right]$$
(1.1)

where  $|t|^2$  is the transmission probability of electrons with  $|E-E_F| > E_{Gap}$  crossing the energy barrier, Rc is the contact resistance, h is plank's constant, e is the charge on an electron, k is Boltzmann's constant, and T is temperature in degree Kelvin. The Gauge factor of a structure is defined as the change in resistance to the amount of volumetric strain acting on it and is given by

$$GF = \frac{\frac{dE_{Gap}}{d\varepsilon} \exp\left(\frac{E_{Gap}^{0} + \frac{dE_{Gap}}{d\varepsilon}}{kT}\right)}{kT \left[\exp\left(\frac{E_{Gap}^{0} + \frac{dE_{Gap}}{d\varepsilon}}{kT}\right) + 1\right]}$$
(1.2)

From the above equations it can be concluded that, as the strain applied increases, the band gap increases. Similarly as

#### IEEE - 45670

## Hybrid Algorithm for Data Confidentiality in Internet of Things

Poornima M. Chanal, Mahabaleshwar S. Kakkasageri Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, INDIA poornima.chanal@yahoo.com, mahabalesh\_sk@yahoo.co.in

Abstract-Internet of Things (IoT) is a network of globally connected physical objects, which are associated with each other via Internet. The IoT foresees the interconnection of few trillions of intelligent objects around us this interconnection produce large amount of private information which needs to be processed, transferred and stored. So to provide security and privacy to this information in IoT is a very challenging task, which is to be concerned at highest priority for several current and future applications. Devices such as smart phone, WSNs and RFIDs etc., are the major components of IoT network which are basically resource constrained devices. Design and development of security and privacy management schemes for these devices are guided by factors like good performance, low power consumption, robustness to attacks, tampering of the data and end to end security. Security schemes in IoT provide unauthorized access to information or other objects by protecting against alterations or destruction. In this paper, we propose hybrid confidentiality algorithm which is a combination of Advanced Encryption Standard (AES), Elliptic Curve Cryptography (ECC) and Message-Digest algorithm (MD5) algorithms. Geo encryption or location based encryption is also integrated with hybrid algorithm to give confidentiality to every devices. The hybrid algorithm enables strong confidentiality on the data transmission for Internet of things. Our proposed algorithm exhibits better performance in terms of encryption time, decryption time w.r.t. various file sizes.

Index Terms—Internet of Things (IoT), Confidentiality, Advanced Encryption Standard, Elliptic Curve Cryptography, Message-Digest algorithm

### I. INTRODUCTION

The IoT is a global network infrastructure, linking physical and virtual things through the exploitation of data capture and communication capabilities. The IoT infrastructure includes existing Internet and other network elements. IoT devices are characterized by a high degree of autonomous data capture, event transfer, network connectivity and interoperability for the independent cooperative services and applications. The all IoT context consist of billions of individuals, individual devices, and services that can interlink to exchange information and other useful information is described [1].

With the rapid increase in Internet and network based applications a huge amount of data is generated were several security and privacy issues are observed. Every object in such network will create security flaws and privacy weakness, these limitations are exploited by hackers, and in a statistical sense all these flaws and weaknesses may be abused in an environment with billions of devices [2]. However, in the absence of solid security in place, attacks and malfunctions in the IoT may outweigh any of its benefits [3]. Security and privacy which is responsible for some issues like confidentiality, authentication, validation, non-repudiation and integration. So all IoT devices support strong security intelligence for encryption and decryption of confidential data.

Confidentiality is one major challenge of security and privacy and it protects information from wrong people. Traditional confidentiality algorithms exhibit many challenges for this task. For IoT confidentiality, solutions should deal with high scalability requirements, heterogeneity of the involved building blocks, in addition to resources scarcity of the embedded devices such as energy and computational limitations. In this paper, we propose secured hybrid confidentiality algorithm that integrates AES, ECC and MD5 algorithms.

Rest of this paper is organized as follows. Related works are given in section II. Proposed work is presented in section III. Simulation and result analysis are discussed in section IV. Section V Concludes our paper.

#### II. RELATED WORKS

IoT allow people and things to be connected i.e. anything, anyone, any-where and anytime. In very short period Internet has turned into more common in our life than any other technology in the history, it has revolutionized the communication way of people. This will be through the usage of unique Internet protocol that permits things for communicating to each other without human intervention [4]. The evolution and importance of IoT in daily life and its architecture, protocols, numerous possible applications are described in [5].

The different security and protection attacks are discussed in [6]. Different IoT layer architectures to analyze the security problems of each layer and try to get new solutions. And also analyzes the cross-layer mixed integration challenges and security challenges are discussed in [7]. The framework of IoT and the security hazards in IoT, and new multi-layer security display is proposed [8]. The existing research on network security and Internet technology, and new approaches for analysts in certain IoT application and its related security are discussed in [9].

The lightweight implementation and evaluation of Elliptic Curve Cryptography (ECC) algorithm information is presented

## Trust and Delay based Routing for VANETs

Mamata J. Sataraddi, Mahabaleshwar S. Kakkasageri Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, INDIA mjsec@becbgk.edu, mskec@becbgk.edu

Abstract-Vehicular Ad Hoc Networks (VANET) is mainly aimed at providing safe and security related information and traffic management. In future, VANET contribute to smart transportation system. In routing, trust between the vehicles plays an important role to forward safety related information. This paper aims at design of trust based minimum delay routing for Vehicle-to-Vehicle (V2V) and Vehicle-to-Infrastructure (V2I) communication under the VANET constraints like rapid change in topology, high dense network, link failures etc. The algorithm works as follows: 1) trust calculation: using the prior knowledge about the neighbors, determine trust between the source node and its neighbors, 2) route selection: select the node with higher trust value, 3) minimum message reachable time (MMRT) calculation: find the MMRT for all selected paths, and 4) route decision: select the path with high trust and minimum delay. The proposed scheme exhibits better packet delivery ratio and decreased endto-end delay. The results show that our work is better and has only trusted vehicles participation in the routing with minimum delay

1 2

3 4 5

6 7

8

9

10

11 12 13

14

15

16

17

18

19

20

21

22

° 3

4

25

26

27

28

29

30

31

32

33

34

39

47

49

57

60

Index Terms-Vehicular Ad hoc Network (VANET); Trust; Delay; Vehicle-to-Vehicle Communication (V2V); Vehicle-to-Infrastructure Communication (V2I)

#### I. INTRODUCTION

A VANET is an emerging technology with vehicles 35 equipped with a communication device called on-board unit 36 (OBU), and a set of stationary units along the road, referred 37 to as road side units (RSUs) which act as a gateway for 38 connectivity to other communication networks, such as the Internet. Each vehicle OBU has a wireless network interface 40 which allows the vehicle to directly connect to other vehicles 41 and RSUs within its communication range, as well as wireless 42or wired interfaces to which application units can be attached. 43 4 By employing Vehicle-to-Vehicle (V2V) and vehicle-to-RSU (V2R) communications, VANETs can support a wide variety 45 46 of applications in road safety, passenger infotainment, and vehicle traffic optimization which is the main reason that 48 VANETs have received significant support from government, academia, and industrial organizations over the globe. Due to 50 features like high mobility [1] rapid changes in the network 51 takes place which leads to many challenges in VANET like 52 Data Aggregation, Clustering, Data validation, Data Dissem-53 ination, Routing, Security [2]. In this paper design of de-54 lay aware trust based routing for Vehicle to Vehicle (V2V) 55 and Vehicle-to-Infrastructure (V2I) communication under the 56 VANET constraints like dynamic topology, unpredictable vehicles density, link failures etc., to maximize the network performance parameter like packet delivery ratio, bandwidth utilization, throughput and to minimize end-to-end delay, control overhead is proposed.

The proposed delay aware trust based routing scheme works as follows: (1) trust calculation: using the prior knowledge about the neighbors, determine trust between the source node and its neighbors 2) route selection: select the node with higher trust value 3) minimum message reachable time (MMRT) calculation: find the MMRT for all selected paths 4) route decision: select the path with high trust and minimum delay.

The remaining part of the paper is summarized as follows: Details about the literature survey is given in section 2. Section 3 proposes details of proposed work where trust based routing in VANET for V2V and V2I communication is done. Simulation inputs and simulation procedure are presented in Section 4. Simulation results are discussed in Section 5.

### **II. RELATED WORKS**

Proactive and reactive routing algorithms for VANETs with different mobility models and network scenario are presented in [3] - [7]. In [8] - [11], routing protocols using intelligent techniques such as fuzzy logic, norm-aware agents, game theory, etc., are proposed to improve delay, packet delivery ratio, security, throughput for high dense environment. In [12], survey of different techniques to improve secure routing with trust enhancement among vehicles in VANETs is proposed. Trusted and secured routing protocol named as Trusted Vehicular Ad-hoc On-demand Distance Vector (TVAODV) is explained in [13]. A secure group authentication technique for VANET is proposed in [14].

The routing protocol based on trustworthiness of the path and the number of hops to identify the optimal route for transmitting reliable information is proposed in [15] - [16]. Due to selfishness some vehicles exhibit various misbehavior such as dropping packets which degrade the efficiency of network. Hence it is important to stop such vehicles to participate in network communication by proposing an effective revocation scheme for disconnected Delay Tolerant VANETs [17]. Considering the real time traffic conditions in case of link connection and historical traffic condition in case of link disconnection for route selection precess is explained in [18].

To decrease the hop count and delay and to increase the network throughput and the packet delivery ratio a new routing protocol based on cost metric is proposed in [19]. In some situations where the target vehicle is moved away from the expected range and in sparse traffic conditions connectivity is

Authorized licensed use limited to: University of Exeter. Downloaded od June 23,2020 at 04:40:11 UTC from IEEE Xplore. Restrictions apply.

## Connectivity and Delay Aware Reliable Routing in Vehicular Ad hoc Networks

Mamata J. Sataraddi, Mahabaleshwar S. Kakkasageri Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, INDIA misec@becbgk.edu, mskec@becbgk.edu

Abstract-In Vehicular Ad hoc Networks (VANETs) safety routing of information data packets is challenging issue, due to their delay-sensitive nature. The paper aims to resolve the issue of reliable and minimal delay routing in VANET. We propose a routing scheme to identify the reliable route with minimum delay from source to destination in VANETs. The routing scheme is in terms of probabilistic analysis of link duration between source and destination vehicles using border-node based most forward progress mechanism. Also the routing scheme selects best next hop vehicle out of the available neighboring vehicles belonging to source vehicle communication range border area. As the delay aware reliable routing is dynamic in nature handles the sparse and dense VANET scenarios to maximize the packet reception at the destination node. To examine the performance efficiency of the proposed work the performance parameters such as packet delivery ratio, latency and End-to-End delay are simulated and analyzed.

1 2 3

4

9

10

11

12

13 14 15

16

17

18

19

20

21

22

23

.4

25

26

27

28

29

30

31

32

37

39

40

41

42

Index Terms-Vehicular Ad hoc Network (VANET); Routing; Connectivity and Delay aware routing

#### I. INTRODUCTION

33 With pervasiveness of mobile computing technology, Vehic-34 ular Ad hoc Network (VANET) is a basic networking concept 35 for the future vehicular communications. In VANETs, the com-36 munication takes place from vehicle to vehicle, vehicle to fixed 38 Road Side Units (RSUs) to propagate different applications like monitoring of surrounding roads, planning the route, traffic information, vehicle crash intimation, internet access, digital map downloading, e-commerce, co-operative driver assistance entertainment in vehicles, parking availability, user interactions 43 etc. [1]-[6].

1 With respect to Media Access Control (MAC), several 45 VANET issues arises which includes information management, 46 routing, network management, congestion and collision con-47 trol, maximum message dissemination, privacy and security 48 [7]. On account of the high speed of vehicles and fast changes 49 in network topology, construction of a reliable routing protocol 50to transfer the message with negligible delay is thought to be 51a critical issue in VANET.

52 The method of finding optimal path between source and 53 destination vehicle with minimum overhead to disseminate 54data is called as routing. Nowadays researches on path finding 55 protocols depending on position of vehicle have become 56 important because of improved network connectivity for road 57 selection. In such protocols sending of data takes place directly 60 between source vehicle and destination vehicle or through 61

vehicles between source and destination and through roadside units [8].

Many research works have been proposed for the implementation and improvement of the VANET routing mechanism. Few ongoing related research works are given as follows. Analysis of path duration for VANETs is discussed in [9]. The density function of probability is estimated to calculate path duration for more number of intermediate nodes using exponential distribution function. The parameters like velocity of the moving vehicle, signal broadcasting range and number of vehicles between the transmitting vehicle and receiving vehicle are considered in calculation of path duration. Route lifetime and link stability are analyzed in [10] for the decentralized type of wireless networks. Edge Effect event happening in denser VANET is also given. A routing scheme explained in research work [11] uses an on demand reactive routing algorithm to decrease the duration for path calculation and also computes an alternate path if route failure occurs.

In [12], authors present that the number of intermediate vehicles and development in intermediate vehicles is depending on density of the vehicle and distance between vehicles. Routing protocols based on position or location of vehicles like Border-node based Most Forward progress within Radius routing (B-MFR) [13] and Edge-node based Directional Routing (E-DIR) [14] are used to choose the best next node for transmission of packets and are also used for evaluating the path duration in VANETs.

Bio-inspired routing algorithm [15] uses ant colony optimization method to find solutions for vehicle routing problem. This helps to avoid congestion whenever an accident happens. The surrounding road scenario that is represented by an undirected graph is given in Reliable Inter-Vehicular Routing (RIVER) where the graph vertices represents the points where curve in street takes place, and the street sections are represented by edges of the graph [16].

The proposed routing scheme identifies reliable route with minimum delay from source to destination in VANETs. Probabilistic analysis approach is provided for link duration between source and destination vehicles using border-node based most forward progress mechanism. Also the routing scheme selects best next hop vehicle out of the available neighboring vehicles belonging to the border area of the source vehicle transmission range. Proposed delay aware reliable routing scheme is

Authorized licensed use limited to: UNIVERSITY OF BIRMINGHAM. Downloaded on July 22,2020 at 17:54:48 UTC from IEEE Xplore. Restrictions apply.

3/8/2020	Multi-	agent based stable clustering	in VANET - IEE	E Conference	Publication
IEEE.org   IEEE Xploi	re Digital Library   IEEE-SA   IEI	EE Spectrum   More Sites		(	Cart   Create Account   Personal Sign In
		Institutional Sign	In		
Browse	My Settings	Get Help	Subscribe		
					Back to Results   Next >
Conferences > 2017 Int	ernational Conference				
Multi-agent	based stable clust	ering in VANET			
Publisher: IEEE	Cite This				
<< Results   Next	>				
2 Author(s) Anita	Bavalatti : Ashok V. Sutagundar	View All Authors			Mara Like This
					Mobility State Based Routing Method in
267			e (p		Venicular Ad-Hoc Network 2015 IEEE International Conference on Mobile Services
Full Text Views			Colla <sup>1</sup>	Alerts	Published: 2015
				Manage Content Alerts	Neighbor Mobility-Based Clustering Scheme for Vehicular Ad Hoc Networks 2015 International Conference on Platform
				Add to Citation	Technology and Service Published: 2015
				Alerts	View More
Abstract	0				Top Organizations with Patents on Technologies Mentioned in This Article
Document Sections	£ D₂-				ORGANIZATION 4
I. Introduction	Abstract: Vehicular Ad-boc I	Network (\/ANET) is part of M	bile Ad boo Note	(orl)	ORGANIZATION 3
II. Multi-Agent Based Stable Clustering Invanet	(MANET) that facilitates com safety, traffic updat View m	munication between group of	vehicles to provid	e driver	ORGANIZATION 2
III. Algorithm	Metadata				0
IV. Simulation and Results	Vehicular Ad-hoc Network (V, facilitates communication bet	ANET) is part of Mobile Ad-ho ween group of vehicles to pro	c Network (MANE vide driver safety,	ET) that traffic	
V. Conclusion	updates, entertainment, data frequent change in the topolo	sharing etc. Clustering is an e gy of VANET by means of loc	officient method to al co-ordination.	handle the Ve present	
Authors	multi agent based stable clus life time. Proposed approach	tering to provide stability, whic make use of multi-agent inter	h in turn enhance action method coi	e the cluster nsists of	
Figures	static and mobile agents to es RSU agent responsible to dea	stablish the communication be cide the cluster size and selec	tween vehicles and the suitable clus	nd RSU. iter head	
References	based on vehicle speed and o cluster size depending upon v	connectivity with neighboring r rehicle speed, which in turn in	odes. Here we va	ary the er life time	
Keywords	and reduces the routing over hand over to cluster head age	nead. After formation of cluste ent. The proposed method is s	r maintenance of imulated in NS2 t	cluster is by	
Metrics	considering some of the perfo cluster formation time, cluster	rmance parameters such as v head selection time, commun	ehicle speed, noo ication range etc.	de density,	
More Like This	Published in: 2017 Internatio (SmartTechCon)	nal Conference On Smart Tec	hnologies For Sn	nart Nation	

IEEE websites placed and the and the analyzon the drad was a stratige to a stratige of the stratige of the stratic strategy and the strategy a

Same and Same and some

Test State Discourse . . . . .

you agree to the placement of these cookies. To learn more, read our Privacy Policy. ISBN Information: Publisher: IEEE Accept & Close

3/8/2020	Congestion controlled ac	aptive routing in wi	reless sensor net	work - IEEE C	onference Publication
IEEE.org   IEEE Xplo	re Digital Library   IEEE-SA   IEEE Spect	rum   More Sites			Cart   Create Account   Personal Sign In
		Institutional Sig	n In		
Browse	My Settings	Get Help	Subscribe		
					< Previous   Back to Results   Next >
Conferences > 2016 Int	ernational Conference				
Congestion	controlled adaptive rou	ting in wirel	ess senso	r	
Rublisher: IEEE					
Publisher: IEEE	Cite This				
<< Results   < Pre	vious   Next >				More Like This
3 Author(s) Rash	mi M. Kittali ; S. K. Mahabaleshwar ; A. V. S	Sutagundar View All	Authors		DAIPaS: A performance aware congestion
8					Control algorithm in Wireless Sensor Networks 2011 18th International Conference on
2 227			Comercia	Alente	Published: 2011
Paper Full Citations Text Vie	WS			Alerts	Source-Based Routing Trees for Efficient Congestion Control in Wireless Sensor
				Manage Content Alerts	2012 IEEE 8th International Conference on Distributed Computing in Sensor Systems
				Add to Citation	Published: 2012
				Alerts	View More
Abstract					Top Organizations with Patents on Technologies Mentioned in
Document Sections	73-				This Article
I. Introduction					ORGANIZATION 3
II. Congestion	Abstract: Control of congestion in cas crucial area of interest, where an incre	se of wireless senso ease in data transmi	r networks (WSNs ssion by sensor no	s) is a very odes results	ORGANIZATION 2
Adaptive Routing Scheme	in a p View more				ORGANIZATION 1
III. Simulation	Metadata Abstract:				0
IV. Conclusions	Control of congestion in case of wirele of interest, where an increase in data	ss sensor networks transmission by sen	(WSNs) is a very sor nodes results i	crucial area in a	
Authors	proportionately smaller increase, or ex sophisticated techniques are required	en a reduction in the to combat with netw	roughput. Therefor ork congestion. Va	re, more arious	
Figures	constrained resources of the WSN are mechanisms. Different approaches for	to be taken into acc congestion detectic	count while devisir on and avoidance I	ig such nave been	
References	introduced in the past few years. The f cause loss of data. To tackle some of t	traditional flooding a the issues, this pape	nd gossiping algor r presents a novel	ithm can I approach	
Citations	for congestion control based dynamic packet traversal experiences the cong	routing. In the propo estion, an alternate	sed work, whenev	ver the	
Keywords	the packets on the new path. The prop detection, alternate path computation	osed work has three and re-routing the pa	e phases namely c	congestion	
Metrice	Congestion in the path can be detected Alternate path is computed based on a	d based on the free	space available in	the buffer.	
More Like This	energy on the path. This work is simula	ated using C for its e	effectiveness. Som	e of the	
	ratio, delay etc.	Tana work are foulle	uiscovery time, co	ngesuon	

Published in: 2016 International Conference on Signal Processing, Communication, IEEE websites place cookies on your device to give you the best user experience. By using our websites, Power and Embedded System (SCOPES) you agree to the placement of these cookies. To learn more, read our Privacy Policy.

Accept & Close

IEEE - 45670

## Hybrid Algorithm for Data Confidentiality in Internet of Things

Poornima M. Chanal, Mahabaleshwar S. Kakkasageri Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, INDIA poornima.chanal@yahoo.com, mahabalesh\_sk@yahoo.co.in

Abstract-Internet of Things (IoT) is a network of globally connected physical objects, which are associated with each other via Internet. The IoT foresees the interconnection of few trillions of intelligent objects around us this interconnection produce large amount of private information which needs to be processed, transferred and stored. So to provide security and privacy to this information in IoT is a very challenging task, which is to be concerned at highest priority for several current and future applications. Devices such as smart phone, WSNs and RFIDs etc., are the major components of IoT network which are basically resource constrained devices. Design and development of security and privacy management schemes for these devices are guided by factors like good performance, low power consumption, robustness to attacks, tampering of the data and end to end security. Security schemes in IoT provide unauthorized access to information or other objects by protecting against alterations or destruction. In this paper, we propose hybrid confidentiality algorithm which is a combination of Advanced Encryption Standard (AES), Elliptic Curve Cryptography (ECC) and Message-Digest algorithm (MD5) algorithms. Geo encryption or location based encryption is also integrated with hybrid algorithm to give confidentiality to every devices. The hybrid algorithm enables strong confidentiality on the data transmission for Internet of things. Our proposed algorithm exhibits better performance in terms of encryption time, decryption time w.r.t. various file sizes.

Index Terms—Internet of Things (IoT), Confidentiality, Advanced Encryption Standard, Elliptic Curve Cryptography, Message-Digest algorithm

### I. INTRODUCTION

The IoT is a global network infrastructure, linking physical and virtual things through the exploitation of data capture and communication capabilities. The IoT infrastructure includes existing Internet and other network elements. IoT devices are characterized by a high degree of autonomous data capture, event transfer, network connectivity and interoperability for the independent cooperative services and applications. The all IoT context consist of billions of individuals, individual devices, and services that can interlink to exchange information and other useful information is described [1].

With the rapid increase in Internet and network based applications a huge amount of data is generated were several security and privacy issues are observed. Every object in such network will create security flaws and privacy weakness, these limitations are exploited by hackers, and in a statistical sense all these flaws and weaknesses may be abused in an environment with billions of devices [2]. However, in the absence of solid security in place, attacks and malfunctions in the IoT may outweigh any of its benefits [3]. Security and privacy which is responsible for some issues like confidentiality, authentication, validation, non-repudiation and integration. So all IoT devices support strong security intelligence for encryption and decryption of confidential data.

Confidentiality is one major challenge of security and privacy and it protects information from wrong people. Traditional confidentiality algorithms exhibit many challenges for this task. For IoT confidentiality, solutions should deal with high scalability requirements, heterogeneity of the involved building blocks, in addition to resources scarcity of the embedded devices such as energy and computational limitations. In this paper, we propose secured hybrid confidentiality algorithm that integrates AES, ECC and MD5 algorithms.

Rest of this paper is organized as follows. Related works are given in section II. Proposed work is presented in section III. Simulation and result analysis are discussed in section IV. Section V Concludes our paper.

#### II. RELATED WORKS

IoT allow people and things to be connected i.e. anything, anyone, any-where and anytime. In very short period Internet has turned into more common in our life than any other technology in the history, it has revolutionized the communication way of people. This will be through the usage of unique Internet protocol that permits things for communicating to each other without human intervention [4]. The evolution and importance of IoT in daily life and its architecture, protocols, numerous possible applications are described in [5].

The different security and protection attacks are discussed in [6]. Different IoT layer architectures to analyze the security problems of each layer and try to get new solutions. And also analyzes the cross-layer mixed integration challenges and security challenges are discussed in [7]. The framework of IoT and the security hazards in IoT, and new multi-layer security display is proposed [8]. The existing research on network security and Internet technology, and new approaches for analysts in certain IoT application and its related security are discussed in [9].

The lightweight implementation and evaluation of Elliptic Curve Cryptography (ECC) algorithm information is presented

16

#### Chapter 4 An Experimental Study on Benzo[a] Pyrene Concentration in Particulate Matter at Industrial Area of Bangalore



### Prashant Basavaraj Bhagawati, Satish G. Muttagi, Poorna B. Bhagawati, Sandip S. Sathe and Abhijit M. Zende

Abstract Assessment of benzo[a]pyrene concentrations, with particularly preferred metals in polluted air, is significant issue for estimating counter health effects. Polycyclic aromatic hydrocarbons (PAHs) recognized one of the major crucial toxic air pollutants in industrials cum urban regions. With partial combustion of organic materials such as motor oils, gasoline, tobacco, cooking oils, butter, and other food leads to the formation of PAHs. With considering the importance of tracing the concentration of PAHs the experimental analysis work is carried out, to recognize the adverse effect on atmosphere. The air samples were collected (as 24 h sample once in a month) from eight specific spots within Peenya industrial monitored network Bangalore, India. The qualitative and quantitative analysis tests were carried out with modern scientific instrumental tool GC-MS. The concentration of B[a]P in eight measuring locations ranged from below detectable limit (BDL) to 0.0490 ng/m<sup>3</sup>. From the results obtained, there was noticeable variation in B[a]P concentration, with respect to measuring locations.

Keywords Polycyclic aromatic hydrocarbons + benzo[a]pyrene + Particulate matter

 P. B. Bhagawati (ES) · S. S. Sathe

 Department of Civil Engineering, Annasabeb Dange College of Engineering and Technology,

 Ashta, Maharashtra, India

 e-mail: prashantbhagawati@gmail.com

 S. G. Muttapi (ES)

 Scientific and Industrial Research Centre, Bangalore, Karnataka, India

 e-mail: researchin.em@gmail.com

 P. B. Bhagawati

 Department of Electronics and Instrumentation Engineering, Basaveshwar Engineering College,

 Bagawati

 Department of Civil Engineering, Daulatrao Aher college of Engineering, Karad, Maharashtra, India

 O Springer Nature Switzerland AG 2020
 53

 R. M. Singh et al. (ds.), Environmental Processes and Management,
 53

 Water Science and Technology, Library 91,
 11

 https://doi.org/10.1007/978-3-030-38152-3\_4
 4

an informa business		&F eBooks V Search for keyword	s, authors, titles, ISBN	Q	Login
				Advanced Search	
About Us Subjects	Browse Pro	ducts Request a trial Lik	rarian Resources What's Nev	w!	
Home > Engineering & T Solution for Agriculture	echnology > Electrica	I & Electronic Engineering > Comm	unications & Information Processing	> ICT for Competitive Strategies >	One Step Technological
In the Competitive StateGale	<sup>Chapter</sup> One Ste Agriculti	p Technological S ure	Solution for	Your institution has no Please get in touch wit this.	t purchased this content. h your librarian to recommend
LOT The Star No. The Star Star Star Star Star Star Star Excession	By Abhijeet R. Kumbalavati, Book <u>ICT fo</u>	Talawar, Abhishek R. Talawa Jayashree D. Mallapur r Competitive Strategies	r, S. B.	To purchase a print ve use or request an insp	rsion of this book for personal ection copy ≫
	Edition First Published	1st Edition 2020		GO TO ROUTLED	GE.COM
	Imprint Pages	CRC Press 7	~		
	eBook ISBN	9781003052098	Share		







Abstract:

94 Full Text Views

Abstract

I. Introduction

IV. Simulation

V Conclusion

IL Related Works

III. Hungarian Model Based

Resource Allocation in VCN

Document Sections



0

The computing resources of a cloud are dispersed geographically in a different servers, vehicles obtain the services through

the network. It is important to device the efficient and cost effective resource allocation schemes in Vehicular Cloud Network (VCN). Innovative schemes are required to allocate the resources so that efficient utilization of resources is obtained in VCN environment. Managing the vehicle demands, minimization of the processing time, reduction of the resource cost are key

parameters to be addressed in VCN. In this paper, we propose the Hungarian method based resource allocation scheme in VCN. The objective of the proposed scheme is to effectively provide the computing resources to the task. The proposed method

scheme optimally allocates the computing resources with significant reduction in the cost. The proposed method improves the

resource utilization efficiently with optimization of resource cost.

Published in: 2020 International Conference on Inventive Computation Technologies (ICICT)

#### More Like This

Implementation of evolutionary algorithms in vehicular ad-hoc network for cluster optimization 2017 Intelligent Systems Conference (Intelligys) Published: 2017

Cloud Computing Model for Vehicular Ad hoc Networks 2018 IEEE 7th International Conference on Cloud Networking (CloudNet) Dublished 7018 Feedback:

Taylor & Francis Group an informa business	T&F eBooks V Search for keywords, authors,	titles, ISBN <u>Advance</u>	Q A Logn
About Us Subjects Browse	Products Request a trial Librarian Re	sources What's New!	
Home > Engineering & Technology > Solution for Agriculture	Electrical & Electronic Engineering > Communications	& Information Processing > ICT fo	or Competitive Strategies > One Step Technological
LE FRA CONTINUE SINIERS LE FRA CONTINUE SINIERS UNIT OF CONTINUE SIN	Step Technological Soluti culture	on for	Your institution has not purchased this content. Please get in touch with your librarian to recommend this.
the shared carbon where the shared carbon share the shared carbon share the shared carbon share the shared carbon	ijeet R. Talawar, Abhishek R. Talawar, S. B. lavati, Jayashree D. Mallapur		To purchase a print version of this book for personal use or request an inspection copy $\gg$
BOOK Edition First Pu	ICT FOR COMPETITIVE Strategies 1st Edition		GO TO ROUTLEDGE.COM
Imprint Pages	CRC Press 7	ح	
eBook I	BN 9781003052098	Share	

### A Framework for image selection for image fusion using crowdsourced data

Pavan Kunchur, Rohan Raju Dhanakshirur, Sameer Ambekar, Sadhana Bangarashetti

Department of Computer Science, KLS Gogte Institute of Technology, Belagavi-590010, Karnataka, India Indian Institute of Technology Delhi (IITD), HauzKhas, New Delhi, Delhi- 110016 Basaveshvara Engineering College, Vidayagiri, Bagalkot, Karnataka, India- 587102

### \* Corresponding author

Mailing address: Department of Computer Science, KLS Gogte Institute of Technology, Belagavi-590010, Karnataka, India Phone: +91- 8095958867 E-mail: pnkunchur@git.edu

Abstract: The proposed method in the paper actions the design an ensemble model based deep learning framework to choose the most relevant images from the crowdsourced data for efficient image fusion. Image fusion is a technique of combining multiple registered images to get more informative or High Resolution (HR) image. Proposed method makes use of crowdsourcing to obtain multiple images. Crowdsourcing is a process of turning to a body of people to obtain relevant images. The images obtained through crowdsourcing may belong to different classes and may also contain multiple noisy and unwanted images and may be of different orientation. Also, the images obtained through crowdsourcing may be captured from different sources and also they may be in multiple resolution. The proposed method consists of training multiple deep learning models to estimate the probability of the image being relevant independently. Proposed method fuses the independent probabilities obtained from different algorithms using the decision fusion algorithm. Followed by elimination ofredundant images and the images containing significant amount of blur/noise. The selected images are then fused to generate the High Resolution (HR) image.The comparison of the results of proposed algorithms with different state-ofthe-art fusion techniques using different qualitative and quantitative techniques yields increased efficiency through the proposed algorithm.

### **Introduction:**

In this paper, the proposed method addresses the problem of image selection with an ensemble model based deep learning framework to choose the most relevant images from the crowd-sourced data for efficient image fusion. The main objective of the image fusion is to combine multiple images of the same scene taken at different instances, from different view-points and/orby different sensors in order to provide more information and semantic interpretation. The spatial and spectral resolution of monocular imaging system limits the information of single image.

#### 978-1-7281-3241-9/19/\$31.00 © 2019 IEEE

167 - 1-7281-3241-9/19/531.00 © 2019 IEEE

## A Survey of Cloud Computing Security Challenges, Issues and their Countermeasures

Lokesh B. Bhajantri

Department of Information Science & Engineering Basaveshwar Engineering College Bagalkot, India lokeshcse79@gmail.com

Abstract-Cloud computing is an emerging paradigm that provides on demand access to various resources such as servers, storage, applications, networking etc., over internet. Cloud computing enables access to shared pool of resources efficiently, which are managed by third party cloud service providers. It comes with wide range of benefits such as low operating cost, efficiency, scalability, flexibility, etc. The cloud users can access different services and/or resources at anytime and from anywhere on pay per use basis. Hence many individual users, organizations, and businesses are moving towards cloud computing and sharing large amount of data on cloud. However, the security concern is the biggest barrier for wide adoption of cloud computing. The security issues exist in context of infrastructure, data and storage, access control in cloud environment. This paper explores various security issues in all these perspectives. The paper briefly reviews the security issues at infrastructure level, data level, and also discusses the concept of Identity and Access Control in cloud. Also the different solutions to avoid or alleviate the security issues in cloud environment are discussed.

*Keywords*—Cloud computing, Security, Access Control, Privacy, Integrity, Identity Management, Authentication, Authorization

### I. INTRODUCTION

In recent years there is a massive evolution in information technology, which has made information technology an important aspect of all individuals. In todays modern era of computation, which includes mobile computing, Internet of Things, cloud computing etc., a large volume of data is generated. Cloud computing has given an efficient way to process, store, and manage such large amount of data over internet. Cloud Computing provides access to various resources that are scalable and easily available. The different forms of cloud services include computation, applications and services over the Internet [1]. The cloud service provider is responsible to deliver the requested service to the cloud user. With the help of cloud computing, users can access shared resources from anywhere and at anytime without any restriction of space, resources, and cost. Further attractions of cloud computing includes, services that are less costly, scalable, flexible, and elastic. Hence, Cloud computing is trending in todays technology-driven world and organizations of all scales are using cloud computing [2]. According to the survey done almost 96% of IT organizations are adopting cloud technology because of the numerous advantages offered by it [3].

Tabassum Mujawar

Research Scholar, Department of Computer Science & Engineering, Basaveshwar Engineering College Bagalkot, India tabbu3002@gmail.com

The cloud computing architecture consists of different deployment models and service delivery models. The types of cloud deployment models are public cloud, private cloud and hybrid cloud. The type depends on the environment the cloud is deployed. The public cloud is accessible to every user, as it does take into consideration the affiliation of user to any specific organization. The private cloud is can be used only by members or users of any particular organization. The hybrid cloud presents the mixture where both public and private cloud is available for use. The service delivery models are classified on basis of type of service delivered to users. The classification includes Software as a Service (SaaS), Platform as a Service (PaaS), and Infrastructure as a Service (IaaS). In SaaS, the different software applications are provided as a service to cloud users through internet. The PaaS model delivers different platforms and resources as a service to users over internet, which is required to build applications. IaaS model offers the hardware and necessary infrastructure as a to cloud users, to facilitate development and service deployment of applications. Though cloud computing is booming and its demand is increasing with rapid speed, it comes with its share of security issues. There exists some security concern for cloud computing that need to be addressed. The security concern arises because user loses control over the data as the data is now available in cloud service providers domain. The different applications and users are able to access the users data. It is complete responsibility of cloud service provider to apply proper security measures. Because of the shared and un-trusted environment of cloud computing, some new security violations are introduced. Hence the wide adoption of cloud computing is limited and many users do not completely trust the cloud computing environment [4-5]. In this paper the most promising challenges for cloud computing and the security issues at different levels are presented. The major focus is given to infrastructure security, data security and access control. The possible countermeasures for the security issues are also discussed in the respective sections.

The rest of the paper is organized as, the Section II describes the security challenges present in cloud computing, which are listed by Cloud Security Alliance. In Section III the different security issues and possible solutions are described. The overview of trust based access control framework that can

### Cluster Based Data Aggregation in Wireless Sensor Networks: A Bayesian Classifier Approach

### Lokesh B. Bhajantri, Basavaraj G. Kumbar

<sup>1</sup>Department of Information Science and Engineering, Basaveshwar Engineering College, Bagalkot, Karnataka, India. lokeshcse@yahoo.co.in

<sup>2</sup>Department of Computer Science and Engineering, BGMIT, Mudhol, Karnataka, India gkbasavaraj@gmail.com

Abstract. The network is composed of wireless distributed sensor nodes with data computational capabilities. In each cluster, the cluster member nodes are used to send the sensed data to its respective cluster head to aggregate and classify the data as effectively. In this work, the algorithm for cluster based aggregation of data using the Naive Bayesian Classifier is proposed. The proposed scheme provides better performance rather than existing algorithms with accuracy, efficient energy utilization, and computation overhead.

**Keywords:** Wireless Sensor Networks, Data Aggregation, Cluster, Cluster Head, Naive Bayesian Classifier.

### **1** Introduction

A Wireless Sensor Network (WSN) contains the number of distributed sensor nodes with high computation network. The sensors are scattered over the network, each of these nodes has the capability to process or route the data to sink node or base station through cluster head (CH) nodes. A sink node collects the data from its CHs from each cluster of the WSN for enhancing and decision making of data. The WSN has been used for various real time applications such as military applications, habitat monitoring, environmental monitoring, and so on. In WSN cluster member nodes in each cluster periodically send the data to its respective CHs, then CHs transmits the data over the network. The frequency of reporting sensed data usually depends on the particular applications to the sink node.

In WSN, a grouping of sensor nodes into clusters to satisfy the load balancing, scalability, the efficiency of energy, and better network lifetime. There are several problems that impacts the design and performance of the WSNs are as follows [1][2]: 1) Quality of Service based communication and computing, 2) fault tolerance management issues, 3) routing issues, 4) context aware issues in

Materials Science and Engineering Conference Series

2019 First International Conference on Recent Advances in Materials and Manufacturing (ICRAMM 2019)

Proceedings of the ICRAMM 2019 - Volume 1 Belagavi I Karnataka I India 12 - 14, September 2019 I www.icramm.com



Scanned by CamScanner

### CONTENT

### RECENT ADVANCES IN MATERIALS & MANUFACTURING

EMTP 100	D4 Effective study of thermal behavior on membrane stratified portable solar pond D Sathish and S Jegadheesaran	001
EMTP 100	07 Growth and characterization of Ag-Cu <sub>2</sub> O thin films: DC magnetron sputtering method M Nagendra Vara Prasad and Y Munikrishna Reddy	006
EMTP 101	2 Design and development of intelligent handling system for components in small and medium scale industries S A Soundattikar, V R Naik and C V Adake	013
EMTP 102	O Growth, characterization and influence of gamma irradiation on strontium chloride doped urea: An NLO crystal M S Dileep, Sindhu Tilak and H M Suresh Kumar	024
EMTP 102	1 Design of a magnetically sensitive monotube damper for leg prostheses applying magnetic materials Oscar Arteaga, Marcelo Ortiz, Eduardo Cárdenas, Juan Rocha-Hoyos, Katherine Amores, Andrés Balarezo, Jason A. Rodriguez	033
EMTP 1030	Facile synthesis of micro / mesoporous functional carbon from turbinaria conoides seaweed for high performance of super capacitors P Divya, A Prithiba and R Rajalakshmi	041
EMTP 1031	Performance evaluation of composition graded layer of aSi <sub>1-x</sub> Ge <sub>x</sub> : H in n+aSi:H/i- aSi:H/p+aSi <sub>1-x</sub> Ge <sub>x</sub> :H Graded Band gap Single Junction Solar Cells J Fatima Rasheed and V.Suresh Babu	053
EMTP 1034	Numerical method based parametric investigation of Isotropic annular plate C Anand, Mattikalli, V Rajashekar and Kurahatti	063
EMTP 1038	Optimization of wear behaviour on AA6061/Al2O3/SiC metal matrix composite using squeeze casting technique - Statistical analysis L Natrayan and M Senthil kumar	072
EMTP 1043	Enhanced thermal properties of silica nanoparticles and chitosan Bio-Based intumescent flame retardant polyurethane coatings <i>TentuNageswara Rao, Imad Hussain and Bon Heun Koo</i>	079
EMTP 1055	Experimental investigation into helical hole milling of fiber metal laminates Krishna Hemant, Anish Kona, S A Karthik, Gururaj Bolar	088
EMTP 1056	Electromechanical characterization of Nitinol based RF MEMS Switch Ajay Sudhir Bale, Suhaas V Reddy, Shivashankar A Huddar	098
EMTP 1060	5-Element Biconcave patch antenna array with DGS for Ku band application Ribhu Abhusan Panda, Sravya Gudla, Preetibardhan Dash, Swati Rani, Debasis Mishra	102
EMTP 1061	Compressive property of newly developed Composite material from polyethylene terephthalate (PET) waste and mild Steel powder Yogesh Dandekar, R Suresh Kumar and Mridul Singh Rajput	108

i | ICRAMM 2019 - VOLUME 1

2019 First International Conference on Recent Advances in Materials and Manufacturing

## Publication Partner: Elsevier

EMTP 1034

## Numerical method based parametric investigation of isotropic annular plate

Anand C Mattikalli<sup>4</sup> and Rajashekar V Kurahatti<sup>b</sup>

\*Maratha Mandal Engineering College, Belagavi-590005,India Basaveshwar Engineering College, Bagalkot-587102, India E-mail: mattikallianand@gmail.com

Abstract : Annular plates play a vital role as a part of machine elements for different applications, hence characterization and investigation of annular plate for different parameters is of crucial importance. The current work deals with investigation of isotropic annular plate to study the behavior of annular for variation in boundary conditions, effect of cutout size, number of cutouts and type of cutouts in the annular plate on its static and dynamic behavior. The numerical method is successfully utilized for determination of effect of various parameters on stresses induced, deflection, natural frequency and mode shape.

Keywords: Annular plate; Finite element method; Free vibration; Deformation; Static analysis; Numerical method.

### **1. INTRODUCTION**

Annular plates find several applications in practical world, circular cutting blades, tiles cutter, flywheels, clutch plates, compact discs may be well-thought-out as annular plate for the purpose of analysis [11]. Stress analysis is aimed at determining the deflection and stresses developed in reaction to various applied loads [9]. Modal analysis is carried out to determine to determine the effect of variation in different parameters of the annular plate on fundamental frequency & different modes of vibration

A good amount of work is carried out on study of the annular plate. Annular disc cutters were analyzed for vibration characteristics [1,10]. Finite element method based analysis of polar orthotropic annular plate with an electro rheological (ER) fluid core and constraining layer was investigated to find that ER fluid core was found to have a significant effect on the vibrational behaviors of the sandwich annular plate[2]. Experimental analysis of circular disc with diametral slots was carried and results obtained were compared with numerical method results [3]. Free vibrational behavior of ST37 steel and the steel annular plates is investigated for determining the natural frequency by experimental method [4]. Circular plates is investigated for determining the natural frequency by experimental plates with multiple circular holes were analyzed for free vibration using indirect BIEMs [5]. In the present work attempt is made to carryout characterization of annular plate by investigating the effect of disc effect of different boundary conditions, cutout size, number of cutouts and type of cutouts in the annular plate, on its static and dynamic behavior.

# <sup>2. THEORETICAL FORMULATION</sup>

Deflection and stresses

The maximum deflection and stresses developed in the annular plate of outer radius 'a' with inner edge completely. edge completely constrained made of isotropic material is given by [6].
# ACADEMIC YEAR 2018-19

CERTIFICATE
OF PARTICIPATION SAIRAM College of Engineering
6 <sup>th</sup> International Conterence on <i>connecting engineersdeveloping</i> restant
28 <sup>th</sup> - 29 <sup>th</sup> November, 2018 at Bengaluru, Karnataka
This is to certify that Basaveshwar Engineering College, Vidyagiri presented his/her
research paper titled
Chip, Circuitry, Current, Coding, Combustion & Composites" - i7C-2018 held in Sri Sairam College of Engineering at Bengaluru, Karnataka
Mr.Rudra Bhanu Satpathy Director IFERP Director IFERP
Anekal, Bengaluru - 562 106. Anekal, Bengaluru - 562 106. Anekal, Bengaluru - 562 106.









Department of CSE & Department of ISE Organized by for the paper titled ... EMAIL ... SPAM .. DETECTION .. AND ... PREVENTION ... USING ... NAIVE

BAYES CLASSIFIER

of ...&h.sw.ESHNAR...ENGG .....Could GE ...... has participated / presented a paper in the

"7<sup>th</sup> NATIONAL CONFERENCE ON RECENT ADVANCES IN TECHNOLOGY AND ENGINEERING (RATE - 2018)"

T.JOHN

**T. JOHN INSTITUTE OF TECHNOLOGY** #86/1, Gottigere, Bannerghatta Road, Bengaluru-560083

- Jaco Colore

wall of act.

RATE 2018

CERTIFICATE

**INSTITUTE OF TECHNOLOGY** 

### The Effects of GGBFS on Strength Properties of Geopolymer Concrete Cured at Ambient Temperature



Aslam Hutagi 💿 and R. B. Khadiranaikar 💿

**Abstract** Geopolymer concrete is an eco-friendly alternative to conventional Portland cement concrete for civil infrastructure applications. Lot of works is available by many researchers to understand the hardened properties of geopolymer concrete by heat curing method. However, in situ condition does not favour the heat curing. Thus, the present investigation is carried out to achieve desirable properties of geopolymer concrete at ambient curing. In the present work, the geopolymer concrete containing 60–100% fly ash (FA) and 0–40% GGBFS is investigated. The study also includes the role of GGBFS on the microstructure. The testing results obtained after 3, 7 and 28 days of ambient curing indicate that there is an increase in the compressive strength, splitting tensile strength and flexural strength significantly. The addition of GGBFS to geopolymer concrete enhances the setting time, rapid strength development is noticed and carbon footprint is significantly reduced.

Keywords Geopolymer concrete · Microstructure · Fly ash · GGBFS Compressive strength



33rd Indian Engineering Congress, The Institution of Engineers (India) Udaipur, 2018 Theme : Integration of Technologies: Emerging Engineering Paradigm

#### Some Major Properties of High Strength Self-Compacting Concrete

#### Aijaz Zende<sup>1</sup>, Dr R B Khadiranaikar<sup>2</sup>

Abstract: An attempt is made to understand the behaviour of High Strength Self-Compacting Concrete in its fresh and hardened state incorporating mineral admixtures with different percentages of silica fume and fly ash. Experiments were carried out using 12 mixes with different w/c ratios. Four replacements of OPC with fly ash and silica fume of 5%, 10%, 15% and 20% were cast cured for 28 days. The experimental program is carried out in three stages. First, for each mix, fresh properties (slump flow, T-500, V-funnel test and L-box test) are checked. In the second stage, cubes and cylinders were cast and kept in fresh water for curing. Then in the third stage, these samples were used for testing mechanical properties (compressive strength, split tensile strength and modulus of elasticity). The results obtained indicate a direct impact of percentage content of fly-ash and silica fume on the properties of fresh and hardened SCC.

Keywords: Self-Compacting Concrete: Fly ash-Silica fume: Slump flow: T-500: V-funnel test: Lbox test special construction methods and improvement in concrete properties. Given the dense reinforcement around which concrete is required to move, or the complicated geometry of the formwork, and the distances over which concrete needs to be pumped, makes a high demand on the workability of the concrete, and practically "flowing" concrete is required. Needless to say, in order that the concrete does not segregate, this workability and flowability strains the ability of the fluid phase to carry with it the denser aggregate fraction to the limit.

Self-Compacting Concrete (SCC) provides a solution by overcoming this problem as it can flow, compact by itself without any need of vibration or another mean of compaction and fills completely on to the formwork with no segregation [1-4]. SCC can reduce the considerable number of the skilled work force and need of good quality control; thereby it reduces the time of construction. It was also reported that Self-Compacting concrete is more economical than conventional concrete [5]. However, because of its requirement of "highly flowing nature", proper care should be taken so as to achieve filling and









# Heliyon



Received: 14 June 2018 Revised: 13 August 2018 Accepted: 19 December 2018

#### Cite as:

Satyanarayan K. Padaganur, Jayashree D. Mallapur. Neural embedded smart link generation scheme for heterogeneous network. Heliyon 4 (2018) e01089. doi: 10.1016/j.heliyon.2018. e01089



# Neural embedded smart link generation scheme for heterogeneous network

### Satyanarayan K. Padaganur<sup>10,10</sup>, Jayashree D. Mallapur<sup>10</sup>

<sup>a</sup> Department of Electronics & Communications, BLDEA's Dr. P.G.Halakatti CET, Vijayapur, Karnataka, India

<sup>b</sup> Department of Electronics & Communications, Basaveshwar Engineering College, Bagalkot, Karnataka, India \*Corresponding author.

E-mail address: padaganursk@gmail.com (S.K. Padaganur).

#### Abstract

The Long Term Evolution (LTE) network is a very much popular network in heterogeneous network. Heterogeneous networks provide maximum data rate by integrating various technologies and channels, based on appropriate network selection. For the sensible data transmission in the LTE network, noise plays an vital role as channel is a free space. The minimum noise channel selection is a decision of present and previous status of network channel. Proposed scheme develop neural network model, which will act as a smart link generation scheme for computing minimum noise channel path for sensible data transmission. Hence, proposed scheme will improve performance of the network. Result indicates that, proposed scheme improves throughput and system reliability. Proposed scheme is also reduces packet loss rate and energy consumption in contrast with conventional techniques.

Keywords: Computer science

#### 1. Introduction

LTE network or future generation network are typically based on integration of GSM and UMTS system, which will form the heterogeneous system. Basic LTE architecture is shown in Fig. 1.

# Intelligent Information Gathering Scheme in Internet of Things (IoT)

Sharanappa P. H.

Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, INDIA phsharanu@gmail.com

Abstract—In recent years, Internet of Things (IoT) has grabbed the attention of the entire world. IoT is a network of physical objects, connected to the Internet, which can be discovered and queried globally. IoT can also be defined as an advanced Wircless Sensor Network (WSN), where the networked sensors are sensing data and transmitting continuously. So, it becomes an obvious challenge to deal with the enormous amount of data that is being gathered. In this paper, we have proposed an intelligent scheme for information gathering in IoT through agent based approach. The main contribution of our proposed research work is to design and develop an intelligent solution for information gathering in IoT.

Index Terms-Internet of Things (IoT), Information Gathering.

#### I. INTRODUCTION

The IoT concept is being revolutionized because of its huge distributed information, where any real-world physical thing can take part in the Internet automatically, so that the 'thing'can be identified and queried over worldwide. The IoT surrounds with different types of technologies. It also has research scope that focuses on extending the current Internet to real-world physical things. It has got open challenges to address, that includes information gathering, information validation, information aggregation, privacy, data analytics, heterogeneous architecture, energy efficiency, security, Quality of Service, etc. In recent years, IoT has really enhanced the living standard of people and transformed their life style which is an indication of promoting the advancement of science and technology that develops the society [1].

A very basic IoT architecture usually contains three layers: sensing, network, and user application. The sensing layer consists of sensor nodes, and identifies each of the things in the network. It contains RFID tags, sensors, cameras, GPS, etc. The perception layer is the physical layer responsible to identify each object/thing in IoT. This is done with the help of sensors and gathering information about each object. The network layer is the major layer of IoT and has the responsibility of connecting to other smart objects, network devices, and servers and gateways. And responsible for transmission of the information gathered by the perception layer. The application layer converges IoT application services, such as smart cities, smart health, smart homes, etc., to the end user [2].

Compared to other networks, IoT is a complex network as all physical objects participate in the network. Obviously the amount of data collection is also enormous. So it is necessary to adapt some intelligent data gathering techniques. An object may send the information it has observed or sensed, directly to its neighboring sensor nodes. It may hoppen-these information to

978-1-7281-5286-8/19/\$31.00 ©2019 IEEE

Mahabaleshwar S. Kakkasageri Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, INDIA mahabalesh sk@yahoo.co.in

mislead the users. During these situations certain information validation methods should be adapted before forwarding it to the other sensor nodes. Information aggregation approach usually combines several messages pertaining to same event to form a single consolidated message, thereby reducing the number of packet transmissions. These techniques are useful to reduce overheads. Data aggregation mechanism also improves the lifetime of nodes by eliminating redundant data transmission.

The fundamental technologies and architectures of IoT are proposed in [3] - [5]. The architecture describes the encapsulation of sensor technology, RFID, logic for embedding the objects, ad-hoc network conditions, information infrastructure for Internet for the use of smart object framework. The objective is to make the reader understand the current stage of the IoT, methodologies that assists, the applications, challenge and the developments through a complete analysis and categorizing the literature [6].

The rest of the paper is organized as follows. Related work is discussed in Section II, the proposed data gathering scheme is illustrated in Section III, simulation and result analysis are discussed in sections IV and V respectively, and conclusions are drawn in Section VI.

#### **II. RELATED WORKS**

The necessary technical issues of IoT like exchange of information within massive different and unrelated network components, proper combination and interaction adjustment of varying data and service adjustment in the dynamic environment are addressed in [7] [8]. The perspective, issues, possible usage situations and IoT technological building blocks are discussed in [9]. Particularly, the RFID and scientific improvements like Internet protocol stacks and web servers are concentrated for smart everyday objects. Societal and ethical issues which will commonly arise are presented in [10].

Different challenges and orientation for advancement in mobile cloud computing for IoT, like safety issues along with its feasible clarification,managing the storage issues, continuous connection of network are discussed in [11]. The work mentioned in [12] describes a routing protocol which organizes by itself that is based on the geography. It presents an approximate algorithm that estimates how many number of nodes are supervising the area with approximate shape and size. To build the Infrastructure for prototyping the system that focuses to design basis model for IoT information is proposed in [13]. It illustrates important building blocks in the 1framework and their jobs and provides information regarding

# Cascaded Structure of Noise Reduction and Multiband Frequency Compression of speech signal to Improve Speech Perception for monaural Hearing Aids

Rajani S. Pujar, P. N. Kulkarni Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587103, Karnataka, India rajani\_pujar@rediffmail.com, pnk\_bewoor@yahoo.com

Abstract - This paper presents a cascaded structure of noise reduction and multiband frequency compression in enhancing speech perception for sensorineural hearing impaired persons using monaural hearing aid. In multiband frequency compression scheme, input signal is divided into 18 frequency bands ranging from 0 to 5 KHz based on auditory critical bandwidths (ACB). Frequency components of each band are compressed at the center of their respective bands using frequency segment mapping method. This results in concentration of all spectral energy at the center of each band to minimize the effect of spectral masking. Noise reduction scheme, namely wiener filter is used to reduce the background noise. Evaluation process involves i) Speech quality assessment (using PESQ and MOS) on normal -hearing subjects ii) Speech intelligibility assessment (using MRT) on hearing-impaired subjects. The test material consists 50 sets of consonant-vowelconsonant (CVC) words. Each set includes six words. Each subject is responded for 1800 presentations (300 CVC words x 6 different SNR conditions). The MOS, PESQ and MRT scores obtained for the SNR values ranging from -6 dB to + 6 dB indicates enhancement in speech quality and intelligibility.

Keywords- Sensorineural Hearing Loss (SNHL); Spectral Masking; Multiband Frequency Compression; MOS; PESQ

#### I. INTRODUCTION

Hearing loss mainly happens when the working of cochlea and the auditory nerve does not function properly. Usually auditory filter bandwidth gets widened and results in reduced frequency selectivity because of the masking of frequency components [1] [2] [3]. This results in listening discomfort by hearing impaired people. Furthermore, intelligibility of the speech gets reduced in the presence of background noise. Hence under noisy environments hearing impaired people find difficulty in understanding speech signal. Therefore Noise reduction techniques as well as hearing aid algorithms are essential to improve quality and the intelligibility of speech under adverse listening conditions.

Many single and multi channel noise reduction techniques have been proposed in past which are applicable for both monaural and binaural hearing aids. Noise reduction techniques helps in achieving similar speech intelligibility as that of normal hearing people in noisy environments. Some of the single channel noise reduction techniques are Spectral subtraction [4], spectral subtraction using spectral floor and over-subtraction [5], wiener filter [6] etc. Multichannel noise reduction techniques exploit spectral and temporal differences and also the spatial separation between sound sources to improve SNR [7]. Spatial domain (direction or location) difference between a target and noise signal can be used to achieve noise reduction in different noisy conditions using a system with a microphone array [8].

Hearing aids help in amplification (frequencydependent) of speech signals for hearing-aid users. Existing hearing aids do not provide sufficient amplification of high frequency speech sounds [9]. This restricts the audibility of high frequency sounds, predominantly for severely hearing impaired subjects. High frequency samples of speech signal are shifted to low frequency regions by using frequency transposition methods [10]. Transposition of frequency sample methods proposed in [11]-[14] were useful in speech recognition, but distorts the speech quality. Multi-band frequency compression method to reduce frequency masking associated with SNHL persons is discussed [15]. By using frequency compression, frequency lowering or frequency transposition algorithms the effect of frequency masking can be minimized to some extent in noisy environments, but results in degraded speech quality. In order to improve speech perception in noisy conditions, and to overcome spectral masking associated with sensorineural hearing loss (SNHL) listeners, a cascaded structure of noise reduction (using wiener filter) and multiband frequency compression of the speech signal is proposed.

Remaining part of the paper is as follows. Section II explains signal processing. Tests and result analysis are given in section III. Spectrograms are explained in section IV. Section V concludes our research work.

#### II. SIGNAL PROCESSING

This study presents a cascaded structure of (i) Noise reduction and (ii) Multi band frequency compression of the speech signal for SNHL listeners. Wiener filter is used in our cascaded scheme to reduce the noise present in a speech signal. Wiener filter [16] minimizes the mean square error (MSE) between the desired signal and an estimated signal. Wiener filter requires separate estimate of the clean speech and the noise power which can be estimated using a Voice Activity Detector (VAD). Proposed VAD method to discriminate between speech and noise is shown in Figure 2.

In multi band compression scheme, the speech signal sampled at 10 KHz sampling frequency is segregated into number of frames (frame duration of 20ms and frame overlap of 50%). Hamming widow is applied for each of these frames. The frequency scale is then partitioned into 18 bands based on Auditory Critical Bandwidths (ACB) [17].

# EFFECT OF NOISE REDUCTION ALGORITHMS ON TEMPORAL SPLITTING OF SPEECH SIGNAL TO IMPROVE SPEECH PERCEPTION FOR BINAURAL HEARING AIDS

WESPAC

Rajani S. Pujar, Pandurangarao N. Kulkarni Department of Electronics and Communication Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, India e-mail: rajani\_pujar@rediffmail.com, pnk\_bewoor@yahoo.com

Sensorineural hearing loss which occurs due to the damage of cochlea, auditory nerves or both, is characterized by elevated hearing thresholds, loudness recruitment, reduced temporal and frequency resolution, and increased temporal and spectral masking. Increased temporal masking affects the speech perception in persons with sensorineural hearing impairment especially under adverse listening conditions. Hearing aid users have problem in speech understanding particularly in the presence of background noise. Hence, hearing aids should include noise reduction techniques for enhancing the speech perception. This paper presents a cascaded scheme, which employs a noise reduction algorithm as well as temporal splitting of the speech signal. The main objective of the paper is to improve quality of perceived speech for SNHL listeners by reducing the temporal masking and improve the speech perception in the background of additive noise. Earlier investigations have shown that by splitting the speech temporally and presenting alternate segments to the two ears help in reducing the effect of temporal masking. In this technique, the speech signal is processed by two fading functions, complementary to each other, and presented to left and right ears for binaural dichotic presentation. In the present study, half cosine signal is used as fading function with crossover gain of 6 dB for perceptual balance of loudness. Temporal splitting is combined with noise reduction algorithm to improve speech perception in the background noise. Two noise reduction schemes, namely spectral subtraction and Wiener filter are used. We have used Voice Activity Detection (VAD) based on short term energy and zero crossing rates to detect noise and speech frames separately. Listening tests were conducted on six normal hearing subjects, with sensorineural loss simulated by adding broadband noise to the speech signal at different signal-to-noise ratios ( $\infty$ , 3, 0, and -3 dB) using mean opinion score(MOS) test. Objective evaluation using PESQ was also carried out. The MOS score for VCV syllable /asha/ for SNR values of ∞, 3, 0, and -3 dB were 5, 4.46, 4.4 and 4.05 respectively, while the corresponding MOS scores for unprocessed speech were 5, 1.2, 0.9 and 0.65, indicating significant improvement in the perceived speech quality for the proposed scheme compared to the unprocessed speech.

### 1. Introduction

Sensorineural hearing loss which occurs due to the damage of cochlea, auditory nerves or both, is characterized by elevated hearing threshold, loudness recruitment, reduced temporal and frequency resolution, and increased temporal and spectral masking [1] [2]. Temporal masking, also known as

1

### **OPTIMIZATION OF LEACH PROTOCOL FOR EFFECTIVE POWER UTILIZATION IN WIRELESS SENSOR NETWORKS**

Ajayakumar C. K<sup>1</sup>, Kirankumar B. B<sup>1</sup>, Supriya B H<sup>1</sup> and Poornima M C<sup>1</sup>

<sup>1</sup> Department of Electronics and Communication Engineering, Basaveshwar Engineering College, Bagalkot, INDIA ajaykatageri@yahoo.co.in, kiranb4004@gmail.com, harlapursupriya@gmail.com, poornima.chanal@yahoo.com

Abstract: Energy utilization of the sensor nodes is a major challenge that is being encountered in lowpowered Wireless Sensor Networks (WSNs) when designing an algorithm, protocol or hardware. Based on the existing LEACH protocol, a new clustering protocol for wireless sensor networks is proposed and implemented in this paper. The outcome of this work is reduction in the energy consumption for sending the sensed data from sensor deployment place to the base station. In the existing LEACH protocol, cluster heads collect the data from sensors in the cluster and aggregate it. Then aggregated data is directly sent to the base station. But in our scheme, we differentiate the cluster heads which are far and near to the base station. Depending upon the distance the far node from the base station will send its aggregated data to the next neighbouring cluster head. Above cycle will be repeated until the entire data is delivered to the base station by its nearest cluster head.

**Keywords:** LEACH, Effective Power Utilization, Cluster Head, TDMA.

#### **INTRODUCTION**

by:

Sensor networks refers [1] to a heterogeneous system consisting of multiple detection stations called sensor nodes with a communications infrastructure intended to monitor and record conditions at diverse locations. Sensor nodes, also known as mote, are small, light weight and portable devices equipped with a transducer, microcomputer, transceiver, and power source. The transducer produces electrical signals based on the sensed physical phenomena. The microcomputer processes and stores the sensed information. The transceiver receives instructions from the base station/central computing system and sends data to it. Each sensor nodes derives its energy usually from a battery or any other embedded form of energy harvesting. The size of the sensor nodes vary from that of a shoebox to that of a minute sandparticle. Similarly their cost also varies from hundreds of dollars to a few pennies. Size and cost constraints result in corresponding constraints on energy, memory, computational speed and communications bandwidth.

Wireless Sensor Networks [2] are characterized

- Limited power they can harvest or store.
- Ability to cope with node failures.

- Heterogeneity of nodes.
- Large scale of deployment.
- Mobility of nodes.
- Communication failures.
- Dynamic network topology.
- Ability to withstand harsh environmental conditions.

Various protocols exist for addressing the issue of utilizing the available limited power they can harvest or store such as TREEPSI (Tree based Energy Efficient Protocol for Sensor Information), TBC (Tree based Clustering for Energy Efficient Wireless Sensor Networks), EDACH (Energy Driven Adaptive Clustering Hierarchy), TEEN (Threshold Sensitive Energy Efficient Sensor Networks), LEACH (Low Energy Adaptive Clustering Hierarchy), etc. LEACH is an effective energy efficient hierarchical-based routing protocol. Our prime focus is on the analysis of LEACH based upon certain parameters like network lifetime, stability period, etc. and also the effect of selective forwarding attack and degree of heterogeneity on LEACH protocol. Energy is not properly utilized and throughput is decreased in networks compromised by selective forwarding attack but the number of cluster-heads per round remains unaffected in such networks.

The wireless sensor node, being a microelectronic device, can only be equipped with a limited power source ( $\leq 0.5$  Ah, 1.2 V). In some application scenarios, replenishment of power resources might be impossible. Sensor node lifetime, therefore, shows a strong dependence on battery lifetime. Power consumption can hence be divided into three domains:

- 1. Sensing Power
- 2. Communication Power
- 3. Data Processing Power

LEACH reduces the amount of energy consumed during data transmission and as a consequence, the energy that the radio consumed. Thus the energy utilization is reduced. In our scheme, we are further reducing the amount of energy spent by each node to transmit its data by more than 40\$\%\$ approximately as compared to the existing LEACH protocol.

Data gathering is the main objective of sensor nodes. The sensors periodically sense the data from the surrounding environment, process it and transmit it to the base station or sink node. The frequency of reporting the data and the number of sensors which report the data depends on the particular application. Data gathering



2018 IEEE 25th International Conference on High Performance Computing Workshops (HiPCW)

### Health Management of a Typical Small Aircraft Fuel System using an Adaptive Technique

Vijaylakshmi S. Jigajinni Dept. of Electronics and Communication Basaveshwar Engineering College, Bagalkot, Karnataka, India talk2vijusj@gmail.com

Abstract— Faults in the aircraft fuel system will degrade its performance and may lead to the complete system failure. In commercial aircraft system, efficient diagnosis can optimize the line to return the aircraft to service, thus allowing less disruption to passenger travel. In this work, an adaptive fault dingnosis technique is developed for a typical small aircraft fuel system, which facilitates efficient learning procedure to forecast is to utilize the system answers for non-linear situations. This adaptive

system, which facilitates efficient learning procedure to forecast the system parameters for non-linear situations. This adaptive leshnique represents the integration of the Fuzzy Logic and Support Vector Machine (SVM) algorithms in the field of fault disgnosis. Using this adaptive technique health monitoring of aireraft fuel system is discussed. In an aircraft fuel tank, the fault is effectively located by assessing and contrasting the actual parameters and set point parameters related to the system for various time-frames. The fuzzy logic controller is enaligured with the logical rules as per the required target migut. It relies on the aircraft fuel system parameters like the find flow rate, level of fuel in the tank, fuel temperature, and fuel From the logical rules, the control signals related to in sireraft fuel system are derived by the SVM technique. The effect ney in execution of this fault diagnosis tool-based aircraft aysiem gets authenticated in the MATLab/Simulink mailurm. The simulation is carried by assuming normal menting conditions of aircraft in the laboratory environment.

www.erds-aircraft fuel system, system health management, in management, fault prognosis, SVM, fuzzy logic

#### I. INTRODUCTION

Nowadays, aviation is becoming an integral part of our daily lives and airplane is becoming an essential method of mender and cargo transports. Integrated Vehicle Health Management (IVHM) of airplanes strongly influences the maintenance efficiency of civil and military fleets. Guigang and al. [1] suggested a framework of system health mont that provides integrated health management of running worldwide. This monitors the health status In order to build a suitable mechanism for managing the full-time fault diagnosis, fault prediction and intelligent instance decision of airplanes. Model-based simulation and an implemented is a robust framework for the analysis of complex systems in recent years [2,3]. the approach is also used to manage failures and complexities during the design and simulation phases of the The selection of a particular design of a system for

Vanam Upendranath Aerospace Electronics and Systems Division, CSIR-National Aerospace Laboratories, Bengaluru, Karnataka, India

fault diagnosis and prognosis has to be made through a tradeoff between safety, cost, and performance of the system [4].

The positive impact on the development of system health management is on the maintenance cost and the extra cost due to the unscheduled maintenance of the aircraft. The purpose of the System Health Management (SHM) process is to utilize the best possible fault identification, detection and prediction of remaining useful life of the system, subsystem or a component. In this work, a simulation model is built for the fuel system of a typical small aircraft that helps to correct and mitigate the current fault status. An adaptive health management methodology is developed by integrating fuzzy logic and SVM techniques which facilitates to diagnose the faults and take corrective measures to meet the required rate of fuel to the engines.

#### II. RELATED WORK

This section provides a brief literature review on system health management techniques. R. Guo et al. [5] presents a fault diagnosis expert system for different faults in a civil aircraft which is based on the fault tree analysis. The designing method of each module is given in detail, and the process of calculation is simplified through Direct Binary Decision Diagrams (DBDD) method, which investigates component probability importance. They presented an analysis of landing gear retraction system and explained the feasibility of the design of an expert system-based fault diagnosis method for civil aircraft. Regenerative braking is one of the vital technologies utilized in hybrid electric automobiles to improve the stability of the vehicle and also to analyze its energy efficiency. Chaitanya Sankavaram et al. [6] have presented an efficient data-driven technique for detecting and diagnosing faults in the regenerative braking system of hybrid electric vehicles where it includes signal processing and statistical techniques for feature extraction and fault classification methodologies to identify, detect and isolate faults in the regenerative braking system. The results demonstrate that highly accurate fault diagnosis is possible with the classification methodologies. The process can be employed for fault analysis in a wide variety of systems from automobiles to aerospace sectors.

The synergistic integration of information from electronic sensors and human sources is called hard/soft

14-9/18/\$31.00 ©2018 IEEE 111PC'W.2018.00025 104

Computer society

# Performance Analysis of Ant Colony based Routing Approach for VANETs using VanetMobiSim and NS2

Poornima M. Chanal Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot – 587102 Karnataka, India poornima.chanal@yahoo.com Mahabaleshwar S. Kakkasageri Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot – 587102 Karnataka, India mahabalesh\_sk@yahoo.co.in

Abstract— Vehicular networks are a class of wireless networks built dynamically in an ad hoc network manner. Routing mechanism is an important component for the success of any vehicular applications. The routing mechanism in vehicular networks is critical because of VANET constraints such as unpredictable network topology, frequent disconnections, varying network densities, etc. Bio-inspired algorithms are the part of the natural computations. These algorithms are basically inspired by the insects like bee, ants, fish etc. by nature. Ant Colony Optimization (ACO) technique has proved as a good technique for developing routing algorithms in ad hoc networks. In this paper, we have applied the ACO technique for routing in car to car and car to road side communication infrastructure under VANET constraints. Performance parameters analyzed are packet delivery ratio, throughput, End-to-End delay, and Control Overheads. The comparison analysis is done using network simulators NS-2 and VanetMobiSim to generate realistic vehicular traffic of a particular area i.e., Bagalkot city, Karnataka state of India.

#### Keywords- V2V, V2I, VANETs, Ant Colony Optimization

#### I. INTRODUCTION

Modern vehicles with capability of sophisticated computing and communication are a good sign for Vehicular Adhoc Networks (VANETs). VANETs communicate with IEEE 802.11p (dedicated short range communication/Wireless access for vehicular environment).Information aggregation, validation and dissemination in VANETs are usually built with routing protocols. In VANETs, efficient routing protocols should address the issues such as unpredictable topology, density of vehicles, mobility, heterogeneous communication environment, etc. [1] - [3].

Many research works have been proposed for bio inspired routing in VANET for improvement of routing mechanism. Few on-going related research works are summarized as follows. The work given in [4] mentioned the various information management schemes considering four network architectures of VANET as Vehicle-to-Vehicle (V2V), Vehicle-to-Infrastructure (V2I), Vehicle-to-Pedestrian (V2P) and hybrid network architecture. Ad-hoc On demand Distance Vector(AODV) routing mechanism is analysed and compared with Dynamic Source Routing(DSR), Temporally Ordered Routing Algorithm (TORA), Associativity Based Routing (ABR) in [5]. Comparison of different routing protocols like AODV, DSR, Destination Sequence Distance vector Routing (DSDV), and TORA for different parameters is mentioned in [6]. Bio inspired routing protocol characters, advantages and disadvantages are discussed in [7]. For better performance of routing protocol, it is better to optimize the results using bioinspired algorithms like Ant Colony based Optimization

Rahul G. Jahagirdar Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot – 587102 Karnataka, India rahuljahagirdar006@gmail.com Mamata J. Sataraddi Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot – 587102 Karnataka, India kdmamata@rediffmail.com

(ACO) techniques and Particle Swarm Optimization(PSO) technique. This helps to find multiple paths and avoids link failure. Performance analysis of routing protocol and its optimization is also evaluated on the basis of QoS parameters such as energy consumption, jitter, throughput, delay, routing load, packet loss, and packet delivery ratio. The performance parameters using Zone Routing Protocol (ZRP), DSR, DSDV and Distance Routing Effect algorithm for Mobility (DREAM)& Simple Forwarding over Trajectory (SIFT) are analysed in [8].

Secured routing protocol based on the calculation of the trust worthiness of the path is explained in [9]. ACO is used as an intelligent technique to calculate the trust paths. Improvement of QoS in terms of route selection over the network is discussed in [10]. ACO approach is used for the Vehicle Routing Problem. Dynamic MANET On demand (DYMO) protocol avoids the issue of network congestion if an accident happens in VANET.

Opposition based Ant Colony Optimization (OACO) algorithm for highways is proposed in [11]. The research works mentioned in [12] - [13] discuss ACO algorithm with the available vehicle position and mobility that performs well in the dynamics of such vehicular networks. DYMO routing protocol is adopted for ACO algorithm leading to Mobility Aware ant colony optimization Routing DYMO (MAR-DYMO). MAR-DYMO performance is evaluated for city scenario and analysed with standard routing schemes.

Ant colony algorithm also prevents attacks and helps in establishing secure and efficient route from source to destination is proposed in [14]. Ontology and vehicle traffic information based routing mechanism ensures the packets transmission with reliable connectivity is discussed in [15]. This paper is organised as follows. In section II, a brief explanation of ACO for VANETs is presented. Section III presents simulation model, simulation inputs and result analysis. Finally, section IV concludes the proposed scheme.

#### II. ANT COLONY OPTIMIZATION FOR VANET

Ant Colony Optimization (ACO) technique based on travelling salesman problem is proposed for VANET environment (i.e., V2V & V2I).

#### A. Ant Colony Optimization (ACO)Technique

Bio-inspired algorithms are the part of the natural computations. They use bottom up approach and decentralized approach. These algorithms are basically inspired by the insects like bee, ants, fish etc. by nature. These algorithms are basically having some set of rules and

#### 978-1-7281-5286-8/19/\$31.00 ©2019 IEEE

#### A Survey on Information Disssemination in VANETs

M. S. Kakkasageri Dept. of Electronics and Communication Engineering Basaveshwar Engineering College Bagalkot, Karnataka, India mahabalesh\_sk@yahoo.co.in

Abstract—Vehicular Ad hoc Network (VANET) has become an active area of research, standardization, and development because next generation vehicles will have capability of sensing, computing, and communicating capabilities. Different components in a vehicle have to constantly exchange available information with other vehicles on the road and cooperate for the purpose of ensuring safety and comfort of users using VANET. In this chapter, we survey some of the ongoing recent research efforts in information dissemination in VANETs. We also outline some of the research challenges that still need to be addressed to enable efficient information dissemination in VANET.

#### I. INTRODUCTION

A Mobile Ad-hoc Network (MANET) is comprised of a group of mobile nodes which have the capability of selforganization in a decentralized fashion and without fixed infrastructure. A Vehicular Ad hoc Network (VANET) is an example of a MANET where the mobile nodes are the vehicles themselves. Communication is possible between vehicles within each other's radio range as well as with fixed road side infrastructure components. The VANET concept is an integral part of the intelligent transportation system (ITS) architecture, which aims to improve road safety, optimize traffic flow, reduce congestion, and so on [1]-[6].

The key differences of VANET as compared to MANET environment are as follows: 1) components building the network are vehicles, 2) restricted mobility constraints, 3) extremely high mobility and time-varying vehicle traffic density, 4) most of the vehicles provide sufficient computational and power resources, thus eliminating the need for introducing complicated energy-aware algorithms, 5) vehicles will not be affected by the addition of extra weight for antennas and additional hardware.

VANET raises several interesting issues with regard to Media access control (MAC), Mobility management, Data aggregation, Data validation, Data dissemination, Routing, Network Congestion, Performance analysis, Privacy and Security.

With the increase of portable devices as well as progress in wireless communication, vehicular ad hoc networking is gaining importance with the increasing number of widespread applications. Some of the important applications S. S. Manvi Dept. of Electronics and Communication Engineering Reva Institute of Technology and Management Bangalore, Karnataka, India agentsun2002@yahoo.com

of VANETs are as follows: 1) message and file delivery, 2) providing location-dependent services like the location of the nearest facilities like fuel stations, parking zones, entertainment places and restaurants, etc., 3) Internet connectivity, 4) information and warning functions, 5) dissemination of road information (including incidents, congestion, surface condition, etc.) to vehicles distant from the subjected site, 6) co-operative assistance systems, 7) traffic monitoring and management services, 8) other advanced services like interactions between VANET nodes and the road infrastructure, for example traffic calming measures (including automatic speed limiters), intelligent road signs, tolling (congestion charges for entering urban areas at peak times, cargo monitoring, etc.,).

VANET architectures for vehicular communication may be classified into three types: purely ad hoc based, infrastructure based and hybrid type. In the purely ad hoc based VANET architecture, vehicle to vehicle (V2V) communication exists. In the infrastructure based VANET, communication is possible in between vehicles using infrastructure (V2I) such as base stations or access points on the road. Combination of V2V and V2I architectures leads to the hybrid mode of VANET.

VANET scenario (hybrid) is as shown in the figure 1. All vehicles are equipped with General Positioning Systems (GPS) and on-board communication devices for communication. Each vehicle is loaded with location digital map and is concerned about road information ahead of it on its direction. Each vehicle communicates with other vehicle and road side unit (like base station, gateway, etc.) within its communication range R. Vehicles within the communication range R form clusters (C1, C2,..CN). All the road side unit are connected to regional transportation system.

#### **II. RELATED WORKS**

Researchers have started to address information dissemination related issues in VANET. Some of the research works for information dissemination in VANETs are as follows. A method for cars to autonomously and cooperatively collect traffic jam statistics to estimate arrival time to destination for each car using inter-vehicle communication is proposed in [7]. Optimal next-hop selection in a route between two

# Cascaded Structure of Noise Reduction and Multiband Frequency Compression of speech signal to Improve Speech Perception for monaural Hearing Aids

Rajani S. Pujar, P. N. Kulkarni Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587103, Karnataka, India rajani\_pujar@rediffmail.com, pnk\_bewoor@yahoo.com

Abstract - This paper presents a cascaded structure of noise reduction and multiband frequency compression in enhancing speech perception for sensorineural hearing impaired persons using monaural hearing aid. In multiband frequency compression scheme, input signal is divided into 18 frequency bands ranging from 0 to 5 KHz based on auditory critical bandwidths (ACB). Frequency components of each band are compressed at the center of their respective bands using frequency segment mapping method. This results in concentration of all spectral energy at the center of each band to minimize the effect of spectral masking. Noise reduction scheme, namely wiener filter is used to reduce the background noise. Evaluation process involves i) Speech quality assessment (using PESQ and MOS) on normal -hearing subjects ii) Speech intelligibility assessment (using MRT) on hearing-impaired subjects. The test material consists 50 sets of consonant-vowelconsonant (CVC) words. Each set includes six words. Each subject is responded for 1800 presentations (300 CVC words x 6 different SNR conditions). The MOS, PESQ and MRT scores obtained for the SNR values ranging from -6 dB to + 6 dB indicates enhancement in speech quality and intelligibility.

Keywords- Sensorineural Hearing Loss (SNHL); Spectral Masking; Multiband Frequency Compression; MOS; PESQ

#### I. INTRODUCTION

Hearing loss mainly happens when the working of cochlea and the auditory nerve does not function properly. Usually auditory filter bandwidth gets widened and results in reduced frequency selectivity because of the masking of frequency components [1] [2] [3]. This results in listening discomfort by hearing impaired people. Furthermore, intelligibility of the speech gets reduced in the presence of background noise. Hence under noisy environments hearing impaired people find difficulty in understanding speech signal. Therefore Noise reduction techniques as well as hearing aid algorithms are essential to improve quality and the intelligibility of speech under adverse listening conditions.

Many single and multi channel noise reduction techniques have been proposed in past which are applicable for both monaural and binaural hearing aids. Noise reduction techniques helps in achieving similar speech intelligibility as that of normal hearing people in noisy environments. Some of the single channel noise reduction techniques are Spectral subtraction [4], spectral subtraction using spectral floor and over-subtraction [5], wiener filter [6] etc. Multichannel noise reduction techniques exploit spectral and temporal differences and also the spatial separation between sound sources to improve SNR [7]. Spatial domain (direction or location) difference between a target and noise signal can be used to achieve noise reduction in different noisy conditions using a system with a microphone array [8].

Hearing aids help in amplification (frequencydependent) of speech signals for hearing-aid users. Existing hearing aids do not provide sufficient amplification of high frequency speech sounds [9]. This restricts the audibility of high frequency sounds, predominantly for severely hearing impaired subjects. High frequency samples of speech signal are shifted to low frequency regions by using frequency transposition methods [10]. Transposition of frequency sample methods proposed in [11]-[14] were useful in speech recognition, but distorts the speech quality. Multi-band frequency compression method to reduce frequency masking associated with SNHL persons is discussed [15]. By using frequency compression, frequency lowering or frequency transposition algorithms the effect of frequency masking can be minimized to some extent in noisy environments, but results in degraded speech quality. In order to improve speech perception in noisy conditions, and to overcome spectral masking associated with sensorineural hearing loss (SNHL) listeners, a cascaded structure of noise reduction (using wiener filter) and multiband frequency compression of the speech signal is proposed.

Remaining part of the paper is as follows. Section II explains signal processing. Tests and result analysis are given in section III. Spectrograms are explained in section IV. Section V concludes our research work.

#### II. SIGNAL PROCESSING

This study presents a cascaded structure of (i) Noise reduction and (ii) Multi band frequency compression of the speech signal for SNHL listeners. Wiener filter is used in our cascaded scheme to reduce the noise present in a speech signal. Wiener filter [16] minimizes the mean square error (MSE) between the desired signal and an estimated signal. Wiener filter requires separate estimate of the clean speech and the noise power which can be estimated using a Voice Activity Detector (VAD). Proposed VAD method to discriminate between speech and noise is shown in Figure 2.

In multi band compression scheme, the speech signal sampled at 10 KHz sampling frequency is segregated into number of frames (frame duration of 20ms and frame overlap of 50%). Hamming widow is applied for each of these frames. The frequency scale is then partitioned into 18 bands based on Auditory Critical Bandwidths (ACB) [17].

# Cascaded Structure of Wiener Filter with FBS based Spectral Splitting and Dynamic Range Compression For Listeners with Sensorineural Hearing Loss

Rajani S. Pujar, Pandurangarao N. Kulkarni Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, India rajani\_pujar@rediffmail.com, pnk\_bewoor@yahoo.com

Abstract- This paper presents a cascaded structure of wiener filter followed by spectral splitting using Filter Bank Summation (FBS) method and dynamic range compression with constant compression factor for binaural dichotic presentation. This helps to compensate the effect of spectral masking, reduced dynamic range for moderate sensorineural hearing loss and also improves speech perception in the adverse listening conditions by removing unwanted noise. Wiener filter is used as noise reduction algorithm. Using auditory critical bandwidths, the signal is separated into eighteen frequency bands, varying from 0-5 KHz. Spectral splitting is performed by combing odd and even numbered filter bands and presented to the left and right ears respectively that help in reducing the effect of frequency masking. Amplitude compression using fixed compression factor (0.4, 0.6 and 0.8) in each band was carried out that helps in compensating the reduced dynamic range of the hearing impaired people. Listening tests using Mean Opinion Score (MOS) on normal-hearing subjects were conducted to test the perceived quality of the processed speech. VCV syllable /aba/ and sentences were used as the test material. The results of the listening tests showed that the MOS scores for the processed speech (4.31, 4.06, 3.68, 3.5, 3.067 and 2.612) were higher as compared with those for unprocessed speech (4.48, 1.2, 1.13, 1.06, 0.8 and 0.45) for SNR values of  $\infty$ , +6, +3, 0, -3 and -6 dB respectively, indicating the improvement in the perceived quality for different SNR conditions for the compression factor of 0.6. To estimate the intelligibility of the perceived speech, listening test was also carried out on hearing impaired (moderate SNHL) subjects in the presence of background noise using Modified Rhyme Test (MRT). The test material consists 50 sets of monosyllabic words of consonant-vowelconsonant (CVC) form with six words in each set. Each subject is responded for total of 1800 presentations (300 words x 6 different SNR conditions). Results of the listening tests (using MRT) showed maximum improvement in speech recognition scores (23.49 - 27.29%) compared to unprocessed speech, indicating enhancement in intelligibility of the speech at lower SNR values.

Keywords-Wiener Noise Reduction Algorithm, Filter bank summation (FBS) method, Dynamic Range Compression (DRC), Spectral Splitting, Mean Opinion Score (MOS), Modified Rhyme Test (MRT)

#### I. INTRODUCTION

Degraded speech perception, increased temporal and spectral masking and reduced dynamic range between threshold of hearing and discomfort level are some of the problems associated with sensorineural hearing impaired people [1]. Furthermore, speech intelligibility gets reduced in the presence of background noise. Therefore hearing impaired people need a higher signal-to-noise-ratio (SNR) to effectively communicate.

Noise reduction in hearing aids is a very challenging issue because the properties of the noise signal change significantly with time. Hence it is very difficult to develop an algorithm that works in different listening environments. The nature and properties of noise sources are very important in deciding which noise reduction algorithm works well to enhance speech signal by eliminating the noise from the corrupted signal. Some of the noise reduction schemes applicable for hearing aids are spectral subtraction, wiener filter, and beam former approaches [2] [3] [4]. Comb filter pairs (with complementary magnitude responses) were used to enhance the intelligibility of perceived speech for bilateral hearing (moderate SNHL) impaired subjects [5].

Dynamic range Compression can be implemented in either within a single frequency band or multiple frequency bands i.e., multiband compression. Hearing aids employing dynamic compression split the signal into number of frequency bands, and amplitude compression is carried out in these bands. Filter bank summation technique (FBST) is used for multiband amplitude compression [6] to partly compensate the effect of reduced dynamic range for the people with sensorineural hearing loss. Therefore it is essential for hearing aids to improve speech perception under adverse listening conditions, and also to overcome the effects of reduced dynamic range and spectral masking effect associated with SNHL listeners.

Hence in our proposed work, we have cascaded wiener filter followed by spectral splitting using Filter Bank Summation (FBS) method and dynamic range compression using constant compression factor. Wiener filter obtains the enhanced signal by optimizing the mean square error criteria. In FBS method, frequencies ranging from 0-5000Hz are partitioned into 18 bands, depending on auditory critical bands. Each band is amplitude compressed with the constant compression factor of 0.4, 0.6 and 0.8. Spectral splitting of compressed speech is carried by combining odd numbered and even numbered filters and presented them to two ears.

Rest of the paper is organized as follows. Related works are discussed in section II. Proposed algorithm is given in section III. Tests and Result analysis are given in section IV. Spectrographic analysis is discussed in section V. Finally, section VI concludes the work.

#### **II. RELATED WORKS**

Many studies have investigated the different signal processing techniques to overcome the problem of SNHL

## EFFECT OF NOISE REDUCTION ALGORITHMS ON TEMPORAL SPLITTING OF SPEECH SIGNAL TO IMPROVE SPEECH PERCEPTION FOR BINAURAL HEARING AIDS

ia

Rajani S. Pujar, Pandurangarao N. Kulkarni Department of Electronics and Communication Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, India e-mail: rajani\_pujar@rediffmail.com, pnk\_bewoor@yahoo.com

Sensorineural hearing loss which occurs due to the damage of cochlea, auditory nerves or both, is characterized by elevated hearing thresholds, loudness recruitment, reduced temporal and frequency resolution, and increased temporal and spectral masking. Increased temporal masking affects the speech perception in persons with sensorineural hearing impairment especially under adverse listening conditions. Hearing aid users have problem in speech understanding particularly in the presence of background noise. Hence, hearing aids should include noise reduction techniques for enhancing the speech perception. This paper presents a cascaded scheme, which employs a noise reduction algorithm as well as temporal splitting of the speech signal. The main objective of the paper is to improve quality of perceived speech for SNHL listeners by reducing the temporal masking and improve the speech perception in the background of additive noise. Earlier investigations have shown that by splitting the speech temporally and presenting alternate segments to the two ears help in reducing the effect of temporal masking. In this technique, the speech signal is processed by two fading functions, complementary to each other, and presented to left and right ears for binaural dichotic presentation. In the present study, half cosine signal is used as fading function with crossover gain of 6 dB for perceptual balance of loudness. Temporal splitting is combined with noise reduction algorithm to improve speech perception in the background noise. Two noise reduction schemes, namely spectral subtraction and Wiener filter are used. We have used Voice Activity Detection (VAD) based on short term energy and zero crossing rates to detect noise and speech frames separately. Listening tests were conducted on six normal hearing subjects, with sensorineural loss simulated by adding broadband noise to the speech signal at different signal-to-noise ratios ( $\infty$ , 3, 0, and -3 dB) using mean opinion score(MOS) test. Objective evaluation using PESQ was also carried out. The MOS score for VCV syllable /asha/ for SNR values of ∞, 3, 0, and -3 dB were 5, 4.46, 4.4 and 4.05 respectively, while the corresponding MOS scores for unprocessed speech were 5, 1.2, 0.9 and 0.65, indicating significant improvement in the perceived speech quality for the proposed scheme compared to the unprocessed speech.

### 1. Introduction

Sensorineural hearing loss which occurs due to the damage of cochlea, auditory nerves or both, is characterized by elevated hearing threshold, loudness recruitment, reduced temporal and frequency resolution, and increased temporal and spectral masking [1] [2]. Temporal masking, also known as

1

/8/2020	Agent Based Job Clas	sification and Reso	urce Allocation in I	oT - IEEE Coi	nference Publication
	e Digital Library   IEEE-SA   IEEE Spec	Inotitutional Sides		,	Cart   Create Account   Personal Sign I
		institutional Sig	in m		
Browse	My Settings	Get Help	Subscribe		
					< Previous   Back to Results   Next >
Conferences > 2017 Int	ernational Conference				
Agent Base	d Job Classification and	d Resource	Allocation	in loT	
Publisher: IEEE	Cite This				
<< Results   < Pre	vious   Next >				
2 Author(s) Dane	eshwari I. Hatti ; Ashok V. Sutagundar Vie	w All Authors			More Like This
					Cloud computing and Internet of Things fusion: Cost issues 2017 Eleventh International Conference on
193			- Gr	Alorte	Sensing Technology (ICST) Published: 2017
Full Text Views	1.03			Aierts	Architecture Design of the Internet of
				Manage Content Alerts	Things Based on Cloud Computing 2015 Seventh International Conference on Measuring Technology and Mechatronics
				Add to Citation	Automation Published: 2015
				Alerts	View More
Abstract					Top Organizations with Patents
	Downi				on Technologies Mentioned in This Article
L Introduction	PDF				ORGANIZATION 4
II. Proposed Work	Abstract: Internet of Things (IoT) is	combination of thing	s and Internet. Thi	ngs in IoT	ORGANIZATION 3
III. Results	issues View more	everal neterogeneou	is devices and rest	its to many	ORGANIZATION 2
IV. Conclusion	Metadata				
Authors	ADSTRACT: Internet of Things (IoT) is combination	n of things and Inte	met. Things in IoT		•
Figures	communicate via Internet. IoT has se issues. In this work we have propose	everal heterogeneou d agent based job o	s devices and resu classification and re	Its to many source	
References	allocation in IoT. In the proposed me System. Agents are defined in two hi	hodology agent use erarchy namelv at L	es the Fuzzy Interfe ocal Processor Act	rence ency (LPA)	
1010101000	and IoT Cloud Agency (ICA). Agents allocation. The jobs are classified an	interact both for job	classification and	resource asks are	
Keywords	allocated with sufficient resources av	ailable in devices ar	nd ICA facilitates re	sources if	
Metrics	simulated using C++. Some of the pe	es. To test the perfo rformance paramete	rmance, the proposers considered are	sed work is response	
More Like This	time, energy consumption and bandy	vidth utilization.			
	Published in: 2017 International Con Communication Technology (ICRAE)	nference on Recent CT)	Advances in Electr	onics and	
	Date of Conference: 16-17 March 2	017 INSPEC Acc	cession Number:	17289604	
	Date Added to IEEE Xplore: 26 Oct	ober <b>DOI:</b> 10.110	9/ICRAECT.2017.2	2	
2 Martin Providence	2017	Puolisiien I			

IEEE websites place control of these cookies. To learn more, read our Privacy Policy.

Accept & Close

3/8/2020	Parabola structure based information gathering using AUV for underwater sensor	network - IEEE Conference Publication
IEEE.org   IEEE Xploi	re Digital Library   IEEE-SA   IEEE Spectrum   More Sites	Cart   Create Account   Personal Sign Ir
	Institutional Sign In	
_		
Browse	My Settings Get Help Subscribe	
		< Previous   Back to Results   Next >
Conferences > 2017 Int	ernational Conference	
Parabola str	ucture based information gathering using AUV for	
underwater	sensor network	
Publisher: IEEE	Cite This	
<< Results   < Prev	vious   Next >	
3 Author(s) Neela	akanth L. Jadhav : Ashok V. Sutagundar : B.S. Halkarpinath View All Authors	More Like This
		consumption in wireless sensor networks 2015 IEEE International Conference on
		Electronics, Circuits, and Systems (ICECS) Published: 2015
91 Full	Aler	ts Energy consumption in long-range linear
Text Views	Manage	wireless sensor networks using LoRaWan and ZigBee 2017 IEEE 60th International Midwast
	Content A	Symposium on Circuits and Systems (MWSCAS)
	Add to Ci	Published: 2017 lation
	Alerts	View More
		Top Organizations with Patents
Abstract		on lechnologies Mentioned in This Article
Document Sections		ORGANIZATION 4
I. Introduction	Abstract: This paper intende to provide on ensure officiant each all a leader have	ORGANIZATION 3
II. Basics of Acoustic Communication	information gathering to enhance the network life of Underwater Sensor Networks	O ORGANIZATION 2
III. Data Gathering	(UWSN). In UWS View more	ORGANIZATION 1
Using Traditional Method for	Metadata Abstract:	${\boldsymbol{\heartsuit}}$
UWSN	This paper intends to provide an energy efficient parabolic structure based information	on
IV. Proposed	gathering to enhance the network life of Underwater Sensor Networks (UWSN). In UWSN, sensor nodes closer to the sink node diminish energy sooner since they sha	re
Structure Based	larger load of packets relay. This problem worsens when water to be monitored is ve deep which is also known as botspot problem. Hence in this paper an effort has been	ry
Information Gathering in	made to mitigate imbalance in the energy consumption using a parabolic structure.	11
UWSN	Existing studies deals with multi-hop data transmission in a network. Proposed meth creates a parabolic structure in all three layers such as surface layer, middle layer.	od
V. Performance Evaluation	bottom layer and sensed data is collected at the parabolic head nodes[PHN]. Furthe	r
	travels in a predefined path to collect the data from parabolic head nodes [PHN] from	1
Authors	each layer. Proposed method is to improve the problem of unbalanced energy consumption. Simulation results show that the proposed model effectively and unifor	mly
Figures	balances the energy consumption and thereby increasing network lifetime.	,
i iguica	Published in: 2017 International Conference on Computing Methodologies and	
References	Communication (ICCMC)	

Keywords IEEE websites place DetoRice one of the state of the second s

you agrees to the placement of these cookies. To learn more read our Privary 2017:3282707

3/8/2020	Energy Efficient I	Dynamic Navigation S	ystem - IEEE	Conference F	Publication
IEEE.org   IEEE Xplore	Digital Library   IEEE-SA   IEEE Spectrur	n ⊢ More Sites		С	art   Create Account   Personal Sign In
	I	Institutional Sign In			
Browse	Mv Settinas G	iet Help	Subscribe		
		h			< Previous   Back to Results   Next >
Conferences > 2018 Sec	ond International Con				
Energy Effici	ient Dynamic Navigation	System			
Publisher: IEEE	Cite This				
<< Results   < Prev	ious   Next>				
( Ashel	- V Sutosundos - Doduncio Kurska - Misur All	Authors			Manu Like This
Ashor	v Sulagundar ; Padmaja Kuruba – <b>view Ali</b>	Authors			Early alert system using relative positioning
			Tome: *c		in Vehicular Ad-hoc Network 2014 Annual International Conference on Emercing Response Areas: Magnetics
37 Full				Alerts	Machines and Drives (AICERA/ICMMD) Published: 2014
Text Views				Manage	Localization of vehicular ad-hoc networks
				Content Alerts	with RSS based distance estimation 2018 International Conference on
				Add to Citation	Technologies (iCoMET) Published: 2018
				Alerts	View More
Abstract					Top Ormanizations with Patents
Document Sections	Ourry				on Technologies Mentioned in This Article
I. Introduction	ŶО <del>г</del>				ORGANIZATION 4
II. Related Work	Abstract: In the recent years there is c	onsiderable growth in t	he Vehicular	system,	ORGANIZATION 3
III. Proposed Work	(GPS) is on View more	the current trend, Glob	ai positioninį	j system	ORGANIZATION 2
IV. Results	Metadata				ORGANIZATION 1
V. Conclusion And	Abstract: In the recent years there is considerable	e growth in the Vehicul	ar system, dr	iver safety	
Future Scope	and assistance system. In the current tr	rend, Global positioning	) system (GP	S) is one	
Authors	of many people in the world today. If bo	th cell phone and GPS	receiver are	operational,	
Figures	it leads to increase in the cost and ener	gy of navigation. The n amount of energy consi	nain objective umed in the r	e of this work avigation	
References	system. In this paper we propose, an er navigation system architecture for the d	nergy efficient, high acc lynamic navigation usin	curacy, low co ig the existing	ost J	
Kevwords	infrastructure. It has three techniques for system namely: (1) uses reservoir as th	or accessing location m	aps for the n	avigation	
	Internet connection). It uses cell phone	as interface between th	he navigation	system and	
Metrics	the reservoir. (2) It is assumed that som (VANET) have location information. In the	ie of the vehicles in Vel his case cell phone is u	hicular Adhoo ised as an inf	e Network terface	
More Like This	between VANET and navigation system	for transmission of loc	ation map an	id (3) cell	
	navigation system. Proposed navigation	system comprises of	memory, Gra	phical User	
	Interface (GUI), audio/visual system and communication system.	a short range Radio Fre	equency(RF)		

Published in: 2018 Second International Conference on Advances in Electronics,

IEEE websites placement of these cookies. To learn more, read our Privacy Policy.

Accept & Close



Imp rint Any bis nd names and product names mentioned in this book are subject to trademark, brand or patent protection and are trademarks or registered trademarks of their respective holders. The use of brand names, product names, common names, trade names, product descriptions etc. even without a particular marking in this work is in no way to be construed to mean that such names may be regarded as unrestricted in respect of trademark and brand protection legislation and could thus be used by anyone.

Cover image: www.ingimage.com

Publisher: LAP LAMBERT Academic Publishing is a trademark of International Book Market Service Ltd., member of OmniScriptum Publishing Group 17 Meldrum Street, Beau Bassin 71504, Mauritius

Printed at: see last page ISBN: 978-3-659-76258-1

Zugl. / Approved by: Bangalore, JU, Diss., 2018

Copyright © Bassnagouda FRonad, Suresh H Jangamshetti Copyright © 2018 International Book Market Service Ltd., member of OmniScriptum Publishing Group All rights reserved. Beau Bassin 2018

Basanagouda F Ronad Suresh H Jangamshetti

#### **Optimum Sizing of SPV Irrigation** Systems based on Field Conditions

LAP LAMBERT Academic Publishing

#### Contents

	Ack Ove	mowledgement rview of the Work Presented tents	1 3 7
1	Intro	oduction	9
	1.1	Overview of Agriculture in India	9
		1.1.1 Agriculture and Imgation – Indian Scenano	9
		1.1.2 SPV Powered Water Pumping Systems	12
		1.1.3 Operation of SPV Water Pumping Systems	14
		1.1.4 Advantages of SPV Water Pumping Systems	16
	1.2	Literature Review	18
	1.3	Motivation	22
	1.4	Research Objectives	23
2	Cro	p Water Assessment for the Selected Agriculture	24
	Lane	ds	2.4
	2.1	Introduction	24
		2.1.1 Evaporation and Transpiration	26
		2.1.2 Factors Affecting Crop Evapotran spiration	26
		2.1.3 Reference Crop Evapotranspiration	26
	2.2	Meteorological Data for Selected Site	30
	2.3	Different Methods for Assessment of Evapotran spiration	32
	2.4	Results and Discussions	36
3	Ene	rgy Conservation by Sizing Grid Connected AC	
	Irrig	ation Pumps based on Crop Water and Hydraulic Head	40
	at Fi	ield Conditions	
	3.1	Introduction	41
		3.1.1 Types of Irrigation	41
		3.1.2 Types of Pumps Employed in Irrigation Systems	44
	3.2	Methodology for Optimal Sizing of the Pumps	50
		3.2.1 Assessment of Flowrate & Diameter of PVC Pine Network	52
		3.2.2 Assessment of Hydraulic Head Offered to Pump	53
		7	

4	3.3 3.4 3.5 Opti	Energy Conservation, Cost Savings and CO <sub>2</sub> Reduction Survey at Malaprabha Riverbed Results Analysis and Discussions	59 60 64
4	3.4 3.5 Opti	Survey at Malaprabha Riverbed Results Analysis and Discussions	60 64
4	3.5 Opti	Results Analysis and Discussions	64
4	Opti		
		mal Sizing of SPV Powered DC Irrigation Pumps	75
	4.1	Introduction	75
	4.2	New method for Optimum Sizing of SPV Powered DC Pumps	77
	4.3	Validation of Proposed Methodology by Performance Analysis	78
	4.4	Results and Discussions of Performance Analysis	80
	4.5	Sizing SPV Panels based on Local Solar Potential	88
	4.6	Cost Analysis-Selected Cases in Malaprabha Riverbed	89
	4.7	Results Analysis and Discussions	90
5	Con	clusions and Future Scope	102
	5.1	Overall Conclusions	102
	5.2	Future Scope	104
Ref	ie ne no	es	105
Ima	geso	f Experimental Setup	109
Pho	otos v	vith Farmer's in Malaprabha Riverbed during Survey	112

a

# FPGA Based Implementation of Symmetrical Switching in SVPWM for Three Level NPC Converter

🖪 PDF

Publisher: IEEE Cite This

Raghuram L Naik ; Kavita B. Hunasikatti All Authors

	◙ <\$ © ⇒ ♠	
90 Full	Alerts	More Like This
Text Views	Manage Content Alerts Add to Citation Alerts	Field Programmable Gate Array (FPGA) Based Neural Network Implementation of Stator Flux Oriented Vector Control of Induction Motor Drive 2006 IEEE International Conference on
Abstract		Industrial Technology Published: 2006
Document Sections I. Introduction II. Symmetrical Switching in SVPWM (SSSVPWM)	Downl PDF Abstract: This paper presents the FPGA based implementation of symmetrical switching strategy in SVPWM for three level NPC converter. A simplified switching strategy is proposed by View more Metadata	A Comparison Between the GPI and PID Controllers for the Stabilization of a DC-DC "Buck" Converter: A Field Programmable Gate Array Implementation IEEE Transactions on Industrial Electronics Published: 2011
III. FPGA Implementation	This paper presents the FPGA based implementation of symmetrical switching strategy in SVPWM for three level NPC converter. A simplified switching strategy in proposed by exploiting the redundant states of existing updates.	
IV. Experimental Validation of Sssvpwm	available in three level space vector resulting into symmetrical switching vectors available. This switching strategy leads to generate switching sequence for all regions of every sector with each switch has to be turned "ON" once in time period regulting into laws daying strategy and reduced suitables long. The	
V. Conclusion	proposed switching strategy is developed using FPGA based Xilinx block sets and tested on bardware test banch consists of Induction motor integrated to the	
Authors	three level Neutral Point Clamped (NPC) converter controlled by FPGA based waved controller. It is observed from the results that the complexity of SVPWM	
Figures	for three level inverter is reduced to that of two level inverter and address the problem of parrow julke width at the midpoint of NPC converter.	
References		
Keywords	Published in: 2019 Third International Conference on Inventive Systems and Control (ICISC)	

### Performance Analysis of Directly Coupled Solar Photovoltaic Powered Irrigation Pumps

Publisher: IEEE PDF Cite This << Results | < Previous | Next > Maruti Naik ; Shashikanth Kori ; Basanagouda F Ronad ; Suresh H. Jangamshetti All Authors **0**<۩⇒♠ More Like This 26 Alerts Full Text Views Paper Citation Some studies on performance analysis of Manage Content Alerts two different laboratory scale solar photovoltaic water pumping in irrigation systems Add to Citation Alerts 2016 21st Century Energy Needs - Materials, Systems and Applications (ICTFCEN) Published: 2016 Abstract Down Energy, Economic, and Environmental (3E) Document PDF Analysis of PV Water Pump: Case Study of Sections the Tapioca Drip Irrigation System Abstract:SPV based irrigation pumps are installed across India. However I. Introduction 2021 International Conference on Power, these irrigation pumps are not widely employed. One of the probable reasons Energy and Innovations (ICPEI) II. Experimental behind this is mismatch of the ... View more Published: 2021 Setup Metadata III. Methodology Show More Abstract: IV. Results and SPV based irrigation pumps are installed across India. However these irrigation Discussions pumps are not widely employed. One of the probable reasons behind this is mismatch of the need and availability of pump ratings to the farmers. V. CONCLUSIONS Performance results of SPV based pumps vary with solar radiations, hydraulic head offered, pump rating etc. This necessitates the need for actual Authors performance results of pumps and selection criteria. This paper presents the detailed performance analysis of the 0.5HP, 1HP and 2HP directly coupled DC Figures irrigation pumps installed in the Energy park, Basaveshwar Engineering College[A], Bagalkot. The system is employed with pressure sensor, flow rate References sensor, speed sensor and suitable software to monitor and document the data. Performance data are logged for every 5 minutes. Results are tabulated and Citations analyzed. Further characteristics of centrifugal pumps are drawn. The presented results can be effectively used for suggesting irrigation pump Keywords capacity to farmers.

		ADVANCED SEARCH	
Conferences > 2018 IEEE Electron	n Devices Ko 🔞		
Design Optimiza Publisher: IEEE Cite Thi Maniula Sutagundar : B.G. Sh	ation of Microdisk Resonator	Using Interior Point Algorithm	IEEE Networking Letters Journal Call For Papers Now Open
29 Full Text Views		0 < © 🗁	Learn More
Abstract	Abstract:		More Like This
Document Sections	This paper presents design optimization of MEMS of dimensions of disk resonator for a particular resona	lisk resonator using interior point method. Determining the optimized nce frequency and quality factor along with minimum possible motion	d Optimization of Q factor in complementary spiral resonator
I. Introduction	resistance is attempted. The algorithm is implement	ed using MATLAB. The results obtained are compared with a fabric	for RFID application
II. Mems Disk Resonator	device. The developed method can provide faster d reduction of design time.	esign optimization compared to full wave simulators resulting in sign	2013 21st Telecommunications Forum Telfor (TELFOR) Published: 2013
III. Design Optimization of Disk Resonator	Published in: 2018 IEEE Electron Devices Kolkata	Conference (EDKCON)	Optimization of the Q factor in
IV. Results	Date of Conference: 24-25 Nov. 2018	INSPEC Accession Number: 18851444	Photonic crystal microcavities IEEE Journal of Quantum Electronics
V. Conclusion	Date Added to IEEE Xplore: 25 July 2019	DOI: 10.1109/EDKCON.2018.8770422	Published: 2002
			Show More



# A Comprehensive Review on Automatic Diagnosis of Diabetic Maculopathy in Retinal Fundus Images

I. S. Rajesh<sup>1(\Box)</sup>, M. A. Bharathi<sup>1</sup>, and Bharati M. Reshmi<sup>2</sup>

<sup>1</sup> Department of CSE, BMSIT & M, Bengaluru, Karnataka, India is.rajesh08l@gmail.com Department of Information Science and Engineering, BEC, Bagalkot, Karnataka, India

Abstract. Diabetic Maculopathy (DM) is one of the major problems of diabetes mellitus and it is one of the key reasons for the vision problem. It arises due to the leakage of blood from injured retinal veins. The development of DM is moderate and soundless and it is found in 10% of the world diabetic population. If diabetic maculopathy is not noticed in the underlying stage the effect this on macula is irreversible and can prompt vision loss. Therefore, screening of diabetic eye helps in finding diabetic maculopathy at the beginning stage which prevents the vision loss. In this review paper, the anatomy of the human eye and a brief overview of diabetes, diabetic retinopathy and diabetic maculopathy is presented. The literature review of various methods/techniques used for detection of DM in retinal fundus images and the performance metrics used to measure these methods are discussed in details. Issues involved in DM detection are also mentioned in this paper.

**Keywords:** Retinopathy  $(DR) \cdot Diabetic maculopathy <math>(DM) \cdot Optic disc (OD)$ Hard exudates  $(HEs) \cdot Blood vessels (BVs)$ 

### **1** Introduction

The human eye is a fundamental body part associated with vision [1]. Figure 1 demonstrates the retinal anatomical structure. The lens of the human eye helps in focusing light beams onto the retina and iris decides how much light is let into the eye. Optic nerve associates the eye to the cerebrum and conveys the electrical signals framed by the retina to the visual cortex of the mind.

The retina is a sensitive layer situated at the posterior of the eye. Uncountable photoreceptors present in the retina catch the light beams and change them into electrical signals. These electrical signals drive to the brain along the optic nerve where they changed into pictures. Figure 2 exhibits the structure of the retina and its primary parts. The optic nerve is the brightest region of the retina where Blood Vessels (BVs) begin. BVs are responsible for supply of nutrition and oxygen to the retina and it must be normal for the proper working of the retina and macula. The central portion of the macula is called fovea [2].

e retina and has no BVs in its surroundings. It acts as a natural sun block because of

© Springer Nature Singapore Pte Ltd. 2019

S. M. Thampi et al. (Eds.): SIRS 2018, CCIS 968, pp. 410–420, 2019. https://doi.org/10.1007/978-981-13-5758-9\_35

# Trust Computation Framework based on User Behavior and Recommendation in Cloud Computing

#### Tabassum. Mujawar

Department of Computer Engineering Ramrao Adik Institute of Technology Nerul, Navi Mumbai, Maharashtra, India tabbu3002@gmail.com

Abstract-Cloud computing provides shared environment for different resources and services that are available for users at anytime and from anywhere. Cloud computing has gained considerable attention of users and businesses. However, security concern is one of the major hurdles for acceptance of cloud computing. In order to guarantee security of data, it is necessary to grant access of data, only to authorized users. The traditional system applies different access policies and permission while granting access to any user. The analysis of user behavior is also important aspect, which can be integrated into access control model. In this paper, the trust computation model is presented that takes user behavior into consideration while providing access to the cloud data. The recommendation for the user is also one of the important components to assess user behavior. The proposed model evaluates trustworthiness of user on basis of reputation and recommendation. With the advent in machine learning techniques, applying learning based techniques in security domain has gained lots of popularity. In the proposed method, the machine learning technique (k-means clustering Algorithm) is incorporated in the trust computation process and the users are classified according their trust values.

Keywords— Cloud computing, Trust, Access control, Machine learning

#### I. INTRODUCTION

Cloud computing delivers various computing services such as storage, server, networks, analytics, platform, and much more over internet. The cloud users can access different services and use required resources through internet whenever needed. The features like scalability, flexible resources, lower operating cost, efficient management, on demand, increased collaboration, etc. have attracted large number of users and also IT industry. With the development of cloud computing, there is tremendous increase in the amount of data stored on remote servers available in cyberspace through cloud. The cloud computing services are available to users through three service delivery models namely, Software as a Service, Platform as a Service and Infrastructure as a Service. The different ways of cloud deployment are public cloud, private cloud and hybrid cloud. Due to immense benefits of cloud computing, large numbers of users are cultivated to store and

978-1-5386-9533-3/19/\$31.00 ©2019 IEEE

Lokesh B. Bhajantri, IEEE Member Department of Information Science and Engineering Basaveshwar Engineering College Bagalkot, Karnataka, India lokeshcse@yahoo.co.in

share their personal or confidential data using cloud storage services. Also, different organizations and business users employ cloud storage due to better utilization of resources at low cost and easy access. Despite of abundant benefits of cloud computing, the cloud has its share of security concerns [1][2]. The data security is one of the important concerns, which holds many users and organizations back from adoption of cloud. There are many organizations that are still reluctant to move towards cloud without appropriate assurance about security of their data. The data stored in cloud is outside the physical control of organization or users and it is managed by the third party cloud service providers. Therefore there is always fear of loss of control over the data and this becomes another important barrier for acceptance of cloud. Hence it is necessary to address the security concerns, before storing sensitive data in shared cloud environment.

The user should have necessary credentials to access data in cloud environment. In traditional system, the access permissions are managed by the service providers through some previously defined access control policies. The traditional solutions are not capable to handle the increasing security threats [3]. As cloud infrastructure is completely different than the traditional, the existing security solutions cannot be applied in same form they exist today. Thus, development of more powerful access control mechanisms for data stored in cloud has become important research topic. Due to the multi-tenancy characteristic of cloud, many users do not trust the cloud environment completely. In order to build trust among users and cloud environment, it is necessary to integrate trust mechanics in traditional access control model. Before providing access to any data, along with the necessary credentials, the trustworthiness of user must be evaluated.

In this paper, the trust evaluation framework for cloud user is presented. The trustworthiness of cloud user is evaluated on basis of user's reputation and the recommendations. The reputation is generated on basis of user's behavior. The recommendation for user is submitted by the other users in the system and the cloud service provider. The behavior data and recommendation data is collected by continuous performance

## **Detection of Face Spoofing using Multiple Texture Descriptors**

Publisher: IEEE Cite This DPF

#### Shanmukhappa A. Angadi ; Vishwanath C. Kagawade All Authors

<b>U</b>	<b>U S</b>		
Ŭ	<b>W N</b>	••••	

Abstract	Abstract:
-	Automatic face recognition systems are being increasingly developed and deployed in various
Document Sections	applications in both civilian and defense sectors. But the emergence of techniques for
I. Introduction	producing fake face images presents a new challenge and demonstrates high security risks
	on such systems thus necessitating imposter detection system. Among the numerous
II. Literature Review	imposter detection methods reported in literature, LBP is often used as one of the best local
	pattern descriptor, as local descriptors are frequently used for imposter detection but the
Approach	traditional LBP descriptor has the limitation of having small spatial support area and being
Αμρισαστι	and the test of the standard test of

← → C ☆ 🌢 ieeexplore.ieee.org/abstract/document/9001636

# Face and Iris Wavelet Feature Fusion through Canonical Correlation Analysis for Person Identification

Publisher: IEEE

Cite This DDF

Shanmukhappa A. Angadi ; Vishwanath C. Kagawade All Authors

	53				
er	Full	0	<	C	
ations	Text Views	\$			

Abstract	Abstract:
	The paper presents, a new technique for person identification using feature level fusion of
Document Sections	face and iris biometric features. The fusion technique employed in the proposed work uses
I. Introduction	wavelet features of face and iris modalities. Canonical Correlation Analysis (CCA) technique is
	used to fuse face and iris wavelet features. CCA is a technique for extracting linearly
II. Related Works	correlated face or iris features from set of features of face or iris images. The technique is able
III Methodology	to extract and enhance the discriminative power for high dimensional feature space for person
m. Methodology	identification from multi-feature information of face and iris. The experimental results on both
IV. Experimental	synthetic and genuine multimodal data sets of face and iris features using SVM classifier
III. Methodology IV. Experimental	correlated face or iris features from set of features of face or iris images. The technique is able to extract and enhance the discriminative power for high dimensional feature space for person identification from multi-feature information of face and iris. The experimental results on both synthetic and genuine multimodal data sets of face and iris features using SVM classifier

Title of the Journal



### Contents Page

1153	Effect of Carbon Black on the Performance of Nitrile Rubber For Analyzing Free Layered Surface Damping Treatment	Prasanth Kumar Mallipudi and N.Ramanaiah	104
1156	Prediction of Temperature Distribution and Displacement of Carbon Steel Plates by FEM	Hitesh Arora, Rupinder Singh, Gurinder Singh Brar	105
1159	Comparative Study of Macroscopic Rupture Criteria	Appurva Jain, Abhishek Mishra	105
1209	Analysis of Steady State Creep Behaviour in Spherical Vessels Made of Composite Material	Sukhjinder Singh Sandhu, Tejeet Singh and V.K. Gupta	106
1215	Design and Analysis of Disk Brake under Tribological Behaviour of Materials	Balaji R, Mukesh Nadarajan,Ashish Selokar, Sreenu S Kumar ,Sivakumar S	106
1225	Finite Element Based Investigation Of Aluminum Annular For Static And Dynamic Behaviour	Anand C. Mattikalli ,Dr. Rajashekar V. Kurahatti	107
1229	A state of art review on the modeling of Contact type Nonlinearities by Extended Finite Element method	Aazim Shafi Lone , Showkat Ahmad Kantha, Azher Jameel, G. A. Harmain	107
1235	Elasto Plastic Crack Growth by XFEM: A Review	Showkat Ahmad Kanth , Aazim Shafi Lone, G. A. Harmain, Azher Jameel	108

9aInternational Conference on Materials Processing and Characterization (ICMPC-2019)

### Finite Element Based Investigation Of Aluminum Annular For Static And Dynamic Behaviour

Anand C. Mattikalli<sup>a</sup>, Dr. Rajashekar V. Kurahatti<sup>b</sup> <sup>a</sup>Maratha Mandal Engineering College, Belagavi-590005,India <sup>b</sup>Basaveshwar Engineering College, Bagalkot-587102,India

#### Abstract

In the field of machine design static and dynamic analysis plays a vital role. The state of resonance is catastrophic to structural elements, which makes the study crucial; we are endeavoring to shield the natural and working frequencies from coincidence. In the present study endeavor is made to do static investigation of annular plate exposed to a uniform loading and analysis of annular plate for free vibration. For the analysis internal boundary of annular plate is constrained & external boundary is let free. For the analysis of stress, maximum transverse deflection and vibration of annular plate finite element analysis is used and effectively executed. The point of study is to discuss the impact of aspect ratio and thickness on the stresses, deflection and frequency.

Keywords: Annular plate; Free vibration Analysis; Finite element method, Aspect Ratio.

### A state of art review on the modeling of Contact type Nonlinearities by Extended Finite Element method

Aazim Shafi Lone<sup>a</sup>, Showkat Ahmad Kanth<sup>a</sup>, Azher Jameel<sup>b</sup>, G. A. Harmain<sup>c</sup>

# ACADEMIC YEAR 2017-18
# **IISST 17 PROCEEDINGS**

### First International Conference on Ideation and Innovations in Sustainable Sciences and Technologies, (IISST 17)

FCSIT (Faculty of CSc and IT), University of Malaya, Kuala Lumpur, MALAYSIA



# **Concepts on Pancha Bhootas**

### Editors

Prof. Pandian Vasant Prof. Junzo Watada Prof. Vish Kallimani Prof. Sujan Chowdhury



http://greenecosociety.weebly.com e

Special Thanks to:

2017







IISST 17 - Malaysia

email:: greenecomax@gmail.com

First International Conference on Ideation and Innovations in Sustainable Sciences and Technologies, (IISST 17), Kuala Lumpur, Malaysia 30 June-2 July 2017

# 4-CLUSTER 2: Green Technology

### Paper Session 3:

Sessio	on Chair: Dr. Moacyr Laruccia, Dr Shivakumar Palaid	ahnakote
		14:30-15:30
NO	Title Papers	Authors
117	Visual Patterns of Rain fall data using SOM for a	Sandeep Maganti, Vish Kallimani, Low Tan jung,
	better water management	Said Jadid, University Technology Petronas,
		Malaysia
118	E-learning: Fishbone Visualizations in Software	Vish Kallimani, UTP, Jyoti Tigadi, MIU, Malaysia
	Engineering Content management	
119	"Maximally Stable Extremal Regions based Method	Mojeed Olatunde, Palaiahnakote Shivakumara,
	for License Plate Recognition	Hamid A. Jalab, , Vish Kallimani, Malaysia
	Key note Speech 3: Performance Enhancement of Tri-	Prof Dr Sivasankaran Harish, International
	generation and Energy Storage Systems Using	Institute for Carbon-Neutral Energy Research
	Innovative Materials	(WPI-I2CNER), Kyushu University, Japan

### Paper Session 4:

Session Chair: Dr Sivasankaran Harish, Dr Sujan Chowdhu		iry	15:45 - 17:30	
NO	Title Papers	Authors		
120	Smart street light for energy saving	Huin Achu, UTP, Malaysia, Gautam, India		
121	Automated EEG Feature Extraction using EMD and	Srushti Bekal, India		
	Classification using ANN to detect whether a person is			
	seizure or a non-seizure patient			
122	Study on Awareness level and Attitude of regional	Nagaraj Navalgund, R. Srivats R, K.L.E		
	SME's towards Green Marketing Practices	Technological University, CMR University, India		
123	A Review of Applications of Nature Inspired Algorithms	Parminder Kaur, Khalsa College for Wome		
	in Building Energy Efficient Systems	Ludhiana, India		
124	Review of Data Integrity In Cloud Storage	Suresh Talwar, S. V. Saboji , C. B. Akki, BEC		
		Bagalkot	, Dayananda Sagar University,	
		Bengalur	u INDIA	
125	A Literature Review on Brain tumor Detection Using	Raviraj (	Chougala, Raviraj Havaldar, AITM, India	
	MRI images			
126	A Review on object detection in Image Processing	SriramK	V, Raviraj Havaldar, AITM, India	
127	A review on Image dehazing Techniques	Vishalkir	thi Patil, RavirajHavaldar, AITM, India	
128	Experimental study of enhancing the performance of	Malagou	da Patil, Dr S.S.Alur, AITM, India	
	photovoltaic panel by water cooling Technique			
129	Studies on Ballistic Impact of Ceramic Coated	Chennap	a Korishetti, Deepak Huddar,	
	Composite for Armour Protection	Rajashek	argouda Angadi, AITM, India	

### **Review of Data Integrity In Cloud Storage**

<sup>[1]</sup>Suresh Talwar, <sup>[1]</sup>S. V. Saboji , <sup>[2]</sup>C. B. Akki Department of Computer Science and Engineering <sup>[1]</sup>Basaveshwar Engineering College, Bagalkot, INDIA <sup>[2]</sup>Dayananda Sagar University, Bengaluru INDIA Email: akmasuresh@gmail.com, saboji\_skumar@yahoo.com , akki.channappa@gmail.com

Abstract— In the past few years' cloud has become the powerful service provider. It provides a way to move the applications and sensitive databases to the large data centres. However there is no guarantee that the data stored in the cloud are secure or not and the contents are original or altered by any unauthorised. Many schemes have been proposed to deal with security issues in an un-trusted server. Integrity of user data in the cloud servers is one of the most important concerns of users nowadays. In this paper we presented few important methodologies and protocols, which the users can use to check the correctness of their data with the simplest possible way and less overhead at the customer side. At the end comparative analysis of data integrity scheme make researcher to take further study in data integrity in cloud.

Index Terms- cloud storage, integrity, Third Party Auditor (TPA).

#### I. INTRODUCTION

Cloud computing is growing technology; provide different services like as software service. These services are provided through Internet. Some people buy hard disk and external device USB. Some people prefer to store data on Cloud storage. It allows storing and managing the data remotely, which will easily scale up and down. Users do not have to buy the expensive hardware and manage the data. Users have to pay only for what they use. Nature of the cloud allows accessing the data from anywhere. These makes cloud storage very popular. As cloud is distributed once user uploads the data user has no control over the data. Data are remotely stored. Users do not have physical access to the data. So user is not aware of any security threats at storage. Cloud storage is visualized pools where data and applications are stored who desires to store their data in the cloud. Some of the cloud storage benefits are reduced costs; provide more flexibility, which reduce management of web applications through automated updates, and storage capacity.

Data integrity is defined as the accuracy and consistency of stored data, in absence of any alteration to the data between two updates of a file or records. Cloud services should ensure data integrity and provide trust to the user privacy. Outsourcing data into the cloud is economically attractive for the cost and complexity. It's offering strong assurance of data integrity. Data Integrity is very important among the cloud storage issues. After moving the data to the cloud, owner hopes that their data and applications are in secured manner. But some times the owner's data may be altered or deleted. In that scenario, it is important to verify if one's data has been deleted. To validate data, often user must download the data. If the outsourced data is very large files such downloading to determine data integrity may become prohibitive in terms of increased cost of bandwidth and time, especially if frequent data checks are necessary. Data integrity provides the proofs that data is stored at a remote storage in the cloud is not modified by anyone and there by integrity of the data is assured. Some of the best examples for cloud storage are Amazon S3, Windows Azure Storage, EMC Atmos, Files Anywhere, Google Cloud Storage, Google App Engine Blobstore, and iCloud by Apple.

The remainder of the paper is organized as follows: in section two review of literature survey is presented. Section three describes system models of data integrity. Section four compares the data integrity schemes. We concluded the paper in section five.

#### II. RELATED WORK

There exist different security problems in cloud computing. Paper [1] proposes the novel security concept of Identity-based Proxy-Oriented Data Uploading and Remote Data Integrity Checking in Public Cloud (ID-PUIC). It formalizes ID PUIC's system and security model. Then, the first concrete ID-PUIC protocol is designed by using the bilinear pairing technique. The concrete ID-PUIC protocol is provably secure and efficient by using the formal security proof and efficiency analysis. Remote data integrity checking is an important security problem. On the other hand, the proposed ID-PUIC protocol can also realize private remote data integrity checking, delegated remote data integrity checking and public remote data integrity checking and public remote data integrity checking based on the original client' authorization. Proof of irretrievability (POR) scheme is proposed by Shacham. POR is a stronger model, which makes the checker, not only check the remote data integrity but also retrieve the remote data. Many POR schemes have been proposed.

Paper [2] proposed to ensure data integrity. A TPA can verify data storage correctness on behalf of the client with use of a public key. This key generated during the setup process. This scheme based on homomorphic linear authentication (HLA). The TPA monitors the stored data in the remote server and notifies the client regarding data security. New approach introduce to data integrity. MTPAs are employed to overcome the limitations of a STPA. This scheme applies the concept of POR for large files to archive encrypted data in cloud storage. Remote data-storage-correctness checking scheme based on HLA and ECDSA to support public verifiability. We suggest a remote data storage correctness-checking scheme based on homomorphic linear authentication and an elliptic curve digital signature algorithm to support public verifiability.

Paper [3] investigates data files on un-structed storage, which provide RSA based homomorphic linear authenticator for giving randomly sampling a few block of the file. Thus it may leak user data information to external parties because of their protocol, which is not secure. The Proof of Retrievability Model in which spot-checking and error correcting code are used. A skip list based mechanism which then makes the use of provable data as client request the cloud server to provide service in which cloud server initially authenticate the client and then provide a virtual machine by means of Software as a service. In Virtual Machine RSA is used for secure communication between user and cloud server. SHA-512 algorithm is used for data integrity but problem over here is that performed operation as static configuration, information leaks to the external parties, insecure transmission of data, system requires user data information to the third party which violates the privacy preserving guarantee, single sign on capability to access the cloud data storage.

Paper [4] addresses the challenges and proposes an efficient public integrity-checking scheme for cloud data sharing that supports multiple writers. This prove the security of our design based on the Computational Diffie-Hellman (CDH) problem, the Bilinear Diffie-Hellman (BDH) problem and the t-Strong Diffie-Hellman (t-SDH) problem. This scheme also supports public checking and efficient user revocation and is provably secure. Numerical analysis and extensive experimental results show the efficiency and scalability of proposed scheme. It achieves this through innovative design on polynomial-based authentication tags, which allows aggregation of tags of different data blocks. For system scalability, it is empower the cloud with the ability to aggregate authentication tags from multiple writers into one when sending the integrity proof information to the verifier.

Paper [5] will give a self-certified public auditing scheme for data integrity. Then evaluate its performance in term of basic cryptographic operations and provides intuition behind the construction. Public auditing scheme for data integrity, after data owner outsourced data to the cloud server, it deletes the local copy of these data. A verifier can periodically check the data integrity, the verifier is the auditor in Cloud storage is more and more attractive because it can realize the sharing and storing of data files among companies and corporations. However, for the data owner, the most concern is the integrity of data file. Fortunately, Proofs of Retrievability technique enables the data owners verify the integrity of their outsourced data on the cloud server. Most of the existing POR schemes still exist efficiency.

Paper [6] investigated data privacy issues in remote data integrity-checking protocols. The existing privacy-preserving remote data integrity-checking protocol not achieves the desired goal of leaking and no information to a third party. It formalized zero-knowledge privacy and proposed an improved the protocol to achieve the performance analysis and the implementation, it is observed that all existing protocols with public verifiability supporting data update, proposal require the data owner to







Geetha Shishu Shikshana Sangha (R)

### **GSSS** Institute of Engineering and Technology For Women

(Affiliated to VTU Belagavi, Approved by AICTE-New Delhi & Govt. of Karnatak MYSURU- 570 016 | KARNATAKA | INDIA http://geethashishu.in/, http://iceeccot.geethashishu.in/

> Accredited Branches by NBA, New Delhi UG - ECE, CSE, ISE, TE & IT (Validity: 01.07.2017 - 30.06.2020)

International Conference on Electrical, Electronics, Communication Computer Technologies and Optimization Techniques ICEECCOT - 2017



### 15-16 December, 2017

Publication Chairs: Dr. Parameshachari B D | Dr. Reshma Banu

SL I	No	Page No
157	Modified Topology-Hiding Multipath Routing (M-THMR) Protocol: Solution to the Topology Exposure Problem in MANET Akshay Suhas Phalke and Jyoti Lagad	157
158	Review on Anomaly based Network Intrusion Detection System Rafath Samrin and D Vasumathi	158
159	Low-energy Adaptive Clustering Hierarchy Protocol and Optimal Number of Cluster Head Algorithm in a Randomized Wireless Sensor Network Deployment <i>Gino Alvarado, Carlos bosques, Fernando Palacios and Luis Córdoba</i>	159
160	Crop Production Rate Estimation Using Parallel Layer Regression with Deep Belief Network Pushpa Mohan and Dr. Kiran Kumari Patil	160
161	Application of MQTT Protocol for Real time Weather Monitoring and Precision Farming Nagesh Uluvar, Uday D.V, Shamitha Gurunath Talekar and Pooja S	161
162	An Efficient Algorithm for Intra Prediction in HEVC/H.265 Standard for 16 x 16 Pixels Swamy T N, Ramesha K and Diwakar K R	162
163	Secured Data transmission using Knight and LSB Technique Laxmi Gulappagol and Dr. K B Shivakumar	163
164	Routing in Opportunistic Networks : Taxonomy, Survey Shobha Bharamagoudar and Shivakumar Saboji	164
165	An Energy-Aware Node Disjoint Multipath Routing Protocol for MANETs with DynamicTransmission Range Adjustment Benakappa S M and Dr. Kiran M	165
166	Routing Protocol for Life Time Enhancement in MANET Shridhar Kabbur, Dr. Ali Ahammed G. F and Dr. Reshma Banu	166
167	Assortment of MANET routing protocols and its parameters for different environments Mallikarjuna A and Dr. Venkanagouda C Patil	167
168	A Comprehensive Analysis of MAC Protocols for MANET Rajeev Kumar	168
169	Experimenting with Scalability of Floodlight Controller in Software Defined Networks	1/0
	Saleh Asadollahi and Bhargavi Goswami	169

# Routing in Opportunistic Networks: Taxonomy, Survey

S R Bharamagoudar Dept. of Electronics & Communication Engineering Basaveshwar Engineering College (Autonomous) Bagalkot, India srb bec@yahoo.com

Abstract- The accelerated growth of wireless communication such as personal digital assistants, tablets, smart phones, that are carried by the people may arrange themselves in to a special network for the communication of messages. The moving nodes communicate with the nearby nodes using node contact opportunity through Wi-Fi or blue tooth. These nodes suffer from frequent disconnections. The network with frequent disconnections is known as opportunistic network or intermittently connected network. It is a unique wireless network architecture which enables the mobile nodes to have communication with each other in environment where there is no continuous route exists between source and destination .Such networks espouse store-carry-forward habit for the message transfer and utilize mobility of moving nodes for message communication. These networks are featured with network topology which is unstable and also long latency. Routing requires the nearby neighborhood nodes information for the message transfer. Depending on the routing action, we categorize routing schemes as social aware information schemes and social zero information schemes. The schemes such as social zero information do not require any past information of the nodes behavior. Subject to the content of information, social aware schemes are furthermore categorized as microscopic and macroscopic schemes. Microscopic schemes utilize mobile user's movement. Whereas macroscopic schemes have integrated social relationship with the mobile user's movement. We explicate about the routing solutions for the microscopic and macroscopic schemes. Additionally we briefly discussed evaluations in routing schemes.

Keywords— Opportunistic Networks, Intermittently connected network, Store-carry-forward.

#### I. INTRODUCTION

In the present networking world, wireless communication plays a predominant role. The accelerated growth of mobile devices like laptops, smart phones, notebooks etc makeup mobile communication area a more fascinating research in recent years. Another extension of the mobile has inclined to the Mobile Ad-hoc Network (MANET) model known as Opportunistic Networks (ONs). These networks make use of nodes mobility characteristic, to enter the communication range of each other. This helps to establish communication path between the nodes to transfer the information. ONs suffer from unstable network topology as well as prolonged latency because of frequent disconnections [1]. Such networks suffer S V Saboji Department of Computer Science & Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587103, India saboji\_skumar@yahoo.com

from abruptly intermittent connections because of moving nodes. These networks are also entitled by intermittently connected networks. Nodes suffer from the initial knowledge of the structure of the network because of random movement of the nodes. So the routes are created dynamically and relay nodes or intermediate nodes take active part in the transferring of messages to respective destinations. If the intermediate nodes cannot find any forwarding opportunity within its communication range then, the nodes store the messages and carry the messages until they find next upcoming contacts to forward the messages [2]. Using Wi-Fi, Bluetooth, the built in technologies, nodes communicate with each other. The network is accustomed by mobile devices which are carried by human beings and hence routing is a challenging task. The major concern in ONs is reliable transfer of message to the destination. Hence co-operation between the nodes and storage of messages are the primary requirement [3].



Fig.1 Message forwarding in opportunistic networks

NCA2CS - 2017 - National Conference on Advanced Computing and Communication Systems-New Horizon College, Depar...



Notes/All Educational News. Subscribe for Latest Updates Click here or Enter your email id below

Click here to get Advertisement P

student technical symposium for free of

NCA2CS - 2017 - National Conference on Advanced Computing and Com Systems-New Horizon College, Department of Computer Science-Bengal 043.Karnataka, India-Feb 23,2017.

### NCA2CS – 2017 - National Conference on Advanced Computing and Communication Systems College, Department of Computer Science-Bengaluru - 560 043. Karnataka, India-Feb 23, 2017

Subscribe

1. Name of the Conference/Workshop/Training Program: NCA2CS - 2017 - National Conference on Advanced Computing and Communication Systems

#### 2. Objective of the Event

The aim of the conference is to provide an intellectual forum for exchange of knowledge and information relating to latest developments in the field of computing We are sure that this National Conference will prove to be a valuable learning opportunity for all the participants.

3. College Details (Name of the College, Address, Dept. etc..)

Enter your Email ID: >

New Horizon College

Department of Computer Science

3 rd 'A' Cross, 2 nd 'A' Main.

Kasturinagar,

Bengaluru - 560 043.

Karnataka, India

4. Call for Papers (Paper Submission Email id. Paper Themes, etc...)

Call for papersTopics of interest for submission include, but are not limited to: Data Warehousing and Big DataData Mining and Knowledge MiningGrid, Cloud a source TechnologySoftware EngineeringArtificial Intelligence & Neural NetworksWireless, Sensor Networks & SecurityMobile communication and computingIn Pervasive computing

5. Important Dates

Dates to RememberFull Manuscript Submission Jan 21st, 2017Intimation of

Acceptance Jan 31st, 2017Registration Deadline Feb 07th, 2017

6. Who Can attend? (UG/PG/Research Scholars)

The First National Conference on Advanced Computing and Communication Systems (NCA2CS - 2017) aims to provide a platform for researchers, academiciar postgraduate and under graduate students to present their research ideas in the area of computing and communication technology.

We are awaiting all the delegates to be present in the conference for impeccable

exchange of knowledge with fusion of ideas and thoughts throughout this day.

7. Event Registration Details.(Payment Details)

Delegates Amount (in Rs)UG / PG Students 400/-

Research Scholar 500/-Faculty 600/- Industry

Personnel 1000/-

8. Address for communication

S.Thangadurai, Asst. Professor,

# Pose Invariant Face Recognition System – A Review

Madhura Shettar PG Student of Computer Science and Engineering Basaveshwar Engineering College Bagalkot Prof S.M.Hatture Professor of Computer Science and Engineering Basaveshwar Engineering College Bagalkot

### Abstract

Face recognition is an important computer vision and pattern recognition technology that is used in various applications like organizing photo albums to surveillance. Face recognition in pose variations is one of the challenges of automatic face recognition, as intra variation caused by pose difference sometimes is better than that caused by distinctiveness difference. However, facial feature changes with pose can be regarded as a continuous process, meaning that a person's face image changes from one pose to another pose gradually. Most of the developed techniques of face recognition are only have capacity of recognizing the frontal views of faces and assuming that the person was looking straight into the camera. The frontal face recognition method is right for certain applications where the client poses constantly the same way from session to session. However, a user might not pose to a camera for the intention of being recognized, definitely not even knowing that a face image is being captured. In these cases it is essential for the system to handle faces with in plane and in depth rotations. Pose invariant face recognition has many real-world applications, specially in creating a more robust recognition system for commercial and government applications.

### Keywords

Face recognition, Pose-invariant, Classifier, Datasets

### I. INTRODUCTION

Face recognition (FR) is a process in which face of individual is identified by a system. However, identifying faces via as digital eye is not an easy nut to crack. Whenever face recognition is used transversely in the surveillance system it is regularly very difficult to get the faces in restricted environment. Hence the system has to be capable of recognizing the captured faces even in poor lightning conditions and variations in poses as beside the faces taken in restricted environment. Moreover, all the techniques are very much affected by variations and their routine get degraded when pose variations are present. The face

Bangalore

New Horizon College – Kasturinagar -

NCA2CS - 2017 - National Conference on Advanced Computing and Communication Systems-New Horizon College, Depar...



Notes/All Educational News. Subscribe for Latest Updates Click here or Enter your email id below

Click here to get Advertisement P

student technical symposium for free of

NCA2CS - 2017 - National Conference on Advanced Computing and Com Systems-New Horizon College, Department of Computer Science-Bengal 043.Karnataka, India-Feb 23,2017.

### NCA2CS – 2017 - National Conference on Advanced Computing and Communication Systems College, Department of Computer Science-Bengaluru - 560 043. Karnataka, India-Feb 23, 2017

Subscribe

1. Name of the Conference/Workshop/Training Program: NCA2CS - 2017 - National Conference on Advanced Computing and Communication Systems

#### 2. Objective of the Event

The aim of the conference is to provide an intellectual forum for exchange of knowledge and information relating to latest developments in the field of computing We are sure that this National Conference will prove to be a valuable learning opportunity for all the participants.

3. College Details (Name of the College, Address, Dept. etc..)

Enter your Email ID: >

New Horizon College

Department of Computer Science

3 rd 'A' Cross, 2 nd 'A' Main.

Kasturinagar,

Bengaluru - 560 043.

Karnataka, India

4. Call for Papers (Paper Submission Email id. Paper Themes, etc...)

Call for papersTopics of interest for submission include, but are not limited to: Data Warehousing and Big DataData Mining and Knowledge MiningGrid, Cloud a source TechnologySoftware EngineeringArtificial Intelligence & Neural NetworksWireless, Sensor Networks & SecurityMobile communication and computingIn Pervasive computing

5. Important Dates

Dates to RememberFull Manuscript Submission Jan 21st, 2017Intimation of

Acceptance Jan 31st, 2017Registration Deadline Feb 07th, 2017

6. Who Can attend? (UG/PG/Research Scholars)

The First National Conference on Advanced Computing and Communication Systems (NCA2CS - 2017) aims to provide a platform for researchers, academiciar postgraduate and under graduate students to present their research ideas in the area of computing and communication technology.

We are awaiting all the delegates to be present in the conference for impeccable

exchange of knowledge with fusion of ideas and thoughts throughout this day.

7. Event Registration Details.(Payment Details)

Delegates Amount (in Rs)UG / PG Students 400/-

Research Scholar 500/-Faculty 600/- Industry

Personnel 1000/-

8. Address for communication

S.Thangadurai, Asst. Professor,

## Performance Evaluation of Fusion Techniques in Biometric Systems-A Review

Poornima Byahatti PG Student of Computer Science and Engineering Basaveshwar Engineering College Bagalkot Prof. S. M. Hatture Professor of Computer Science and Engineering Basaveshwar Engineering College Bagalkot

### Abstract

Biometrics is a discipline of recognizing an individual depends on his / her behavioral or physiological characteristics. Various biometric systems established in real world applications are unimodal which depend on the proof of single source of information for authentication such as face, voice etc. These systems are susceptible to variety of problems. Multiple sources of information for establishing individuality solve some of the limitations of unimodal biometric systems. These systems permit the combination of two or more types of biometric systems known as Multimodal biometric systems. These systems are more reliable due to the presence of multiple, independent biometrics. These systems are capable to meet the rigid performance requirements forced by various applications. Various fusion techniques are available such as Sensor level, score level, feature level, rank level and decision level fusion. It is planned to work on performance evaluation of various fusion techniques by taking into consideration of the multiple biometric traits.

### Keywords

Multimodal biometrics, feature level fusion, match score level fusion, Decision level fusion.

### I. INTRODUCTION

In the present epoch of information technology, there is a need to realize authentication and authorization techniques for security of resources. There are various ways to prove authentication and authorization. But all other techniques are beaten by biometric authentication. Based on the usage of number of traits, the biometric systems are partitioned into two categories: Unimodal biometric systems (use of only single trait) and Multimodal biometric systems (use of two or more traits or algorithms or samples). Unimodal biometric systems encompass many implicit problems in their applications. Unimodal systems have the lots of limitations like Noisy data, Intra-class variation, Intra-class similarities, Non-

New Horizon College – Kasturinagar - Bangalore

T. N. Nagabhushan V. N. Manjunath Aradhya Prabhudev Jagadeesh Seema Shukla Chayadevi M. L. (Eds.)

Communications in Computer and Information Science

# Cognitive Computing and Information Processing

Third International Conference, CCIP 2017 Bengaluru, India, December 15–16, 2017 Revised Selected Papers



Х

801

Contents	XXI

Hybrid Optimization in Big Data: Error Detection and Data Repairingby Big Data Cleaning Using CSO-GSAK. V. Rama Satish and N. P. Kavya	258
<ul> <li>Application of Optimization Technique for Performance and Emission</li> <li>Characteristics of a CNG-Diesel Dual Fuel Engine: A Comparison Study</li> <li>A. Adarsh Rai, B. R. Shrinivasa Rao, Narasimha K. Bailkeri, and P. Srinivasa Pai</li> </ul>	274
Use of Hybrid Algorithm for Surface Roughness Optimization in Ti-6Al-4V Machining <i>Grynal D'Mello, P. Srinivasa Pai, and Adarsh Rai</i>	285
Eigenvalue Analysis with Hough Transform for Shape Representation and Classification Bharathi Pilar and B. H. Shekar	299
Event Data Analysis in Large Virtualized Environment M. B. Bharath and D. V. Ashoka	313
Noisy Speech Recognition Using Kernel Fuzzy C Means H. Y. Vani and M. A. Anusuya	324
An Enhanced Water Pipeline Monitoring System in Remote Areas Using Flow Rate and Vibration Sensors Praveen M. Dhulavvagol, K. R. Ankita, G. Sohan, and Renuka Ganiger	331
Smart Helmet with Cloud GPS GSM Technology for Accident and Alcohol Detection	346
Text-Dependent Speaker Recognition System Using Symbolic Modelling	
of Voiceprint	358
An IOT Based Smart Shopping Cart for Smart Shopping Srinidhi Karjol, Anusha K. Holla, C. B. Abhilash, P. V. Amrutha, and Y. V. Manohar	373
Pathnet: A Neuronal Model for Robotic Motion Planning V. M. Aparanji, Uday V. Wali, and R. Aparna	386
Impact of Named Entity Recognition on Kannada Documents Classification R. Jayashree, Basavaraj S. Anami, and S. Teju	395
Big Data Analysis - An Approach to Improve Power System Data Analysis	402
Sandhya S. Shankarlinga, K. T. Veeramanju, and R. Nagaraja	403



### Text-Dependent Speaker Recognition System Using Symbolic Modelling of Voiceprint

Shanmukhappa A. Angadi<sup>1</sup> and Sanjeevakumar M. Hatture<sup>2((\Box)</sup>)

 <sup>1</sup> Visvesvaraya Technological University, Belagavi 590018, Karnataka State, India vinay\_angadi@yahoo.com
 <sup>2</sup> Basaveshwar Engineering College, Bagalkot 587103, Karnataka State, India smhatture@yahoo.com

Abstract. Speaker recognition system automatically recognize/identify a speaker by their combined behavioral and physiological characteristics. A symbolic inference system for text-dependent speaker recognition system by exploring the physiological characteristics embedded in the user utterance is presented in this paper. The inter-lexical pause position, complementary spectral features such as spectral centroid, spectral entropy and spectral flatness, loudness, pitch and formants features are extracted from the voiceprint and symbolic data object is constructed. These features are explored in this work as interlexical pause position provides the articulation capability of user vocal tract. The functional properties of the human ear is modelled with spectral characteristics and loudness feature provides the strength of ear's perception. The relation between physical and perceptual properties of sound is estimated through pitch whereas formants provide the acoustic reverberation of the human vocal tract. The variability in features of user/speaker utterance of words is represented with symbolic data. The speaker identification is performed using modified span, content and position symbolic similarity measures [3], modified for the current work. The proposed method is evaluated on 100 users of voice corpus of VTU-BEC-DB multimodal biometric database and achieves an overall identification rate of 90.56%.

**Keywords:** Speaker identification · Symbolic object · Voice biometric Complementary spectral features · Symbolic similarity measure

### 1 Introduction

Speaker recognition system identify an individual by analyzing the human voice. Voice is a biometric trait which exhibits combined behavioral and physiological characteristics used to alleviate the problem of spoof attack in biometric system. Voice of an user is unique as the knowledge/codes used in the utterances are user specific. The physical differences in users' voiceprints are characterized by measuring the amplitude, frequency, duration and spectral distribution. The information embedded in the voice signal is extracted at six different levels such as spectral, prosodic, phonetic, idiolectal (i.e. syntactical), dialogic and semantic [1]. The usage of the text in the testing phase of the speaker recognition system categorize it as text-dependent and text-independent

<sup>©</sup> Springer Nature Singapore Pte Ltd. 2018

T. N. Nagabhushan et al. (Eds.): CCIP 2017, CCIS 801, pp. 358–372, 2018. https://doi.org/10.1007/978-981-10-9059-2\_32

Lecture Notes on Data Engineering and Communications Technologies 9

Nabendu Chaki Agostino Cortesi Nagaraju Devarakonda *Editors* 

Proceedings of International Conference on Computational Intelligence and Data Engineering





# About this book

The book presents high quality research work in cutting edge technologies and most-happening areas of computational intelligence and data engineering. It contains selected papers presented at International Conference on Computational Intelligence and Data Engineering (ICCIDE 2017). The conference was conceived as a forum for presenting and exchanging

### **Opinion Content Extraction from Web Pages Using Embedded Semantic Term Tree Kernels**

Veerappa B. Pagi and Ramesh S. Wadawadagi

Abstract Rapid proliferation of user-generated content (UGC) published over the Web in the form of natural language has made the task of automatic Information Extraction (IE) a challenging issue. Despite numerous models proposed in the literature to address Web IE issues, still there is a growing demand for researchers to develop novel techniques to cope up with new challenges. In this paper, an approach to extract opinion content from Web pages using Embedded Semantic Term Tree Kernels (ESTTK) is addressed. In traditional tree kernels, the similarity of any two given production rules is determined based on exact string comparison between the peer nodes in the rules. However, semantically identical tree fragments are forbidden, even they can contribute to the similarity of two trees. A mechanism needs to be addressed, which accounts for the similarity of nodes with different vocabulary and phrases holding knowledge that are relatively analogous. Hence, the primitive tree kernel function is reconstructed to obtain the similarity of nodes by searching keywords in opinion lexicon embedded as vectors. Experimental results reveal that ESTTK results in better prediction performance compared to the conventional tree kernels.

**Keywords** Opinion extraction • DOM tree • Tree kernels • Classification Sentiment analysis

Department of Computer Science and Engineering, Basaveshwar Engineering College, Bagalkot, India e-mail: rswlib@yahoo.co.in

V.B. Pagi e-mail: veereshpagi@gmail.com

© Springer Nature Singapore Pte Ltd. 2018

345

V.B. Pagi · R.S. Wadawadagi (🖂)

N. Chaki et al. (eds.), *Proceedings of International Conference on Computational Intelligence and Data Engineering*, Lecture Notes on Data Engineering and Communications Technologies 9, https://doi.org/10.1007/978-981-10-6319-0\_29

		e
	2017 International Conference On Smart Technologies For Smart Nation (SmartTechCon)	
	REVA University 17–19 August 2017, REVA University, Bangalore, India SmartTech	Con
R	This is to certify that PRAVEEN S. CHALLAGIDAD	
A A	from BASAVESHWAR ENGINEERING COLLEGE, INDIA	has
3)		
$\gtrsim$	participated and presented a paper titlea	
	participated and presented a paper titlea	
	in International Conference on Smart Technologies for Smart Nation (SmartTechCon 2017) held	at
	in International Conference on Smart Technologies for Smart Nation (SmartTechCon 2017) held REVA University, Bangalore, Karnataka, India during 17–19 August 2017.	at
	participated and presented a paper titlea <u>Trust MANAGEMENT IN CLOUD COMPUTING</u> in International Conference on Smart Technologies for Smart Nation (SmartTechCon 2017) held REVA University, Bangalore, Karnataka, India during 17–19 August 2017. S.L. Nigen	at
	participated and presented a paper titlea <u>Tevst Management In Clove Computing</u> in International Conference on Smart Technologies for Smart Nation (SmartTechCon 2017) held REVA University, Bangalore, Karnataka, India during 17–19 August 2017. <u>Dr. Mallikarjun M. Kodabagi</u> <u>Organizing Chair</u> <u>Dr. S. K. Niranjan</u> <u>TPC Chair</u> <u>TPC Chair</u>	at wi

# 2017 International Conference on Energy, Communication, Data Analytics and Soft Computing (ICECDS 2017)

Chennai, India 1-2 August 2017

Pages 1-745



IEEE Catalog Number: ISBN: CFP17M55-POD 978-1-5386-1888-2

# Copyright © 2017 by the Institute of Electrical and Electronics Engineers, Inc. All Rights Reserved

*Copyright and Reprint Permissions*: Abstracting is permitted with credit to the source. Libraries are permitted to photocopy beyond the limit of U.S. copyright law for private use of patrons those articles in this volume that carry a code at the bottom of the first page, provided the per-copy fee indicated in the code is paid through Copyright Clearance Center, 222 Rosewood Drive, Danvers, MA 01923.

For other copying, reprint or republication permission, write to IEEE Copyrights Manager, IEEE Service Center, 445 Hoes Lane, Piscataway, NJ 08854. All rights reserved.

### \*\*\* This is a print representation of what appears in the IEEE Digital Library. Some format issues inherent in the e-media version may also appear in this print version.

IEEE Catalog Number:	CFP17M55-POD
ISBN (Print-On-Demand):	978-1-5386-1888-2
ISBN (Online):	978-1-5386-1887-5

### Additional Copies of This Publication Are Available From:

Curran Associates, Inc 57 Morehouse Lane Red Hook, NY 12571 USA Phone: (845) 758-0400 Fax: (845) 758-2633 E-mail: curran@proceedings.com Web: www.proceedings.com



### TABLE OF CONTENTS

A CLOUD-BASED DASHBOARD FOR TIME SERIES ANALYSIS ON HOT TOPICS FROM SOCIAL MEDIA	1
Ehsan Akbari Dahouei	
DEVELOPMENT OF A SYSTEM TO IMPROVE SAFETY IN SWITCHING OF HIGH VOLTAGE CIRCUIT BREAKERS	7
Prashobh Karunakaran ; Mohammad Shahril Osman ; Ian Rino Anak Boniface ; Atif Norzan Shah	
SIMULATION AND IMPLEMENTATION OF MICROCONTROLLER BASED GATE DRIVE CIRCUIT FOR THREE-PHASE MOSFET INVERTER	
Tapan Kumar Chakraborty ; Shahinur Rahman ; Sakhawat Hossen Rakib	
AN APPROACH OF EFFICIENT SECURITY ALGORITHMS FOR DISTRIBUTE	
ARCHITECTURES	
Toapanta T. Moisés ; Mafla G. Enrique ; Orizaga T. Antonio	
APPLICATION OF OPTIMIZATION TECHNIQUE IN SHE CONTROLLED MULTILEVEL	
INVERTER	
Sarika D Patil ; Sumant G Kadwane	
A NEW GENERALIZED CASCADED MULTILEVEL STRUCTURE FOR SINGLE PHASE H BRIDGE INVERTER	
Sreenivasulu Mamilla ; Suresh Kumar Anisetty ; K. Sri Gowri	
GREEN COMPUTING BASED COST OPTIMIZATION IN CLOUD COMPUTING	
Varsharani Londe ; M. M. Math	
A SURVEY ON RELAY SELECTION IN COOPERATIVE DEVICE-TO-DEVICE (D2D)	
COMMUNICATION FOR 5G CELLULAR NETWORKS	
K. Shamganth; Martin. J. N. Sibley	
A NOVEL THRESHOLD-BASED CLUSTERING METHOD TO SOLVE K-MEANS	47
WEAKNESSES	
S. Ehsan Yasrebi Nayini ; Somayeh Geravand ; Ali Maroosi	
PREDICTIVE MODELING FOR IDENTIFYING THE HUMAN BRAIN DEVELOPMENTAL	50
STAGES	
FUTHAN KHUN ; TAPAN GUNANI MADLADLE TADIEE ENIEDON METED MUTILAUTOMATIC DOMED ELOM CONTDOL	50
VARIABLE TARIFF ENERGY METER WITH AUTOMATIC POWER FLOW CONTROL	
SOLID STATE TDANSCODMED CONTDOLLED AND IT'S INTEDEACINC WITH WIND	
FNFRCY CONVERSION RASED DISTRIBUTION SYSTEMS	62
Paresh P Luhana · Apurva N Christian	
PERFORMANCE OPTIMIZED DIGITAL OPSK MODULATOR	68
Mansi Jain : Palbha Kesharwani : Ashwini Kumar Malviva : Kavita Khare : Pratvush Shandilva : Sushmita	
Haldar ; Himanshu Rai ; Supriya Aggarwal	
HYBRID GA-MEANS FOR ENERGY EFFICIENT CLUSTERING IN WIRELESS SENSORS	
NETWORK	72
Rajesh Kumar Yadav ; Mikrant Sharma	
RESEARCH ON SOFTWARE TESTING TECHNIQUES AND SOFTWARE AUTOMATION	
TESTING TOOLS	77
Karuturi Sneha ; Gowda M Malle	
FUZZY LOGIC FOR DECISION EXTRACTION FROM PRODUCT REVIEWS	
Mhd Tahssin Altabbaa ; Tarik Ayabakan ; Arif Selcuk Ogrenci	
COLLABORATION TECHNOLOGIES FOR SECURE ROAD TRAFFIC CONGESTION	
CONTROL SYSTEM IN EGYPT	
Heshem A. El Zouka	
DESIGN OF A VOICE CONTROLLED ROBOTIC GRIPPER ARM USING NEURAL	
NETWORKS	94
Fariha Musharrat Haque ; Asif Shahriyar Sushmit ; M. A. Rashid Sarkar	
MULTI-OBJECTIVE OPTIMIZATION OF A CHP-PV-BATTERY ISLANDED MICROGRID Nnamdi Nwulu	98
LANDAU-GINZBURG-DEVONSHIRE BASED MODELLING AND ITERATIVE LEARNING	
COMPENSATION OF PHASE TRANSITIONING SMART MATERIAL STRUCTURE FOR 1 DOF	
OSCILLATOR APPLICATION	109
C Sailalitha C Sathagh Kuman S Karthigh I A Dalani	

G. Sailalitha ; G. Satheesh Kumar ; S. Karthick ; I. A. Palani

AUTOMATIC TRAFFIC CONTROL SYSTEM USING PCA BASED APPROACH	
D OSMA IVANAMI, M. SANAVANAN, J ALDEH MAYAN, MARAIL DEVAKAMINAN KAMADAN, K. MONARA I HASAA	
IEEE 802.13.4 WI-FI MODULE FOR WIRELESS COMMUNICATION AMONG SPECH BROCESSOD AND HE ADDIECE (TD ANSMITTED) OF COCHT EAD IMDIANTS	2202
K Denselv Skingened D Lengen Sensker S D Killemi	
K. Ramesn , Smyanana d Lamani , Sumaya Sanooei , s. d. Kukarni Matyi Ad Cda Adulata i Jisedd Inteedda Celudi to Cytudy dy Modelii e	
MAILAB-GKAPHICAL USEK INTERFACE (GUI) TO STUDY PV MODULE	2200
CHARACTERISTICS	
Amiruaana Manaai ; Adaulian Alom ; Kayes Ali Khan ; Sakkai Pal ; Amiraana De	• • • •
FEATURE SELECTION AND CLASSIFICATION OF EPILEPSY FROM EEG SIGNAL	
S. Priyanka; Dawa Dema; 1. Jayanthi	
EXTRACTION OF DIFFERENT FEATURES OF ECG SIGNAL FOR DETECTION OF CARDIAC	<b>•</b> 40 <b>-</b>
ARRHYTHMIAS BY USING WAVELET TRANSFORMATION DB 6	
S. T. Sanamdikar ; S. T. Hamde ; V. G. Asutkar	
IMPROVED ALGORITHM BASED ON MULTISCALE MORPHOLOGY FOR	
INTENSIFICATION OF GRAY SCALE IMAGES	
Mananpreet Kaur ; J. S. Sohal	
REAL TIME PERFORMANCE EVALUATION OF MODIFIED PMDC MOTOR, BLDC MOTOR	
AND DUAL COMMUTATOR PMDC MACHINE	
Rahat Ullah Khan	
SOLAR POWERED AUTO WATERING SYSTEM FOR IRRIGATION USING EMBEDDED	
CONTROLLER	
Gouthami Eragamreddy ; K. Ramya Sree	
LINE TRAP AND ARTIFICIAL INTELLIGENCE BASED DOUBLE CIRCUIT TRANSMISSION	
LINE FAULT CLASSIFICATION	
Sagar D. Nimkar ; Vikramsingh R. Parihar ; Anagha P. Dhote ; Kishor G. Dhake	
AD ANALYSIS USING MACHINE LEARNING: CLASSIFYING AND RECOMMENDING	
ADVERTISEMENTS FOR A GIVEN CATEGORY OF VIDEOS. USING MACHINE LEARNING	
R Vinit Kaushik : R. Raghu : L Maheshwar Reddy : Ankita Prasad : Sai Prasanna	
DATA ANALYTICS ON PLACEMENT DATA IN A SOUTH ASIAN UNIVERSITY	2438
Anindva Oiha : Upasana Patinaik : Sasi Rekha Sankar	
A NOVEL TRIPLE DES TO ENHANCE E-COVERNANCE SECURITY	2443
I Raia Sekhar Reddy - G Murali	
AN EMPIRICAL STUDY OF APPLICATION LAVER PROTOCOLS FOR IOT	2447
Unendra Tandale · Rashirahamad Momin · Deva P. Seetharam	
OPPHILA TANALE, BANNANA MANNAN, BOTT I SCHWARM	
FEATURE DASED EVALUATION OF HOTEL REVIEWS AND RATINGS FOR DIFFERENTLY	2452
Hungira G. Momin - Chandrabhana S. Kondhawala - Ruhina F. Shaikh - Kiran G. Gawandha	
Initiation of Monore And Society with a new initiate financial of our matter as a	2455
ANALISIS OF MOOFF AND SOUFF WITH AND WITHOUT IFFC USING FSO	
5. Fraveen, 5 K Narenara, Konaadoona sanega, N. vena dada openicu, AND MATH AD DASED CIAD DA DVINC SYSTEM MODUL E FOD SMADT CUTY	
OPENCY AND MAILAB BASED CAR PARKING SYSTEM MODULE FOR SMART CITY	0461
USING CIRCLE HOUGH TRANSFORM	
Janak Frivedi ; Mandalapu Sarada Devi ; Dave Dhara	2465
AN EXPERT SYSTEM FOR FAULT LOCATION IN DISTRIBUTION FEEDER LINE	
Dusharla Venkata Sunil ; Narri Yadaiah	
DYNAMIC BEHAVIOR OF WIND DRIVEN DOUBLY FED INDUCTION GENERATOR WITH	
ROTOR SIDE CONTROL FOR WIND POWER APPLICATION	
T. Samina ; S. Ramalyer ; A. Bisharathu Beevi	
IOT BASED WASTE MANAGEMENT: AN APPLICATION TO SMART CITY	
B. S. Malapur ; Vani R. Pattanshetti	
POWER QUALITY IMPROVEMENT OF A NON-STIFF SYSTEM USING DSTATCOM WITH	
STATE FEEDBACK SWITCHING CONTROL	
Nivedita Pande ; S. P. Gawande ; P. D. Debre ; M. R. Ramteke ; Akshay Kadu	
ROAD QUALITY ANALYSIS AND MAPPING FOR FASTER AND SAFER TRAVEL	
S B Prapulla ; Sriram N Rao ; Vivek A Herur	
STRUCTURAL, OPTICAL AND PHOTOELECTROCHEMICAL PROPERTIES OF TRIVALENT	
IMPURITY (B3+) DOPED ZNO NANORODS GROWN BY FACILE HYDROTHERMAL	
TECHNIQUE	
Akash Sharma ; Mohua Chakraborty ; Zeeshan Anwar ; Pooja Sahoo ; R. Thangavel	
OFF-SHORE WIND POWER AS A PILLAR OF ENERGY TRANSMISSION USING IOT	
(OSWPETIOT)	
N. P. G. Bhavani ; P. Vaishnavi ; K. Sujatha	

# IoT based Waste Management:An Application to Smart City

Prof. B.S.Malapur Computer Science and Engineering Basaveshwar Engineering College Bagalkot, Indian. e-mail:basu.malapur@gmail.com

Abstract—Emerging Technologies of IoT are transforming slowly with Cities administration. As cities will generate waste at an alaraming rate which needs collection of waste in smarter way, this collection of waste must be within time and trip planning should be done in real time, based on the status of waste. Earlier efforts were on collection of waste with smart bins but garbage collection to their places and plan trip in an optimal path is not much considered. In this paper, proposed IoT technologies with management of waste and trip management in cities is done, so that cost and time are reduced with optimized path for waste collection. Thus proposed effective results for same.

## Keywords— Smart bin, Global Positing System(GPS); Internet of Things(IoT), Ultrasonic sensor, Genetic Algorithm(GA).

### I. INTRODUCTION

Internet of Things is a connection of devices. Devices connected to network, transfer data and communicate each other provided with a Unique Identifier (UID), with or without interaction from human to human/human to computer. Automation of human life is goal of an IoT. It's an ever growing Technology connecting sensors, object extends to computing capability and also allowing devices to generate, exchange and data consumption[8].

Urban development is the vision of smart city, to Integrate Information and Communication Technology(ICT) and IoT Technology to remain safe and to manage a city's asset. Smart city includes smart economy, smart people, smart governance, smart environment, smart business, smart living and smart mobility. Smart city provides quality of life[9].

Waste management is a consistently developing issue at local and global levels. Normally, human and animal activities emerge has solid wastes and are futile and undesirable. The residential waste items are gathered through waste bins at a typical place at a specific spot for a road/area. Waste bins checking procedure for waste collection is one of the major troublesome tasks. The typical technique by which, a man needs to wander through the distinctive spots and check the spots for waste accumulation. This is to some degree complex and time consuming process. Presently, waste management system is not as effective as it ought to have been taken over the progressions in the advances and technologies that emerged in the current years. There is no surety about the Vani R.Pattanshetti(PG) Computer Science and Engineering Basaveshwar Engineering College Bagalkot, Indian. e-mail:vanimp464@gmail.com

administration/clearing of wastes at all the spots[10]. To conquer this issue another approach, proposed automated waste management system.

### II. LITERATURE SURVEY

Review of papers is carried to know the background, current working procedure and existing system flaws, where we can workout on unsolved problems. A variety of related papers have been reviewed and summarized as follows.

They focus mainly on collection. Current management activity of system and Waste characterization was the main objective to study and to determine[1]. It was demonstrated at thoubal muncipalty, which highlights municipal solid waste management (MSWM) system. And also leaves with suggestions, present management system to be improved which will be useful for authorities to work further also. Smart bin is not used here.

In paper[2] Author introduces technologies like Global Position System, Radio Frequency Identification, General Packet Radio Service, Geographic Information System and web camera are integrated devices. RFID reader device in truck will read both customer and also bin information. Efficient waste collection is achieved. Truck management is also done but is not in optimized way.

Writer of this paper introduces two models of routing they are dynamic and semi static[3]. Further, accumulation of waste in city is done by facilitating IoT a routing model dynamically is designed and in case of tragedy it's robust. Research related to waste collection concentrates on routing and scheduling dynamic design. Though lesser amount of research indicates gathering waste for smarter cities assistance.

Author explains[4] that sensor helps to know the level of garbage at maximum level, Micro- controller acts has interface between sensor and GSM system. GUI is to monitor garbage information for different locations selected. If garbage is not collected when maximum filled message reaches higher authority to take serious action which inturn will help to reduce number of trip for garbage collection and also can reduce expenditure. Collecting garbage is efficiently managed.

This work express to find the shortest route[5] using route algorithm and predictions, which will reduce the work force,







Advances in Intelligent Systems and Computing 479

Rajesh Singh Sushabhan Choudhury Editors

Proceeding of International Conference on Intelligent Communication, Control and Devices

ICICCD 2016

2 Springer

### Advances in Intelligent Systems and Computing

Volume 479

### Series editor

Janusz Kacprzyk, Polish Academy of Sciences, Warsaw, Poland e-mail: kacprzyk@ibspan.waw.pl

xviii	Contents
Fuzzy-Based Multi-Objective Optimization for Subjection and Diagnosis of Hybrid Energy Storage System of an Electric Vehicle Kurukuru Varaha Satya Bharath and Kamlesh Pandey	299
MDS <sup>3</sup> C: Modified Digital Signature Scheme for Secure Communication Gulista Khan, Bhumika Gupta and Kamal Kumar Gola	309
Autonomous Operation of Wind-Battery Hybrid Power System with Intelligent Power Management Capability Diwaker Pathak, Ankit Gupta, Rupendra Kumar Pachauri and Yogesh K. Chauhan	317
Performance Analysis of a Novel Text Watermarking Technique for Devanagari Text Nitin Namdeo Patil and Jayantrao Bhaurao Patil	325
Evaluation of Entropy-Based Segmentation Techniques for Automated Skin Disease Detection. Ishita Bhakta and Santanu Phadikar	335
Futuristic Automobile Accident and Theft Notifier with Location Tracker Archit Agarwal and Ayush Agarwal	343
Design and Comparative Analysis of Controllers for Continuous Stirred Tank Reactor (CSTR) Parvesh Saini, Rajesh Kumar, Priyanka Sharma and Nalini Rajput	351
Performance Analysis of Reactive, Proactive and Hybrid Routing Protocol Used in Petroleum Tank Over Network Control Systems S.K. Agarwal, Sachi Bansal and Anwar S. Siddiqui	367
Dead Time Compensation in Sugar Crystallization Process Sandeep Kumar Sunori, Pradeep K. Juneja, Mayank Chaturvedi and Jeevanshi Mittal	375
Performance Analysis of AMC-Based Bowtie-Shaped Slotted Antenna for Terahertz (THz) Applications Devesh Kumar, Malay Ranjan Tripathy, Manish Sharma, Amit Kumar and Umesh Kumar	383
ANFIS-Based Fault Diagnosis Tool for a Typical Small Aircraft Fuel System Vijaylakshmi S. Jigajinni and Vanam Upendranath	391
Control of Reactive Power of RES-Based Hybrid Isolated Power System Using UPFC Abhishek Gandhar, Balwinder Singh and Rintu Khanna	407

### ANFIS-Based Fault Diagnosis Tool for a Typical Small Aircraft Fuel System

Vijaylakshmi S. Jigajinni and Vanam Upendranath

Abstract In the present paper, an Adaptive Neuro-Fuzzy Inference System (ANFIS) based intelligent diagnosis tool for investigating the health of a typical small aircraft fuel system simulation was proposed. The system was designed for identifying the faults present in the aircraft fuel system and to diagnose those conditions with a proper fuel flow to the engine. The ANFIS intelligent tool works based on the logical rules of an expert system, which are developed as per the engine's fuel consumption and the fuel flow from the tanks. The inputs to train the ANFIS are the fuel flow at the previous instant and the engine's fuel consumption and the fuel tank's control signals. Training of ANFIS, generates the control signals as per the fuel requirement of the engine and the fuel flow to the tanks. The proposed intelligent controller model was implemented in the platform of MATLAB/Simulink and a comparison with the other techniques allowed the effectiveness of the proposed model.

Keywords Aircraft fuel system · Fuel flow · Fault diagnosis · ANFIS · Fuel tank

#### 1 Introduction

The aircraft industry has an important socio-economic impact on society and world economy [1]. Based on the nonlinear systems, the aircraft systems are proposed as hybrid systems [2]. The normal cooperating systems in an aircraft are specified as, Flight Management System (FMS), Flight Warning Computer (FWC), Display

Vanam Upendranath IVHM, Structural Technologies Division (STTD), CSIR-National Aerospace Laboratories, Bangalore 560 017, Karnataka, India e-mail: vanam@nal.res.in

 Springer Science+Business Media Singapore 2017
 R. Singh and S. Choudhury (eds.), Proceeding of International Conference on Intelligent Communication, Control and Devices, Advances in Intelligent Systems and Computing 479, DOI 10.1007/978-981-10-1708-7\_45 391

V.S. Jigajinni (🖂)

Department of Electronics and Instrumentation Engineering, Basaveshwar Engineering College, Bagalkot 587 102, Karnataka, India e-mail: talk2vijusj@gmail.com

International Conference on Electrical, Electronics, and Optimization Techniques (ICEEOT) - 2016

# LNLCS: Leveraging Network Lifetime using Clock Synchronization in Wireless Sensor Network

K. Nagarathna Research Scholar Basaveshwar Engineering College Bagalkot-587102, Karnataka, India Email: nagarathnavtu@gmail.com

Abstract— Clock synchronization significantly assists in enhancing the data quality during the data dissemination phase from sensors to the base station. The existing studies towards addressing time synchronization errors don't address the associated latent problems in it. Therefore, this paper introduces a novel technique called as LNLCS i.e. Leveraging Network Lifetime using Clock Synchronization. The technique is a novel attempt to integrate multihop routing with clock synchronization. LNLCS also presents a novel clustering technique that makes the sensor to play multiple roles during data aggregation as well as minimization of clock synchronization errors based on threshold based technique. The outcome of the study was gauged using clock synchronization error and amount of energy being dissipated to find that LNLCS outperforms the most standard and existing clock synchronization errors in dense and large scale WSN.

Keywords-; Clock Synchronization, Energy Efficiency, Network Lifetime, Routing Protocol, WSN

#### I. INTRODUCTION

With the advancement of the modern communication system, WSN has been playing a contributory role since more than a decade. A WSN is basically formed from the network established among the sensor nodes [1]. A sensor has the inbuilt capability of sensing a particular form of physical attributes from the surrounding environment. Normally, sensors are used in area where it is quite difficult of human to intervene in the communication system. Various applications of the sensors lie in habitat monitoring, environmental monitoring, industrial applications, healthcare monitoring system etc [2] [3]. Based on the deployment, there are two types of WSN i.e. random-based deployment and uniform-based deployment [4]. The uniformbased deployment lets the user know about the location of the sensors and usually deployed for small area of surveillance. The random deployment is meant for distribution of the sensor when the deployment area is quite large. Based on types, WSN is also classified as homogeneous and heterogeneous WSN [5] [6]. The homogeneous WSN consist of the entire sensor bearing the same physical and networking configuration, whereas heterogeneous network is formed by multiple forms of sensors. According to our observation, we find more research papers inclined towards

Jayashree D. Mallapur Dept. of Electronics & Communication Engg. Basaveshwar Engineering College Bagalkot-587102, Karnataka, India Email: bdmallapur@yahoo.co.in

homogenous sensor network in comparison to heterogeneous sensor network.

The WSN is also shrouded with various problems e.g. routing problems, energy problems, security problems [7] [8] [9]. Out of all the problems, energy problems in wireless sensors are quite critical in nature owing to the close connection between battery lifetime and communication requirements. However, from the data quality viewpoint, there exists another problem which is called as Time Synchronization problems [10]. Basically, a sensor has an inbuilt hardware clock which performs time stamping, i.e. a mechanism to attach time with the outgoing data packets. Time stamping is the best mechanism to understand the freshness of data as well as to understand the data redundancy problems. During data fusion technique, a member node when it forwards a fused data to cluster leader, it is essential that cluster leader must only accept the fused packet with different time stamp. The problem is capturing and processing of data in WSN is done by TDMA scheduling, where there are fair chances of data redundancies [11]. Moreover during intra and inter clustering mechanism, it is essential to incorporate perform time stamping. But normally, the process is inflicted with problems of energy depletion due to spontaneous data dissemination process in WSN. This problem leads to a significant difference of clock time which is called as clock drift and clock offset [10]. Hence, it is essential to perform minimization of clock drift and clock offset using a standard process called clock synchronization. Majority of the experiments done till date have focused on reducing the clock synchronization errors. But it is also significant to understand that there is a close relationship among routing, clock synchronization, and energy efficiency. We do have existing literatures, but they are more or less focused on single problem.

Therefore, this research paper will introduce an analytical technique that jointly addresses multiple hops, clock synchronization errors, and energy depletion. The proposed system is also compared with the existing standard studies towards clock synchronization techniques with respect to synchronization errors and energy consumption. Section II discusses about the existing literatures on the topic followed by discussion of problem identification in Section III that discusses

#### 978-1-4673-9939-5/16/\$31.00 ©2016 IEEE

This full-text paper was peer-reviewed and accepted to be presented at the IEEE WiSPNET 2017 conference.

# Frequency Compression of Speech for Improving Speech Perception in Sensorineural Hearing Loss: FBS Approach

Rajani S. Pujar<sup>1</sup> and P. N. Kulkarni<sup>2</sup>

Electronics and Communication Engineering Department, Basaveshwar Engineering College (Autonomous), Bagalkot-587103, Karnataka, India Email: <sup>1</sup>rajani\_pujar@rediffmail.com <sup>2</sup>pnk\_bewoor@yahoo.com

501

Abstract-Filter-bank summation (FBS) method is a commonly used technique for multi-band processing of speech and audic signals, especially in digital hearing aids. In multi-band speech processing techniques, filter bank summation provides the convenient way of processing the auditory information present in different bands based on their perceptual significance. People with hearing loss have issue in perception of speech due to widening of the auditory filter bandwidth leading to increased frequency masking. Previous studies have shown that spectral splitting of speech signal for binaural dichotic presentation helps to reduce the effect of frequency masking. Also studies showed that using multiband frequency compression it is possible to compensate the effect of widened auditory filters. This paper presents a filter bank summation method to perform dichotic spectral splitting of input speech signal followed by frequency compression to enhance speech perception for hearing impaired. In the present study, the speech signal is split into eighteen frequency bands ranging from 0-5000 Hz based on auditory critical bandwidths and frequency samples of every band compressed in the direction of center of each band using spectral segment frequency mapping technique. Performance of the algorithm was evaluated using MOS test for subjective assessment of speech quality and Perceptual Evaluation of Speech Quality (PESQ) scores for objective assessment of speech quality. The results showed a significant improvement in speech quality as indicated by MOS and PESQ scores for the SNR values in the range of 6 dB to 6 dB.

¥. .

Index Terms-Sensorineural Hearing Loss (SNHL); Spectral Masking; Filter Bank Summation Method; Dichotic spectral splitting; Frequency Compression.

### I. INTRODUCTION

Sensorineural hearing loss (SNHL) occurs when the functioning of the cochlea is affected or when there is dysfunction of the auditory nerve or higher centres in the auditory pathway. The bandwidth of auditory filter normally widened and selection of frequency gets reduced due to masking of spectral components [1], [2]. So SNHL persons have problem in understanding the speech due to broader auditory filters, leading to smoother representation of speech spectrum [3]. This causes poor perception of speech by hearing impaired people.

Hearing loss commonly occurs at higher frequencies. Hearing aid technology provides frequency-dependent amplification for hearing-impaired (HI) persons. Existing hearing devices have a constrained capacity to give adequate gain for high-frequency sounds [4]. This confines the perceptibility of significant sounds, particularly for persons with critical hearing impairment. High frequency components of the speech signal gives essential linguistic information (especially for fricatives) to the listeners.

In frequency transposition [5] method and slow playback high-frequency speech samples shifts to the regions of low frequency by a constant factor These traditional methods of transposition of frequency [6-9] were helpful in recognition of speech, but results in degradation of sound quality. The investigation of the effectiveness of multi-band frequency compression methods using different compression values, to enhance the speech audibility is discussed in [10]. In this method, the spectrum of speech is split into various frequency bands and frequency samples in every band are compressed in the direction of center of band by fixed compression rates. The technique helps in improving the speech quality for hearing impaired people. The work given in [11] reduces the effect of spectral masking using filter bank summation method. A signal is split into 15 frequency sub bands using an analysis filter banks. These 15 frequency bands were designed with sampling frequency of 12 kHz, pass band ripple and stop band attenuation less than 1 dB and 40 dB. These sub band signals were combined by using synthesis filter bank which accomplishes the frequency compression. The problem of masking of spectral components, for listeners with bilateral SNHL, can be reduced with binaural dichotic spectral splitting, using auditory critical bandwidth (ACB) based comb filters [12]. Speech perception can be improved by using a pair of complementary comb filters by adjusting interband crossover gain (between 4 and 6 dB) below pass band gain.

The paper [13] discussed the effect of linear frequency transposition (LFT) and nonlinear frequency compression (NFC) in improving the speech perception for school aged children. Paediatric participants with varying audiometric characteristics such as total distortion of the auditory system, amount of hearing loss, and the usage of hearing aid were considered for evaluating the efficiency of NFC and LFT. The analysis of both the methods was conducted for each subject based on above mentioned levels. The consequences of this examination proposes that LFT and NFC can enhance the perceptibility of high frequency (consonant) sounds and improve understanding

# Cascaded Structure of Wiener Filter with FBS based Spectral Splitting and Dynamic Range Compression For Listeners with Sensorineural Hearing Loss

Rajani S. Pujar, Pandurangarao N. Kulkarni Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, India rajani pujar@rediffmail.com, pnk bewoor@yahoo.com

Abstract- This paper presents a cascaded structure of wiener filter followed by spectral splitting using Filter Bank Summation (FBS) method and dynamic range compression with constant compression factor for binaural dichotic presentation. This helps to compensate the effect of spectral masking, reduced dynamic range for moderate sensorineural hearing loss and also improves speech perception in the adverse listening conditions by removing unwanted noise. Wiener filter is used as noise reduction algorithm. Using auditory critical bandwidths, the signal is separated into eighteen frequency bands, varying from 0-5 KHz. Spectral splitting is performed by combing odd and even numbered filter bands and presented to the left and right ears respectively that help in reducing the effect of frequency masking. Amplitude compression using fixed compression factor (0.4, 0.6 and 0.8) in each band was carried out that helps in compensating the reduced dynamic range of the hearing impaired people. Listening tests using Mean Opinion Score (MOS) on normal-hearing subjects were conducted to test the perceived quality of the processed speech. VCV syllable /aba/ and sentences were used as the test material. The results of the listening tests showed that the MOS scores for the processed speech (4.31, 4.06, 3.68, 3.5, 3.067 and 2.612) were higher as compared with those for unprocessed speech (4.48, 1.2, 1.13, 1.06, 0.8 and 0.45) for SNR values of ∞, +6, +3, 0, -3 and -6 dB respectively, indicating the improvement in the perceived quality for different SNR conditions for the compression factor of 0.6. To estimate the intelligibility of the perceived speech, listening test was also carried out on hearing impaired (moderate SNHL) subjects in the presence of background noise using Modified Rhyme Test (MRT). The test material consists 50 sets of monosyllabic words of consonant-vowelconsonant (CVC) form with six words in each set. Each subject is responded for total of 1800 presentations (300 words x 6 different SNR conditions). Results of the listening tests (using MRT) showed maximum improvement in speech recognition scores (23.49 - 27.29%) compared to unprocessed speech, indicating enhancement in intelligibility of the speech at lower SNR values.

Keywords-Wiener Noise Reduction Algorithm, Filter bank summation (FBS) method, Dynamic Range Compression (DRC), Spectral Splitting, Mean Opinion Score (MOS), Modified Rhyme Test (MRT)

#### INTRODUCTION

Degraded speech perception, increased temporal and spectral masking and reduced dynamic range between threshold of hearing and discomfort level are some of the problems associated with sensorineural hearing impaired people [1]. Furthermore, speech intelligibility gets reduced in the presence of background noise. Therefore hearing impaired people need a higher signal-to-noise-ratio (SNR) to effectively communicate.

 $\frac{c}{2}$ 

Noise reduction in hearing aids is a very challenging issue because the properties of the noise signal change significantly with time. Hence it is very difficult to develop an algorithm that works in different listening environments. The nature and properties of noise sources are very important in deciding which noise reduction algorithm works well to enhance speech signal by eliminating the noise from the corrupted signal. Some of the noise reduction, schemes applicable for hearing aids are spectral subtraction, wiener filter, and beam former approaches [2] [3] [4]. Comb filter pairs (with complementary magnitude responses) were used to enhance the intelligibility of perceived speech for bilateral hearing (moderate SNHL) impaired subjects [5].

Dynamic range Compression can be implemented in either within a single frequency band or multiple frequency bands i.e., multiband compression. Hearing aids employing dynamic compression split the signal into number of frequency bands, and amplitude compression is carried out in these bands. Filter bank summation technique (FBST) is used for multiband amplitude compression [6] to partly compensate the effect of reduced dynamic range for the people with sensorineural hearing loss. Therefore it is essential for hearing aids to improve speech perception under adverse listening conditions, and also to overcome the effects of reduced dynamic range and spectral masking effect associated with SNHL listeners.

Hence in our proposed work, we have cascaded wiener filter followed by spectral splitting using Filter Bank Summation (FBS) method and dynamic range compression using constant compression factor. Wiener filter obtains the enhanced signal by optimizing the mean square error criteria. In FBS method, frequencies ranging from 0-5000Hz are partitioned into 18 bands, depending on auditory critical bands. Each band is amplitude compressed with the constant compression factor of 0.4, 0.6 and 0.8. Spectral splitting of compressed speech is carried by combining odd numbered and even numbered filters and presented them to two ears.

Rest of the paper is organized as follows. Related works are discussed in section II. Proposed algorithm is given in section III. Tests and Result analysis are given in section IV. Spectrographic analysis is discussed in section V. Finally, section VI concludes the work.

#### **II. RELATED WORKS**

Many studies have investigated the different signal processing techniques to overcome the problem of SNHL

Proceedings of the 10<sup>th</sup> INDIACom; INDIACom-2016; IEEE Conference ID: 37465 2016 3<sup>rd</sup> International Conference on "Computing for Sustainable Global Development", 16<sup>th</sup> - 18<sup>th</sup> March, 2016 Bharati Vidyapeeth's Institute of Computer Applications and Management (BVICAM), New Delhi (INDIA)

# Improved Cluster Head Selection in LEACH Algorithm for Heterogeneous Wireless Network using NS2

Jayashree Mallapur Dept. of Electronics and Communication Basaveshwar Engineering College, Bagalkot - 560085, Karnataka, INDIA Email ID: bdmallapur@yahoo.co.in Satyanarayan Padaganur

Abstract- With the fast development in wireless applications, services and devices, used in a single radio technology such as a second generation (2G) and third generation (3G) wireless technology may not be efficient to deliver data with high speed and services related to quality for mobile users in a seamless way. The next generation wireless systems (Heterogeneous Wireless Network (HWN)) are being formulated with the image of heterogeneity, where a mobile user will be able to access multiple wireless networks. Coupling the high data rates of IEEE 802.11based MANETs and the wide coverage area of 3G networks (e.g., UMTS), this paper explain MANET-UMTS integrated network architecture. Mobility management is required to provide seamless mobility. In this scheme we consider the cluster based LEACH algorithm for Ad-hoc network, where Cluster Head (CH) is defined. But if CH is moved from cluster, then entire cluster becomes unused. So we define the new CH and its position in such a way that it can improve parameters like throughput, energy consumption, stability and configuration time. We design proposed scheme and implemented with certain simulation parameters using Network Simulator (NS-2) tool.

# Keywords: Cluster Head (CH), Heterogeneous Wireless Network (HWN), LEACH, Mobility Management, NS-2

#### I. INTRODUCTION

This From the history, the progress of wireless communication brings handiness to our routine life. Moreover, various wireless communication networks have been used globally across different location so that, users can converse with others and access the internet seamlessly [1, 2]. Therefore, wireless network must provide ubiquitous communication capability and information access regardless of user's location. There are many wireless terminals used with respect to the requirements and the features of networks, among them can be categorised according to their transmission rate and coverage area. The Wireless Local Area Network (WLAN) operate at Mbps rate (typically 1~2Mb/s) over a range of hundreds meters (typically 150~250 meter using IEEE 802.11b protocol). The dynamism of LAN has different effect such as reconfiguration [3]. The popular 3G network has the coverage in the range of several kilometers with the transmission rate of about 200~300 kbps or

Dept. of Electronics and Communication Engineering Engineering BLDEA Dr P.G.Halakatti College of Engineering, Vijayapur -586103, Karnataka, INDIA Email ID: padaganursk@gmail.com

higher. Furthermore, the Mobile Ad-hoc Network (MANET) is another wireless communication network without infrastructure [4, 5]. It is characterized by dynamic topology with node mobility and limited bandwidth. Each mobile node will function as both host and router. It send the packet to other nodes using one-to-one communication [6].

• We propose Heterogeneous network which is the hybrid network of 3G cellular network and MANET. It extracts the features of the area coverage from 3G system and the high data rate in Mobile Ad-hoc network. We also analyze the mobility management to provide seamless mobility. For this we considered the cluster based Leach algorithm for Ad-hoc network, where CH is moved then entire cluster becomes unused. So we define the new CH in such a way that it has been repositioned (reconfigured) with same old CH position, so that parameters like throughput, energy consumption and stability should be improved, and we also calculate the settling time (reconfiguration time) for new node where it is joining the new cluster.

### II. RELATED WORK

In Clustering technique, mobile nodes of network system arrange themselves into hierarchical structures [3, 7]. Using clustering, mobile nodes can use the limited network resources such as bandwidth, battery energy efficiently. In individual cluster, aggregation of data and fusion are done by clusterhead, to decrease the amount of data forwarding to the Base Station. Cluster creation is usually performed based on residual energy of nodes, stability and selection of cluster-head [1]. Normal nodes select their CH right after employment and transmit data to the CH. The function of CH is to send its own data and external data from different nodes to the base station after doing fusion and aggregation of data. LEACH is one among the hierarchical protocols for routing in wireless networks. The scheme proposed in LEACH has encouraged many other protocols.

In no paper they are discussing about the movement of CH itself from the cluster. But in our proposed scheme we

Copy Right © INDIACom-2016; ISSN 0973-7529; ISBN 978-93-80544-20-5

1239

# FPGA-Based Design of a Hearing Aid with Frequency Response Selection through Audio Input

Shankarayya G. Kambalimath<sup>1</sup>, Prem C. Pandey<sup>2</sup>, Pandurangarao N. Kulkarni<sup>1</sup>, Shivaling S. Mahant-Shetti<sup>3</sup>, and Sangamesh G. Hiremath<sup>4</sup>

<sup>1</sup> Dept. of Electronics & Comm. Engg., Basaveshwar Engg. College, Bagalkot, Karnataka, India

<sup>2</sup> Dept. of Electrical Engg., Indian Institute of Technology Bombay, Mumbai, India

Karnataka Microelectronics Design Center, Manipal, Karnataka, India

<sup>4</sup>G.M. Institute of Technology, Davangere, Karnataka, India

<kambalimathsg@rediffmail.com>, <pcpandey@ee.iitb.ac.in>, <pnk\_bewoor@yahoo.com>, <mahant@karmic.co.in>, <dr\_sgh@yahoo.co.in>

Abstract—Hearing aids provide frequency-selective amplification to compensate for the elevated hearing thresholds and automatic volume control to partly compensate for the significantly reduced dynamic range of hearing. The response selection controls of these devices are not easy to access during their normal use. To address this problem, an FPGA-based design of a hearing aid is presented, with frequency response selection by DTMF coded sound from a hand-held device without using switches, ports, or additional hardware. The audio input is given to two parallel processing blocks. The first block is a DTMF detector employing Goertzel algorithm and outputs control bits for selecting one of the pre-stored responses. The second block processes the signal for frequency-selective amplification and produces the audio output. The filter is designed as an FIR filter with its magnitude response approximating the product of the responses required for hearing-loss compensation and noise attenuation. It is efficiently realized using a processing architecture employing sequential multiply-accumulate operations. The design is implemented using FPGA "Altera Cyclone IV EP4CE115F29C7" and an audio codec and verified for satisfactory operation.

Keywords—DTMF detector; FPGA based signal processing; hearing aid.

#### I. INTRODUCTION

Persons with sensorineural hearing loss experience degraded speech perception and find it to difficult to adapt to different listening conditions. The currently available digital hearing aids provide frequency-selective amplification to compensate for the elevated hearing thresholds and automatic volume control to partly compensate for the significantly reduced dynamic range of hearing [1]. They generally have a set of filter responses to partly compensate for the different adverse listening conditions and for suppressing significant spectral components of the noise. However, the selection controls are not easy to access during their normal use. As optical and RF links require additional hardware in the device and are not useable when it resides in the ear canal, an audio controlled selection can be of great help to the hearing aid user.

For reducing the size and power, hearing aids are designed using ASICs and an FPGA-based prototyping and testing is helpful in significantly reducing the development time and nonrecurring cost [2]. Hence, an FPGA-based design of a hearing aid is presented, with the response selection by dualtone multi-frequency (DTMF) coded sounds from a hand-held device. The control signal is input through the same microphone, pre-amplifier, and ADC as used for speech. Goertzel algorithm-based DTMF detection [3]-[5] is used as it is not affected much by speech and environmental noise.

#### II. DESIGN AND IMPLEMENTATION

Realization of hearing loss compensation filter as a parallel bank of IIR band-pass filters, with adjustable gains to obtain the desired magnitude response, results in nonlinear phase response. FIR filter design, although requiring a relatively higher order, permits coupling an almost arbitrary magnitude response with a linear phase response. We use a linear-phase FIR filter, with the magnitude response approximating the product of the responses required for hearing-loss compensation and noise attenuation. A single filter helps in reducing the signal delay. The loss compensation response is obtained from the audiogram, by linear interpolation of the gains desired at the audiometric test frequencies. Four noise attenuation responses are devised: neutral or all pass (AP), low pass (LP) with 6 dB/octave tilt above 2 kHz to suppress highfrequency noise, high-pass (HP) with 6 dB/octave tilt below 500 Hz to suppress low-frequency noise, and band-pass (BP) as a cascade of LP and HP. The sets of filter coefficients are calculated by iterative application of frequency sampling technique and are stored. The set of coefficients for the appropriate response is selected through DTMF coded sounds from a handheld device.

The design is implemented using an FPGA board with audio codec "WM8731" and FPGA "Altera Cyclone IV FPGA EP4CE11529C7". The audio input with sampling frequency of 10 kHz is applied, as shown in Fig. 1, to two parallel processing blocks: (i) DTMF detector with output control bits for selecting one of the pre-stored set of filter coefficients and (ii) FIR filter for frequency-selective amplification.

A DTMF signal is a sum of two harmonically unrelated tones, one from four "row" frequencies and the other from four "column" frequencies. The DTMF detector, shown in Fig. 2, has a set of four Goertzel filter blocks for row frequencies and another such set for column frequencies, each with a second order IIR filter and 250-sample energy calculation. The detector generates 2-bit row and 2-bit column output. Only 4 out of 16 codes are used at present. The bits are applied as input to the ROM containing four sets of filter coefficients each consisting of 513 coefficients.



3rd International Conference on Research Trends in Engineering, Applied Science and Management (ICRTESM-2017)at Institution of Electronics and Telecommunication Engineers, (IETE) , Erandwane, Pune, Maharashtra, Indiaon 28th May 2017ISBN: 978-81- 934083-1-5

## Design of an Optimized Piezoresistive Pressure Sensor for a Pressure Range of 0 to 1 MPa

Shweta Nayak K, Ajayakumar C Katageri, B G Sheeparamatti Department of Electronics and Communication Engineering, Basaveshwar Engineering College, Bagalkot, Karnataka, INDIA

### ABSTRACT

The design of the piezoresistive pressure sensor considers the geometry of the membrane, size, shape and placement of the piezoresistor on the surface of the diaphragm. when pressure is applied, it is converted to output voltage by the pizoresistros which are connected in wheatstone bridge form. this paper describes the optimum techniques to enhance the performance of the sensor. Sensitivity can be enhanced by placing the PZR on proper location and selecting geometry of the sensor. Deflection of diaphragm, change in the output voltage and sensitivity are the various parameters used for the analysis of the designed sensor. The simulation results show a highest sensitivity of  $0.016 \mu V/V/kPa$ .

Keywords: Optimum, pressure sensor, Piezoresistive, Sensitivity, Comsol multiphysics.

### **I.INTRODUCTION**

MEMS devices like pressure sensors have modified the way that application engineers and system designers measure pressure. The ease of use, ruggedness, low cost and small size allow these sensors as an application in medical as well as industrial and automobile process control and portable products. A pressure sensor basically consists of a sensing element (diaphragm) and certain transduction mechanism such as capacitive, piezoelectric, Piezoresistive, etc.

Pressure sensors can be classified in terms of pressure ranges they measure, temperature ranges of operation, and most importantly the type of pressure they measure. Pressure sensors are variously named accordingly to their purpose, but the same technology may be used under different types. The three different categories of pressure measurements are Absolute sensor, Gauge pressure sensor and Differential pressure sensor. Pressure sensors can also be classified depending upon the pressure sensing technology based on transducer used: there are many types of pressure sensors available based on transducer used such as Capacitive pressure sensor, Piezoelectric pressure sensor, Resonant pressure sensor, Piezoresistive pressure sensor, a capacitance to voltage conversion is required, resulting in added electronics. Fabrication of capacitive pressure sensors is more complex, more susceptible to electromagnetic interference. Nonlinear relationship between the capacitance and the displacement, hence require force-balancing and linearization electronics to be fabricated. Similarly resonant pressure sensors are complicated in processing, they need to be packaged them in a vacuum to maintain high quality factor. While the piezoelectric pressure sensors are not suitable for static pressure sensing because piezoelectric materials only respond to changing strains.

When compared with piezoresistive mechanism, it has mature fabrication process, linear operation for different pressure ranges, simple read out circuitry, can be made highly sensitive. Hence this paper includes mechanism of Piezoresistive transduction because of it's simple fabrication and measurement.

### **II.LITERATURE SUEVEY:**

Various articles on Piezoresistive pressure sensor has been studied with respect to their sensing mechanism, sensing pressure, material, pressure range, merits and demerits.

# Sensitivity Enhancement of Piezoresistive Pressure Sensor with Meander Shape Piezoresistor

Shwetha Meti Department of Electronics & Communication Engineering, Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India E-mail: siddarud.becbgk@gmail.com

Ajayakumar C. Katageri Department of Electronics & Communication Engineering, Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India E-mail: ajaykatageri@yahoo.co.in Kirankumar B. Balavalad Department of Electronics & Communication Engineering, Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India E-mail: kiranb4004@gmail.com

B. G. Sheeparamatti Department of Electronics & Communication Engineering, Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India E-mail: sheepar@yahoo.com

Abstract—Amongst various transduction principles of pressure sensor, piezoresistive method provides high sensitivity and linear operation over a wide range of pressure. In this paper, piezorestive transduction mechanism is employed for design of a pressure sensor. The diaphragm of the proposed sensor is designed using n-type of Si with copper material used as a connecting arms for the p-type meander shaped piezoresistors placed on the surface of the diaphragm in a Wheatstone bridge configuration at the high strain region of the diaphragm. Meander shape piezoresistors of different length are simulated, in order to find out the best configuration for high sensitivity and linearity. The proposed design is analysed to study the deflection of the diaphragm and output voltage across the bridge. Results reveal that the 50  $\mu$ m length of piezoresistors is found to have the best sensitivity.

*Keywords*—Pressure sensor; Meander shape piezoresistors; Linearity; Sensitivity.

### I. INTRODUCTION

Micro pressure sensors were the first MEMS based devices to be fabricated. So far about 18% of the MEMS based devices available in the world are pressure sensors [1]. MEMS pressure sensors work on the principle of the mechanical deformation of a thin diaphragm due to the pressure exerted by the contact medium. The mechanical stress induced due to the applied pressure is converted into an electrical signal using piezoresistive, capacitive, optical and resonant sensing mechanisms. Among the various transduction mechanisms available for the sensor, piezoresistive type is the most widely used due to various advantages such as good linear input/output relationship, high reliability, small size, easy integration with electronics, simple compensation circuitry and a well matured fabrication process Piezoresistance is the phenomenon of change in resistivity of certain materials (like silicon, polysilicon, SiC etc.) on application of stress/strain [2]. Piezoresistive pressure sensors employ this principle to measure the applied pressure. Piezoresistive sensors also have much higher sensitivity than the metal strain gauges which work on the basis of change in resistance due to geometrical deformation. Meander shape of piezoresistive pressure sensor configuration constitutes an important part to a pressure sensor market. These types of sensors are helpful in placing the entire length of the piezoresistors inside the long and narrow high stress regions and they are used mainly because of their high sensitivity and good linearity.

Piezoresistive pressure sensors employing the use of silicon, poly silicon as piezoresistive materials, because of its high sensitivity and repeatability. Pressure sensors employing the use of other materials like SiC, and diamond for piezoresistors are not very popular for commercial usage because the fabrication technology is not mature for such materials [3]. Also silicon based piezoresistors can be easily fabricated using ion implantation and the sheet resistance of the resistors can also be carefully controlled using ion implantation. In this work, we will discuss the design methodology and simulation of a meander shape piezoresistive pressure sensor based on piezoresistance in silicon.

### **II. LITERATURE SURVEY**

S Santosh Kumar et.al [4] discusses the silicon piezoresistive pressure sensor for barometric applications in the range of 0 to 1.1 bar pressure. When a pressure is applied on the diaphragm with four piezoresistors, the stress induced

# Design of Piezoresistive Pressure Sensor for Enhanced Sensitivity

Siddarud Bannikoppa Department of Electronics & Communication Engineering, Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India E-mail: siddarud.becbgk@gmail.com

Kirankumar B. Balavalad Department of Electronics & Communication Engineering, Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India E-mail: kiranb4004@gmail.com

Abstract- Piezoresistive pressure sensor is a significant transduction mechanism for measuring pressure due to the fact that it is simpler to integrate with electronics, its response is more linear, they are inherently shielded from RF noise and fabrication is easy compared to other transduction mechanism. In this paper, piezoresistive transduction mechanism is employed for design of a pressure sensor. The diaphragm of the proposed p-type Silicon sensor is designed using n-type Silicon with piezoresistors placed on the surface of a diaphragm in a Wheatstone bridge configuration at the most sensitive region of the diaphragm. The proposed design is analyzed to study the deflection of the diaphragm and the output voltage across the bridge. The effect of change in piezoresistor length on the output voltage is also investigated. The results reveal that the proposed sensor provides highest sensitivity for the piezoresistor length of 50um.

*Keywords*— Piczoresistance; Pressure sensor; Sensitivity; Stress

#### I. INTRODUCTION

A pressure sensor measures pressure, typically of gases or liquids. Pressure is an expression of the force required to stop a fluid from expanding, and is usually stated in terms of force per unit area. Semiconductor sensors are widely used in current measurement systems because of their low cost, small size and easy integration with electronic circuits [1]. In the automotive industry, for example, mechanical semiconductor sensors have become very widespread, especially those intended for the measurement of pressure and acceleration [2].

Pressure sensors can be classified based on the transduction mechanism such as capacitive, piezoresistive, optical, resonant, Thermal sensors. In piezoresistive pressure sensor, the piezoresistive effect is a change in the electrical resistivity of a piezoresistive material when mechanical strain is applied. Ajayakumar C. Katageri Department of Electronics & Communication Engineering, Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India E-mail: ajaykatageri@yahoo.co.in

B. G. Sheeparamatti Department of Electronics & Communication Engineering, Basaveshwar Engineering College, Bagalkot-587103, Karnataka, India E-mail: sheepar@yahoo.com

In contrast to the piezoelectric effect, the piezoresistive effect causes a change only in electrical resistance, not in electric potential. The mechanism is in conducting and semiconducting materials, changes in inter-atomic spacing resulting from strain affect the band gaps, making it easier for electrons to be raised into the conduction band. This results in a change in resistivity of the material.

The sensors combine the excellent mechanical properties and piezoresistive effect of single crystal silicon and related process technologies of integrated circuits. In order to enhance the piezoresistive effect and also the sensitivity of the sensors, a thin diaphragm is formed by silicon micro machining technology so that the stress resulting from applied pressure is concentrated on the piezoresistors located on the diaphragm. Piezoresistive pressure sensors are used in a wide variety of applications including tire pressure meters, pressure gauges, pressure switches, blood pressure monitoring, process control and automobile parts, etc.

The design of piezoresistor is vital in determining the performance of pressure sensors. In view of that, this paper proposes Piezoresistive pressure sensor with a Different length of piezoresistors to examine the enhance in sensitivity and linearity of pressure sensor.

### **II. LITERATURE SURVEY**

In this section we discuss about the papers on piezoresistive pressure sensor with respect to pressure sensing mechanism, materials used for the design of the sensor, parameters used for analysis, pressure sensor range, merits and demerits of the work.
## Design Simulation and Analysis of Piezoresistive Micro Pressure Sensor for Pressure range of 0 to 1MPa

Kirankumar B. Balavalad

Electronics & Communication Engineering Department Basaveshwar Engineering College, Bagalkot-587102 (Affiliated VTU Belagavi) Karnataka, India kiranb4004@gmail.com

Abstract: In this work, we present the design, simulation and analysis of piezoresistive micro pressure sensor operable at pressure range of 0 to 1MPa. Piezoresistors are placed and are connected in the form of Wheatstone Bridge, on the diaphragm. For the design and simulation various aspects like diaphragm dimensions, placement of piezoresistors, dimensions of piezoresistors and different shapes of piezoresistors have been considered to find the best possible configuration for better sensitivity. Simulation and analysis for all the parameters mentioned above has been done using Finite element method based COMSOL Multiphysics. Simulation results show that the sensor design with two-turn meander configuration with the dimensions mentioned in the paper provide better sensitivity 35.54 mV/V/MPa over the pressure range of 0 to 1MPa.

Keywords: Micro Pressure Sensors, Piezoresistive, Pressure Range, COMSOL/Multiphysics, Sensitivity, Linearity.

#### I. INTRODUCTION

MEMS Pressure sensors are the first commercially successful devices. Pressure sensors are used in many applications such as, industry, automotive, chemical, oil fields and water treatment etc. Conventional pressure sensors are replaced almost completely due to their advantages viz., small size, low cost, ease of fabrication, high reliability, less stray factors. Based on the method of sensing micro pressure sensors are classified as absolute, differential and gauge sensors and based on transduction mechanisms, mechanism like capacitive, piezoelectric, resonant and piezoresistive have been reported [1]. the various transductions mechanisms, Amongst piezoresistive mechanism is widely preferred because of its ease of fabrication, high reliability, better sensitivity, high linearity, simple voltage readout and compensation circuitry [2]. Micro pressure sensors design and miniaturization has been changed considerably after finding the piezoresistivity silicon and germanium [3-4]. Semiconductor in

978-1-5090-4697-3/16/\$31.00 ©2016 IEEE

Sheeparamatti B G

Electronics & Communication Engineering Department Basaveshwar Engineering College, Bagalkot-587102 (Affiliated VTU Belagavi) Karnataka, India sheepar@yahoo.com

piezoresistive pressure sensors provide high sensitivity then the metal strain gauges, because of their high gauge factors. Piezoresistive sensors explore the piezoresistive effect, where the resistance of piezoresistors changes with applied pressure. Silicon based micro piezoresistive pressure sensors are very popular because of many advantages of silicon. Silicon can be easily doped and controlled to develop required piezoresistors and their resistivity. In this paper we explore design, simulation and analysis of MEMS based piezoresistive pressure sensors using silicon over a pressure range of 0 to 1MPa. COMSOL simulator has been used for the simulation. Piezoresistors are placed and are connected in the form of Wheatstone Bridge, on the diaphragm. For the design and simulation various aspects like diaphragm dimensions, placement of Piezoresistors, dimensions of Piezoresistors and different shapes of Piezoresistors have been considered to find the best possible configuration for better sensitivity and linearity. Meander shape has been explored to enhance the sensitivity of the designed sensor [5].

#### II. PIEZORESISTIVE MICRO PRESSURE SENSOR

Typical piezoresistive pressure sensor (PPS) consists of a diaphragm most of the time made up of silicon, on top of which four Piezoresistors (usually formed by controlled doping of silicon) are either embedded or placed (deposited). Then each of the piezoresistors are connected in the form of Wheatstone bridge. The simple diagram is shown in fig. 1. For the configuration shown in fig. 1, pressure is applied on the diaphragm, which will deflect. The deflection causes stress on the piezoresistors and results in change in the resistance, which will be further sensed by the bridge. Diaphragm dimensions, size and Piezoresistors dimensions, shape and placement are very important design considerations. Burst pressure, sensitivity and linearity are important performance parameters of the sensors.

345

## LPC Based Speech Recognition for Kannada Vowels

Anand H. Unnibhavi Dept. of E & C Basaveshwara Engineering College, Bagalkot, Karnataka, India anandhu.rampur@gmail.com D S Jangamshetti Dept. of E & E Basaveshwara Engineering College, Bagalkot, Karnataka, India adsi1869@rediffmail.com

Abstract— Automatic speech recognition (ASR) is most promising area which is of special attention today. This paper proposes speech recognition system for Kannada vowels. The proposed framework consists of preprocessing unit and classification unit. The preprocessing unit segments the speech signal into proper frames and extracts the features using Linear Predictive Coding (LPC). The Kannada vowels are classified into the proper classes in the classification unit where Euclidean distance is used. The recognition accuracy obtained is about 40 %.

Keywords— Linear Predictive Coding; Mel frequency cepstral coefficients; Isolated Word Recognition

#### I. INTRODUCTION

Speech recognition is used to translate words spoken by humans to a machine understandable format. The recognition process is still a challenging research area because existing methods are not fast and precise enough compared to human recognition abilities. Keyboard acts as a barrier between computer and the user which is true in rural India. This barrier can be overcome by speech recognition system. There is a difficulty in implementing ASR because different regions peoples of Karnataka have different speaking style, accents, and different voice patterns. Along with these challenges, the presence of background noises, changing of signal properties over time and differences between the training and testing environments also pose major problems in speech recognition [1]. The speech signals are continuously changing in nature. which also affects the speech recognition process. In case of isolated word recognition, signals are recorded in intervals. The signal is preceded by silence and followed by silence therefore the speech segment need to be separated from non speech segment i.e. silence region because this requires storage space and increased computation time. This paper proposes an approach for identifying isolated words corresponding to Kannada vowels. The isolated speech recognition system is divided into three parts. The first part is to develop the database of spoken words. The second part deals with the extraction of features. Finally the third part helps in classification of the spoken words. Literature survey, proposed work, analysis of results and conclusions and outlines are discussed in the following sections.

#### II. LITERATURE SURVEY

The wavelet packet based Kannada speech recognition is discussed in paper [2]. Wavelet Packet Decomposition (WPD)

is a speech compression technique. This method yields four ets of WPD coefficients. A Mel frequency cepstral coefficient (MFCC) is the feature extractor that was used to compute the coefficients from WPD coefficients and Euclidian distance was used to classify the speech. 90 % of accuracy was achieved using Daubechies8, 5-level decomposition technique. Paper [3] presents an Isolated Word Recognition (IWR) system for identification of spoken words. MFCC was used as feature extractor and SVM was used as a classifier. This developed system was tested and evaluated with a performance of 79% accuracy.

In paper [4], a statistical technique was used to remove the silence from the speech signal. This method was used to identify the minimum speech patterns that are required while creating the training set of the speech samples. MFCC was used as a feature extractor and Vector Quantization (VQ) was used as classifier. The recognition error rate was decreased from 2.59 to 1.56 in the case of VQ1 clustering algorithm and 2.5 to 1.45 for VQ2 algorithm. Also the speaker dependent error recognition rate was decreased from 2.5 to 1.45.

A Kannada isolated digit recognition system using MFCC as feature vector was discussed in paper [5]. The system was designed to recognize isolated Kannada numbers. Hidden Markov Model (HMM) was used as recognizer. An average efficiency of 90 % was achieved. Sarika Hegde, et. al. have designed speech recognition system as Multiple Classifier System (MCS). MFCC is used as feature extractor and HMM and Support Vector Machine (SVM) classifiers are used where both of which are trained separately with the same training dataset [6]. While testing, each of these classifier tests the data in parallel. The combined classifier takes the final decision which also combines the technique of Dynamic Time Warping (DTW). An automatic phoneme recognition system based on Gaussian Mixture Modeling (GMM) which is a static modeling scheme is presented in paper [7]. 15 Kannada phonemes were used to note and less the model. Caussian multivariate model was used to recognize an unknown phoneme from the testing database from the given MFCC for each class by calculating the mean and covariance matrices of corresponding phoneme sequences.

## **Comparative Analysis of Different Micro-Pressure Sensors Using Comsol Multiphysics**

Rashmi. S. Jakati Department of ECE,Basaveshwar Engineering College, Bagalkot (Affiliated VTU Belagavi) Karnataka, India

rashmijakati@yahoo.com

Kirankumar B. Balavalad Department of ECE,Basaveshwar Engineering College, Bagalkot(Affiliated VTU Belagavi) Karnataka, India kiranb4004@gmail.com

B. G. Sheeparamatti Department of ECE, Basaveshwar Engineering College, Bagalkot (Affiliated VTU Belagavi) Karnataka, India sheepar@yahoo.com

Abstract: Amongst MEMS technology, pressure measuring technology is one of the most important parameter, which is measure of force per unit area. In this work a comparative analysis of different pressure sensor mechanisms is carried out. Capacitive pressure sensors work on the principle of electrostatic transduction mechanism, piezoresistive pressure sensor employs a change in resistance values of piezoresistive elements placed on the surface of the diaphragm. The piezoelectric pressure sensor works based on piezoelectric effect, where the sensing element is stressed due to the applied pressure, positive electric charge is generated which in turn results to induced voltage. The simulation and modeling of proposed pressure sensors is done using COMSOL Multiphysics. Based on simulation results, capacitive pressure sensor exhibits non-linear response. Piezoresistive pressure sensor provides high linearity and better sensitivity. Piezoelectric pressure sensors achieve high sensitivity but are applicable to only dynamic pressure conditions.

Keywords: Pressure, Capacitive pressure sensor, Piezoresistive pressure sensor, piezoelectric pressure sensor, Linearity, Sensitivity

#### I. INTRODUCTION

Pressure is inevitable used in many industrial and day to day life applications. There is always increasing need for pressure measurement techniques. MEMS based Pressure sensors are highly popular in the domain of pressure measurement due to their reliable and accurate measurements. Pressure (P) is defined as the force applied per unit area which is given by,

P = F / A

Where F is force applied and A is the area where the force is applied.

MEMS pressure sensor converts mechanical into electrical signal based on particular transduction mechanism.



The figure 1 shows the basic MEMS Pressure sensor. The pressure sensors have their wider application in the field of automotive industry, bio-medical and weather forecasting.

#### A. TYPES OF PRESSURE SENSORS

The pressure sensors can be classified based on different parameters. The pressure sensors classified based on their sensing element are as follows.

- a) Capacitive Pressure Sensor
- b) Piezoresistive Pressure Sensor
- c) Piezoelectric Pressure Sensor

#### a) Capacitive Pressure Sensor

This Pressure sensor works based on the principle of electrostatic transduction mechanism. The design of the pressure sensor consists of pair of parallel diaphragms separated by dielectric medium in between them. Capacitive pressure sensor when subjected to pressure, the upper diaphragm bends and the distance between the diaphragms changes hence, change in capacitance can be observed, which is given by,

$$C = \varepsilon_0 \varepsilon_r A / D$$

Where, C – capacitance,  $\epsilon_o$ - dielectric constant,  $\epsilon_r$  – relative permeability, A-area of the plate, D- Distance between two parallel diaphragms.

#### b) Piezoresistive pressure sensor

Piezoresistive pressure sensor works based on Piezoresistive effect. Pressure sensor when subjected to mechanical strain the effective mass of silicon atoms either increase or decrease which in turn changes the mobility of the silicon carriers, hence the resistance(R) of the material changes which is given by,

$$\mathbf{R} = \rho l / A$$

Where, A-cross-sectional area of the piezoresistive material, l-length of the piezoresistive material,  $\rho$ - Resistivity of the material.

The piezoresistive pressure sensor consists of silicon diaphragm with piezoresistive elements mounted on it. Silicon is used as sensing element. The Piezoresistors are placed in the form of Wheatstone bridge circuit on the

355

## Routing in Opportunistic Networks: Taxonomy, Survey

S R Bharamagoudar Dept. of Electronics & Communication Engineering Basaveshwar Engineering College (Autonomous) Bagalkot, India srb bec@yahoo.com

Abstract— The accelerated growth of wireless communication such as personal digital assistants, tablets, smart phones, that are carried by the people may arrange themselves in to a special network for the communication of messages. The moving nodes communicate with the nearby nodes using node contact opportunity through Wi-Fi or blue tooth. These nodes suffer from frequent disconnections. The network with frequent disconnections is known as opportunistic network or intermittently connected network. It is a unique wireless network architecture which enables the mobile nodes to have communication with each other in environment where there is no continuous route exists between source and destination .Such networks espouse store-carry-forward habit for the message transfer and utilize mobility of moving nodes for message communication. These networks are featured with network topology which is unstable and also long latency. Routing requires the nearby neighborhood nodes information for the message transfer. Depending on the routing action, we categorize routing schemes as social aware information schemes and social zero information schemes. The schemes such as social zero information do not require any past information of the nodes behavior. Subject to the content of information, social aware schemes are furthermore categorized as microscopic and macroscopic schemes. Microscopic schemes utilize mobile user's movement. Whereas macroscopic schemes have integrated social relationship with the mobile user's movement. We explicate about the routing solutions for the microscopic and macroscopic schemes. Additionally we briefly discussed evaluations in routing schemes.

Keywords— Opportunistic Networks, Intermittently connected network, Store-carry-forward.

#### I. INTRODUCTION

In the present networking world, wireless communication plays a predominant role. The accelerated growth of mobile devices like laptops, smart phones, notebooks etc makeup mobile communication area a more fascinating research in recent years. Another extension of the mobile has inclined to the Mobile Ad-hoc Network (MANET) model known as Opportunistic Networks (ONs). These networks make use of nodes mobility characteristic, to enter the communication range of each other. This helps to establish communication path between the nodes to transfer the information. ONs suffer from unstable network topology as well as prolonged latency because of frequent disconnections [1]. Such networks suffer S V Saboji Department of Computer Scieffce & Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587103, India saboji\_skumar@yahoo.com

from abruptly intermittent connections because of moving nodes. These networks are also entitled by intermittently connected networks. Nodes suffer from the initial knowledge of the structure of the network because of random movement of the nodes. So the routes are created dynamically and relay nodes or intermediate nodes take active part in the transferring of messages to respective destinations. If the intermediate nodes cannot find any forwarding opportunity within its communication range then, the nodes store the messages and carry the messages until they find next upcoming contacts to forward the messages [2]. Using Wi-Fi, Bluetooth, the built in technologies, nodes communicate with each other. The network is accustomed by mobile devices which are carried by human beings and hence routing is a challenging task. The major concern in ONs is reliable transfer of message to the destination. Hence co-operation between the nodes and storage of messages are the primary requirement [3].



Fig.1 Message forwarding in opportunistic networks

This full-text paper was peer-reviewed and accepted to be presented at the IEEE WiSPNET 2017 conference.

## BDI Agent based Routing Scheme in VANETs

Mahabaleshwar S. Kakkasageri\*, Mamata J. Sataraddi\*, Poornima M. Chanal\*, Gururaj S. Kori\*\* \*ECE Dept., Basaveshwar Engineering College (Autonomous)

Bagalkot - 587103, Karnataka, India

mahabalesh\_sk@yahoo.co.in, kdmamata@rediffmail.com, poornima.chanal@yahoo.com

\*\*ECE Dept., Biluru Gurubasava Mahaswamiji Institute of Technology

Mudhol-587313, Karnataka, India

korigurus@yahoo.com

Abstract - Vehicular Ad hoc Networks (VANETs) are a class of wireless networks built dynamically in an ad hoc network manner. VANET is a key technology for smart and Intelligent Transportation System. Efficient and intelligent algorithms for routing in VANETs are critically required for safety information propagation. Routing in highly dynamic VANET topology is important for the achievement of any vehicular safety applications. The routing mechanism in vehicular networks is critical because of VANET constraints unpredictable network topology, such frequent disconnections, varying network densities, etc. Multi agent scheme based routing scheme for VANETs is addressed in the paper. We address the challenge of how to route the data with short communication delay, overhead, and the complexity. Specifically, we consider the scenario for (V2V) vehicle-tovehicle communication. Multi agent scheme consists of static agent and mobile agents. The scheme operates as follows: (1) establishing a connectivity pattern between the vehicles; (2) creating a set of beliefs; (3) desire development; (4) finalizing the intention. We validate the proposed scheme by simulation. Evaluated simulation results are packet delivery ratio, bandwidth utilization, and route identification delay.

#### Index Terms- VANETs, V2V, Routing, Multi-Agent System

#### I. INTRODUCTION

The VANETs are best example for (MANETs) mobile ad hoc network. Modern vehicles with capability of sophisticated computing and communication are a good sign for Vehicular Adhoc Networks (VANETs). VANETs communicate with IEEE 802.11p, were communication is conceivable between vehicles and road side infrastructure. Design of systematically organised VANET environment improves the motorway safety and optimizes the traffic conditions. Auto collision happens because of heavy traffic on the road. These accidents and other incidents on the roads occur due to absence of safety and other necessary services information. Overcrowding of vehicles results in heavy delay of information for the travellers and creates harmful to the environment on the roads. Valid safety data related issues are given most noteworthy priority in VANETs data aggregation, validation and dissemination. In VANETs, efficient routing protocols should address the issues such as unpredictable topology, density of vehicles, mobility, heterogeneous communication environment, etc. are discussed in papers [1] - [7].

Execution assessment of the distinctive vehicular framework circumstances is appeared in [8] [9]. Both topologies based routing algorithms and position-basedrouting are assessed in urban and city situations. Dynamic security and smart transportation are essential applications of VANET are presented in [10]. The paper additionally gives a point by point overview of routing mechanisms and mobility prototypes which are reasonable for efficient V2V and V2RSU communication.

The results of the paper [11] show that multicast routing algorithms, including geocast and mobility based, are more convincible than other routing algorithms. In paper [12] the researchers showed the recent taxonomy of the routing protocols for VANETs. The paper [13] gives a review of VANETs plus opens current challenges in routing issues, primarily paper focus on V2V communication. Finally author summarise the strength and limitations of different classes in routing protocols.

The investigation in paper [14] is predominantly on the review of new outcomes of the routing schemes in VANETs. The difficulties and viewpoints of routing protocols are finally discussed by author. The author proposes multi agent based approach for intelligent information propagation in VANETs. The paper [15] explained information dissemination using software agents in VANET.

Aggregation of critical information and minimization of redundant data dissemination can efficiently done by regression based cognitive agent approach is presented in [16]. DSDV protocols performance can be improved by using agent based approach is basically discussed in [17].

Paper [18] investigates that the conventional and customary routing algorithms of MANETs and proved that efficiency is less in VANETs due to VANETs architecture and environment. Multi agent system approach improves the efficiency of the routing protocol. A vehicular information ad-hoc network that consists of three tier network architecture using multi agent system offers adaptability, versatility and viability for movement of data in VANETs and also supports hearty and deft system administration [19].

A framework for keen learning and choice making techniques to build smart, intelligent and centralized traffic management model is proposed and discussed in [20], further paper shows a cognitive radio enabled VANET for multi-agent based ITS. The scheme demonstrates two model for data trade, one is dynamic (V2V) and another is semi-dynamic (V2RSU).

Static and mobile software agents to address the traffic problem in city and urban areas are discussed in [21]. Agent based approach results in best QoS in VANET and also provides the safety information to the drivers for comfort driving in accident prone areas.

A software agent is a software entity to function autonomously in a continuous inputs environment. Belief-Desire-Intention agent architecture represents agent three key mental structures: enlightening states of mind around the world (beliefs or convictions), motivational approaches

#### 978-1-5090-4442-9/17/\$31.00 ©2017 IEEE

## **Review of Microscopic and Macroscopic Routing Algorithms in Opportunistic Networks**

S. R. Bharamagoudar<sup>1</sup> Department of Electronics & Communication Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587103, India srb bec@yahoo.com

Abstract- Opportunistic networks are highlighted with long latency and unstable network topology. It is a unique wireless network architecture which enables the mobile nodes to have communication with each other in environment where there is no continuous route between source and destination. In this network scenario there is no continuous routing path exist between source and destination. Using intermediate nodes information is conveyed to the destination. These networks espouse capture move-forward behavior arising from the mobility of the mobile nodes for message relying. In this article we explicate about the challenges of the opportunistic networks, classification of routing protocols, also confer about the metrics required for the performance measurement of routing algorithms. For the message transfer, routing requires information about the neighborhood nodes. Based on routing behavior, we classify routing protocols as social zero information schemes and social aware information schemes. Social aware information schemes are further classified into microscopic schemes and macroscopic schemes depending upon content of the information. In this paper we discuss some of recent microscopic and macroscopic routing protocols and also confer the comparison among the reviewed algorithms.

#### Keywords—Opportunistic Networks, Routing, Mobility pattern

#### I. INTRODUCTION

In our daily life, wireless communication plays an important role. Recent progress in wireless technologies such as 3G, 4G, and Bluetooth, Wi-Fi, another advancement of the Mobile has given rise to Ad-hoc Network (MANET) model called Opportunistic Networks (ONs). ONs are challenged MANETs and are characterized by opportunistic connectivity i.e. data is forwarded opportunistically whenever contact between the nodes take place. ONs are one of the types of Delay Tolerant Networks (DTNs) that are highlighted with long latency and unstable network topology. Due to the node mobility, these networks may have unexpectedly intermittent connections. So these networks are also called by intermittently connected networks.

ONs are one of the interesting advancements of MANETs where the nodes do not have any prior knowledge about the structure of the network. In such scenario routes are built dynamically and the communication between mobile nodes are enabled via the intermediate nodes called as relay nodes. When no forwarding opportunity exists within its transmission

S. V. Saboii<sup>2</sup> Department of Computer Science and Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587103, India saboji skumar@yahoo.com

range intermediate node captures the message and waits for the upcoming contacts to forward the information. ONs are also called as Packet Switched Networks (PSN) i.e. networks formed by people carrying mobile devices [1]. Fig. 1 how routing in ONs take place to illustrates an example transfer the messages between disconnected nodes. We referred node B as source and node C as destination. In fig.1 (a) node B initially carries the message and fig. 1(b) node B relays the message to node A since it is in node's B proximity. Then message is delivered to destination node C when node A encounters node C in fig.1(c).



(c) Finally message (a) Initially B carries message (b)B relays message to A is delivered to C

#### Fig. 1 Routing in opportunistic network

One of the important and compelling challenges in opportunistic networks is to find routes from sourcedestination, because of the lack of direct connectivity between the node pairs for transferring of message. The routing protocols in ONs are classified into infrastructure-less protocols and infrastructure based protocols [2]. The routing protocols should provide maximum delivery ratio, minimum overhead ratio, minimum cost, minimum delivery delay, maximum route quality, maximum node popularity etc. In this paper we review the recent progress in opportunistic networks routing/forwarding protocols. The organization of the paper is as follows. The challenges of Routing in opportunistic networks are described in Section 2. Section 3 reviews the related work. In section 4 we compared the performance parameters of reviewed protocols and section 5 describes few applications of opportunistic networks. Conclusion is depicted in section 6.

## LPC Based Speech Recognition for Kannada Vowels

Anand H. Unnibhavi Dept. of E & C Basaveshwara Engineering College, Bagalkot, Karnataka, India anandhu.rampur@gmail.com

Abstract— Automatic speech recognition (ASR) is most promising area which is of special attention today. This paper proposes speech recognition system for Kannada vowels. The proposed framework consists of preprocessing unit and classification unit. The preprocessing unit segments the speech signal into proper frames and extracts the features using Linear Predictive Coding (LPC). The Kannada vowels are classified into the proper classes in the classification unit where Euclidean distance is used. The recognition accuracy obtained is about 40 %.

Keywords— Linear Predictive Coding; Mel frequency cepstral coefficients; Isolated Word Recognition

#### I. INTRODUCTION

Speech recognition is used to translate words spoken by humans to a machine understandable format. The recognition process is still a challenging research area because existing methods are not fast and precise enough compared to human recognition abilities. Keyboard acts as a barrier between computer and the user which is true in rural India. This barrier can be overcome by speech recognition system. There is a difficulty in implementing ASR because different regions peoples of Karnataka have different speaking style, accents, and different voice patterns. Along with these challenges, the presence of background noises, changing of signal properties over time and differences between the training and testing environments also pose major problems in speech recognition [1]. The speech signals are continuously changing in nature, which also affects the speech recognition process. In case of isolated word recognition, signals are recorded in intervals. The signal is preceded by silence and followed by silence therefore the speech segment need to be separated from non speech segment i.e. silence region because this requires storage space and increased computation time. This paper proposes an approach for identifying isolated words corresponding to Kannada vowels. The isolated speech recognition system is divided into three parts. The first part is to develop the database of spoken words. The second part deals with the extraction of features. Finally the third part helps in classification of the spoken words. Literature survey, proposed work, analysis of results and conclusions and outlines are discussed in the following sections.

#### II. LITERATURE SURVEY

The wavelet packet based Kannada speech recognition is discussed in paper [2]. Wavelet Packet Decomposition (WPD)

D S Jangamshetti Dept. of E & E Basaveshwara Engineering College, Bagalkot, Karnataka, India adsj1869@rediffmail.com

is a speech compression technique. This method yields four ets of WPD coefficients. A Mel frequency cepstral coefficient (MFCC) is the feature extractor that was used to compute the coefficients from WPD coefficients and Euclidian distance was used to classify the speech. 90 % of accuracy was achieved using Daubechies8, 5-level decomposition technique. Paper [3] presents an Isolated Word Recognition (IWR) system for identification of spoken words. MFCC was used as feature extractor and SVM was used as a classifier. This developed system was tested and evaluated with a performance of 79% accuracy.

In paper [4], a statistical technique was used to remove the silence from the speech signal. This method was used to identify the minimum speech patterns that are required while creating the training set of the speech samples. MFCC was used as a feature extractor and Vector Quantization (VQ) was used as classifier. The recognition error rate was decreased from 2.59 to 1.56 in the case of VQ1 clustering algorithm and 2.5 to 1.45 for VQ2 algorithm. Also the speaker dependent error recognition rate was decreased from 2.5 to 1.45.

A Kannada isolated digit recognition system using MFCC as feature vector was discussed in paper [5]. The system was designed to recognize isolated Kannada numbers. Hidden Markov Model (HMM) was used as recognizer. An average efficiency of 90 % was achieved. Sarika Hegde, et. al. have designed speech recognition system as Multiple Classifier System (MCS). MFCC is used as feature extractor and HMM and Support Vector Machine (SVM) classifiers are used where both of which are trained separately with the same training dataset [6]. While testing, each of these classifier tests the data in parallel. The combined classifier takes the final decision which also combines the technique of Dynamic Time Warping (DTW). An automatic phoneme recognition system based on Gaussian Mixture Modeling (GMM) which is a static modeling scheme is presented in paper [7]. 15 Kannada phonemes were used to train and test the model. Gaussian multivariate model was used to recognize an unknown phoneme from the testing database from the given MFCC for each class by calculating the mean and covariance matrices of corresponding phoneme sequences.

## A Survey of Speech Recognition on South Indian Languages

Anand H.Unnibhavi Dept. of Electronics and Communication Engineering Basaveshwar Engineering College Bagalkot, Karnataka, India Anandhu.rampur@gmail.com

Abstract— Automatic Speech Recognition is an active field of research to identify speech patterns for providing the equivalent text. Many types of interactive software applications are available and the uses of these applications are limited due to language barriers. Therefore development of speech recognition systems in local languages will help anyone to make use of this technological advancement. This paper presents a brief survey of Automatic Speech Recognition System of south Indian Languages and discusses the advances made in the recent years of research and compares some of the well known methods used in various stages of speech recognition system of south Indian languages.

Keywords— Speech Recognition; Feature Extraction; MFCC; Support Vector Machine; RBF kernel; Hidden Markov Model; Neural Network; Dynamic Time Warping.

#### I. INTRODUCTION

Speech recognition is the translation of spoken words into text. It is also known as Automatic Speech Recognition (ASR), computer speech recognition or just Speech To Text (STT). Speech recognition system performs two fundamental operations: signal modeling and pattern matching. Signal modeling represents a process of converting voice signal into a set of parameters. The signal modeling involves four basic operations: Spectral shaping, feature extraction, parametric transformation, and statistical modeling. Spectral shaping is the process of converting the voice signal to a digital signal and emphasizing important frequency components in the signal. Feature extraction is a process of obtaining different features such as power, pitch, and vocal tract configuration from the speech signal. Parameter transformation is the process of converting these features into signal parameters. Statistical modeling involves conversion of parameters to observation vectors [1, 2]. The pattern matching approach involves two essential steps namely, pattern training (this is the method by which representative sound patterns are converted into reference patterns for use by the pattern matching algorithm) and pattern comparison. The essential feature is that it uses a well-formulated mathematical framework and establishes consistent speech pattern representations for reliable pattern comparison from a set of labeled training samples via a formal training algorithm. A speech pattern representation can be in the form of a speech template or a statistical model, and can be applied to a sound, a word or a phrase. In the pattern comparison stage of the

D.S. Jangamshetti Dept. of Electronics and Communication Engineering Basaveshwar Engineering College Bagalkot, Karnataka, India dsj1869@rediffmail.com

approach, a direct comparison is made between the unknown speech patterns with each possible pattern learnt in the training stage, in order to determine the identity of the unknown according to the goodness of match of the patterns.



Fig.1: Basic Block diagram of speech recognition

Figure 1 shows different stages of Automatic Speech Recognition System. The microphone will pick up the analog signal and convert it into electrical signal and is passed to a processing unit called as signal analyzer (feature extractor), which processes the speech signal. The acoustic model compares the phonemes with the previous results from the trained models and highest probability match of a phoneme is selected as being the correct phoneme. Language model consist of two parts Dictionary file and Grammar file. Dictionary file matches the phoneme patterns to particular words and decides the correct pattern for the user input. Grammar file is made up of many templates which map the process of conversation that is made of limited number of paths which allows the Language Model to match a sentence to the user's response i.e Text output [3, 4].

#### A. Types of speech recognition

Speech recognition systems can be classified as:

#### Isolated word recognition

Isolated word recognizers usually require each utterance to have quiet on both sides of the sample window. It doesn't mean that it accepts single words, but does require a single utterance at a time. This is fine for situations where the user is required to give only one word responses or commands, but is very unnatural for multiple word inputs. It is comparatively simple and easiest to implement because word boundaries are obvious and the words tend to be clearly pronounced which is the major advantage of this type. The disadvantage of this type is choosing different boundaries affects the results.

#### • Connected word recognition

In connected word systems separate utterances are 'runtogether' with a small pause between them.

#### 978-1-5090-4620-1/16/\$31.00 ©2016 IEEE

## BDI Agent based Dynamic Routing Scheme for Vehicle-to-Vehicle Communication in VANETs

Mamata J. Sataraddi, Mahabaleshwar S. Kakkasageri Electronics and Communication Engineering Department Basaveshwar Engineering College (Autonomous) Bagalkot-587102, Karnataka, INDIA kdmamata@rediffmail.com, mahabalesh\_sk@yahoo.co.in

Abstract—Design of routing protocols to send message for safety applications in VANET is an essential and required issue to support smart and intelligent transportation system. This paper aims to design and develop a dynamic routing scheme for Vehicleto-Vehicle (V2V) communication in VANET under the constraints such as dynamic topology, unpredictable vehicle density, link failure etc., using Belief-Desire-Intention (BDI) model of cognitive agent. The proposed dynamic routing scheme works as follows: (1) collecting the information such as end-to-end delay and available bandwidth; (2) generation of beliefs based on the collected information; (3) develop the desire based on the belief; (4) select the route if desire is achieved and intention is executed; and (5) reselect the route if intention is not executed. Proposed routing scheme is analyzed in-terms of network life time and packet delivery ratio.

100

Index Terms-Vehicular Ad hoc Network (VANET); Cognitive agent; BDI model

#### I. INTRODUCTION

Now a days because of more demand for safety in vehicles irrespective of the cost much of research is going on Vehicular Ad hoc Networks (VANETs). Basically VANETs are distributed self organizing, self-healing communication network formed by vehicles. It involves communication among vehicles, vehicles and Road Side Units (RSU) and sometimes among Wireless Access in Vehicular Environment (WAVE) devices.

With respect to Media Access Control (MAC), several VANET issues arises which includes information management, routing, network management, congestion and collision control, maximum message dissemination, privacy and security [1]. Nowadays because of increasing in the demand for safety in vehicles irrespective of the cost much of research is going on VANET. To do communication between vehicles, routing takes an important role which can be defined as selection of best path for a message to reach its destination. An agent is software which functions under self governance in a particular environment. The Belief-Desire-Intention (BDI) model is considered as the execution of an agent's belief, desire, and intention to find a solution for a particular issue in agent programming.

Many research works have been proposed for VANET routing mechanisms. Few ongoing related research works are given as follows. In [2], an outline of the VANET technology, VANET standards and issues, with some suggestions for the most necessary applications is proposed along with the future research. Different existing routing protocols with their merits and demerits for Vehicle-to-Vehicle (V2V) and Vehicle-to-Infrastructure (V2I) communication is proposed in [3]. Application of cellular networks to present a novel communication for vehicles is presented in [4]. It combines both V2V and V2I into one system to propagate the message among vehicles (V2V) and between the car and the road side infrastructure (V2I).

12

In [5], design and evaluation of DSR (Dynamic Source Routing) through simulation for different communication environments is presented. A fast path recovery scheme for V2V communication using AODV (Ad-hoc On Demand Distance Vector) protocol is proposed in [6], where basic AODV is modified whenever there is a link failure occurs. At the time of link failure, this protocol is used to find the direction of destination vehicle and broadcasts it to its neighbors to find the new path.

A dynamic Routing-Tree based Data Forwarding (RTDF) algorithm that focuses with a static hotspot on road networks is presented in paper [7]. When vehicle receives more number of inquiries at a time the hotspot will transmit messages to build a path-tree for the delivery of data. In [8] an efficient multi hop reliable broadcast routing algorithm for VANETs based on mobility prediction and relative velocity is presented. Because of high mobility there is frequent changes in network topology of VANET. Hence in [9] a Reliability model is applied to the Ad-hoc On Demand Distance Vector (AODV-R) is proposed. This scheme is based on link between vehicles which is calculated by means of the position, direction and speed of vehicles along the path.

A survey of vehicular communication systems, with all technical details to develop projects in industry and academia is given in [10]. In [11] a trusted geographic information routing protocol (TGPSR) is proposed which provides protection against malicious nodes. The work given in [12] explains current research issues on information management techniques considering four network architectures of VANET as V2V, V2I, Vehicle-to-Pedestrian (V2P) and hybrid employed for applications related to safety in VANETs, which covers gathering, validation, aggregation and dissemination. To establish the communication between vehicles using fish swarm algorithm for hybrid communication is proposed in [13].

This full-text paper was peer-reviewed and accepted to be presented at the IEEE WiSPNET 2017 conference.

## BDI Agent based Routing Scheme in VANETs

Mahabaleshwar S. Kakkasageri\*, Mamata J. Sataraddi\*, Poornima M. Chanal\*, Gururaj S. Kori\*\* \*ECE Dept., Basaveshwar Engineering College (Autonomous) Bagalkot – 587103, Karnataka, India

12

mahabalesh\_sk@yahoo.co.in, kdmamata@rediffmail.com, poornima.chanal@yahoo.com \*\*ECE Dept., Biluru Gurubasava Mahaswamiji Institute of Technology

surve of reemiology

Mudhol-587313, Karnataka, India korigurus@yahoo.com

Abstract - Vehicular Ad hoc Networks (VANETs) are a class of wireless networks built dynamically in an ad hoc network manner. VANET is a key technology for smart and Intelligent Transportation System. Efficient and intelligent algorithms for routing in VANETs are critically required for safety information propagation. Routing in highly dynamic VANET topology is important for the achievement of any vehicular safety applications. The routing mechanism in vehicular networks is critical because of VANETconstraints such 28 unpredictable network topology, frequent disconnections, varying network densities, etc. Multi agent scheme based routing scheme for VANETs is addressed in the paper. We address the challenge of how to route the data with short communication delay, overhead, and the complexity. Specifically, we consider the scenario for (V2V) vehicle-tovehicle communication. Multi agent scheme consists of static agent and mobile agents. The scheme operates as follows: (1) establishing a connectivity pattern between the vehicles; (2) creating a set of beliefs; (3) desire development; (4) finalizing the intention. We validate the proposed scheme by simulation. Evaluated simulation results are packet delivery ratio, bandwidth utilization, and route identification delay.

Index Terms- VANETs, V2V, Routing, Multi-Agent System

#### I. INTRODUCTION

The VANETs are best example for (MANETs) mobile ad hoc network. Modern vehicles with capability of sophisticated computing and communication are a good sign for Vehicular Adhoc Networks (VANETs). VANETs communicate with IEEE 802.11p, were communication is conceivable between vehicles and road side infrastructure. Design of systematically organised VANET environment improves the motorway safety and optimizes the traffic conditions. Auto collision happens because of heavy traffic on the road. These accidents and other incidents on the roads occur due to absence of safety and other necessary services information. Overcrowding of vehicles results in heavy delay of information for the travellers and creates harmful to the environment on the roads. Valid safety data related issues are given most noteworthy priority in VANETs data aggregation, validation and dissemination. In VANETs, efficient routing protocols should address the issues such as unpredictable topology, density of vehicles, mobility, heterogeneous communication environment, etc. are discussed in papers [1]-[7].

Execution assessment of the distinctive vehicular framework circumstances is appeared in [8] [9]. Both topologies based routing algorithms and position-based routing are assessed in urban and city situations. Dynamic security and smart transportation are essential applications of VANET are presented in [10]. The paper additionally gives a point by point overview of routing mechanisms and mobility prototypes which are reasonable for efficient V2V and V2RSU communication.

The results of the paper [11] show that multicast routing algorithms, including geocast and mobility based, are more convincible than other routing algorithms. In paper [12] the researchers showed the recent taxonomy of the routing protocols for VANETs. The paper [13] gives a review of VANETs plus opens current challenges in routing issues, primarily paper focus on V2V communication. Finally author summarise the strength and limitations of different classes in routing protocols.

The investigation in paper [14] is predominantly on the review of new outcomes of the routing schemes in VANETs. The difficulties and viewpoints of routing protocols are finally discussed by author. The author proposes multi agent based approach for intelligent information propagation in VANETs. The paper [15] explained information dissemination using software agents in VANET.

Aggregation of critical information and minimization of redundant data dissemination can efficiently done by regression based cognitive agent approach is presented in [16]. DSDV protocols performance can be improved by using agent based approach is basically discussed in [17].

Paper [18] investigates that the conventional and customary routing algorithms of MANETs and proved that efficiency is less in VANETs due to VANETs architecture and environment. Multi agent system approach improves the efficiency of the routing protocol. A vehicular information ad-hoc network that consists of three tier network architecture using multi agent system offers adaptability, versatility and viability for movement of data in VANETs and also supports hearty and deft system administration [19].

A framework for keen learning and choice making techniques to build smart, intelligent and centralized traffic management model is proposed and discussed in [20], further paper shows a cognitive radio enabled VANET for multi-agent based ITS. The scheme demonstrates two model for data trade, one is dynamic (V2V) and another is semi-dynamic (V2RSU).

Static and mobile software agents to address the traffic problem in city and urban areas are discussed in [21]. Agent based approach results in best QoS in VANET and also provides the safety information to the drivers for comfort driving in accident prone areas.

A software agent is a software entity to function autonomously in a continuous inputs environment. Belief-Desire-Intention agent architecture represents agent three key mental structures: enlightening states of mind around the world (beliefs or convictions), motivational approaches

#### 2017 IEEE 7th International Advance Computing Conference

## Intelligent Routing for Hybrid Communication in VANETs

#### Mamata J. Sataraddi, Mahabaleshwar S. Kakkasageri

Department of Electronics and Communication Engineering Basaveshwar Engineering College (Autonomous), Bagalkot, Karnataka, India. kdmamata@rediffmail.com, mahabalesh\_sk@yahoo.co.in

#### Gururaj S. Kori –

Department of Electronics and Communication Engineering Biluru Gurubasava Mahaswamiji Institute Technology, Mudhol, Karnataka, India. korigurus@yahoo.com

#### Rajeshwari V. Patil

Department of Information Technology JSS International School, Dubai, United Arab Emirates rajveer.gvp@gmail.com

Abstract- Vehicular Ad hoc Network (VANET) is an emerging technology with some unique characteristics that makes it different from other ad-hoc networks. VANETs could be a key networking technology of the future vehicle communications, where the communication is possible between V2V, V2I and other hybrid architectures. One of the major issues in VANET is routing due to high mobility and dynamic topology. Since the routing protocol decides how better the communication occurs between the vehicles, the design of routing protocol is very important. We bring the current research efforts on the intelligent routing scheme for VANET. In this research paper, we determine the position of vehicles using GPS, in both city and highway scenarios and establish the communication links between vehicles based on fish swarm algorithm. Our simulation results shows that the proposed intelligent routing algorithm performs better in terms of bandwidth utilized, packet delivery ratio and end-to-end delay etc.

Index Terms-Vehicular Ad hoc Network, Intelligent Routing, Fish Swarm Optimization and Hybrid Communication

#### I. INTRODUCTION

The main task of the VANET is to receive and transmit information from all the neighboring vehicles in case of emergency situation such as heavy traffic jams, accidents and during climatic disasters. The main aim of our work is to design an intelligent routing scheme between two vehicles. This work is based on the scenario, where vehicles move at a different velocity along two different or same directions on the same or two different lanes road. Simulation experiments are performed to evaluate the performance of our proposed scheme in comparison with the other existing routing protocols in VANETs.

In paper [1], author has discussed the main characteristics and the research challenge of routing in VANETs, which may be considered in designing various routing protocols, they surveyed and compared symbolized instances for all the classes of protocols. The paper [2] provides an overview on current research state, challenges, and potentials of VANETs as well as the ways forward to achieving the long awaited Intelligent Transportation System (ITS).

In VANETs, to have smart ITS, survey of different routing algorithms in different scenario is an essential and important issue. From the survey done it was seen that the existing routing protocols are not effective to meet all traffic scenarios. Thus the authors in [3] presented the design of an efficient routing protocol. The advantages and disadvantages of present routing protocols are also given in this paper which can be considered for further advancement of any new routing protocol.

Authors proposed a new vehicular reliability model to find the reliable route in VANETs. The reliable path is calculated using the location, direction and velocity information of vehicles along the road in [4].

Paper [5], presents an enhanced version of the well-known Dynamic Source Routing (DSR) scheme based on the Ant Colony Optimization (ACO) algorithm, which can produce a high data packet delivery ratio in low end to end delay with low routing overhead and low energy consumption.

A novel routing scheme, Anchor-Geography based routing protocol (AGP), designed specifically for VANET communication in city environment is proposed in [6], reactive broadcasting is used for both getting destination location and routing discovery and they show that AGP protocol gains improvement in packet delivery ratio and average hops.

For VANET routing a Hybrid Bee swarm Routing (HyBR) algorithm is presented in [7]. HyBR considers the rapid changes in environment which is the key feature of VANETs in real time for continuous learning paradigm. The features of geographic routing and topology routing are combined in this algorithm. HyBR considers both urban and rural road scenario

978-1-5090-1560-3/17 \$31.00 © 2017 IEEE DOI 10.1109/IACC.2017.79 This full-text paper was peer-reviewed and accepted to be presented at the IEEE WiSPNET 2017 conference.

## Frequency Compression of Speech for Improving Speech Perception in Sensorineural Hearing Loss: FBS Approach

Rajani S. Pujar<sup>1</sup> and P. N. Kulkarni<sup>2</sup>

Electronics and Communication Engineering Department, Basaveshwar Engineering College (Autonomous), Bagalkot-587103, Karnataka, India

Email: <sup>1</sup>rajani\_pujar@rediffmail.com <sup>2</sup>pnk\_bewoor@yahoo.com

Abstract-Filter-bank summation (FBS) method is a commonly used technique for multi-band processing of speech and audic signals, especially in digital hearing aids. In multi-band speech processing techniques, filter bank summation provides the convenient way of processing the auditory information present in different bands based on their perceptual significance. People with hearing loss have issue in perception of speech due to widening of the auditory filter bandwidth leading to increased frequency masking. Previous studies have shown that spectral splitting of speech signal for binaural dichotic presentation helps to reduce the effect of frequency masking. Also studies showed that using multiband frequency compression it is possible to compensate the effect of widened auditory filters. This paper presents a filter bank summation method to perform dichotic spectral splitting of input speech signal followed by frequency compression to enhance speech perception for hearing impaired. In the present study, the speech signal is split into eighteen frequency bands ranging from 0-5000 Hz based on auditory critical bandwidths and frequency samples of every band compressed in the direction of center of each band using spectral segment frequency mapping technique. Performance of the algorithm was evaluated using MOS test for subjective assessment of speech quality and Perceptual Evaluation of Speech Quality (PESQ) scores for objective assessment of speech quality. The results showed a significant improvement in speech quality as indicated by MOS and PESQ scores for the SNR values in the range of -6 dB to 6 dB.

\*\*

33

Index Terms—Sensorineural Hearing Loss (SNHL); Spectral Masking; Filter Bank Summation Method; Dichotic spectral splitting; Frequency Compression.

#### I. INTRODUCTION

Sensorineural hearing loss (SNHL) occurs when the functioning of the cochlea is affected or when there is dysfunction of the auditory nerve or higher centres in the auditory pathway. The bandwidth of auditory filter normally widened and selection of frequency gets reduced due to masking of spectral components [1], [2]. So SNHL persons have problem in understanding the speech due to broader auditory filters, leading to smoother representation of speech spectrum [3]. This causes poor perception of speech by hearing impaired people.

Hearing loss commonly occurs at higher frequencies. Hearing aid technology provides frequency-dependent amplification for hearing-impaired (HI) persons. Existing hearing devices have a constrained capacity to give adequate gain for high-frequency sounds [4]. This confines the perceptibility of significant sounds, particularly for persons with critical hearing impairment. High frequency components of the speech signal gives essential linguistic information (especially for fricatives) to the listeners.

In frequency transposition [5] method and slow playback high-frequency speech samples shifts to the regions of low frequency by a constant factor These traditional methods of transposition of frequency [6-9] were helpful in recognition of speech, but results in degradation of sound quality. The investigation of the effectiveness of multi-band frequency compression methods using different compression values, to enhance the speech audibility is discussed in [10]. In this method, the spectrum of speech is split into various frequency bands and frequency samples in every band are compressed in the direction of center of band by fixed compression rates. The technique helps in improving the speech quality for hearing impaired people. The work given in [11] reduces the effect of spectral masking using filter bank summation method. A signal is split into 15 frequency sub bands using an analysis filter banks. These 15 frequency bands were designed with sampling frequency of 12 kHz, pass band ripple and stop band attenuation less than 1 dB and 40 dB. These sub band signals were combined by using synthesis filter bank which accomplishes the frequency compression. The problem of masking of spectral components, for listeners with bilateral SNHL, can be reduced with binaural dichotic spectral splitting, using auditory critical bandwidth (ACB) based comb filters [12]. Speech perception can be improved by using a pair of complementary comb filters by adjusting interband crossover gain (between 4 and 6 dB) below pass band gain.

The paper [13] discussed the effect of linear frequency transposition (LFT) and nonlinear frequency compression (NFC) in improving the speech perception for school aged children. Paediatric participants with varying audiometric characteristics such as total distortion of the auditory system, amount of hearing loss, and the usage of hearing aid were considered for evaluating the efficiency of NFC and LFT. The analysis of both the methods was conducted for each subject based on above mentioned levels. The consequences of this examination proposes that LFT and NFC can enhance the perceptibility of high frequency (consonant) sounds and improve understanding

## Routing in Opportunistic Networks: Taxonomy, Survey

S R Bharamagoudar Dept. of Electronics & Communication Engineering Basaveshwar Engineering College (Autonomous) Bagalkot, India srb bec@yahoo.com

Abstract- The accelerated growth of wireless communication such as personal digital assistants, tablets, smart phones, that are carried by the people may arrange themselves in to a special network for the communication of messages. The moving nodes communicate with the nearby nodes using node contact opportunity through Wi-Fi or blue tooth. These nodes suffer from frequent disconnections. The network with frequent disconnections is known as opportunistic network or intermittently connected network. It is a unique wireless network architecture which enables the mobile nodes to have communication with each other in environment where there is no continuous route exists between source and destination .Such networks espouse store-carry-forward habit for the message transfer and utilize mobility of moving nodes for message communication. These networks are featured with network topology which is unstable and also long latency. Routing requires the nearby neighborhood nodes information for the message transfer. Depending on the routing action, we categorize routing schemes as social aware information schemes and social zero information schemes. The schemes such as social zero information do not require any past information of the nodes behavior. Subject to the content of information, social aware schemes are furthermore categorized as microscopic and macroscopic schemes. Microscopic schemes utilize mobile user's movement. Whereas macroscopic schemes have integrated social relationship with the mobile user's movement. We explicate about the routing solutions for the microscopic and macroscopic schemes. Additionally we briefly discussed evaluations in routing schemes.

Keywords— Opportunistic Networks, Intermittently connected network, Store-carry-forward.

#### I. INTRODUCTION

In the present networking world, wireless communication plays a predominant role. The accelerated growth of mobile devices like laptops, smart phones, notebooks etc makeup mobile communication area a more fascinating research in recent years. Another extension of the mobile has inclined to the Mobile Ad-hoc Network (MANET) model known as Opportunistic Networks (ONs). These networks make use of nodes mobility characteristic, to enter the communication range of each other. This helps to establish communication path between the nodes to transfer the information. ONs suffer from unstable network topology as well as prolonged latency because of frequent disconnections [1]. Such networks suffer S V Saboji Department of Computer Science & Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587103, India saboji\_skumar@yahoo.com

from abruptly intermittent connections because of moving nodes. These networks are also entitled by intermittently connected networks. Nodes suffer from the initial knowledge of the structure of the network because of random movement of the nodes. So the routes are created dynamically and relay nodes or intermediate nodes take active part in the transferring of messages to respective destinations. If the intermediate nodes cannot find any forwarding opportunity within its communication range then, the nodes store the messages and carry the messages until they find next upcoming contacts to forward the messages [2]. Using Wi-Fi, Bluetooth, the built in technologies, nodes communicate with each other. The network is accustomed by mobile devices which are carried by human beings and hence routing is a challenging task. The major concern in ONs is reliable transfer of message to the destination. Hence co-operation between the nodes and storage of messages are the primary requirement [3].



Fig.1 Message forwarding in opportunistic networks

## **Review of Microscopic and Macroscopic Routing** Algorithms in Opportunistic Networks

#### S. R. Bharamagoudar<sup>1</sup>

Department of Electronics &Communication Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587103, India srb bec@yahoo.com

Abstract- Opportunistic networks are highlighted with long latency and unstable network topology. It is a unique wireless network architecture which enables the mobile nodes to have communication with each other in environment where there is no continuous route between source and destination. In this network scenario there is no continuous routing path exist between source and destination. Using intermediate nodes information is conveyed to the destination. These networks espouse capture move-forward behavior arising from the mobility of the mobile nodes for message relying. In this article we explicate about the challenges of the opportunistic networks, classification of routing protocols, also confer about the metrics required for the performance measurement of routing algorithms. For the message transfer, routing requires information about the neighborhood nodes. Based on routing behavior, we classify routing protocols as social zero information schemes and social aware information schemes. Social aware information schemes are further classified into microscopic schemes and macroscopic schemes depending upon content of the information. In this paper we discuss some of recent microscopic and macroscopic routing protocols and also confer the comparison among the reviewed algorithms.

#### Keywords—Opportunistic Networks, Routing, Mobility pattern

#### I. INTRODUCTION

In our daily life, wireless communication plays an important role. Recent progress in wireless technologies such as 3G, 4G, and Bluetooth, Wi-Fi, another advancement of the Mobile has given rise to Ad-hoc Network (MANET) model called Opportunistic Networks (ONs). ONs are challenged MANETs and are characterized by opportunistic connectivity i.e. data is forwarded opportunistically whenever contact between the nodes take place. ONs are one of the types of Delay Tolerant Networks (DTNs) that are highlighted with long latency and unstable network topology. Due to the node mobility, these networks may have unexpectedly intermittent connections. So these networks are also called by intermittently connected networks.

ONs are one of the interesting advancements of MANETs where the nodes do not have any prior knowledge about the structure of the network. In such scenario routes are built dynamically and the communication between mobile nodes are enabled via the intermediate nodes called as relay nodes. When no forwarding opportunity exists within its transmission

#### S. V. Saboji<sup>2</sup>

Department of Computer Science and Engineering Basaveshwar Engineering College (Autonomous) Bagalkot-587103, India saboji skumar@yahoo.com

range intermediate node captures the message and waits for the upcoming contacts to forward the information. ONs are also called as Packet Switched Networks (PSN) i.e. networks formed by people carrying mobile devices [1]. Fig. 1 illustrates an example how routing in ONs take place to transfer the messages between disconnected nodes. We referred node B as source and node C as destination. In fig.1 (a) node B initially carries the message and fig. 1(b) node B relays the message to node A since it is in node's B proximity. Then message is delivered to destination node C when node A encounters node C in fig.1(c).



(a) Initially B carries message (b)B relays message to A (c) Finally message is delivered to C

#### Fig. 1 Routing in opportunistic network

One of the important and compelling challenges in opportunistic networks is to find routes from sourcedestination, because of the lack of direct connectivity between the node pairs for transferring of message. The routing protocols in ONs are classified into infrastructure-less protocols and infrastructure based protocols [2]. The routing protocols should provide maximum delivery ratio, minimum overhead ratio, minimum cost, minimum delivery delay, maximum route quality, maximum node popularity etc. In this paper we review the recent progress in opportunistic networks routing/forwarding protocols. The organization of the paper is as follows. The challenges of Routing in opportunistic networks are described in Section 2. Section 3 reviews the related work. In section 4 we compared the performance parameters of reviewed protocols and section 5 describes few applications of opportunistic networks. Conclusion is depicted in section 6.

## Implementation of FPGA Based Closed Loop V/f Speed Control of Induction Motor Employed for Industrial Applications

Kavita B. Hunasikatti, Student member IEEE

Basaveshwar Engineering College (Autonomous), Bagalkot, Karnataka India Raghuram L Naik, <sub>Member IEEE</sub> Basaveshwar Engineering College (Autonomous), Bagalkot, Karnataka India Basayya V Hiremath Member IEEE Basaveshwar Engineering College (Autonomous), Bagalkot, Karnataka India

Abstract- In recent days, Field Programmable Gate Array (FPGA), provides the ability to integrate multiple tasks through the parallel operations. This processor is preferred over DSP for the implementation of controller to control speed of induction motor through power converter employed in industrial applications. This paper presents the implementation of FPGA based closed loop V/F speed control of induction motor. The existing approaches having the limitations of regulating slip and conversion of frequency in proportion to the slip. This issue can be addressed by summing the slip with actual speed during conversion of speed into frequency. Further, this paper focuses on the development of the PI controller using Xilinx block sets to improve the tracking ability of the controller. The developed controller is validated on hardware test bench consisting of FPGA based controller connected to Induction motor drive. Finally, it can be concluded that FPGA processor provides a real deterministic behavior of induction motor employed in industrial applications

Keywords- FPGA, Induction Motor, v/f speed control, wavect, Xilinx

#### I. INTRODUCTION

Energy is the indispensable factor in the development and economic growth of industry and country. The majority of energy consumption in industries is dominated by drive systems such as pumps, fans and compressors. These systems are continuously operated to improve the productivity and quality of products in various industrial applications Viz Process control, automotive control, Robotic and electric vehicle operation[1]. There is a need of efficient controller for variable or adjustable speed drive to optimize the power fed to these systems, which leads to energy conservation and reduction in overall cost of the system. The Induction Motor (IM) accounts 85% of all motors and has dominated over a number of fixed- speed applications because of its reliability and low maintenance operation compared to DC motors.

The V/F controlling strategy is most popular method as the ratio of voltage to frequency is kept constant, thereof flux remains constant. Torque speed characteristic of V/F control implies that starting current is low, the stable operating region of motor is increased and speed range of motor becomes wider [1]. The V/F control strategies are implemented effectively using power converters [2-7].

The various processors like Micro-controllers and DSPs are commonly used to control the speed of induction motors [2,12,14].Considerable research has been done in using DSP processors for digital implementation of V/F control. This method involves various computations like fractional multipliers, high-resolution integer, math with fractional numbers. To obtain this task an additional DSP processor is used, but this complicates the design considerably and also increases the size of the system [14]. In view of this, there is need of shifting to more advanced technologies like FPGA processors.

FPGA processor offers more flexibility in designing of PWM generator for power converters connected to Induction motor [9-11]. FPGA basically having interconnection between logic gates, which combined to form a combinational logic blocks. Customized architecture reduces the execution time of control algorithm. The processor offers direct hardware implementation while maintaining the high control performance systems.

This paper presents the design and implementation of FPGA modified closed loop V/F control of induction motor. The modified approach addresses the issue of limiting the slip value for the conversion of new frequency value at the inverter. Further the contribution of this paper is to implement PI controller using Xilinx block sets. The purpose of this controller is to improve the tracking ability and instantaneous response of the controller during the change in mechanical load on the induction motor. The entire Xilinx model is validated on hardware test bench consisting of FPGA based controller connected to Induction motor drive. Finally, it is observed from the hardware results that, FPGA processor provides a real deterministic behavior of induction motor employed for industrial applications.

#### II. CLOSED LOOP V/F SPEED CONTROL

The generalized block diagram of V/F closed loop control of Induction motor drive is shown in Fig.1. The actual speed is compared with reference speed in rad/sec, this frequency is

## Critical Review of Control Strategies for Switched Reluctance Motor Employed in Electric Vehicle

Kavita B. Hunasikatti, Student member IEEE Basaveshwar Engineering College (Autonomous) Bagalkot, Karnataka India

Raghuram L Naik, Member IEEE Basaveshwar Engineering College (Autonomous), Bagalkot, Karnataka India

Veena D Nayak Member IEEE Basaveshwar Engineering College (Autonomous), Bagalkot, Karnataka India

strategies for switched reluctance motor employed in electric vehicle. The survey is carried out on various control strategies namely average torque control, instantaneous torque, torque is not constant in the period of commutation between two sharing and direct instantaneous torque control. The purpose of the survey is to identify the capability of control strategies in terms of speed and torque of SRM for electric vehicle application. The preliminary observation from the study is that, SRM have high torque ripples and it is severe during commutation between the adjacent phases. Further, it also leads to high acoustic noise, vibration and harshness of the electric vehicle system. In view of this, there is need to develop an appropriate control method to overcome the said problems of SRM drive employed in EVs.

Keywords: Switched Reluctance Motor, Electric Vehicle, Average Torque Control, Instantaneous Torque Control, DITC.

#### I. INTRODUCTION

The first electric vehicle (EV) was built in 1834 and these were preferred for road transportation among the wealthy privileged. However, usage of EV was limited due to superior performance of internal combustion engine vehicles (ICEVs). Further, EVs have gained importance during the year 1970s due to energy crises and inefficient oil storage facilities. Traditionally, Electric Drive System (EDS) converts the electrical energy in to mechanical energy which is performed by the electric motors. Selection of electric motor is the critical dynamics [24-27]. part in designing overall EV system.

DC motor, induction, permanent magnet synchronous motors and brushless DC motors. The DC motors have simple control, decoupling of flux and torque but their construction poses maintenance problems [1-2]. Even though squirrel cage induction motor is reliable in terms of its speed regulation under different loading conditions, but it has drawback of poor starting forme due to low rotor resistance and low efficiency at high speed [1-2]. Permanent magnet synchronous (PMS) motors are good for traction application but require a complicated controller to control field flux, magnetic characteristics and speed of the motor [1-3]. Brushless DC motor has high initial cost due to its construction and it requires complicated controllers to determine the rotor position [1].

Recently switched reluctance motor (SRM) are gaining much interest for the operation of electric vehicle due to its simple and rugged construction, ability of extremely highspeed operation and hazard-free operation [4]. These features of SRM make it possess strong competition in EV applications.

Abstract- This paper presents a critical review of control Major shortcoming of SR motor is high torque ripple as compared to conventional AC machines. These torque ripples are due to the discrete excitation of phase windings and torque adjacent phases. In general, there are two approaches to reduce torque ripple, first approach is to modify the hardware design of the motor and second approach is to design various control strategies to control speed and torque [4].

> Hardware design does not guarantee the performance of SRM as it involves time-varying complex non linear speed control system. Design of control strategies to minimize the torque ripple for SRM drive is presented [7-27]. The design of control strategy is mainly based on various parameters like supply voltage, position sensors, turn-on and turn-of angle and current. The optimal control strategy has to be developed for best motoring operation of the EVs. The best motoring operation is considered as high motoring torque, high operating efficiency and low torque ripple [5].Few literatures presents a method that determines the optimal turn-on and turn-off angles to accomplish an acceptable balance between energy efficiency and torque ripple criteria [7-14]. Instantaneous torque control methods are proposed for fast time response and low torque ripple [15-18]. Four types of torque sharing functions are implemented to minimize the torque ripple in SRM drives [19-22]. Several studies presented DITC in which torque is used as the direct control variable in order to achieve fast torque

In view of issues related with EVs, there is need for In recent past several motors were employed in EVs Viz. detailed analysis of control methods to control the speed of SRM drive employed in EVs. An effort is made to study the performance of various control strategies for SRM drive in terms of higher efficiency and low torque ripple. Observation from the survey will help the designers to develop an appropriate control method to overcome the said problems of SRM drive based EVs.

#### II. REVIEW OF CONTROL STRATEGIES

Traditionally, the control strategies are proposed in order to improve the performance of SRM drive for EVs in terms of higher efficiency, constant torque, low torque ripple and ability to operate in wide speed range. In view of this, broad classification of control strategies are presented in this paper Viz Average torque control, Instantaneous torque control, Torque sharing function (TSF) and Direct instantaneous torque control (DITC). A critical analysis of the above said control strategies are discussed in the following sections.



#### D. G. MALLAPUR



Available online at www.sciencedirect.com ScienceDirect

Materials Today: Proceedings 4 (2017) 10921-10926

materialstoday: ROCEEDINGS www.materialstoday.com/proceedings

#### AMMMT 2016 Influence of heat treatment on the volumetric wear rate of Al-25Mg<sub>2</sub>Si2Cu4Mn alloy

#### Neeta S. Mathapatia\*, Virupaxi Auradib and D. G. Mallapurc

<sup>4</sup>Department of Mechanical Engineering, BLDEA's College of Engineering and Technology, Flipsyapura - 584103, Karnataka, India <sup>1</sup>Department of Mechanical Engineering, Stadaganga Institute of Technology, Tanukaro-572103, Karnataka, India arment of Industrial and Production Engineering, Basaveshwar Engineering College (Autonomas), Bagako-57103, Karnataka, India

Abstract

Strength of a material is very important factor in selecting the material or alloy for various applications, with this it should also possess good mechanical as well as tribological properties. In this paper, the influence of the heat treatment, load and sliding velocity on the volumetric wear rate of AA-25Mg2SiZVaMh alloy was evaluated using a pin-on-disc ware testing machine. Dry sliding wear test were conducted for four different loads (1, 2, 3 and 4kg), for four different velocities (1, 2, 3 and 4ms)), It was observed that the volumetric wear rate decreased with increasing pade the volumetric wear rate decreased.

#### © 2017 Published by Elsevier Ltd.

Selection and Peer-review under responsibility of [Conference Committee Members of International Conference on Advanced Materials, Manufacturing, Management and Thermal Sciences-2016]

rds: Heat treatment, sliding velocity, volumetric wear rate and Al25Mg2Si2Cu4Mn alloy

#### 1. Introduction

Potential applications in areas such as automotive, aerospace and sporting goods industries require light weight and strength combination along with good wear and corrosion resistance [1]. Fatigue strength and good impact strength are available due to the addition of alloying elements in the AI-Si alloys [2]. Alloys have a high resistance to corrosion and others possessing significant strength at elevated temperature is invaluable for chemical plants and for the boilers which are working at high temperature and pressure [3]

\* Corresponding author. Tel.:+91 9538 037994 E-mail address:neeta mathpati@rediffmail.com

2214-7853© 2017 Published by Elsevier Ltd.

Selection and Peer-review under responsibility of [Conference Committee Members of International Conference on Advanced Materials, Manufacturing, Management and Thermal Sciences-2016]



#### Available online at www.sciencedirect.com ScienceDirect

Materials Today: Proceedings 4 (2017) 11010-11014

#### materialstoday: ROCEEDING:

www.materialstoday.com/proceedings

#### AMMMT 2016

#### Wear behavioural studies on 3Hr homogenized Al-25Mg2Si2Cu4Ni alloy at constant speed

#### M.D. Harlapur<sup>a\*</sup>, D. G. Mallapur<sup>b</sup>

<sup>4</sup>Department of Mechanical Engineering, Rural Engineering College, Huikoti, 582205, Karnataka, India.
<sup>4</sup>R& D Centre, Department of Industrial and Production Engineering, Bazavezinwar Engineering College, Bagalkot-587103, Karnataka, India.

#### Abstract

In the present work, a study on the wear behaviour of as cast aluminium (Al-25Mg.512Cu-4Ni) alloy which has been homogenized for 3hr with T6 heat treatment is studied. Various parameters like friction force, firction co-efficient, speed and volumetric wear rate and their relationship has been studied. It is observed that the fiction force and the volumetric wear rate are high a lower looks and reduce accordingly with an increase in load. Whereas, the interface temperature is lower at the low loads and increases accordingly with the increase in loads. Results of SEM structure show rounded corners of Mg.Si intermetallic blocks embedded with Ni in it due to the effect of 3hour homogenizing.

© 2017 Elsevier Ltd. All rights reserved

Selection and Peer-review under responsibility of Advanced Materials, Manufacturing, Management and Thermal Science (AMMMT 2016).

Keywords: Volumetric wear rate, coefficient of friction, solutionizing, Interface temperature.

1 Introduction

Wide applications in the field of aerospace, automobile, infrastructure and marine industries for aluminium alloys is due to its light weight, high strength and corrosion resistance properties. Therefore, aluminium alloys contribute in reducing fuel consumption, cost and achieving better design compared to stells. Major components manufactured with Al-Si alloys have applications in the process that requires smooth operation below and at 250°C. Due to incoherency in the lattice structure, between the aluminium solid solution lattice and the precipitates such an But of incontrated in the finite structure, or units in a manufacture of a sile on finite cardinal properties and the presence of a sile on (more than 12%) low wear rate, lower thermal coefficient of expansion lower thermal coefficient of expansion have been observed. Fluidity and machinability are excellent that gives good chip removal rate.

M.D.Harlapur Tel.:+91 9448862675 ;Fax; 08372-289427

2214-7853@ 2017 Elsevier Ltd. All rights reserved

Selection and Peer-review under responsibility of [Cont Manufacturing, Management and Thermal Sciences-2016] ence Committee Members of International Conference on Advanced Materials,

#### Available online at www.sciencedirect.com

ScienceDirect Materials Today: Proceedings 4 (2017) 11023-11027

materialstoday: www.materialstoday.com/proc

#### AMMMT 2016

A comparative study of the tribological properties of as cast and 3Hr T6 treated Al-25Mg2Si2Cu4Fe alloy

D.G. Sondur<sup>a</sup>\*, D.G. Mallapur<sup>b</sup>

"Department of Mechanicai Engineering, Rurai Engineering College, Hulkoti, 582205, Karnataka, India. "Department of Industrial and Production Engineering, Basaveshwar Engineering College, Bagalkot - 587103, Karnataka, India.

#### Abstract

ELSEVIER

Research work has been carried out to ascertain the effects of T6 heat treatment on the A125Mg\_512Cu4Fe alloy and compare the results with their as cast counterpart. It is observed that the precipitation of the intermetalics and their growth with increase in the homogenizing hours leads to improvement in the ware resistance. Blocks of MGS, is intermetalics are found distributed unevelvel in the aluminium matrix shows though and coarse needles of Fe in the matrix in as cast structure. In T6 treated structure, Chinese script structure of ALCu precipitates are also distributed innohmly in the aluminium matrix. SBM structures shows rounded comers Te needles with ALCu precipitates are also distributed innohmly in the aluminium matrix. SBM structures shows rounded comers of MgS3 blocks embedded with Fe in the Let of the interaction with the Fe needles Mg.53 blocks get fragmented into smaller sized pieces with rounded comers. Fe needles get reduced in diameter increasing their speer ratio. Thus Fe gets distributed in the matrix get strengthess the matrix. Therefore, volumetric war net at constant sliding speed increases only slightly and remains clustered for all the sliding speeds with increasing load.

© 2017 Elsevier Ltd. All rights reserved.

Keywords: Volumetric wear rate, Sliding speed, Load,

der responsibility of Advanced Materials, Manufacturing, Management and Thermal Science (AMMMT 2016).

#### 1. Introduction

Highly alloyed 6xxx series alloys have complex intermetallic phases that originate from the casting stage in the form of ingot. Since iron is the compresent impurity having very low solubility in aluminium, iron ich phases are seen in aluminium alloys that adversely affect the ductility and castability. Presence of copper, manganese leads to the formation of  $(Fe_M n.C.u.)$  Skill<sub>1</sub> phase. Mg,Si is the other phase that easily discolves during solutionizing and contribute to the precipitation hardening process. The type of intermetallic compounds formed depends on the

\* D.G. Sondur. Tel.: +919980661963; fax: 08372289427. E-mail address: dgsondur\_tce@yahoo.co.in

2214-7853 © 2017 Elsevier Ltd. All rights reserved. Selection and Peer-review under responsibility of Advanced Materials, Manufacturing, Management and Thermal Science (AMMMT 2016).



Available online at www.sciencedirect.com ScienceDirect Materials Today: Proceedings 4 (2017) 10696-10703

#### materialstoday: Rocffdings

www.materialstoday.com/proceedings

#### AMMMT 2016

Stress and damage tolerance analysis of stiffened panel with passenger door cutout in airframe structure using FEA

Sharaschandra Kusugal<sup>a</sup>, R. S. Kadadevarmath<sup>b</sup> and D.G. Mallapur<sup>c</sup> "Department of Machanical Engineering, Gogre Institute of Technology, Belagani- 190008, Karnataka, India. "Department of Industrial Degineering and Management, Saldaganga Institute of Technology, Tunksrus 571108, Karnataka, India. "Department of Industrial and Production Engineering: Glaces, Bagalios-17108, Karnataka, India.

#### Abstract

It is very essential to design the airframe structure with minimum possible weight without compromising on the structural safety. Fuselage has many stress concentration areas. Passenger door cutout region is one among such critical regions. Current work includes stress and damage tolerance analysis of a critical region in the fuselage structure. The structure considered for the analysis includes a striffered panel with a passenger door cutout and surrounding area with the presence of stiffering members. The present work also includes the dumage tolerance analysis of the panel in presence of a crack examating from the high stress jocation. The pressurfation load case is one of the critical load cases considered for the design of the aurcraft survature. The tensile loads at the edge of the panel corresponding to pressuriation will be considered for the insura struc-analysis of the panel. A location of maximum tensile stress and its magnitude will be disemified from the stress analysis, stress intensity factor calculation is carried our for different incremental crack lengths for a crack emanating from the high stress location. Modified virtual crack closure integral methods is used for substructure classified on a considered for the insure intensity factor calculation is carried our for different incremental crack lengths for a crack emanating from the high stress location. Modified virtual crack

© 2017 Elsevier Ltd. All rights reserved. Selection and Peer-review under responsibility of Advanced Materials, Manufacturing, Management and Thermal Science

Keywords: Fuselage: damage tolerance: pressurization: stress intensity factor: MVCCL

#### 1. Introduction

Safety and weight are two major considerations in aircraft design. There should not be any compromise between safety and weight. During its life cycle it will be subjected to many load cases such as drag force, lift force, thrust force, pressurization load etc. The load-bearing members of these main sections, those subjected to major forces are called the airframe [1] and [12].

\*Sharaschandra Kusugal. Tel.: +918105254566; *E-mail address:* sharachandrask@git.edu

2214-7853 © 2017 Elsevier Ltd. All rights reserved. Selection and Peer-review under responsibility of Advanced Materials, Manufacturing, Management and Thermal Science (AMMMT 2016).





Dr. Dayanand G.	Wear and m alloy at cons	echanical property studies on ascast and 3H forged homogenized Al25Mg2Si2Cu4Ni stant speed	Dr. Dayanand G.	Wear behavioral studies of as cast and 7H forged homogenized Al25Mg2Si2Cu4Ni alloy at constant load based on Taguchi method			
манариг	Authors	MD Harlapur, DG Mallapur	Mallapur	Authors	MD Harlpur, DG Mallapur		
	Publication date	2018/1/1		Publication date	2018/1/1		
	Journal	Materials Today: Proceedings		Journal	Materials Today: Proceedings		
	Volume	5		Volume	5		
	Issue	9		Issue	9		
	Pages	17989-17995		Pages	17820-17827		
	Publisher	Elsevier		Publisher	Elsevier		
	Description	In present study, AI25Mg <sub>2</sub> SI2Cu4Ni alloy is experimental used for investigating wear (triction coefficient) and mechanical properties in ascast and 3h homogenized forged conditions at constant speed. The sliding wear behavioral tests were carried out on a pin on disc apparatus for mechanical property studies universal testing machine was used. The hardness tests were carried out on Brinell hardness tester. To analyze worn out surfaces of as cast as well as forged samples. Scanning Electron Microscope was used. Wear and mechanical properties study of 3r homogenized forged AI25Mg <sub>2</sub> Si2Cu4Ni aloy samples reveal better tensile strength, yield stress, compression strength and coefficient of friction as compared to the as cast ones, due to formation of secondary precipitate in the matrix		Description	In the current research work, an experimental study of wear behavior of aluminium (Al25Mg <sub>2</sub> Si2Cu4Ni) alloy in as cast and 7hr honogenized forged at constant load is carried out. The sliding wear tests were carried out on a pin on disc apparatus. To evaluate the data on the friction force Taguchi method based on L-16 orthogonal array was employed. Objective of smaller the better and mean of means results are used for signal to noise ratio. For correlation general regression model was used. Confirmation test were carried out between the experimental results forcescen from the mention correlation. Load has maximum contribution on the friction force compared to speed as per the mathematical model. The worn out wear surfaces are analyzed using scanning electron microscope. Wear results of 7 hr homogenized forged sample are lower compared to as cast samples. Pooled error is less than 0.005%. Anova		
	Scholar articles	Wear and mechanical property studies on ascast and 3H forged homogenized ADS/MgIS/2Cu-INi alloy at constant speed MD Harlapur, DG Mallapur - Materials Today: Proceedings, 2018 Related articles		Scholar articles	Wear behavioral studies of as cast and 7H forged homogenized AI25Mg2Si2Cu4Ni alloy at constant load based on Taguchi method MD Hartpur, DG Mallapur - Materials Today: Proceedings, 2018 Related articles		





# Metadata of the chapter that will be visualized in SpringerLink

Book Title	Advances in Manufact	uring and Industrial Engineering			
Series Title					
Chapter Title	Theoretical and Statistical Analysis of Inventory and Warehouse Management in Supply Chain Management—A Case Study on Small-Scale Industries				
Copyright Year	2021				
Copyright HolderName	The Editor(s) (if applicable) and The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd.				
Author	Family Name	Latte			
	Particle				
	Given Name	Mahesh R.			
	Prefix				
	Suffix				
	Role				
	Division				
	Organization	K.L.E. College of Engineering and Technology			
	Address	Chikodi, Belgavi, India			
	Email	latthemahesh@gmail.com			
Corresponding Author	Family Name	Javalagi			
	Particle				
	Given Name	Channappa M.			
	Prefix				
	Suffix				
	Role				
	Division				
	Organization	Basaveshwar Engineering College			
	Address	Bagalkot, India			
	Email	cmjavalagi@gmail.com			
Abstract	The main function of the supply chain is to satisfy the customer by making use of all the parties such as the manufacturers and suppliers, transporters, warehouses, retailers, and even customer themselves. Within small-scale organization, the supply chain includes all those functions which are involved in receiving and filling a customer request. These functions include, but are not limited to, desired new product development, innovative and cost-effective marketing, systematic operations, well-planned distribution, wisely managed finance, and dedicated customer service. The present case study is the analysis of supply chain management (SCM) in small-scale functional industries, located at two industrial estates, namely Gokul Shirgaon MIDC and Shiroli MIDC in Kolhapur, Maharashtra. These operational units were divided majorly into two parts, manufacturing and logistics, and the study is mainly focused on theoretical & statistical analysis of Inventory management & Warehouse management of small-scale industries. For an effective supply chain model, every company has adopted its own supply chain strategy on the basis of their needs, capacity, and available resources. So, to analyse the SCM in small-scale industry, the relevant information from these industries was collected both from primary and secondary sources. And the relevant information is statistically analysed and verified statistically by using Minitab software, and conclusion was drawn.				
Keywords (separated by '-')	Inventory management	t - Warehouse management			

## Theoretical and Statistical Analysis of Inventory and Warehouse Management in Supply Chain Management—A Case Study on Small-Scale Industries



#### Mahesh R. Latte and Channappa M. Javalagi

**Abstract** The main function of the supply chain is to satisfy the customer by making 1 use of all the parties such as the manufacturers and suppliers, transporters, ware-2 houses, retailers, and even customer themselves. Within small-scale organization, the 3 supply chain includes all those functions which are involved in receiving and filling a Δ customer request. These functions include, but are not limited to, desired new product 5 development, innovative and cost-effective marketing, systematic operations, well-6 planned distribution, wisely managed finance, and dedicated customer service. The 7 present case study is the analysis of supply chain management (SCM) in small-8 scale functional industries, located at two industrial estates, namely Gokul Shirgaon 9 MIDC and Shiroli MIDC in Kolhapur, Maharashtra. These operational units were 10 divided majorly into two parts, manufacturing and logistics, and the study is mainly 11 focused on theoretical & statistical analysis of Inventory management & Warehouse 12 management of small-scale industries. For an effective supply chain model, every 13 company has adopted its own supply chain strategy on the basis of their needs, 14 capacity, and available resources. So, to analyse the SCM in small-scale industry, 15 the relevant information from these industries was collected both from primary and 16 secondary sources. And the relevant information is statistically analysed and verified 17 statistically by using Minitab software, and conclusion was drawn. 18

19 Keywords Inventory management · Warehouse management

M. R. Latte

K.L.E. College of Engineering and Technology, Chikodi, Belgavi, India e-mail: latthemahesh@gmail.com

C. M. Javalagi (⊠) Basaveshwar Engineering College, Bagalkot, India e-mail: cmjavalagi@gmail.com

© The Editor(s) (if applicable) and The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2021 R. M. Singari et al. (eds.), *Advances in Manufacturing and Industrial Engineering*, Lecture Notes in Mechanical Engineering, https://doi.org/10.1007/978-981-15-8542-5\_52 1

### The Value Research of Communication and Information Sharing in Supply Chain Management (SCM) for enhancing the Supply Chain Performance (SCP) - A SSI Case Study

Asst. Prof. Mahesh R Latte<sup>1</sup> and Dr. Channappa M Javalagi<sup>2</sup>

<sup>1</sup>K.L.E. College of Engg. & Tech., Chikodi, Belgavi <sup>2</sup>Basaveshwar Engineering College (A), Bagalkot. latthemahesh@gmail.com

**Abstract.** The importance of communication & information sharing is increasing day by day in today's world of digital supply chain management & there are lots of value research is going on to increase the supply chain performance. In this paper an attempt is made to analyse the communication channels in small scale industries (SSI) & information sharing methods & techniques employed by the SSI. Also the work carried out is a value research for enhancing the supply chain performances.

In this paper a survey is made across selected 7 SSI in the Kolhapur MIDC area. The questionnaire is been prepared for the managers/ Owners & Whole-salers/retailers & Customers for the preliminary data for analysis along with that secondary data is also been collected from different sources on digital plat-forms & brain storming sessions with the company owners. For statistical analysis we have used SPSS software for factor analysis & used statistical tools such as mean, standard deviation, percentage, t-test, correlation, multiple regression, one way Anova etc and the conclusion is drawn out of the total value research & also some of the useful suggestions were drawn out for the positive increase in the supply chain performance.

Keywords: Communication, information sharing, supply chain performance.

#### 1 Introduction to importance of Communication & Information Sharing

An information and good communication can help to improve the consumption of the supply chain assets and the synchronization of supply chain flows to increase sensitivity and diminish cost. Even though the sharing of data can help a supply chain more but it is better to satisfy customer needs at lower cost, there is a threat in the assumption that additional data is always better. As extra information is shared across the supply chain, the complicity and expenditure of the both required infrastructure and the follow-up analysis rise exponentially. The trivial value provided by the information share, however, lessens as more and more information become obtainable. It is

## Assessing the potentiality of agricultural residues in extending operational period of bagasse-based co-generation plants in India

Parashuram R. Madar<sup>1</sup> and Suresh T. Dundur<sup>2,\*</sup>

<sup>1</sup>Research Scholar, <sup>2</sup>Professor, Department of Industrial & Production Engineering Basaveshwar Engineering College, Bagalkot -587103, Karnataka, India, \*Corresponding Author Email-Id: <u>sureshdundur@gmail.com</u>

Abstract: Under the signs of extinction of conventional fossil fuels in near future, more emphasis lies now on exploitation of renewable energy sources. The energy production in India from the biomass for the financial year 2017-18 is merely 2.06 percentage of total potential derived from all the renewable sources out of which 0.46 percentage is from cogeneration which is far below their installed capacity. This may be attributed to the fact that cogeneration is pertaining to the utilization of only the bagasse for power production. However, the supply of bagasse is adequate only during crushing season but its non availability for rest of the period leaves those plants inoperative. This void created in the supply of fuel for cogeneration in plants can be filled up through alternative agricultural residues and by managing properly, continued production of power can be ensured. The residues from the nine prime crops are considered for the investigation based on their available energy potential. The power potential of these residues, determined theoretically on the basis ultimate and proximate analyses figures out approximately one million MW of power year on year.

Key words: Agricultural residue, Biomass Energy, Cogeneration, Power Potential

#### 1. INTRODUCTION

Under the signs of extinction of conventional fossil fuels in near future, more emphasis lies now on exploitation of renewable energy sources. The acclaimed reliable sources of renewable energy include solar, wind, biomass and hydro. The estimated energy potential in India from the biomass for the financial year 2017-18 is merely 2.06 percentage of total potential derived from all the renewable sources [1]. In contrast to it, agriculture is a major occupation in India since ages and is capable of producing more and more to meet the ever increasing demand due to swelling population. It also produces large quantity of agricultural residues that are usually considered as waste and the formers burn it in the fields to reduce the cost of cleaning. It is worthwhile estimating the potential of this huge biomass produced in the form of agricultural residue for boosting the productivity in co-generation plants. Though found in abundance, agricultural residues except bagasse were never been considered as the persuasive sources of energy for power generation. Major agricultural residues that are produced in large quantities every year other than bagasse include sugarcane trash, arecanut shell, coconut shell/coir, groundnut shell, cotton stalk, rice/wheat husk, corn cob etc. Proper utilization biomass results in the production of electricity at cheaper rate and benefits the environment by reducing the greenhouse gases (GHGs) in atmosphere due to their lower emissions as compared to conventional fossil fuels [2-3].

India being a major consumer of electricity in South-Asia contributing to 85% of its total production, is still facing sever power crisis due to its own production falling short by 30% of its demand. In filling this gap cogeneration of electricity using biomass is very attractive due to its low capital investment, short gestation period, reduced fuel consumption and associated environmental pollution and increased fuel diversity [ESCAP, 2000] [4]. It is estimated that the sugarcane production in India is expected to touch 38 million tons during kharif season of 2018-19 [5-6]. Never the less the power produced from cogeneration is just amounts to 0.46 percent of the total power produced from the renewable energy sources. The figure indicates that the potentiality of cogeneration in India has not been exploited to its fullest extent. This may be attributed to the fact that cogeneration is pertaining to the utilization of only the bagasse for power production. However, the supply of bagasse is adequate only during crushing season but its non availability for rest of the period leaves those plants inoperative. This void created in the supply of fuel for cogeneration in plants can be filled up through alternative agricultural residues and by managing properly, continued production of power can be ensured.

2. POTENTIALITY OF AGRO-BASED BIOMASS RESOURCES

About 58 % of population in India is depending on agriculture for their income and contribute to 18 % of

## **Routing Protocols in VANET- A Survey**

Anil D. Devangavi anildevangavi\_s@yahoo.co.in Basaveshwar Engineering College, Bagalkot

Abstract--- Intelligent Transportation Systems (ITS) is an integrated approach being developed to exchange relevant information to increase the safety and efficiency of the road transportation systems. Vehicular Ad hoc Network (VANET), a variant of mobile ad hoc networks (MANET), is a core component of ITS. Performance of this smart ITS mainly owes to the design of efficient routing protocols in VANETs. Distinct features of VANETs like unsteady connectivity, high mobility and partitioning of the network have made routing of the information in VANETs difficult and challenging, hence dictating the development of efficient routing protocols. The computation of the best route measures the performance of communication whereas routing protocols takes care of communication & routing of the data. Provision of smart communication, necessitates the analysis of routing protocols in VANET. Accordingly in this paper we have reviewed various types of existing routing protocols in VANET, listing their advantages and disadvantages. Lastly possible inclinations of forthcoming research linked to VANET routing are discussed.

Keywords: VANET, Characteristics, Routing protocols, Perspectives.

#### I. INTRODUCTION

The increasing percentage of people depending upon vehicular transportation has steered issues related to safety and traffic jams. VANET is a variant of MANET in which nodes are replaced by vehicles. VANET is the primary component in ITS that is envisioned towards providing road safety and comfort zone to their users [2]. Apart from safety VANET also provides several value added services like audio/video sharing and other multimedia applications etc.

VANET makes the communication between the vehicle drivers, to evade any acute situation e.g. accidents on roads, roadblocks, speed control, unrestricted way for ambulances and concealed obstacles etc. Vehicles communicate among themselves directly making vehicle to vehicle communication (V2V), else communicate with fixed equipment next to the road (RSU) constituting vehicle to infrastructure communication (V2I) [10]. Through these communication variants vehicles exchange varieties of information. Thus Information propagation is very important in a VANET environment. Routing plays a vibrant role in information propagation. Routing in VANET is categorized based on topology, position, clustering, geocast and broadcasting.

163

978-1-5386-0569-1\$31.00 © 2017 IEEE

Dr. Rajendra Gupta rajendragupta I @yahoo.com AISECT University, Madhya Pradesh

The design of dynamic routing protocol itself is the principal challenge in VANET. Exceptional features of VANET like ever altering topologies, varying speed & density of vehicles, etc. makes routing in VANET very puzzling. The designed routing protocols should deal with the unanticipated and pulsating behavior of VANET to yield optimal performance. Above all, computing and sustaining ideal paths is very complicated task associated with routing in VANET. These arresting features and unfamiliar characteristics have indeed made VANET thrilling.

In this paper we have reviewed various types of existing routing protocols in VANET. The benefits and drawbacks of various routing protocols are also discussed. Rest of the paper is structured as: Unit 1.1 presents the distinct characteristics of VANET. In unit 2 various routing protocols in VANET are reviewed. Probable track of future trends related to VANET routing is offered in unit 3. Conclusion is provided in unit 4.

**I.1 Characteristics of VANETs:** 

- High Mobility: Owing to the high speed of vehicles in VANETs, taking decisions related to routing and security issues based on guessing node's position is very tricky [6].
- Rapidly changing Network topology: In VANET vehicles travel continuously and that too with high speed. Thus the position of vehicle changes repeatedly leading to recurrent topology changes.
- Unbounded network size: The range of the VANET is limitless. Thus network extent in VANET is physically uncontrolled [3].
- Frequent exchange of information: Owed to the ad hoc nature of VANET frequent information exchange between vehicles and also with RSUs is desired.

#### II. CATEGORIES OF ROUTING PROTOCOLS

Broadly routing protocols in VANETs are characterized into 5 brands viz. *Topology based*, *Position based*, *Broadcast based*, *Geo cast based* and *Cluster based routing protocols*[1] [5] as shown in Fig. 1. 2018 Second International Conference on Advances in Electronics, Computer and Communications (ICAECC-2018)

## Bezier Curve based Multipath Routing in VANET

Anil D. Devangavi Information Science and Engineering Basaveshwar Engineering College Bagalkot, India anildevangavi\_s@yahoo.co.in

Abstract—In VANET environment multipath routing protocols enhance reliability and fault tolerance. However major drawback of the existing multipath protocols is that all the computed paths are not utilized for communication at any given time. Hence this work proposes Bezier Curve based Multipath Routing (BCMR) in VANET. The proposed work computes multiple paths from source to given destination using cubic Bezier curves and more notably engage all or more than one path during the actual communication. Proposed work performs better w.r.t. parameters like packet delivery ratio, transmission time and path discovery time.

Keywords—VANET; Bezier curve; Degree of bezier curve; Multipath;

#### I. INTRODUCTION

Vehicular Ad-hoc network (VANET) is a network of vehicles as nodes without any centralized infrastructure. In VANETs the high mobility of nodes results in frequent destruction of routes. Further, the links over which data is communicated are highly unreliable. In single path routing end to end transmission delay is thus substantially high due to the time involved in the computation of alternate path, when primary gets broken. Multipath routing protocols are employed to tackle the problems associated with single path routing protocols.

Advantages of multipath routing: (1) Minimize delays in data transmission because of link failure. (2) Packet delivery ratio, throughput and delay are optimized. (3) Increased reliability and fault tolerance. (4) It facilitates load balancing [1] [2]. Thus development of efficient and reliable multipath routing is major apprehension in VANETs. Thus development of efficient and reliable multipath routing is major apprehension in VANETs. This has prompted to the development of various routing protocols and algorithms to realize this objective.

In paper [3] authors present a multipath route protocol based on AODV called Enhanced Multipath AODV (EMP-AODV). The approach employs a node-disjoint multipath strategy to compute routing. Data transmission is centered on expected link lifetime algorithm built on velocity of nodes is employed. Authors in [4] propose an adaptive cross-layer multi-path routing protocol (R-S-AOMDV) which is an improvement over AOMDV. Hop-count, quality of link and motion of vehicle information together constitutes routing metric. Authors in [5] present a protocol known as IMAODV (Improved Multicast-AODV) which integrates routing techniques of DSR and MAODV protocols. The protocol Rajendra Gupta Computer Science and Engineering Research Cell, AISECT University Madhya Pradesh, India rajendragupta1@yahoo.com

works in two phases viz. 1) Route Discovery Mechanism, 2) Route Maintenance Mechanism. Authors in [6] present ondemand multipath protocol, Fast Restoration On-demand Multipath Routing (FROMR). The presented approach focuses on quickly calculating replacement path if the original path is disconnected. The geographic region is divided into equal size square shaped grids. Within every grid, a vehicle which is likely to be there for the maximum period is designated as the leader of the grid. Grid leaders are accountable for all routing activities.

Authors in [7] present a Multipath Route Restoration Protocol (MRRP) that focuses not just on route change in case of link failure due to congestion, but also on the route restoration of the link failure path from source to destination. Authors in [8] paper presents a connectivity-aware intersection based routing (CAIR) protocol. In this technique amongst two intersections on the route packets are communicated by geographical forwarding centered on position prediction. The protocol aims at computing an optimal route with high connectivity and low delay. Authors in [9] present effective approach to forward packets along a trajectory modeled like parametric curve. To encode trajectories into packets at source Bezier parametric curve is employed. Centered on this encoding they have developed and evaluated numerous greedy forwarding algorithms.

However existing systems has certain disadvantages like: (1) Multiple paths are computed, but all the paths are not utilized simultaneously leading to extra overhead of computing multiple paths. (2) They have not taken into account vehicle mobility and issues related to disjoint paths. (3) The authors have not adequately addressed measures to reduce congestion caused by redundant relay messages and (4) Multiple paths are not searched effectively and hence are costly.

In this paper, we propose a new approach which employ cubic Bezier curve to compute multipath between source and destination. The proposed work thus targets to optimize reliability and delay by utilizing all the multiple paths computed. Accordingly information is disseminated to destination vehicle over multipath simultaneously. Our contributions are as follows: (1) Use of Bezier curves to compute multipath. (2) Minimize congestion and network traffic.

The proposed work Bezier Curve based Multipath Routing in VANET (BCMR) is compared with Improved Multicast AODV (IMAODV) [5]. The advantages of proposed approach

#### 978-1-5386-3785-2/18/\$31.00 ©2018 IEEE

## Cluster Based Context Aware Secure Routing in Distributed Sensor Networks

Lokesh B. Bhajantri<sup>1</sup>, Shiddalingesh S. Hiremath<sup>2</sup> <sup>1</sup>Department of Information Science and Engineering, Basaveshwar Engineering College, Bagalkot, Karnataka, India lokeshcse@yahoo.co.in

<sup>2</sup>Department of Computer Science and Engineering, Basaveshwar Engineering College, Bagalkot, Karnataka, India siddalingesh.409@gmail.com

Abstract — Distributed Sensor Network (DSN) built with smart sensor nodes by high computing network. A DSN has been attractive platform for researchers from many years and it's to provide effective computing and making secure communication or routing between source and sink node. However, sensor networks are vulnerable in an environment with respect to attacks.

This paper proposes a cluster based context ware secure routing in DSN by employing Grid Based cluster approach. This approach incorporates efficient cluster head selections, context discovery, context interpretation, and secure routing mechanisms in DSNs. The objective of the proposed work is to improve the effective performance of networks. The proposed work had been simulated in terms of packet delivery ratio, network lifetime, energy consumption, throughput, and packet dropping ratio.

## Keywords— Distributed Sensor Network, Cluster Head, Context Aware, Routing, Cryptography, CMKB, CKB.

#### I. INTRODUCTION

Distributed Sensor Networks (DSNs) come up with many number of smart sensor nodes. All the sensor networks are severely resource constrained such as energy, computing capabilities, and communication resources [1]. All sensor nodes are connected with one or more sink node or base station with high speed networks. In cluster based network, entire network is divided into number of cluster/grids as per the design requirement. Each cluster/grid has its own cluster head (CH), all cluster member (CM) nodes are used to sense the data, forward it to the respective CH.

A CH collects the information from CM nodes through intra-communication, aggregates the data, and removes the redundant information. CH nodes are intern forwards the aggregated information to the sink node or base station (BS) via single hop or multi hops. To increase the life time of the network, and nodes in the network should stay alive for long duration of time to avoid network failure. In the proposed work, selects sensor node as CH based on the highest residual energy among the sensor nodes within the clusters for some number of rounds. Energy is one of the main constraints in the design of DSN, energy consumption can be reduced by allowing only CH nodes to communicate with base station. It is best practice to understand the parameters, which are suitable for sensor network. Once parameters are identified then it can be easy to evaluate the DSN. It uses the following metrics are: ease of deployment, system life time, latency, load balancing, and quality [2] [3].

Context is any information that can be used to characterize the situation of an entity [4]. It gives the meaning to something in an environment. It can be materialized as identity, activity, location, and time [5]. In the proposed work, use of context to determine the context information that will distinguishes the data priority whether high, medium or low level priority data for routing data in DSN.

Routing is the process of data transfer from source sensor node to destination or sink node [6]. In the proposed work, routing is achieved in two phases, first all possible paths have to be identified from the all CHs in the network. In second phase, according to the context path, data transmission is to be done through the best possible path (trust) across the CHs in the network. Each CH transmits the encrypted data with private key from source node to sink node in the network. It preserves the quality of information and to avoid the malicious attacks over the path. Cryptography plays an important role in data security and it is standard way to provide security in network. The symmetric key algorithm is simulated in the proposed work for data security.

The rest of the paper is organized as follows. The overview of related work discussed in section II. A proposed work context aware secure routing is discussed in Section III. Simulation and results analysis are depicted in Section IV and finally conclude the work in section V.

#### II. RELATED WORK

Some of the related works are as follows: the work given in [7] presents context aware data aggregation in distributed sensor network. It proposes four different contexts in DSN by considering environment as temperature, context, air pressure context, energy aware context and object aware context. In

## Data Processing in Semantic Sensor Web: A Survey

#### Lokesh B. Bhajantri<sup>1</sup>

<sup>1</sup>Department of Information Science and Engineering, Basaveshwar Engineering College, Bagalkot, Karnataka, India lokeshcse@yahoo.co.in

#### Pundalik. R<sup>2</sup>

#### <sup>2</sup>Department of Computer Science and Engineering, KLE, CET, Belgaum, Karnataka, India pundalikr@gmail.com

Abstract: - Wireless Sensor Network (WSN) has set of wireless sensor nodes that are used to generate or sense the high volume of raw data from the environment, transmits the data to its respective sink node. The processing of raw data requires more amount of energy from sensor nodes to sink node or base station through single-hop or multi-hop communication over the nodes. In this context, sensor data need to be quality of data with semantic information in terms of semantic data integration and fusion and data processing, and so on.

In order to problem of semantic data and managing raw data in WSN effectively, the Semantic Sensor Web (SSW) is taken into the wireless sensor networks. The SSW is promising technology for realizing the use of sensor data and semantic web that provide the semantic information to existing sensor networks by utilizing domain, spatial and temporal ontologies and other related semantic technology. This paper presents survey of data processing in SSW from various aspects. Besides, it presents the architecture of SSW and its advantages.

*Keywords* – Wireless Sensor Network, Semantic Web, Semantic Sensor Web, and Data Processing.

#### 1. Introduction

A Wireless Sensor Network (WSN) consist of wireless sensors nodes with high speed of network, which are distributed logically, spatially or geographically over an environment. The WSN is used to provide the following benefits are: reliability, accuracy, flexibility, cost effectiveness, and ease of deployment of sensor nodes. Intelligent sensor nodes of WSN can used to detect and monitor the environment [1] [2].

Sensor nodes frequently sense the raw data from the surrounding environment for processing to the sink node. The frequently sensing of raw data and the number of sensors periodically reports the data to the respective base stations that depend on the particular applications. Semantic data aggregation and integration involves systematically collecting the sensed raw data from multiple sensors, aggregating to suppress the redundant data, and transmitting the aggregated data to the sink node for further processing [3] [4]. WSN is used for various applications like area, habitat, and environmental monitoring, vehicular adhoc networks applications, fire detection, battlefield surveillance and assessments, medical applications, industrial monitoring, target tracking, vigilance monitoring, and landslide monitoring [5] [6].

The semantic web (SW) is an analogy of current web that carries the semantic information to users in the real world environment [7] [8]. That is SW process the data from unstructured or semi-structures web data to semantic data or structures type of data in web. The following are the components of SW: (1) Uniform Resource Identifier (URI)-for identifying and locating services from the web. (2) Extensible Markup Language (XML)-for syntactic representation of text in the web. (3) Resource Description Framework (RDF)-is the metadata model for structuring and linking of data that describes the semantic data to users through semantic web. (4) Semantic Ontologies – for providing richer or semantic data. And (5) SPARQL-is the type of query language for RDF. The users can access the semantic data through SPARQL.

The Semantic Sensor Web (SSW) is an emerging area of research in the current world that gives the great benefits for the future. The SW technology has been introduced in the sensor network in categorize to defeat the problem of lack of meaningful data and also difficult to handle the homogeneous or heterogeneous data in WSN effectively. The SSW is

166

International Journal on Future Revolution in Computer Science & Communication Engineering Volume: 3 Issue: 11

ISSN: 2454-4248 475 - 481

#### Multipath Routing in VANET: Multi-Agent based Approach

Anil D. Devangavi Information Science and Engg., Basaveshwar Engineering College, VTU, Bagalkot, India anildevangavi\_s@yahoo.co.in Rajendra Gupta Research Cell, AISECT University, Madhya Pradesh, India rajendragupta1@yahoo.com

Abstract— In VANET routing of data is a exciting task owing to the high dynamics involved in this network. Delivery of data to the projected destination turns out to be very puzzling. Single path routing suffers from drawbacks like unreliability and etc. To manage such situation multipath data delivery is very nominal. In multipath routing more than one path discovered between source and destination node. Data packet can be sent simultaneously in all paths or data packet can be send by selecting path one after another. It is up to the routing algorithm to select path thoughtfully to deliver data proficiently. However existing multipath routing protocols even though compute multipath, only one path will be engaged in actual communication at any given time. Hence this work proposes Multipath Routing in VANET: Multi-agent based Approach which calculates multiple paths amongst source and destination. Further, all such computed paths will be employed for information dissemination. NS2 simulation of the proposed approach in realistic mobility models show that it can select more stable link and improve the network performance.

Keywords- Multipath, Node-disjoint, Congestion.

#### I. INTRODUCTION

Vehicular Ad hoc Network (VANET) is a variety of network intended for V2V (vehicle-to-vehicle) and V2I (vehicle to infrastructure) communication It is a core part of Intelligent Transportation System (ITS) and is being extensively researched now a days. Unique characteristics of VANET are no energy resource constraint, high mobility patterns, relatively high moving speed. During the transit of information through multi-hop routing, if an intermediate link was broken, the path needs to be rebuilt or restored. Hence, efficient routing protocols are essential to function properly. Multi-path routing protocols aim to compute several paths during the discovery phase between sender and receiver. Such network has been extensively used in battle field, disaster management, under water network etc. Advantages of multipath routing are 1) Load Balancing 2) Fault Tolerance 3) Higher Aggregate Bandwidth 4) Lower Delay 5) Reduced Congestion and improved QoS 6) Higher resiliency 7) Increased network lifetime [1].

Multipath routing protocols compute node disjoint, link disjoint or non-disjoint routes in the course of route discovery Process. Node-disjoint routes have completely disjoint routes where there are no nodes or links in common. Link-disjoint routes have no links in common but may have nodes in common. In an ad hoc network, identification of all nodedisjoint paths between a given pair of nodes is a challenging task [2].

Rest of the paper is organized as follows. Section I. A discusses related works. In section I. B methodology is presented. In section II structure of the proposed work is discussed. Network environment is mentioned in section II. A. Section II. B discusses mathematical models. Section II. C describes the proposed agency. The proposed scheme for the work is stated in section II. D. Section II. E describes the algorithm for the proposed work. Simulated model for the proposed work is discussed in section III. Results and analysis

are discussed in section IV. Conclusion is presented in section V.

#### A. Related Work

Authors in [3] propose a novel road topology-aware routing (RTR) protocol for VANETs. To adequately utilize connected routing paths RTR creates two junction-disjoint paths. RTR alternately transfers data packets through each established routing path and dynamically changes the routing path based on the connectivity of the current path. Authors in [4] propose Junction-based Multipath Source Routing (JMSR) a geographic routing protocol which exploits the location of the nodes and street junctions. Two concurrent paths from the source to the destination are preserved as a series of junctions the packets should pass through. Routing information is injected in each packet. As such every node on the path is aware of the route, packets to go ahead. Authors in [5] propose distributedly finding disjoint paths (DFDP) to compute k disjoint paths from source to destination. Further they analyze the relationship between disjoint paths and network parameters like robustness, throughput, and load balancing. Authors in [6] propose linkdisjoint, node-disjoint and zone disjoint multi-path algorithms aiming on the compromise between lifetime and hop count of routes for MANET. Authors in [7] evaluate the applicability of node disjoint multipath in VANET environment. Through simulation results author's present that it improves the packet delivery ratio and reduces the end-to-end delay. Authors in [8] present a modified form of AOMDV known as Maximally Spatial Disjoint Multipath routing protocol (MSDM) for MANETs. The data is communicated employing multiple paths which are mostly spatially separated and node-disjoint paths. Results show that spatially node-disjoint routes perform better than AOMDV which selects only link-disjoint paths.

IJFRCSCE | November 2017, Available @ http://www.ijfrcsce.org

SJCSE International Journal of Computer Sciences and Engineering Open Access

**Research Paper** 

Volume-5, Issue-10

E-ISSN: 2347-2693

#### Multi-Agent Based Context Aware Multicast Routing in VANET

A. D. Devangavi<sup>1\*</sup>, Rajendra Gupta<sup>2</sup>

<sup>1\*</sup>Information Science and Engineering, Basaveshwar Engineering College, VTU, Bagalkot, India <sup>2</sup>Research Cell, AISECT University, Bhopal, India

<sup>1</sup>Corresponding Author: anildevangavi s@yahoo.co.in, Mob: +919343186108

Available online at: www.ijcseonline.org

Received: 21/Sep/2017, Revised: 30/Sep/2017, Accepted: 15/Oct/2017, Published: 30/Oct/2017

Abstract— Vehicular Ad-hoc Network (VANET) is a variant of Mobile Ad-hoc Network (MANET). The key goal of VANET is to facilitate communications among vehicles and also amongst vehicles and fixed infrastructure. Regardless of the fact that VANET is considered as a sub-class of MANET, it has distinctive features in terms of high mobility of vehicles, fabricating to recurrent topology changes, random node density and boundless network size. Hence most of the clustering algorithms considered for MANET are inappropriate for VANET. Much of the literature recently published focus on clustering in VANETs. However most of them are concentrated on diminishing network overhead value, number of clusters formed and do not consider the vehicles interests (viz. traffic congestion, looking for petrol pumps, free parking space, etc.). To decrease the complexity of transmissions, only context-aware data is required to be communicated to the intended recipients as needless information may cause a performance bottleneck in VANETs. Hence in this paper, we propose a context aware multicast/clustered routing algorithm based on agent technology which addresses above mentioned issues and improves the performance parameters associated with routing in VANET. The performance of the proposed scheme is tested with respect to bandwidth consumption, cluster formation time, multicast grouping time and communication overhead.

Keywords- VANETs, Cluster, Multicast, Context, Interest.

#### I. INTRODUCTION

Vehicular ad-hoc network (VANET) is currently emerging technology which features many safety applications for transportation on roads. However, owing to the random mobility of vehicles in VANETs, communication links are. frequently disconnected leading to regular change in network topology. Many of the existing routing protocols are prone to the consumption of high bandwidth and to the overhead associated to route discovery. Clustering forms a virtual communication backbone that facilitates efficient data delivery in VANETs. The consumption of scarce resources such as bandwidth is minimized [1]. In VANET, broadly two approaches are employed for clustering of vehicles: Static clustering and dynamic clustering. Static clustering is formed between Vehicles and fixed Infrastructure (V2I). In this communication Road Side Units (fixed infrastructure) act as static cluster heads [2]. To facilitate real-time communication and connectivity with the Internet all the vehicles are basically connected with Road Side Units (RSU). However, owing to the large distance between the RSUs, the vehicles with high mobility are not always connected to the RSUs. Dynamic clustering is formed amongst vehicles (V2V) and cluster heads are elected from cluster members. Thus dynamic clustering method eliminates the idea of static cluster heads. The dynamic clusters are in motion on the road and vehicles either join or leave the clusters according to their speed and proximity to identified cluster heads. As V2V communications are more flexible and independent of the roadside conditions, they are particularly attractive for the most developing countries or remote rural areas where the roadside infrastructures are not necessarily available [3] [4]. Some of the advantages of clustering in VANETs are (1) Entire network can be divided into smaller groups of moving vehicles. This also facilitates the usage of different protocols within and outside clusters (hierarchical routing) [5]. (2) Efficient usage of bandwidth, routing, distribution of resources and scalability (3) Better routing and network lifetime, Support scalability of network (4) Routing path is limited to the cluster head, thus leading to small sized routing tables, reducing the bandwidth demand and efficient use of limited channel bandwidth (4) Eliminate the redundant and highly tedious route discovery process. (5) Users can be classified into separate groups based upon their interests.

The main purpose of the clusters will be essentially to manage a specific region. As the vehicles continuously move in and out of the cluster to which they belong, their position also change frequently and links between vehicles will be regularly broken. This necessitates the development of alternative stable approach for the formation of clusters. Thus it is very important to consider the context information, viz.

26

## Face Recognition Through Symbolic Data Modeling of Local Directional Gradient

Shanmukhappa Angadi & Vishwanath Kagawade

Conference paper | First Online: 17 November 2017 406 Accesses

#### Abstract

Face recognition systems are vulnerable to changes in expression, light, and occlusion factors. This paper presents a robust face recognition algorithm that effectively deals with these variations. The algorithm is based on the local directional gradient features that exploit the edge information in multiple directions, modeled as symbolic data object . First, face parts are detected and cropped from the face images, then the cropped images are resized to 64p × 64p grayscale images. Further, local directional gradient-based features are computed based on co-relation between pixel elements in multiple directions. The extracted directional gradient features (GC) are represented as a symbolic data object. For classification , a new symbolic similarity measure based on content is devised and employed. The experimental

Conferences > 2017 International Conference ?							
Iris recognition using savitzky-golay filter energy feature through symbolic data modeling Publisher: IEEE Cite This PDF							
Shanmukhap	pa A. Angac	ii; Vishwanath C. Kagawade All Authors					
<b>2</b> Paper Citations	65 Full Text Views	3 < C 🛌 🌲					
Abstract		Abstract:					
Document Sections		The iris has remained a preferred biometric trait compared to other biometrics because of its uniqueness, stability properties. However, degraded iris images captured under less					
I. Introduction		constrained acquisition setups and varying lighting conditions will affect the performance of iris					
II. Literature Review		symbolic modeling approach and also investigates the applicability of Savitzky-Golay filter for iris feature extraction. The approach also proves that the symbolic representation effectively					
	. – .						

#### ← → C ☆ ⓐ ieeexplore.ieee.org/abstract/document/8358392

Cite This

Conferences > 2017 International Conference... ?

## Iris recognition: A symbolic data modeling approach using Savitzky-Golay filter energy features

Publisher: IEEE

🔓 PDF

Shanmukhappa A. Angadi ; Vishwanath C. Kagawade			All Authors				
4 Bapar	114			0	مر	0	
Citation	s Text Views			•	•	٢	-

Abstract	Abstract:
De sum ent Os stisses	This paper presents a novel approach for iris recognition through symbolic modeling and
Document Sections	symbolic similarity analysis of iris features, which are transformed using Savitzky-Golay filter.
I. Introduction	The proposed approach is based on iris Savitzky-Golay filter energy features. Prior to
	symbolic modeling of iris feature data, Canny edge detector and Hough Transform technique
II. Literature Review	is used to segment iris region from the eye image. In the normalization stage, Daugman's
	rubber sheet model is used to obtain a fixed size rectangular block from segmented iris region

← → C △ a ieeexplore.ieee.org/abstract/document/8126180						
Conferences > 2017 International Conference ?						
Multicla	Multiclass classification of kirlian images using svm technique					
Publisher: I	Publisher: IEEE Cite This DPDF					
Chandrashek	khar S. Jana	adri; B. G. Sheeparamatti; Vishwanath Kagawade All Authors				
3	263					
Paper	Full	💄 📹 © ≽ 🥄				
Citations	Text View	S				
Abstract	:	Abstract:				
Document S	Sections	Every organism emits energy around it which comprises UV-radiation, EM-radiation, infrared and thermal radiation. This energy around human body represents health condition of the				
I. Introductio	n	subject under study. These energy fields are called as aura of the body under consideration.				
II. Methodolo	ogy	Several types of equipments are there to capture such energy. Kirlian camera captures the distribution of energy radiation around each finger of both hands and maps them into energy				
III. Training and Classification		distribution various organs of our body. In this work classification of human diseases based on Kirlian image features is done. Few Kirlian images in each of the 6 different categories of				
ISBN:9788192958057



## INTERNATIONAL CONFERENCE

on

Automation Science and Engineering

# **ICASE-18**

15<sup>th</sup> December - 2018 Bangalore

Published by

## CONTENTS

TITLES AND AUTHORS

5.NO

-

 		_	
		_	

Comparing Subdivision study of Naturnho Nuccifera flower (Petal) Extract as Green	51-55
Intuitions for Mild Read in Hydrochloric Acid	
> ANPaul	
> G.M.Malk	
Taka	56-62
Image Entropy as the Quality Metric in Automating Agricultural integral, view	
<ul> <li>Aran Lamar 2</li> </ul>	
* Ypey 5 Regenerated	
and a second to prove the second seco	63-67
FEM for Stress Radiction by Optimal Automaty France or a control of the	
Elliptical Nets	
> AC Summer bod	
- And Andrews	

# FEM for Stress Reduction by Optimal Auxiliary Holes in a Loaded Plate with Elliptical Hole

<sup>[1]</sup> Basavaraj R.Endigeri, <sup>[2]</sup> S.G.Sarganachari

[1][2] Mechanical Engineering Department of Basaveshwar Engineering College, Bagalkot, Karnataka, India

Abstract: — Steel is widely used in machine parts, structural equipment and many other applications. In many steel structural elements, holes of different shapes and orientations are made with a view to satisfy the design requirements. The presence of holes in steel elements creates stress concentration, which eventually reduce the mechanical strength of the structure. Therefore, it is of great importance to investigate the state of stress around the holes for the safety and properties design of such elements. By providing such and radii in a steel plate. The numerical method can be used to determine the optimum location and radii in a steel plate. The numerical method can be used to determine the optimum location and radii with its effect on stress concentration is graphically represented. The introduction of auxiliary holes at a optimum location and radii is carried out using a plane 42 element. Further the ANSYS optimization model is used to determine the location and radii for optimum values of auxiliary hole to reduce entration. All the results for different diameter to plate with ratio are presented graphically. The results of this study are in the form of the graphs for determining the locations and radii for optimum values of auxiliary hole to reduce stress concentration. All the results for different diameter to plate width ratio are presented graphically. The results of this study are in the form of the graphs for determining the locations and diameter to plate width ratio are presented graphically. The results of this study are in the form of the graphs for determining the locations and radii for optimum values of auxiliary hole to reduce stress concentration. All the results for different diameter to plate width ratio are presented graphically. The results of this study are in the form of the graphs for determining the locations and diameter to plate width ratio are presented graphically. The results of this study are in the form of the graphs for determining the locations and diameter to plate widt

loaded plate can be reduced by introducing auxiliary holes on either side of the central circular hole.

Key words: Finite Element Method, Optimization, Stress concentration factor, auxiliary holes.

### L INTRODUCTION

THE failure of structures due to stress concentration at any discontinuity has been baffling engineers for long. It is found that structure failures in ships, offshore structures, boilers or high rise buildings subjected to natural calamities is due to stress concentration. problem of determining stress distribution around holes in a plate has been treated extensively in the literature. T. Hasan [2] has reported stress analysis of steel plate having holes of various shapes, sizes and orientations using finite element method. Finite element analysis is carried out by using the commercial software. Effects of hole shapes are critically analyzed and results are compared with analytical method which showed good agreement with the results. A.J.Durelli, V.J.Parks and H.C.Feng[3] presented Photoelastic solution of the distribution of stresses around a centrally located elliptical hole in a plate of finite width subjected to uniform axial loading and reported stress distribution at the boundaries for a wide range of the parameters, and stress concentration factors have been computed for the points of maximum tensile and maximum compressive stress.

Ukadgaonkar [4] has given a new approach of stress analysis of an infinite plate with elliptical hole or crack with tensile stress. The closed form equations for SCF are given for anisotropic plate. According to the study SCF in flat plate with different geometry can be reduced by introducing auxiliary holes. The location of auxiliary holes from the center of ellipse has to be reported.

G. V. Kolosoff and C. E. Inglis [5]: proposed boundary element method to the problems of anisotropic plates containing multiple various shaped holes, this research solves the stress and strain fields numerically by embedding the fundamental solutions of polygonal holes into boundary element method (BEM), and using the technique of sub region. In the literature, most of the examples of multi-holes consider elliptical holes due to the geometry restraints; it limits the range of the problems. Since the fundamental solutions have satisfied the boundary condition for holes, it's unnecessary to descretize the hole boundary. Since, the fundamental solutions are for the polygon-like holes, their associated BEM would be able to solve multi-holes much more effectively.

There is extensive literature which uses finite element method (FEM) to optimize the stress concentration but not used for auxillary hole optimization for stress reduction.

In the present work FEM optimization was carried to determine the optimum sizes and location for the auxiliary holes for different center elliptical hole diameter to plate width ratios were minimization was achieved for all possible ratios.

## **II. DESCRIPTION OF THE PROBLEM**

To examine the stress distribution around the original hole and to study the effect of auxiliary hole on stress

NIER International Conference Bangalore 63 ' ISBN: 9788192958057



-			
SI. P	TITLES AND AUTHORS	Page N	
01.	Assessment of Stress Reduction Techniques in an Uniaxially Loaded Plate with a Hole	1-3	
	S.G. Sarganachari, Basavaraj R. Endigeri		
02.	An Novel Idea: Artificial Neural Network	17	
	Ravi Chandra, Reddem Pradeep Kumar, Reddy Shamanths	4-7	
03.	Product Identification and Monitoring System in Shopping Complex	8-10	
	Pragathi S.B., Sridevi H.S., Khatija Faisal, Sheethal P.R.	0-10	
04.	Advancement of Smart Helmet based on IoT Technology for Safety and Accident Detection	11-12	
	Sankalp Kumar Rai, Shivshankar Kumar Yadav, Suraj Kumar		
05.	Parking Assistance System by Mapping Ultrasonic and Camera Data to Detect Objects with Distance	13-14	
	Ayesha Misbah, Deepak V, Gautham N		
06.	-Categorization of Email using Machile Learning Algorithm		
	S Vino David, Rashmi KR, Sandhya V, Rakshitha P	15-16	
07.	Optimum Parametric Analysis of Mechanical Oscillator for Pumping Application	10.24	
	Rejin Balachandran, Nandu A, Nirmal M S	19-24	
08.	Option for Optimal Extraction to Indicate Recognition of Gestures using the Self- Improvement of the Micro-Genetic Algorithm	25-28	
	Haitham Sabah, Omran Al Shamaa		
09.	Pollution Spillover, International Trade and Abatement Policy	20.00	
	Devleena Majumdar, Saibal Kar	29-37	
0.	Company Predictor using Resume	20.40	
	Yusoof Ali, Suraj RK, Noor Ahmed	38-40	
1.	Detection of Fraud Presence based on Stereovision Face Recognition		
	Gayathri RM, Lokesh R, Hemanth Kumar K, Likith S	41-44	
2. (	Closed Loop Performance Evaluation of Line Differential Relay on RTDS	45-49	

## ASSESSMENT OF STRESS REDUCTION TECHNIQUES IN AN UNIAXIALLY LOADED PLATE WITH A HOLE

## <sup>1</sup>S.G. SARGANACHARI, <sup>2</sup>BASAVARAJ R. ENDIGERI

<sup>12</sup>Basaveshwar Engineering College, Bagalkot, Karnataka, India E-mail: <sup>1</sup>sgsarganachari@gmail.com, <sup>3</sup>endigeribasavaraj@gmail.com

Abstract - Stress concentration in a rectangular plate with central hole under static loading, have found widespread applications in various fields of engineering such as Aerospace, Marine and Automobile. For design of plates with hole, accurate knowledge of deflection and stress are required. Stress concentration arises from abrupt change in geometry of plate under static loading. As a result, stress distribution is not uniform throughout the cross section. Various stress reduction techniques are available to determine stress concentration factor under different loading conditions such as experimental, analytical and numerical method. Though experimental methods give the most reliable results, it is very costly, as it requires special equipments, testing facilities etc. Analytical solution of every problem is almost impossible because of complex boundary conditions and shapes. For this reason the numerical methods had become the ultimate choice by the researchers in the last few decades. In the present paper an attempt has been made to review various stress reduction techniques.

Index Terms - Stress Concentration Factor, Auxiliary Holes.

#### I. INTRODUCTION

## Most of the engineering applications consist of assembly of simple elements, which are connected to each other by joints. Joints or connections that are usually made in steel structures are mechanical fastening, holes are made to place the bolts or rivets; these make the structure weak and susceptible to failure. Therefore, it is necessary to investigate the state of stress around the holes for the safety and proper design of such structures. From the point of view of the above facts, it is of great importance to understand the behavior of the steel structures with holes.

## **II. STRESS CONCENTRATION**

The referenced collection of stress concentration factor data is in Peterson [1]. This book compiles the theoretical and experimental results of many researchers in to useful design charts from which the values of stress concentration kt (for normal stress), for various geometric parameters and types of loading can be read. Rorak and Young [2] also provide tables of stress concentration factors for a number of cases. The ratio of the maximum stress over the average stress computed in the critical section of the discontinuity is kt= omax/oapp

Where omax and oapp are the maximum stress and applied stress, respectively. The value of kt was plotted by Frocht (discussed by Beer and Johnston [3]). Amelio M. D'Arcangelo [4], investigated the variation of the maximum stress concentration factor as a function of the corner radius to the opening width ratio in a rectangular plate with a central hole. Therefore, more study or more investigation is required to minimize the stress concentration effects of the structure.

### **III. LITERATURE REVIEW**

The stress distribution around holes in a plate has been treated extensively in the literature. An experimental solution was given by P. E. Erickson and W. F. Riley[5] in the year 1978, which was carried using Photoelastic method. They obtained the photoelastic isochromatic fringe pattern. Fig.1.



Fig.1. Photoelastic isochromatic fringe patterns from a model with auxiliary boles[4]

The results of the study indicates that, the stress concentration effects of a central circular hole in a uniaxially loaded plate can be reduced by 13 to 21 percent by introducing auxiliary holes on either sides of the central hole. Such a reduction of stress amplitude could have a significant effect on the fatigue life of a part.

A considerable reduction in stress concentration can be observed by the experiments conducted by K Rajaiah and N K Naik [6] Fig.2.

Proceedings of ARSSS International Conference, 16th December, 2018, Bengaluru, India.